DOE/CAO-95-1121

TRANSURANIC WASTE BASELINE

INVENTORY REPORT

(REVISION 3)







June 1996

NOTICE TO READERS

This document, Revision 3 of the Transuranic Waste Baseline Inventory Report (TWBIR), has been prepared to document the transuranic (TRU) waste inventory data to be used in the Sandia National Laboratories/New Mexico (SNL/NM) calculations for the Waste Isolation Pilot Plant's (WIPP's) performance assessment (PA). The TWBIR Revision 3, is comprised of previously published information found in Revision 2 of the TWBIR and supplemented with information and data that were specifically requested by the U.S. Department of Energy (DOE) Carlsbad Area Office (CAO) for the SNL/NM PA calculations.

The data contained in this document will also be used as the inventory basis for the WIPP Compliance Certification Application (CCA) to be submitted to the U.S. Environmental Protection Agency. The site information requested in the January 1996 data call has not been included in Revision 3. Future editions of the TWBIR will be identified by the year of data origin.





TABLE OF CONTENTS

| | LIST C | of figu | JRES iv |
|---|--------|---------|---|
| | | OF TAB | LES |
| | ACRO | NYMS | AND ABBREVIATIONS |
| | EXEC | | SUMMARY ES - 1 |
| | 1. | INTRO | DUCTION |
| | | 1.1 | BACKGROUND 1 - 1 |
| | | 1.2 | PURPOSE |
| | | 1.3 | WASTE INVENTORY TERMINOLOGY 1 - 3 |
| | | 1.4 | METHODOLOGY FOR DEVELOPMENT OF DISPOSAL INVENTORY 1 - 8 |
| | | 1.5 | DOCUMENT ORGANIZATION |
| | 2. | SUMN | IARY OF WIPP DISPOSAL INVENTORY INFORMATION |
| | | 2.1 | INTRODUCTION |
| | | 2.2 | WIPP DISPOSAL INVENTORY VOLUMES FOR EACH FINAL WASTE FORM |
| | | 2.3 | ROLL-UP OF WIPP WASTE MATERIAL PARAMETERS BY FINAL WASTE FORM |
| | | 2.4 | SUMMARY OF WIPP ANTICIPATED INVENTORY FROM EACH SITE 2 - 4 |
| | 3. | SUPP | LEMENTAL DISPOSAL INVENTORY INFORMATION |
| | | 3.1 | INTRODUCTION |
| M | | 3.2 | SUPPLEMENTAL RADIONUCLIDE INFORMATION 3 - 1 3.2.1 Revised WIPP Disposal Radionuclide Inventory 3 - 1 3.2.2 Activity Calculations for Waste Streams 3 - 10 |
| | | 3.3 | SUPPLEMENTAL INFORMATION FOR OTHER CONSTITUENTS |
| | | | 3.3.3 Estimate of Cement in TRU Solidified Waste Forms for Disposal in WIPP |

| 4. | REFERENCE | S |
|------|-----------|---|
| 5. | GLOSSARY | |
| APPE | NDIX A. | SNL/NM REQUIREMENTS FOR SUPPLEMENTAL DATA A - 1 A - 1. CH- and RH-TRU Waste Parameters Potentially Important in WIPP PA, November 6, 1995. A - 2. Information Needed from TWBIR (Revision 2/Addendum), January 11, 1996. A - 3. Information Needed from TWBIR (Revision 2/Addendum), January 30, 1996. |
| APPE | NDIX B. | SUPPLEMENTAL TWBIR REVISION 2 INFORMATION B - 1 B - 1. Revised Radionuclide Data in Support of the Compliance Certification Application, June 4, 1996. B - 2. Preliminary Activities for Selected Radionuclides for CH-TRU Waste Streams, June 12, 1996. B - 3. Preliminary Estimates of Complexing Agents in TRU Solidified Waste Forms Scheduled for Disposal in WIPP, March 15, 1996. B - 4. Current Estimate of Complexing Agents in Transuranic Solidified Waste Forms Scheduled for Disposal in WIPP, March 29, 1996. B - 5. Revision of Current Estimate of Complexing Agents in Transuranic Solidified Waste Forms Scheduled for Disposal in WIPP, June 26, 1996. B - 6. Preliminary Estimate for SNL/NM Performance Assessment Calculations of Nitrate, Sulfate, and Phosphate in Transuranic Solidified Waste Destined for Disposal in WIPP, February 20, 1996. B - 7. Estimate of Cement Content in TRU Solidified Waste Forms Scheduled for Disposal in WIPP, April 4, 1996. |
| APPE | NDIX C. | |
| APPE | INDIX D. | CORRECTION FOR Cf-252 DECAY INVENTORY D-1 |

LIST OF FIGURES

| Figure | | | Page |
|--------|-----|--|---------|
| | 1-1 | U.S. Department of Energy Transuranic Waste Sites | . 1 - 2 |
| | 1-2 | Schematic of Waste Stream Profile Methodology | . 1 - 9 |



LIST OF TABLES

| Table | F | ^o age |
|-------|--|------------------|
| ES-1 | WIPP CH-TRU Waste Material Parameter Disposal Inventory | S - 4 |
| ES-2 | WIPP RH-TRU Waste Material Parameter Disposal Inventory | S - 5 |
| ES-3 | WIPP CH-TRU Waste Anticipated Inventory by Site E | S - 6 |
| ES-4 | WIPP RH-TRU Waste Anticipated Inventory by Site E | S - 7 |
| ES-5 | Summary Radionculide Inventory | S - 8 |
| 1-1 | Technical Data Needs for Performance Assessment Waste Material Parameters | 1 - 4 |
| 1-2 | Waste Matrix Codes and Their Anticipated Final Waste Form | 1 - 6 |
| 2-1 | Transuranic Waste Disposal Inventory for WIPP | 2 - 2 |
| 2-2 | WIPP CH-TRU Waste Material Parameter Disposal Inventory | 2 - 5 |
| 2-3 | WIPP RH-TRU Waste Material Parameter Disposal Inventory | 2 - 6 |
| 2-4 | WIPP CH-TRU Waste Anticipated Inventory by Site | 2 - 7 |
| 2-5 | WIPP RH-TRU Waste Anticipated Inventory by Site | 2 - 8 |
| 3-1 | WIPP Disposal Radionuclide Inventory for the CCA | 3 - 3 |
| 3-2 | Estimates of Complexing Agents in Transuranic Waste from RFETS, iNEL, LANL, and Hanford | 3 - 12 |
| 3-3 | Estimates of Total Organic Carbon (TOC) in ORNL Transuranic Sludge Tanks and Possible Complexing Agents that can Contribute to TOC | - V3. |
| | | |



ACRONYMS AND ABBREVIATIONS

Argonne National Laboratory-East site identifier AF Ames Laboratory site identifier AL. ARCO Medical Products Company site identifier AM AW ANL-W site identifier Battelle Columbus Laboratory site identifier BC BΤ Bettis Atomic Power Laboratory site identifier Agreement for Consultation and Cooperation between the Department of C&C Agreement Energy and the State of New Mexico on the Waste Isolation Pilot Plant CAO Carlsbad Area Office **Compliance Certification Application** CCA CFR Code of Federal Regulations CH contact-handled CY calendar year D&D decontamination and decommissioning DOE U.S. Department of Energy EPA U.S. Environmental Protection Agency ER environmental restoration Energy Technology Engineering Center site identifier ET FFCAct Federal Facilities Compliance Act IDB Integrated Data Base IN Idaho National Engineering Laboratory site identifier IT Inhalation Toxicology Research Institute site identifier Knolls Atomic Power Laboratory-Schenectady site identifier KA kilograms kg LA Los Alamos National Laboratory site identifier LANL Los Alamos National Laboratory Lawrence Berkeley Laboratory site identifier LB LL Lawrence Livermore National Laboratory site identifier LWA Land Withdrawal Act MC U.S. Army Material Command Mound Plant site identifier MD m³ cubic meters millirem mrem MU University of Missouri Research Reactor site identifier NT Nevada Test Site site identifier Oak Ridge National Laboratory site identifier OR Oak Ridge Isotope Generation and Depletion Code ORIGEN2 Oak Ridge National Laboratory ORNL performance assessment (in text only) PA Paducah Gaseous Diffusion Plant site identifier (in waste profiles only) PA polychlorinated biphenyls PCB Pantex site identifier PX Resource Conservation and Recovery Act RCRA Rocky Flats Environmental Technology Site site identifier RF Rocky Flats Environmental Technology Site RFETS

vi

- RH remote-handled
- RL Hanford (Richland) site identifier
- SA Sandia National Laboratories/New Mexico site identifier

SNL/NM Sandia National Laboratories/New Mexico

- SR Savannah River Site site identifier
 - SRS Savannah River Site
 - TB Teledyne Brown Engineering
 - TOC total organic carbon
 - TRU transuranic
- TWBIR Transuranic Waste Baseline Inventory Report
 - WAC waste acceptance criteria
- WIPP Waste Isolation Pilot Plant
- WMC waste matrix code
- WMP waste material parameter
 - WV West Valley Demonstration Project site identifier

 (\mathbf{F})



EXECUTIVE SUMMARY

The *Transuranic Waste Baseline Inventory Report* (TWBIR) establishes a methodology for grouping wastes of similar physical and chemical properties from across the U.S. Department of Energy (DOE) transuranic (TRU) waste system into a series of "waste profiles" that can be used as the basis for waste form discussions with regulatory agencies.

The purpose of Revisions 0 and 1 of this report was to provide data to be included in the Sandia National Laboratories/New Mexico (SNL/NM) performance assessment (PA) processes for the Waste Isolation Pilot Plant (WIPP). Revision 2 of the document expanded the original purpose and was also intended to support the WIPP Land Withdrawal Act (LWA) requirement for providing the total DOE TRU waste inventory. The document included a chapter and an appendix that discussed the total DOE TRU waste inventory, including nondefense, commercial, polychlorinated biphenyls (PCB)-contaminated, and buried (predominately pre-1970) TRU wastes that are not planned to be disposed of at WIPP.

Revision 3 of the TWBIR is based on the TWBIR Revision 2 data which are supplemented by data in several memoranda issued during early calendar year (CY) 1996. These memoranda summarize additional data requested by the U. S. Department of Energy/Carlsbad Area Office (DOE/CAO) to support the SNL/NM PA modeling. The primary purpose of Revision 3 is to provide the summary data from TWBIR Revision 2 and the supplemental information used by SNL/NM in the development of the Compliance Certification Application (CCA) to be delivered to the Environmental Protection Agency (EPA), and to support the LWA (Public Law, 1992b). The supplemental information was generated from specific data requests to the TRU waste sites since the publication of Revision 2. The supplemental data discussed in detail in Chapter 3 and Appendices A and B are listed below:

- Radionuclide data in support of the Compliance Certification Application.
- Estimate of complexing agents in TRU solidified waste forms scheduled for disposal in WIPP2
- Estimate for SNL/NM PA calculations of nitrate, sulfate, and phosphate content in transuranic solidified wastes destined for disposal in WIPP.
- Estimate of cement content in TRU solidified waste forms scheduled for disposal in WIPP.

Revision 2 of the TWBIR included both the TRU waste that is allowed to be disposed of in WIPP and the DOE TRU waste that is not currently allowed to be disposed of in WIPP (Public Law, 1992b). Because the primary purpose of this Revision 3 TWBIR is to support the CCA and PA, it includes only the DOE TRU waste that is currently allowed to be disposed of in WIPP.

Revision 3 of the TWBIR is different from previous revisions in that it provides the TRU waste inventory information developed for Revision 2 along with supplemental data. It is necessary for the reader to be familiar with Revision 2 of the TWBIR to understand this TWBIR Revision 3 document. Much of the TWBIR Revision 2 information is referenced, rather than repeated, in this document, resulting in an abbreviated document. Revision 3 of the TWBIR consists of one volume having five chapters and four appendices. There is not a new electronic database for

TWBIR Revision 3 because the data in the Revision 2 database are unchanged; therefore new database diskettes are not being published with this document.

The WIPP anticipated (stored and projected) inventory of TRU waste is defined as the sum of retrievably stored waste plus currently projected TRU waste volumes. Current projections do not include waste generated as a result of future environmental restoration (ER) and decontamination and decommissioning (D&D) activities and have only been developed over a 25 year period, consequently the anticipated inventory for CH-TRU waste is not sufficient to fill the maximum CH-TRU disposal inventory for WIPP (calculated to be approximately 168,500 cubic meters or 5,950,000 cubic feet). Scaling has been developed as a means for SNL/NM to model the impacts of a full repository. Scaling has not been applied to the RH-TRU inventory (approximately 7,080 cubic meters or 250,000 cubic feet).

The TWBIR also estimates the WIPP disposal inventory in terms of 12 waste material parameters and additional packaging materials that have been identified by SNL/NM as necessary for PA. The 12 waste material parameters and additional packaging materials are constituents of TRU waste and are input parameters for one or more PA models or are required to adequately describe the waste form.

The 12 waste material parameters and additional packaging materials are listed below:

Waste Material Parameters

- Iron-base metal/alloys
- Aluminum-base metai/alloys
- Other metal/alloys
- Other inorganic materials
- Vitrified
- Cellulosics
- Rubber
- Plastics
- Solidified inorganic material
- Solidified organic material
- Cement (solidified)
- Soils

Packaging Materials

- Steel
- Plastic
- Lead (for RH-TRU waste only)

The waste material parameters are expressed on a weight/volume (kilograms per cubic meter) basis. The occurrence of more than one waste material parameter at the maximum value within a waste stream is highly unlikely. If required by PA calculations, the sampling statistics must be controlled so that several waste material parameters do not get sampled all at their maximum value (weight/volume), thereby exceeding the average weight/volume.

Attached to this Executive Summary are several summary tables from the body of the TWBIR Revision 3 which are frequently requested by TWBIR users:

ES - 2

- Table ES-1. WIPP CH-TRU Waste Material Parameter Disposal Inventory
- Table ES-2. WIPP RH-TRU Waste Material Parameter Disposal Inventory
- Table ES-3. WIPP CH-TRU Waste Anticipated Inventory by Site
- Table ES-4. WIPP RH-TRU Waste Anticipated Inventory by Site
- Table ES-5. Summary Radionuclide Inventory

Table ES-1. WIPP CH-TRU Waste Material Parameter Disposal Inventory*

| Waste Material Parameters (Kg/m3) | <u>Maximum</u> | Average | <u>Minimum</u> |
|-----------------------------------|-----------------|------------------|-----------------|
| Iron Base Metal/Alloys | 2.6E+03 | 1.7E+02 | 0.0E+00 |
| Aluminum Base Metal/Alloys | 8.0E+02 | 1.8E+01 | 0.0E+00 |
| Other Metal/Alloys | 1.6E+03 | 6.7E+01 | 0.0E+00 |
| Other Inorganic Materials | 1.46+03 | 3.1E+01 | 0.0E+00 |
| Vitrified | 2.5E+03 | 5.5E+01 | 0.0E+00 |
| Cellulosics | 9.6E+02 | 5.4E+01 | 0. 0E+00 |
| Rubber | 6.3E+02 | 1.0E+01 | 0. 0E+00 |
| Plastics | 8.9E+02 | 3.4E+01 | 0.0E+00 |
| Solidified Inorganic Material | 2.2E+03 | 5. 4E+0 1 | 0. 0E+00 |
| Solidified Organic Material | 1.4E+03 | 5.6E+00 | 0.0E+00 |
| Cement (Solidified) | 1. 2E+03 | 5.0E+01 | 0.0 E+00 |
| Soils | 1. 5E+03 | 4.4E+01 | 0.0E+00 |
| Container Materials - Kg/m3 | | | |
| Steel | | 139 | |
| Plastic/ Liners | | 25 | |



*This table is identical to Table ES-1 of TWBIR Revision 2, page ES-4 (DOE, 1995c).

Table ES-2. WIPP RH-TRU Waste Material Parameter Disposal Inventory*

| Waste Material Parameters (Kg/m3 | <u>Maximum</u> | Average | <u>Minimum</u> |
|----------------------------------|-----------------|------------------|-----------------|
| Iron Base Metal/Alloys | 1.7 E+03 | 1.0E+02 | 0.0E+00 |
| Aluminum Base Metal/Alloys | 1.7E+02 | 7.1 E+00 | 0.0E+00 |
| Other Metal/Alloys | 9.1 E+02 | 2.5E+02 | 0. 0E+00 |
| Other Inorganic Materials | 2.0E+03 | 6.4E+01 | 0. 0E+00 |
| Vitrified | 2.5E+03 | 4.7E+00 | 0.0E+00 |
| Cellulosics | 5.7E+02 | 1.7 E+01 | 0.0E+00 |
| Rubber | 4.4E+02 | 3.3E+00 | 0.0E+00 |
| Plastics | 6.2E+02 | 1.5E+01 | 0.0E+00 |
| Solidified Inorganic Material | 6.1E+02 | 2.2E+01 | 0.0E+00 |
| Solidified Organic Material | 8.1E+02 | 9.3 E-0 1 | 0.0E+00 |
| Cement (Solidified) | 5.8E+02 | 1.9E+01 | 0.0E+00 |
| Soils | 2.4E+01 | 1.0E+00 | 0.0E+00 |
| Container Materials - Kg/m3 | | | |
| Steel | | 446 | |
| Plastic/Lipers | | 3.1 | |
| Lead | | 465 | |
| Steel Plug | | 2145 | |

*This table is identical to Table ES-2 of TWBIR Revision 2, page ES-5 (DOE, 1995c).



Table ES-3. WIPP CH-TRU Waste Anticipated Inventory By Site*

| | (| Cubic Meters) | | |
|---|-------------------|----------------------|------------------------|----------|
| Storage/Generator Site | Stored Volumes | Projected Volumes | Anticipated Volumes | |
| Ames Laboratory - Iowa State Univ. | 0.0E+00 | 4.2E-01 | 4.2E-01 | |
| Argonne National Laboratory - East | 1.1E+01 | 1.3E+02 | 1.4 E+02 | |
| Argonne National Laboratory - West | 6.5E+00 | 7.4E+02 | 7.5E+02 | |
| Bettis Atomic Power Laboratory | 0.0E+00 | 1.2E+02 | 1.2E+02 | |
| Energy Technology Engineering Center | 1.7E+00 | 0.0E+00 | 1.7E+00 | |
| Hanford (Richland) Site | 1.2E+04 | 3.3E+04 | 4.6E+04 | |
| Idaho National Engineering Laboratory | 2.9E+04 | 0.0E+00 | 2.9E+04 | |
| Lawrence Livermore National Laboratory | 2.3E+02 | 7.1E+02 | 9.4E+02 | |
| Los Alamos National Laboratory | 1.1E+04 | 7.4E+03 | 1.8E+04 | |
| Mound Plant | 2.7E+02 | 0. 0E+00 | 2.7E+02 | |
| Nevada Test Site | 6.2E+02 | 9.0E+00 | 6.3E+02 | |
| Oak Ridge National Laboratory | 1.3E+03 | 2.6E+02 | 1.6E+03 | |
| Paducah Gaseous Diffusion Plant | 0.0E+00 | 1. 9E+00 | 1.9E+00 | |
| Pantex Plant | 6.2E-01 | 0.0E+00 | 6.2E-01 | |
| Rocky Flats Environmental Technology Site | 7.1E+02 | 4.4E+03 | 5.1E+03 | |
| Sandia National Laboratory - Albuquerque | 6.7E+00 | 7.5E+00 | 1.4 E+01 | · |
| Savannah River Site | 2.9E+03 | 6.8E+03 | 9.6E+03 | 124 |
| Teledyne Brown Engineering | 2.1E-01 | 0.0E+00 | 2.1E-01 | |
| U.S. Army Material Command | 2.5E+00 | 0.0E+00 | 2.5E+00 | |
| University of Missouri Research Reactor | 2.1E-01 | 8.3E-01 | 1.0E+00 | <u> </u> |
| Totai CH Volumes | 5.8E+04 | 5.4E+04 | 1.1 E+05 | |

*This table is identical to Table ES-3 of TWBIR Revision 2, page ES-6 (DOE, 1995c).

Table ES-4. WIPP RH-TRU Waste Anticipated Inventory By Site*

| | | (Cubic Meters) | |
|---------------------------------------|-------------------|----------------------|------------------------|
| Storage/Generator Site | Stored Volumes | Projected Volumes | Anticipated Volumes |
| Argonne National Laboratory - West | 1.9E+01 | 1.3E+03 | 1.3E+03 |
| Battelle Columbus Laboratories | 5.8E+02 | 0.0 E+00 | 5.8E+02 |
| Bettis Atomic Power Laboratory | 0.0E+00 | 6.7E+00 | 6.7E+00 |
| Energy Technology Engineering Center | 8.9E-01 | 0.0E+00 | 8.9E-01 |
| Hanford (Richland) Site | 2.0E+02 | 2.2E+04 | 2.2E+04 |
| Idaho National Engineering Laboratory | 2.2E+02 | 0.0 E+00 | 2.2E+02 |
| Los Alamos National Laboratory | 9.4E+01 | 9.9E+01 | 1.9E+02 |
| Oak Ridge National Laboratory | 2.5E+03 | 4.5E+02 | 2.9E+03 |
| Total RH Volumes | 3.6E+03 | 2.3E+04 | 2.7E+04 |
| Total TRU Waste Volumes | 6.2E+04 | 7.7E+04 | 1.4 E+05 |

*This table is identical to Table ES-4 of TWBIR Revision 2, page ES-7 (DOE, 1995c).

*



ES - 7

| Nuclide | CH-TRU Waste (Ci/m ³) | RH-TRU Waste (Ci/m ³) |
|---------|-----------------------------------|-----------------------------------|
| Am241 | 2.62E+00 | 8.42E-01 |
| Ba137m | 4.53E-02 | 2.89E+01 |
| Cm244 | 1.87E-01 | 4.45E-02 |
| Co60 | 3.83E-04 | 1.47E+00 |
| Cs137 | 4.78E-02 | 3.05E+01 |
| Pu238 | 1.55E+01 | 2.05E-01 |
| Pu239 | 4.66E+00 | 1.45E+00 |
| Pu240 | 1.25E+00 | 7.15E-01 |
| Pu241 | 1.37E+01 | 2.00E+01 |
| Sr90 | 4.07E-02 | 2.95E+01 |
| Y90 | 4.07E-02 | 2.95E+01 |

Table ES-5. Summary Radionuclide Inventory^{1*}

¹Summary shows the ten radionuclides with the highest concentration in curies per cubic meter for both CH-TRU and RH-TRU waste. The list includes eleven radionuclides because the ten radionuclides with the highest concentration are different for CH-TRU and RH-TRU waste.

*This table is an update of Table ES-7, of TWBIR Revision 2, page ES-10 (DOE, 1995c).

1. INTRODUCTION

1.1 BACKGROUND

Transuranic (TRU) waste is defined as waste that is contaminated with alpha-emitting radionuclides with an atomic number greater than 92, with half-lives greater than 20 years, and concentrations of TRU isotopes greater than 100 nanocuries per gram of waste at the time of assay (DOE, 1988). TRU wastes are classified as either contact-handled (CH) waste or remote-handled (RH) waste, depending on the dose rate at the surface of the waste container. CH-TRU wastes are packaged TRU wastes with an external surface dose rate less than 200 millirems (mrem) per hour, while RH-TRU wastes are packaged TRU wastes with an external surface dose rate of 200 mrem or greater per hour (Public Law, 1992b). Unless otherwise indicated, for the purposes of this document, all references to TRU waste include TRU waste and mixed TRU waste (waste that contains both radioactive and hazardous components, as defined by the Atomic Energy Act [Public Law, 1954] and the Resource Conservation and Recovery Act [RCRA] as codified in Title 40 Code of Federal Regulations [CFR] Part 261.3 [EPA, 1980]).

The Waste Isolation Pilot Plant (WIPP) is a TRU waste management facility operated by the U.S. Department of Energy (DOE). The WIPP is currently identified as the permanent disposal site for TRU wastes (in retrievable storage or projected) generated at various DOE sites from defense-related activities of the United States government. The WIPP is scheduled to receive and dispose of TRU defense wastes from 8 major and additional minor DOE TRU waste sites (see Figure 1-1).

The DOE is committed to demonstrating compliance with all applicable regulations prior to permanent disposal of TRU defense wastes in the WIPP repository. These regulations are the environmental standards for management and disposal of TRU defense wastes as mandated in 40 CFR Part 191 (EPA, 1993) and Part 194 (EPA, 1996), and the RCRA regulations. Compliance demonstration through Sandia National Laboratories/New Mexico (SNL/NM) performance assessment (PA) calculations will be based on the inventory of existing and currently projected waste streams compiled in this document and the *Transuranic Waste Baseline Inventory Report* (TWBIR) Revision 2, as reported by the DOE TRU waste sites. Revision 3 of the TWBIR is different from previous revisions in that it provides the TRU waste inventory information developed for Revision 2 along with supplemental data. It is necessary for the reader to be familiar with Revision 2 of the TWBIR (DOE, 1995c) to understand TWBIR Revision 3.

1.2 PURPOSE

The purpose of the TWBIR is to document the total inventory of DOE TRU waste as defined by the DOE TRU waste sites. This document is based on the TWBIR Revision 2 data supplemented by several memoranda prepared during early calendar year (CY) 1996 that summarize additional data requested by the U. S. Department of Energy/Carlsbad Area Office (DOE/CAO) to support the SNL/NM PA modeling. The primary purpose of this document is to provide the summary data from TWBIR Revision 2 and the supplemental information used by SNL/NM for the development of the Compliance Certification Application (CCA) to be delivered to the Environmental Protection Agency (EPA), and to support the Land Withdrawal Act (LWA) (Public Law, 1992b). The supplemental information was generated from specific data requests

1 - 1



Figure 1-1. U.S. DOE Transuranic Waste Sites*

"This fight is identical to Figure 1-1 in TWBIR Revision 2, page 1-2 (DOE, 1995c).

1 - 2

DOE/CAO-95--1121, Rev. 3 June 1996 to the TRU waste sites since the publication of Revision 2.

Revision 2 of the TWBIR included both the TRU waste that is allowed to be disposed of in WIPP and the DOE TRU waste that is not currently allowed to be disposed of in WIPP (Public Law, 1992b). Because the primary purpose of this Revision 3 TWBIR is to support the CCA and PA, it includes only the DOE TRU waste that is currently identified by the sites as being allowed to be disposed of in WIPP.

The TWBIR has been developed from the best available information and acceptable knowledge provided by the DOE TRU waste sites. In support of PA, the TWBIR describes a process for grouping individual waste streams with similar physical and chemical properties into waste profiles, based on their waste matrix code (WMC) (DOE, 1995a) assigned by the DOE TRU waste sites. The individual waste streams are also evaluated to estimate the occurrence and quantities of nonradioactive waste material parameters (WMPs) listed in Table 1-1 (e.g., cellulosics, plastics, iron-base metal/alloys, etc.) that have been identified by SNL/NM as being potentially important to the performance of the WIPP repository. Waste profiles with similar WMCs are then combined across the DOE TRU waste system to provide estimated total volumes and total WMPs.

1.3 WASTE INVENTORY TERMINOLOGY

All terminology in this document is unchanged from the TWBIR Revision 2. A summary of terminology used in this document is provided in this section and in Chapter 5 (Glossary). A list of acronyms and abbreviations used are provided in the front of the document.

Stored Inventory – The part of the TRU inventory currently in retrievable storage at the time of the TWBIR Revision 2 data call for inventory information is known as "stored inventory" in this document. Retrievably stored waste includes waste stored since approximately 1970 in buildings or in berms with earthen cover and does not include any waste that was buried (predominately prior to 1970) (DOE, 1990).

As-Generated Waste – The chemical and physical status of waste when it is generated. The "as-generated" term applies to both stored and projected waste.

Projected Inventory – The part of the TRU waste inventory that has not been generated but is currently estimated to be generated at some time in the future by the TRU waste sites is known as "projected inventory."

Anticipated Inventory – For the TWBIR, this is the sum of the stored and projected inventories, calculated as:

| Stored | L | Projected | - | Anticipated |
|-----------|---|-----------|---|-------------|
| Inventory | Ŧ | inventory | - | Inventory |

Scaling – The process for adjusting, if needed, the projected inventory to the design limit (disposal inventory) of the WIPP repository is called "scaling."

Stored Inventory + Projected Inventory (scaled as needed) = Disposal Inventory

| | Input Variable in <u>Current</u> PA Models | | | |
|------------------------------------|--|-------------------------------|--|--|
| Waste Material Parameter | Gas Generation | Mechanical Characteristics | | |
| Iron-base metal/alloys | YES | YES | | |
| Aluminum-base metal/alloys | - | YES | | |
| Other metal/alloys | - | YES | | |
| Other inorganic materials | - | YES | | |
| Vitrified ¹ | - | YES | | |
| Cellulosics | YES | YES | | |
| Rubber | YES ² | YES | | |
| Plastics | YES ² | YES | | |
| Solidified inorganic material | - | YES | | |
| Solidified organic material | - | YES | | |
| Cement (solidified) ^{3,4} | YES | - | | |
| Soils⁵ | - | YES | | |

TABLE 1-1. TECHNICAL DATA NEEDS FOR PERFORMANCE ASSESSMENT WASTE MATERIAL PARAMETERS

¹ Waste material parameter corresponding to treatment, identified by some sites that plan to treat waste in the future.

² Only one-half of materials assumed to generate gas.

³ Percentage of material to generate gas is unknown at the present time.

⁴ Information on this waste material parameter is needed for non-PA scoping calculations for assessment of its importance.

⁵ May impact colloids.

Disposal Inventory – The inventory volume defined for WIPP emplacement to be used for PA calculations is the "disposal inventory." The LWA defines the total amount of TRU waste allowed for disposal in the WIPP as approximately 175,600 cubic meters (6,200,000 cubic feet) (Public Law, 1992b). The "Agreement for Consultation and Cooperation" (C&C Agreement) limits the RH-TRU inventory to approximately 7,080 cubic meters (250,000 cubic feet) (DOE and State of New Mexico, 1981). Therefore by difference, the CH-TRU inventory will be limited to approximately 168,500 cubic meters (5,950,000 cubic feet) if all of the RH-TRU allowance is filled.

Waste Matrix Code (WMC) - The WMCs were developed by DOE in response to the Federal Facilities Compliance Act (FFCAct) (Public Law, 1992a) as a methodology to aid in categorizing mixed waste streams in the DOE system into a series of five-digit alphanumeric codes (e.g., S3100; Inorganic Process Residues) that represent different physical/chemical matrices (DOE, 1995a).

Final Waste Form – Final waste form of a waste stream refers to the expected physical and chemical form of that stream once the waste has been processed, treated, or repackaged (if necessary) and is ready for disposal. This consists of a series of WMCs that are grouped together. The use of the final waste form helps to group waste streams that are expected to have similar physical and chemical properties at the time of disposal. The final waste form applies to both stored and projected waste. An example of combining three waste streams which either contain particulates or are cemented particulate waste is presented below:

WMC S3100 (inorganic process residues) WMC S3110 (inorganic particulates) WMC S3150 (solidified process residues)

Solidified Inorganics

Particulate waste may be immobilized prior to shipment to WIPP. If so, all three of these waste streams would be the same basic waste form when emplaced in WIPP and have similar physical and chemical properties. The final waste form for this example is solidified inorganics. Table 1-2 presents all anticipated WMCs for TRU waste and indicates the final waste form typically assigned to each WMC for the TWBIR. There are 11 final waste forms used in this TWBIR. The last two rows in Table 1-2, Excluded and Unknown Waste Streams, group WMCs that will not be accepted at WIPP until additional characterization and/or processing occurs to meet the WIPP Waste Acceptance Criteria (WAC) (DOE, 1996).

Waste Material Parameter – This is one (or more) nonradioactive waste constituent(s) that occurs in a TRU waste stream that is an input parameter into one or more PA models or is required to adequately describe the waste form. The waste material parameters and additional packaging materials that are reported in weight/volume (kg/m³) and included in the TWBIR are:

WASTE MATERIAL PARAMETERS

- Iron-base metal/alloys
- Aluminum-base metal/alloys
- Other metal/alloys
- Other inorganic materials
- Vitrified
- Cellulosics
- Rubber
- Plastics
- Solidified inorganic material
- Solidified organic material
- Cement (solidified)
- Soils

WIPP Waste Profile – The WIPP waste profile represents a summary of TRU wastes at all DOE TRU waste sites that have an identical final waste form.

PACKAGING MATERIALS

- Steel
- Plastic
- Lead (for RH-TRU waste only)

1 - 5

| Final Waste Form | Waste Matrix Codes |
|--|--|
| Solidified Inorganics | L1000 ¹ , L1100 ¹ , L1110 ¹ , L1120 ¹ , L1130 ¹ , L1140 ¹ , L1190 ¹ , 1200 ¹ , L1210 ¹ , L1220 ¹ , L1230 ¹ , L1240 ¹ , L1290 ¹ , S3000 ² , S3100 ³ , S3110 ³ , S3111 ³ , S3112 ³ , S3113 ³ , S3115 ³ , S3118 ³ , S3119 ³ , S3120 ¹ , S3121 ¹ , S3122 ¹ , S3123 ¹ , S3124 ¹ , S3125 ¹ , S3129 ¹ , S3130 ¹ ^{\overline 3} , S3131 ^{1 \overline 3} , S3132 ^{1 \overline 3} , S3139 ^{1 \overline 3} , S3144 ³ , S3150, S3160 ³ , S3190 ^{1 \overline 3} , S3900 ² , X6000 ⁴ , X6200 ⁵ , X6300 ⁶ , X6400 ⁵ , X6900 ⁴ , X7300 ³ , X7500 ⁸ , X7510 ⁸ , X7520 ⁸ , X7530 ⁸ , X7590 ⁸ , L9000 ² , Z1110, Z1190 |
| Salt | S3000 ² , S3140, S3141, S3142, S3143, S3149, S3900 ² , L9000 ² |
| Solidified Organics | L2000 ¹ , L2100 ¹ , L2110 ¹ , L2120 ¹ , L2190 ¹ , L2200 ¹ , L2210 ¹ , L2220 ¹ , L2290 ¹ , L2900 ¹ , S3000 ² , S3114 ³ , S3200 ³ , S3210 ³ , S3211 ³ , S3212 ³ , S3219 ³ , S3220 ¹ , S3221 ¹ , S3222 ¹ , S3223 ¹ , S3229 ¹ , S3230 ³ , S3290 ^{1 or 3} , S3900 ² , S5340 ³ , X6000 ⁴ , X6100 ⁵ , X6190 ⁴ , X6900 ⁴ , L9000 ² , Z1110, Z1190 |
| Soils | S4000, S4100, S4200, S4300, S4900, |
| Uncategorized Metal (Metal Waste Other Than Lead and/or Cadmium) | S3116, S5000 ⁹ , S5100 ⁷ , S5110, S5111, S5119, S5190, X6200, X7000 ¹⁰ , X7290, X7400 ¹¹ , X7430, X7490 ¹¹ , X7520 ⁸ , Z1140, Z1190, Z2100 ¹⁰ |
| Lead/Cadmium Metal | S5000 ⁹ , S5100 ⁷ , S5110, S5112, S5113, S5119, S5190, X6220 ⁸ , X7000 ¹⁰ , X7200, X7210, X7211, X7212, X7219, X7220, X7290, X7400 ¹¹ , X7410 ¹¹ , X7420 ¹¹ , X7490 ¹¹ , Z2100 ¹⁰ |
| Inorganic Non-Metal | S3117, S3118, S3160, S5000 ⁹ , S5100 ⁷ , S5120, S5121, S5122, S5123, S5124, S5125, S5126, S5129, S5190, Z1120, Z1150, Z1190 |
| Combustible | S5000 ⁹ , S5300, S5310, S5311, S5312, S5313, S5319, S5320, S5330, S5390, Z1130, Z1190, Z1200 |
| Graphite | S5000°, S5126 |
| Heterogeneous | S5000 ⁹ , S5100 ⁹ , S5400, S5420, S5440, S5450, S5460, S5490, X7520 ⁸ , Z2900 |
| Filter | S5000 ⁹ , S5410 |
| Excluded Waste Streams ¹² | X7000, X7100, X7600, X7700 |
| Unknown ¹³ | S5190, X7900, L9000, S9000, Z9000, U9999 |

TABLE 1-2. WASTE MATRIX CODES AND THEIR ANTICIPATED FINAL WASTE FORM



TABLE 1-2. WASTE MATRIX CODES AND THEIR ANTICIPATED FINAL WASTE FORM (CONTINUED)

¹ Liquid waste streams are assumed to be solidified prior to being sent to WIPP.

² WMCs S3000, S3900, and L9000 are placed in "solidified inorganics," "salt," or "solidified organics," depending on the information provided by the TRU waste site.

³ Particulate waste streams are assumed to be solidified prior to being sent to WIPP.

⁴ WMCs X6000 and X6900 are placed in "solidified organics" or "solidified inorganics" depending on the information provided by the TRU waste site.

⁵ Liquid lab pack waste is assumed to be solidified prior to being sent to WIPP.

⁶ Solid lab packs are assumed to be solidified prior to being sent to WIPP.

⁷ WMC S5100 is placed in "uncategorized metal," "lead-cadmium metal," or "inorganic non-metal" depending on the information provided by the site.

⁸ Waste stream is assumed to be treated prior to being sent to WIPP.

⁹ WMC S5000 is placed in "uncategorized metal," "lead/cadmium metal," "inorganic non-metal," "combustible," "graphite," "heterogeneous," or "filter," depending on the information provided by the site.

¹⁰ WMC Z2100 is placed in "uncategorized metal" or "lead/cadmium metal" depending on the information provided by the site.

¹¹ WMCs X7400, X7410, X7420, and X7490 are assumed to be drained of liquid and contain only metal waste.

¹² These waste streams are excluded from disposal in WIPP at this time, e.g., PCB and elemental mercury.

¹³ If adequate information is provided by the TRU waste site, these WMCs are changed. If there is not enough information, these waste streams remain as "unknown" and are excluded from disposal in WIPP until characterized.



* _ 2

1.4 METHODOLOGY FOR DEVELOPMENT OF DISPOSAL INVENTORY

Development of the WIPP TRU waste disposal inventory is accomplished by a series of steps starting with the individual waste stream profiles submitted by the TRU waste sites. These waste stream profiles are grouped together, based on similar physical and chemical properties, into common "WIPP waste profiles," which should facilitate discussions with regulatory agencies and stakeholders concerning the disposal waste inventory. The process of grouping similar waste streams is exemplified in Figure 1-2. The waste profiles also contain information on waste material parameters that could affect the performance of the WIPP repository and that may be direct inputs to the PA models.

The CH-TRU anticipated inventory consists of up to 11 overall CH-TRU WIPP final waste forms based on the physical and chemical properties of the waste streams. Because the volume of the CH-TRU anticipated inventory is not sufficient to fill the maximum calculated CH-TRU capacity of WIPP, scaling of the projected CH-TRU inventory is necessary to attain the maximum calculated WIPP CH-TRU disposal inventory of approximately 168,500 cubic meters (5.95 million cubic feet). The scaling factor for CH-TRU waste is computed as follows:

The WIPP disposal inventory is the inventory to be used in PA calculations. To calculate the disposal inventory by final waste form for CH-TRU waste, the **projected inventory** is multiplied by the scaling factor, added to the stored inventory for each final waste form, and summed together.

The RH-TRU anticipated inventory is greater than the WIPP C&C Agreement limit (DOE and State of New Mexico, 1981) of approximately 7,080 cubic meters. DOE will abide by the WIPP C&C Agreement for RH-TRU waste volumes and the LWA, which limits the curies of RH-TRU waste allowed in WIPP to 5.1 million curies (Public Law, 1992b). As stated earlier, one purpose of the TWBIR is to report the DOE TRU inventory in such a way that it will facilitate performance assessment by SNL/NM and support development of compliance applications to the appropriate regulatory agencies. Since this is not a WIPP load management document, the RH-TRU inventory has not been scaled back in this document to the regulatory limit. The RH-TRU inventory for WIPP will be averaged across all RH-TRU waste sites and reported as kilograms/cubic meter for the waste material parameters and curies/cubic meter for radionuclides.

1.5 DOCUMENT ORGANIZATION

The TWBIR Revision 3 is organized into chapters of text, figures, tables, and supporting appendices. The contents of remaining chapters in this document are summarized below:

- Chapter 2 provides a summary of the WIPP disposal inventory information previously presented in TWBIR Revision 2.
- Chapter 3 presents supplementary disposal inventory information.
- Chapter 4 provides the document references.





"This figure is identical to Figure 2-3 of TWBIR Revision 2, page 2-10 (DOE, 1995c).

- · Chapter 5 provides a document glossary.
- Appendix A provides the SNL/NM memoranda requesting information to supplement the TWBIR Revision 2.
- Appendix B includes DOE and SNL/NM memoranda that provide information to supplement the TWBIR Revision 2.
- Appendix C provides the site-specific stored radionuclide inventories decayed to December 1995.
- Appendix D provides the correction received from SNL/NM for Cf-252 decayed inventory.

2. SUMMARY OF WIPP DISPOSAL INVENTORY INFORMATION

2.1 INTRODUCTION

The DOE TRU waste sites have assigned an overall final waste form to each waste stream based on the expected physical and chemical form of the waste after the sites process, treat, or repackage the waste (if necessary). Each site provides the stored and projected inventory for each waste stream. The TWBIR generates the WIPP TRU waste inventory by rolling-up the waste stream volumes that have the same final waste form within a site to generate site profiles (see TWBIR Revision 2 [DOE, 1995c] for waste stream and site-specific waste profiles). Then the site-level volumes with the same final waste form are rolled-up to generate the WIPP TRU waste inventory by final waste form (see TWBIR Revision 2 for detailed information on the roll-up methodology).

This chapter summarizes the WIPP-level information for the disposal inventory. The data provided in this chapter are identical to those provided in TWBIR Revision 2. These are the data used by SNL/NM in the WIPP performance assessment to demonstrate regulatory compliance. This chapter will include the following TWBIR Revision 2 information:

- WIPP disposal inventory volumes for each final waste form taken from Table 3-1 (unchanged) in Section 3.2 of TWBIR Revision 2.
- WIPP disposal inventory waste material parameters taken from Tables 3-2 and 3-3 (unchanged) in Section 3.3 of TWBIR Revision 2.
- Summary of WIPP anticipated inventory from each site taken from Tables 4-1 and 4-2 (unchanged) in Chapter 4 of TWBIR Revision 2.

2.2 WIPP DISPOSAL INVENTORY VOLUMES FOR EACH FINAL WASTE FORM

The disposal inventory is defined by the LWA (Public Law, 1992b) and the WIPP C&C Agreement (DOE and the State of New Mexico, 1981) as follows: the maximum allowable WIPP capacity is approximately 175,600 cubic meters, of which RH-TRU disposal inventory is limited to approximately 7,080 cubic meters resulting in a calculated CH-TRU disposal inventory limit of approximately 168,500 cubic meters.

Using volumes for all the retrievably stored and projected defense TRU waste streams (including the mixed and nonmixed TRU waste volumes) a disposal inventory of TRU waste has been developed using the methodology described in Chapter 3 of Revision 2 of the TWBIR. This inventory is presented in Table 2-1 (by final waste forms) and depicts both the anticipated and disposal inventory volumes.

The anticipated CH-TRU inventory volumes are the sum of the stored and projected volumes. Scaling of the disposal inventory is for PA purposes to enable SNL/NM to model a capacity waste load based on currently anticipated profiles.

2 - 1

| Final Waste Forms | Stored Volumes | Projected Volumes | Anticipated Volumes | WIPP Disposal Volumes |
|--------------------------|-------------------|----------------------|------------------------|--------------------------|
| Combustible | 5,8E+03 | 4.6E+03 | 1.0E+04 | 1.4 E+0 4 |
| Filter | 2.2E+02 | 5.1E+02 | 7.3E+02 | 1.2E+03 |
| Graphite | 5.1E+02 | 4.8E+01 | 5.6E+02 | 6.0E+02 |
| Heterogeneous | 2.7E+04 | 1.3E+04 | 4.0E+04 | 5.1E+04 |
| Inorganic Non-Metal | 3.1E+03 | 9.4 E+0 2 | 4.1E+03 | 4.9E+03 |
| Lead/Cadmium Metal Waste | 3.5E+01 | 3. 3E+02 | 3.7 E+02 | 6.6E+02 |
| Sait Waste | 2.1E+01 | 3.3E+02 | 3. 5E+02 | 6.4E+02 |
| Soils | 4.1E+02 | 6.0E+03 | 6.4E+03 | 1.2E+04 |
| Solidified Inorganics | 9.6E+03 | 4.5E+03 | 1.4 E+0 4 | 1.8 E+04 |
| Solidified Organics | 9.1E+02 | 7.5E+01 | 9.8E+02 | 1. 1E+03 |
| Uncategorized Metal | 1.1E+04 | 2.3E+04 | 3.4E+04 | 5.4E+04 |
| Total CH Volumes | 5.8E+04 | 5.4E+04 | 1.1E+05 | 1.6E+05 |
| Remote Handled Waste | | | | |
| Combustible | 3.6E+01 | 4.9E+01 | 8.5E+01 | |
| Heterogeneous | 2.3E+03 | 5.5E+03 | 7.8E+03 | |
| Inorganic Non-Metal | 4.6E+01 | 2.1E+01 | 6.8E+01 | |
| Lead/Cadmium Metal Waste | 7.1E+00 | 6.7E+01 | 7.4E+01 | |
| Solidified Inorganics | 1.1E+03 | 2.3E+02 | 1.3E+03 | |
| Solidified Organics | 3.6E+00 | 0.0E+00 | 3.6E+00 | |
| Uncategorized Metal | 1.2E+02 | 1.7E+04 | 1.8E+04 | |
| Total RH Volumes | 3.6E+03 | 2.3E+04 | 2.7E+04 | |
| Total TRU Waste Volumes | 6.2E+04 | 7.7E+04 | 1.4 E+05 | 1.7E+05 |

TABLE 2-1. TRANSURANIC WASTE DISPOSAL INVENTORY FOR WIPP* Contact Handled Waste (Cubic Meters)

*This table is identical to Table 3-1 of TWBIR Revision 2, page 3-2 (DOE, 1995c).



Applying the formula given in Chapter 1:

-

| • | 1.685 x 10⁵m³ | 5.8 x 10 ⁴ m ³ | |
|---|--|--------------------------------------|------------------|
| | (CH-TRU disposal inver | ntory) – (stored inventory) | _ ≈ 2.05 |
| | 5.4 x 10 ⁴ m ³ (pr | ojected inventory) | (scaling factor) |

 Multiply the CH-TRU waste projected inventory volumes by the scaling factor for all the final waste forms, and add the stored volumes (which results in the numbers in the "Disposal Inventory" column of Table 2-1).

The CH-TRU waste stream volume on a system-wide final waste form basis is increased by approximately 50 percent to account for the difference between the anticipated inventory and the maximum calculated WIPP CH-TRU disposal inventory.

The RH-TRU WIPP inventory has not been scaled. The RH-TRU anticipated inventory is greater than the amount of RH-TRU waste allowed in the WIPP by the C&C Agreement (DOE and the State of New Mexico, 1981). DOE is committed to abide by all agreements and laws regarding RH-TRU limitations. DOE and SNL/NM will evaluate this inventory to determine the disposal options for all DOE RH-TRU waste. This inventory has not been scaled back to the limit imposed by the C&C Agreement so that all available data are presented to DOE and SNL/NM to conduct modeling and other evaluations to determine the disposition of this waste.

2.3 ROLL-UP OF WIPP WASTE MATERIAL PARAMETERS BY FINAL WASTE FORM

The roll-ups of waste material parameters by final waste forms are developed from the volumes presented in the TWBIR Revision 2. The roll-ups by final waste forms require combining data from several waste streams. A weighted average value for the waste material parameters is calculated from the average densities provided by the TRU waste sites modified by the volume fractions and summed as follows:



*where i is an index representing individual waste streams of the same final waste form

The minimum density is chosen as the smallest minimum density of a particular waste material parameter in the TWBIR Revision 2. The maximum density is chosen in a similar manner, except that the largest maximum density is chosen. Thus, the maximum and minimum values reported in Tables 2-2 and 2-3 are the absolute extreme values reported across the system, and in many cases they only apply to a very small volume of waste. If required, the user can use the data in the TWBIR Revision 2 database to calculate a "weighted average maximum" value to obtain a maximum value that may be more representative of the total inventory.

2 - 3

The waste material parameters that are inputs to the PA models are presented in Table 2-2 for CH-TRU waste and Table 2-3 for RH-TRU waste. These tables represent the waste material parameters for the WIPP disposal inventory.

2.4 SUMMARY OF WIPP ANTICIPATED INVENTORY FROM EACH SITE

Each WIPP waste stream from each TRU waste site is characterized in a waste stream profile in TWBIR Revision 2. Summary tables of CH-TRU and RH-TRU WIPP waste volumes by site are provided in Tables 2-4 and 2-5.



Table 2-2. WIPP CH-TRU Waste Material Parameter Disposal Inventory*

| Waste Material Parameters (Kg/m3) | Maximum | Average | <u>Minimum</u> |
|-----------------------------------|-----------------|---------|----------------|
| Iron Base Metal/Alloys | 2.65+03 | 1.7E+02 | 0.0E+00 |
| Aluminum Base Metal/Alloys | 8.0E+02 | 1.85+01 | 0.0E+00 |
| Other Metal/Alloys | 1.6E+03 | 6.7E+01 | 0.0E+00 |
| Other Inorganic Materials | 1.4E+03 | 3.1E+01 | 0.0E+00 |
| Vitrified | 2.5E+03 | 5.5E+01 | 0.0E+00 |
| Cellulosics | 9.6E+02 | 5.4E+01 | 0.0E+00 |
| Rubber | 6.3E+02 | 1.0E+01 | 0.0E+00 |
| Plastics | 8.9E+02 | 3.4E+01 | 0.0E+00 |
| Solidified Inorganic Material | 2.2E+03 | 5.4E+01 | 0.0E+00 |
| Solidified Organic Material | 1.4E+03 | 5.6E+00 | 0.0E+00 |
| Cement (Solidified) | 1.2E+03 | 5.0E+01 | 0.0E+00 |
| Soils | 1. 5E+03 | 4,4E+01 | 0.0E+00 |
| Container Materials - Kg/m3 | | | |
| Steel | | 139 | |
| Plastic/ Liners | | 25 | |

*This table is identical to Table 3-2 in TWBIR Revision 2, page 3-4 (DOE, 1995c).

Table 2-3. WIPP RH-TRU Waste Material Parameter Disposal Inventory*

| Waste Material Parameters (Kg/m3 | Maximum | Average | <u>Minimum</u> |
|----------------------------------|---------|---------|-----------------|
| Iron Base Metal/Alloys | 1.7E+03 | 1.0E+02 | 0.0E+00 |
| Aluminum Base Metal/Alloys | 1.7E+02 | 7.1E+00 | 0. 0E+00 |
| Other Metal/Alloys | 9.1E+02 | 2.5E+02 | 0.0E+00 |
| Other Inorganic Materials | 2.0E+03 | 6.4E+01 | 0.0E+00 |
| Vitrified | 2.5E+03 | 4.7E+00 | 0.0E+00 |
| Cellulosics | 5.7E+02 | 1.7E+01 | 0.0E+00 |
| Rubber | 4.4E+02 | 3.3E+00 | 0.0E+00 |
| Plastics | 6.2E+02 | 1.5E+01 | 0.0E+00 |
| Solidified Inorganic Material | 6.1E+02 | 2.2E+01 | 0.0E+00 |
| Solidified Organic Material | 8.1E+02 | 9.3E-01 | 0.0E+00 |
| Cement (Solidified) | 5.8E+02 | 1.9E+01 | 0. 0E+00 |
| Soils | 2.4E+01 | 1.0E+00 | 0.0E+00 |
| Container Materials - Kg/m3 | | | |
| Steel | | 445 | |
| | | | |

| Plastic/Liners | 3.1 |
|----------------|------|
| Lead | 485 |
| Steel Plug | 2145 |

*This table is identical to Table 3-3 in TWBIR Revision 2, page 3-5 (DOE, 1995c).



| Table 2-4. W | IPP CH-TRU | Waste Anticipated | Inventory By Site* |
|--------------|------------|-------------------|--------------------|
|--------------|------------|-------------------|--------------------|

| | (| | |
|---|-------------------|----------------------|------------------------|
| Storage/Generator Site | Stored Volumes | Projected Volumes | Anticipated Volumes |
| Ames Laboratory - Iowa State Univ. | 0.0E+00 | 4.2E-01 | 4.2E-01 |
| Argonne National Laboratory - East | 1.1E+01 | 1.3 E+02 | 1.4E+02 |
| Argonne National Laboratory - West | 6.5E+00 | 7.4E+02 | 7.5E+02 |
| Bettis Atomic Power Laboratory | 0.0E+00 | 1.2E+02 | 1.2E+02 |
| Energy Technology Engineering Center | 1.7E+00 | 0. 0E+00 | 1.7E+00 |
| Hanford (Richland) Site | 1.2E+04 | 3.3E+04 | 4.6E+04 |
| Idaho National Engineering Laboratory | 2.9E+04 | 0.0E+00 | 2.9E+04 |
| Lawrence Livermore National Laboratory | 2.3E+02 | 7.1E+02 | 9.4E+02 |
| Los Alamos National Laboratory | 1.1E+04 | 7.4E+03 | 1.8E+04 |
| Mound Plant | 2.7E+02 | 0.0E+00 | 2.7E+02 |
| Nevada Test Site | 6.2E+02 | 9.0E+00 | 6.3E+02 |
| Oak Ridge National Laboratory | 1.3E+03 | 2.6E+02 | 1.6E+03 |
| Paducah Gaseous Diffusion Plant | 0.0E+00 | 1.9E+00 | 1.9E+00 |
| Pantex Plant | 6.2E-01 | 0.0E+00 | 6.2E-01 |
| Rocky Flats Environmental Technology Site | 7.1E+02 | 4.4E+03 | 5.1E+03 |
| Sandia National Laboratory - Albuquerque | 6.7E+00 | 7.5E+00 | 1.4E+01 |
| Savannah River Site | 2.9E+03 | 6.8E+03 | 9.6E+03 |
| Teledyne Brown Engineering | 2.1E-01 | 0.0E+00 | 2.1 E-01 |
| U.S. Army Material Command | 2.5E+00 | 0.0E+00 | 2.5E+00 |
| University of Missouri Research Reactor | 2.1E-01 | 8.3E-01 | 1.0 E+00 |
| Total CH Volumes | 5.8E+04 | 5.4E+04 | 1.1E+05 |

*This table is identical to Table 4-1 in TWBIR Revision 2, page 4-2 (DOE, 1995c).

Ŋ

2 - 7

| Storage/Generator Site | Stored Volumes | Projected Volumes | Anticipated Volumes |
|---------------------------------------|-------------------|----------------------|------------------------|
| Argonne National Laboratory - West | 1.9E+01 | 1.3E+03 | 1.3E+03 |
| Battelle Columbus Laboratories | 5.8E+02 | 0.0E+00 | 5.8E+02 |
| Bettis Atomic Power Laboratory | 0.0E+00 | 6.7E+00 | 6.7E+00 |
| Energy Technology Engineering Center | 8.9E-01 | 0.0E+00 | 8.9E-01 |
| Hanford (Richland) Site | 2.0E+02 | 2.2E+04 | 2.2E+04 |
| Idaho National Engineering Laboratory | 2.2E+02 | 0.0E+00 | 2.2E+02 |
| Los Alamos National Laboratory | 9.4E+01 | 9.9E+01 | 1.9 E+02 |
| Oak Ridge National Laboratory | 2.5E+03 | 4.5E+02 | 2.9E+03 |
| Total RH Volumes | 3.6E+03 | 2.3E+04 | 2.7E+04 |
| Total TRU Waste Volumes | 6.2E+04 | 7.7E+04 | 1.4 E+05 |

Table 2-5. WIPP RH-TRU Waste Anticipated Inventory By Site*

*This table is identical to Table 4-2 in TWBIR Revision 2, page 4-3 (DOE, 1995c).

3. SUPPLEMENTAL DISPOSAL INVENTORY INFORMATION

3.1 INTRODUCTION

This chapter summarizes supplemental information about the WIPP disposal inventory that was requested by SNL/NM in support of WIPP PA either after the publication of Revision 2 of the TWBIR or that was not available from the TRU waste sites at the time of publication of Revision 2 of the TWBIR in December 1995 (DOE, 1995c). Appendices A-1 through A-3 are the three memoranda from SNL/NM requesting supplemental information about the WIPP TRU waste inventory.

The first memorandum from SNL/NM (dated November 6, 1995), entitled "CH and RH-TRU Waste Parameters Potentially Important in WIPP PA" (Appendix A-1), was included as Appendix B in Revision 2 of the TWBIR. This memorandum requested information on certain nonradioactive materials present in the TRU waste (nitrates, sulfates, phosphates, cement, and organic ligands), and also requested information on residues present at TRU waste sites other than Rocky Flats Environmental Technology Site (RFETS). The information on residues was provided in Revision 2 of the TWBIR. However, the remainder of the requested information had to be obtained from the sites after the publication of Revision 2 and is presented in this document.

The second and the third memoranda from SNL/NM (dated January 11 and January 30, 1996), both entitled "Information Needed from TWBIR (Revision 2/Addendum)" (Appendices A-2 and A-3), requested additional information about the WIPP disposal radionuclide inventory. This information is also presented in the main body of this document.

The supplemental information provided to SNL/NM in response to the memoranda referenced above is discussed in the following sections:

- Supplemental Radionuclide Information (Section 3.2)
- Supplemental Information for Other Constituents (Section 3.3)

3.2 SUPPLEMENTAL RADIONUCLIDE INFORMATION

In response to the memoranda requesting radionuclide information (Appendices A-2 and A-3), two sets of radionuclide information were provided in support of WIPP PA (Appendices B-1 and B-2). Appendix B-1 is an update of the WIPP disposal radionuclide inventory presented in Table 3-4 of Revision 2 of the TWBIR, while Appendix B-2 presents preliminary activity calculations for seven radionuclides on a waste stream basis. The memoranda reporting these supplemental data and the details of the methodology for calculations are included in Appendices B-1 and B-2. A summary of the information provided by DOE to SNL/NM and the major assumptions used in deriving portions of the data are presented in Sections 3.2.1 and 3.2.2.

3.2.1 Revised WIPP Disposal Radionuclide Inventory

A revised estimate of the WIPP disposal radionuclide inventory (i.e., Table 3-4 in TWBIR Revision 2) was not specifically requested by SNL/NM in the memoranda included in Appendices A-1 through A-3. However, after the publication of TWBIR Revision 2, new and updated

radionuclide information became available from four sites (Hanford, Oak Ridge National Laboratory [ORNL], RFETS, and Savannah River Site [SRS]). A review of the new information indicated that it may result in considerable changes to the WIPP disposal radionuclide inventory published in Revision 2 of the TWBIR. Therefore, the disposal radionuclide inventory was recalculated on the basis of the new information and the results provided to SNL/NM in a format identical to Table 3-4 in TWBIR Revision 2 (see Table 3-1). The methodology and the assumptions used for recalculation of the radionuclide inventory are identical to those described in TWBIR Revision 2, except that the new radionuclide information from the four sites was incorporated. The new information from the four sites is summarized below:

- Hanford Site reported corrections to the values for Cf-252, Cm-244, and Cm-245 from their earlier submittals for the Integrated Data Base (IDB) (DOE, 1995b).
- Preliminary sludge sampling data were obtained for the ORNL RH-TRU sludges, which showed that the primary uranium isotope present in these sludges is U-238 (not U-235, as reported in their previous IDB submittals). The uranium curies reported for RH-TRU waste in previous ORNL IDB submittals were redistributed based on the preliminary sludge sampling data. This corrected the previously high estimates of U-235 in the ORNL RH-TRU inventory.
- The RFETS provided undecayed yearly activity data for the radionuclides present in the RFETS residues, which enabled activity decay calculations for these radionuclides. This was not provided for in TWBIR Revision 2; therefore the radionuclide activity from these residues could not be decayed.
- The SRS provided a break-up of radionuclide activity data for SRS waste between onsite and off-site waste (i.e., waste from other sites that was shipped to SRS for storage in the early 1970s). The activity from the off-site waste was included in the WIPP disposal radionuclide inventory but excluded from any extrapolations for SRS projected waste under the assumption that there would be no future accumulation of off-site Pu-238 dominant waste at SRS.

Based on the above information, Table 3-1 provides the revised WIPP disposal radionuclide inventory estimated in curies per cubic meter and total curies for each radionuclide for both CH-TRU and RH-TRU waste. The revised stored radionuclide inventory for each site in decayed curies is provided in Appendix C for both CH-TRU and RH-TRU waste. Appendix C includes the effect of all corrections, additions, or revisions to the site radionuclide inventories used to develop Table 3-1 and is an update of Appendix D in TWBIR Revision 2. All numbers in Appendix C are decayed to December 1995 using the Oak Ridge Isotope Generation and Depletion Code (ORIGEN 2) (Croff, 1980; 1983).

Based on the total curies shown in Table 3-1, it is estimated that approximately 98.9 percent of the total CH-TRU curies is contributed by Pu-238, Pu-239, Pu-240, Pu-241, and Am-241. In contrast, approximately 96.5 percent of the total RH-TRU curies is contributed by Cs-137, Sr-90, Ba-137m, Pu-241, and Y-90. Thus, the remaining radionuclides contribute a very small fraction of the total curies for the repository.

In comparison to TWBIR Revision 2, the most significant change in the revised disposal radionuclide inventory shown in Table 3-1 is the *decrease* in the estimated concentration of Pu-



3 - 2
| Nuclide | CH-TRU Waste (Ci/m ³) | RH-TRU Waste (Ci/m³) | CH-TRU Waste (Total Curies ²) | RH-TRU Waste (Total Curies ²) |
|---------|--------------------------------------|-------------------------|--|--|
| Ac225 | 1.71E-05 | 1.66E-05 | 2.88E+00 | 1.17E-01 |
| Ac227 | 3.61E-06 | 1.07E-07 | 6.08E-01 | 7.57E-04 |
| Ac228 | 4.43E-06 | 1.10E-05 | 7.47E-01 | 7.77E-02 |
| Ag109m | 9.32E-05 | NR | 1.57E+01 | NR |
| Ag110 | 4.19E-14 | 2.46E-13 | 7.07E-09 | 1.74E-09 |
| Ag110m | 3.15E-12 | 1.85E-11 | 5.31E-07 | 1.31E-07 |
| Am241 | 2.62E+00 | 8.42E-01 | 4.42E+05 | 5.96E+03 |
| Am242 | 1.04E-05 | NR | 1.75E+00 | NR . |
| Am242m | 1.04E-05 | NR | 1.75E+00 | NR |
| Am243 | 1.93E-04 | 3.23E-08 | 3.26E+01 | 2.28E-04 |
| Am245 | 7.89E-15 | 4.06E-20 | 1.33E-09 | 2.87E-16 |
| At217 | 1.71E-05 | 1.66E-05 | 2.88E+00 | 1.17E-01 |
| Ba137m | 4.53E-02 | 2.89E+01 | 7.63E+03 | 2.04E+05 |
| Bi210 | 1.52E-05 | 1.01E-09 | 2.55E+00 | 7.16E-06 |
| Bi211 | 3.61E-06 | 1.07E-07 | 6.09E-01 | 7.58E-04 |
| Bi212 | 1.61E-04 | 1.04E-05 | 2.71E+01 | 7.36E-02 |
| Bi213 | 1.71E-05 | 1.66E-05 | 2.88E+00 | 1.17E-01 |
| Bi214 | 6.91E-05 | 5.05E-09 | 1.16E+01 | 3.58E-05 |
| Bk249 | 5.44E-10 | 2.80E-15 | 9.16E-05 | 1.98E-11 |
| Bk250 | 2.59E-16 | NR | 4.37E-11 | NR |

Table 3-1. WIPP DISPOSAL RADIONUCLIDE INVENTORY FOR THE CCA1+

NR = Not reported by sites.

¹Decayed to December 1995.

²Total curies estimated by assuming a volume of 168,500 cubic meters for CH-TRU waste and 7,080 cubic meters for RH-TRU waste.

³The values for Cf252 and Cm248 are different from the values reported in Attachment A in Appendix B-1 because these incorporate the corrections received from SNL/NM for these isotopes (see Appendix D) after Appendix B-1 was finalized.

*This table is an update of Table 3-4, of TWBIR Revision 2, pages 3-30 through 3-36 (DOE, 1995c)

| Nuclide | CH-TRU Waste (Ci/m ³) | RH-TRU Waste (Ci/m³) | CH-TRU Waste (Total Curies ²) | RH-TRU Waste (Total Curies ²) |
|--------------------|--------------------------------------|-------------------------|--|--|
| C14 | 6.43E-05 | 2.90E-04 | 1.08E+01 | 2.05E+00 |
| Cd109 | 9.31E-05 | NR | 1.57E+01 | NR |
| Cd113m | 1.08E-11 | 7.71E-11 | 1.82E-06 | 5.46E-07 |
| Ce144 | 3.71E-07 | 7.24E-04 | 6.26E-02 | 5.13E+00 |
| Cf249 | 3.81E-07 | 6.31E-07 | 6.42E-02 | 4.47E-03 |
| Cf250 | 1.96E-06 | NR | 3.30E-01 | NR |
| Cf251 | 2.24E-08 | NR | 3.78E-03 | NR |
| Cf252 ³ | 1.44E-05 | 1.82E-04 | 2.43E+00 | 1.29E+00 |
| Cm242 | 6.76E-06 | NR | 1.14E+00 | NR |
| Cm243 | 1.61E-05 | 6.99E-03 | 2.72E+00 | 4.95E+01 |
| Cm244 | 1.87E-01 | 4.45E-02 | 3.15E+04 | 3.15E+02 |
| Cm245 | 6.81E-08 | 2.07E-10 | 1.15E-02 | 1.46E-06 |
| Cm246 | 6.06E-07 | NR | 1.02E-01 | NR |
| Cm247 | 1.91E-14 | NR | 3.21E-09 | NR |
| Cm248 ³ | 2.19E-07 | 2.89E-08 | 3.69E-02 | 2.05E-04 |
| Co58 | 1.81E-18 | 1.75E-15 | 3.05E-13 | 1.24E-11 |
| Co60 | 3.83E-04 | 1.47E+00 | 6.46E+01 | 1.04E+04 |
| Cr51 | NR | 4.29E-10 | NR | 3.04E-06 |
| Cs134 | 7.97E-08 | 2.60E-03 | 1.34E-02 | 1.84E+01 |
| Cs135 | 2.98E-09 | 1.66E-08 | 5.02E-04 | 1.17E-04 |

NR = Not reported by sites.

¹Decayed to December 1995.

²Total curies estimated by assuming a volume of 168,500 cubic meters for CH-TRU waste and 7,080 cubic meters for RH-TRU waste.

³The values for Cf252 and Cm248 are different from the values reported in Attachment A in Appendix B-1 because these incorporate the corrections received from SNL/NM for these isotopes (see Appendix D) after Appendix B-1 was finalized.

* This table is an update of Table 3-4, of TWBIR Revision 2, pages 3-30 through 3-36 (DOE, 1995c)

3 - 4

| Nuclide | CH-TRU Waste (Ci/m ³) | RH-TRU Waste (Ci/m ³) | CH-TRU Waste (Total Curies ²) | RH-TRU Waste (Total Curies ²) |
|---------|--------------------------------------|--------------------------------------|--|--|
| Cs137 | 4.78E-02 | 3.05E+01 | 8.06E+03 | 2.16E+05 |
| Es254 | 2.51E-16 | NR | 4.24E-11 | NR |
| Eu150 | 2.08E-10 | NR | 3.51E-05 | NR |
| Eu152 | 7.46E-06 | 1.73E-01 | 1.26E+00 | 1.22E+03 |
| Eu154 | 6.80E-06 | 8.34E-02 | 1.15E+00 | 5.91E+02 |
| Eu155 | 5.62E-06 | 1.67E-02 | 9.46E-01 | 1.18E+02 |
| Fe55 | 1.13E-10 | 2.38E-05 | 1.91E-05 | 1.69E-01 |
| Fe59 | 1.57E-12 | NR | 2.64E-07 | NR |
| Fr221 | 1.71E-05 | 1.66E-05 | 2.88E+00 | 1.17E-01 |
| Fr223 | 4.98E-08 | 1.48E-09 | 8.39E-03 | 1.04E-05 |
| нз | 5.16E-06 | 9.33E-06 | 8.69E-01 | 6.60E-02 |
| 1129 | 4.18E-12 | NR | 7.05E-07 | NR |
| Kr85 | 1.20E-06 | 2.37E-04 | 2.02E-01 | 1.68E+00 |
| Mn54 | 5.05E-09 | 3.32E-06 | 8.51E-04 | 2.35E-02 |
| Nb95 | 1.51E-14 | 9.45E-05 | 2.54E-09 | 6.69E-01 |
| 'Nb95m | 5.04E-17 | 3.17E-07 | 8.50E-12 | 2.24E-03 |
| Ni59 | 4.47E-08 | NR | 7.52E-03 | NR |
| ·Ni63 | 5.46E-06 | 1.40E-04 | 9.19E-01 | 9.88E-01 |
| Np237 | 3.33E-04 | 4.02E-04 | 5.61E+01 | 2.85E+00 |
| Np238 | 5.20E-08 | NR | 8.77E-03 | NR |

NR = Not reported by sites.

¹Decayed to December 1995.

²Total curies estimated by assuming a volume of 168,500 cubic meters for CH-TRU waste and 7,080 cubic meters for RH-TRU waste.

³The values for Cf252 and Cm248 are different from the values reported in Attachment A in Appendix B-1 because these incorporate the corrections received from SNL/NM for these isotopes (see Appendix D) after Appendix B-1 was finalized.

*This table is an update of Table 3-4, of TWBIR Revision 2, pages 3-30 through 3-36 (DOE, 1995c)

| Nuclide | CH-TRU Waste (Cì/m ³) | RH-TRU Waste (Ci/m ³) | CH-TRU Waste (Total Curies ²) | RH-TRU Waste (Total Curies ²) |
|---------|--------------------------------------|--------------------------------------|--|--|
| Np239 | 1.93E-04 | 3.23E-08 | 3.26E+01 | 2.28E-04 |
| Np240m | 8.91E-12 | 3.12E-15 | 1.50E-06 | 2.21E-11 |
| Pa231 | 2.67E-06 | 2.70E-07 | 4.51E-01 | 1.91E-03 |
| Pa233 | 3.33E-04 | 4.02E-04 | 5.61E+01 | 2.85E+00 |
| Pa234 | 3.05E-07 | 1.92E-06 | 5.14E-02 | 1.36E-02 |
| Pa234m | 2.35E-04 | 1.48E-03 | 3.96E+01 | 1.05E+01 |
| Рь209 | 1.71E-05 | 1.66E-05 | 2.88E+00 | 1.17E-01 |
| Рь210 | 1.52E-05 | 1.01E-09 | 2.55E+00 | 7.16E-06 |
| Pb211 | 3.61E-06 | 1.07E-07 | 6.09E-01 | 7.58E-04 |
| Pb212 | 1.61E-04 | 1.04E-05 | 2.71E+01 | 7.36E-02 |
| Pb214 | 6.91E-05 | 5.05E-09 | 1.16E+01 | 3.58E-05 |
| Pd107 | 4.40E-10 | 2.45E-09 | 7.41E-05 | 1.73E-05 |
| Pm147 | 4.67E-05 | .1.52E-03 | 7.87E+00 | 1.07E+01 |
| Po210 | 1.52E-05 | 1.01E-09 | 2.55E+00 | 7.16E-06 |
| Po211 | 1.01E-08 | 3.00E-10 | 1.71E-03 | 2.12E-06 |
| Po212 | 1.03E-04 | 6.66E-06 | 1.73E+01 | 4.72E-02 |
| Po213 | 1.67E-05 | 1.62E-05 | 2.82E+00 | 1.15E-01 |
| Po214 | 6.91E-05 | 5.05E-09 | 1.16E+01 | 3.57E-05 |
| Po215 | 3.61E-06 | 1.07E-07 | 6.09E-01 | 7.58E-04 |
| Po216 | 1.61E-04 | 1.04E-05 | 2.71E+01 | 7.36E-02 |

NR = Not reported by sites.

¹Decayed to December 1995.

²Total curies estimated by assuming a volume of 168,500 cubic meters for CH-TRU waste and 7,080 cubic meters for RH-TRU waste.

³The values for Cf252 and Cm248 are different from the values reported in Attachment A in Appendix B-1 because these incorporate the corrections received from SNL/NM for these isotopes (see Appendix D) after Appendix B-1 was finalized.

3 - 6

*This table is an update of Table 3-4, of TWBIR Revision 2, pages 3-30 through 3-36 (DOE, 1995c)



| Nuclide | CH-TRU Waste (Ci/m ³) | RH-TRU Waste (Ci/m³) | CH-TRU Waste (Total Curies ²) | RH-TRU Waste (Total Curies ²) |
|---------|--------------------------------------|-------------------------|--|--|
| Po218 | 6.91E-05 | 5.05E-09 | 1.16E+01 | 3.58E-05 |
| Pr144 | 3.67E-07 | 7.16E-04 | 6.18E-02 | 5.07E+00 |
| Pu236 | 6.16E-08 | NR | 1.04E-02 | NR |
| Pu238 | 1.55E+01 | 2.05E-01 | 2.61E+06 | 1.45E+03 |
| Pu239 | 4.66E+00 | 1.45E+00 | 7.85E+05 | 1.03E+04 |
| Pu240 | 1.25E+00 | 7.15E-01 | 2.10E+05 | 5.07E+03 |
| Pu241 | 1.37E+01 | 2.00E+01 | 2.31E+06 | 1.42E+05 |
| Pu242 | 6.96E-03 | 2.11E-05 | 1.17E+03 | 1.50E-01 |
| Pu243 | 1.91E-14 | NR | 3.21E-09 | NR |
| Pu244 | 8.92E-12 | 3.12E-15 | 1.50E-06 | 2.21E-11 |
| Ra223 | 3.61E-06 | 1.07E-07 | 6.09E-01 | 7.58E-04 |
| Ra224 | 1.61E-04 | 1.04E-05 | 2.71E+01 | 7.36E-02 |
| Ra225 | 1.71E-05 | 1.66E-05 | 2.88E+00 | 1.17E-01 |
| Ra226 | 6.91E-05 | 5.05E-09 | 1.16E+01 | 3.58E-05 |
| Ra228 | 4.43E-06 | 1.10E-05 | 7.47E-01 | 7.77E-02 |
| Rh106 | 1.72E-07 | 1.54E-03 | 2.90E-02 | 1.09E+01 |
| Rn219 | 3.61E-06 | 1.07E-07 | 6.09E-01 | 7.58E-04 |
| Rn220 | 1.61E-04 | 1.04E-05 | 2.71E+01 | 7.36E-02 |
| Rn222 | 6.91E-05 | 5.05E-09 | 1.16E+01 | 3.58E-05 |
| Ru106 | 1.72E-07 | 1.54E-03 | 2.90E-02 | 1.09E+01 |

NR = Not reported by sites.

¹Decayed to December 1995.

²Total curies estimated by assuming a volume of 168,500 cubic meters for CH-TRU waste and 7,080 cubic meters for RH-TRU waste.

³The values for Cf252 and Cm248 are different from the values reported in Attachment A in Appendix B-1 because these incorporate the corrections received from SNL/NM for these isotopes (see Appendix D) after Appendix B-1 was finalized.

*This table is an update of Table 3-4, of TWBIR Revision 2, pages 3-30 through 3-36 (DOE, 1995c)

| Nuclide | CH-TRU Waste (Ci/m ³) | RH-TRU Waste (Ci/m ³) | CH-TRU Waste (Total Curies ²) | RH-TRU Waste (Total Curies ²) |
|--------------------|--------------------------------------|--------------------------------------|--|--|
| Sb125 | 7.1 7E -07 | 2.67E-04 | 1.21E-01 | 1.89E+00 |
| Sb126 | 8.02E-10 | 4.46E-09 | 1.35E-04 | 3.16E-05 |
| Sb126m | 5.73E-09 | 3.18E-08 | 9.65E-04 | 2.25E-04 |
| Se79 | 2.58E-09 | 1.44E-08 | 4.35E-04 | 1.02E-04 |
| Sm151 | 8.72E-06 | 5.05E-05 | 1.47E+00 | 3.57E-01 |
| Sn119m | 2.46E-11 | 1.35E-10 | 4.14E-06 | 9.59E-07 |
| Sn121m | 1.58E-07 | 9.45E-07 | 2.66E-02 | 6.69E-03 |
| Sn126 | 5.73E-09 | 3.18E-08 | 9.65E-04 | 2.25E-04 |
| Sr90 | 4.07E-02 | 2.95E+01 | 6.85E+03 | 2.09E+05 |
| Ta182 | NR | 5.95E-12 | NR | 4.21E-08 |
| Tc99 | 1.49E-04 | 8.26E-07 | 2.52E+01 | 5.85E-03 |
| Te125m | 1.75E-07 | 6.57E-05 | 2.95E-02 | 4.65E-01 |
| Te127 | 7.72E-13 | 2.41E-13 | 1.30E-07 | 1.71E-09 |
| Te127m | 7.88E-13 | 2.47E-13 | 1.33E-07 | 1.75E-09 |
| Th227 | 3.56E-06 | 1.06E-07 | 6.01E-01 | 7.47E-04 |
| ⁻ Th228 | 1.61E-04 | 1.04E-05 | 2.71E+01 | 7.36E-02 |
| Th229 | 1.71E-05 | 1.66E-05 | 2.88E+00 | 1.17E-01 |
| Th230 | 4.78E-07 | 1.07E-06 | 8.06E-02 | 7.56E-03 |
| Th231 | 7.59E-05 | 6.53E-04 | 1.28E+01 | 4.63E+00 |
| Th232 | 5.42E-06 | 1.31E-05 | 9.13E-01 | 9.25E-02 |

NR = Not reported by sites.

¹Decayed to December 1995.

²Total curies estimated by assuming a volume of 168,500 cubic meters for CH-TRU waste and 7,080 cubic meters for RH-TRU waste.

³The values for Cf252 and Cm248 are different from the values reported in Attachment A in Appendix B-1 because these incorporate the corrections received from SNL/NM for these isotopes (see Appendix D) after Appendix B-1 was finalized.

* This table is an update of Table 3-4, of TWBIR Revision 2, pages 3-30 through 3-36 (DOE, 1995c)

| Nuclide | CH-TRU Waste (Ci/m ³) | RH-TRU Waste (Ci/m ³) | CH-TRU Waste (Total Curies ²) | RH-TRU Waste (Total Curies ²) |
|---------|--------------------------------------|--------------------------------------|--|--|
| Th234 | 2.35E-04 | 1.48E-03 | 3.96E+01 | 1.05E+01 |
| TI207 | 3.61E-06 | 1.07E-07 | 6.07E-01 | 7.56E-04 |
| T1208 | 5.77E-05 | 3.74E-06 | 9.73E+00 | 2.65E-02 |
| TI209 | 3.69E-07 | 3.58E-07 | 6.22E-02 | 2.53E-03 |
| U232 | 1.53E-04 | NR | 2.58E+01 | NR |
| U233 | 1.06E-02 | 2.23E-02 | 1.79E+03 | 1.58E+02 |
| U234 | 2.76E-03 | 6.03E-03 | 4.65E+02 | 4.27E+01 |
| U235 | 7.59E-05 | 6.53E-04 | 1.28E+01 | 4.63E+00 |
| U236 | 1.98E-06 | 1.37E-05 | 3.33E-01 | 9.68E-02 |
| U237 | 3.36E-04 | 4.91E-04 | 5.66E+01 | 3.48E+00 |
| U238 | 2.35E-04 | 1.48E-03 | 3.96E+01 | 1.05E+01 |
| U240 | 8.91E-12 | 3.12E-15 | 1.50E-06 | 2.21E-11 |
| Y90 | 4.07E-02 | 2.95E+01 | 6.85E+03 | 2.09E+05 |
| Zr93 | 3.34E-08 | 1.86E-07 | 5.63E-03 | 1.32E-03 |
| Zr95 | 6.80E-15 | 4.27E-05 | 1.15E-09 | 3.02E-01 |
| TOTALS | 3.81E+01 | 1.43E+02 | 6.42E+06 | 1.02E+06 |

NR = Not reported by sites.

¹Decayed to December 1995.

²Total curies estimated by assuming a volume of 168,500 cubic meters for CH-TRU waste and 7,080 cubic meters for RH-TRU waste.

³The values for Cf252 and Cm248 are different from the values reported in Attachment A in Appendix B-1 because these incorporate the corrections received from SNL/NM for these isotopes (see Appendix D) after Appendix B-1 was finalized.

* This table is an update of Table 3-4, of TWBIR Revision 2, pages 3-30 through 3-36 (DOE, 1995c)

238 for the CH-TRU waste in the repository. This is primarily due to the exclusion of the SRS off-site waste from any future extrapolations. Since this off-site waste has a high concentration of Pu-238, excluding it from the extrapolations decreases the amount of Pu-238 in the projected portion of the inventory. It should be noted that this off-site waste is included in the stored waste portion of the disposal radionuclide inventory. The decrease in the Pu-238 also causes a decrease in the total estimated curies for CH-TRU waste in the repository.

Based on the data corrections from Hanford Site to the Cm-244 and Cm-245 inventories, the estimated concentration of Cm-244 has increased, while that of Cm-245 has decreased. Similarly, based on the correction to the reported value of Cf-252 from the Hanford Site, the revised concentration of Cf-252 has decreased significantly from the values estimated in Revision 2 of the TWBIR. The effect of decaying the activity from the RFETS residues has resulted in a minor decrease in the estimated concentration of Pu-241. Since Pu-241 decays to Am-241, the decrease in the Pu-241 concentration is also accompanied by a corresponding increase in the concentration of Am-241.

The major change for the RH-TRU waste from TWBIR Revision 2 is the decrease in the estimated concentration of U-235 and an increase in the concentration of U-238. Both are a result of the preliminary sludge sampling data from ORNL mentioned earlier.

3.2.2 Activity Calculations for Waste Streams

As documented in the SNL/NM memoranda in Appendices A-2 and A-3, data on radionuclide activity on a waste stream basis was requested for 21 radionuclides. However, the request was subsequently limited to seven radionuclides by SNL/NM WIPP PA staff (Am-241, Cm-244, Pu-238, Pu-239, Pu-240, Pu-241, and U-234). Appendix B-2 presents the results provided to SNL/NM by DOE/CAO in response to this data request.

Since many sites did not have the ability to provide radionuclide data on a detailed waste stream basis for every waste stream in TWBIR Revision 2, the radionuclide activities for many individual waste streams (especially for projected waste) were not reported by the sites for TWBIR Revision 2. Therefore, the radionuclide activity data for the WIPP disposal inventory cannot be directly obtained on a waste stream basis by running queries on the TWBIR Revision 2 database. Due to the unavailability of detailed radionuclide data on a waste stream basis for many waste streams, the WIPP disposal radionuclide inventory presented in all revisions of the TWBIR has always been developed on the basis of the site-level radionuclide inventories reported by the sites in the IDB.

For the sake of consistency with the revised WIPP disposal radionuclide inventory in Attachment A of Appendix B-1 (which is also based on the site-level IDB data), assumptions were required in order to estimate the waste stream radionuclide activities presented in Appendix B-2. These assumptions can be found in Appendix B-2 and are not reproduced here. Thus, it should be noted that the data in Appendix B-2 are *derived* on the basis of assumptions and not directly obtainable from the TWBIR Revision 2 database. Because of the unavailability of the radionuclide data on a waste stream basis, some of the waste streams from small sites are not included in the activity table in Appendix B-2. Efforts are currently underway to ensure that the sites will be able to provide radionuclide data on a waste stream basis for most waste streams in future updates of the TWBIR so that radionuclide activity data for the WIPP disposal inventory can be directly obtained from the TWBIR database.



3.3 SUPPLEMENTAL INFORMATION FOR OTHER CONSTITUENTS

SNL/NM and DOE/CAO requested supplemental information on several constituents in TRU waste (see Appendix A-1) that were not able to be estimated based on the information reported by the TRU waste sites in Revision 2 of the TWBIR (DOE, 1995c). The information requested can be divided into three general categories which were requested on solidified waste forms destined for disposal in WIPP:

- Complexing Agents
- Nitrate, Sulfate, and Phosphate
- Cement

The TWBIR team worked with those major sites that generate/store most of the solidified waste forms: Los Alamos National Laboratory (LANL), RFETS/INEL, and ORNL. A summary of the results of these supplemental information requests is provided in Section 3.3.1, 3.3.2 and 3.3.3 and the memoranda reporting the data are located in Appendices B-3 through B-7. The detailed methodology for calculating the estimates of these physical/chemical constituents are provided in each memorandum in these Appendices.

3.3.1 Estimate of Complexing Agents in Transuranic Solidified Waste Forms Scheduled for Disposal in WIPP

The information on complexing agents in the waste was provided in a series of three memoranda to DOE/CAO. The initial memorandum, entitled "Preliminary Estimate of Complexing Agents in TRU Waste Forms Scheduled for Disposal in WIPP," provided in Appendix B-3, represents the earliest estimate of complexing agents in the TRU Waste. The Appendix B-3 memorandum was superseded by the second estimate, entitled "Current Estimate of Complexing Agents in Transuranic Solidified Waste Forms Scheduled for Disposal in WIPP," provided in Appendix B-4. After the Appendix B-4 memorandum was issued, preliminary information from the January 1996 data submittal from INEL was received. The data submittal indicated that over 90% of the stored waste at INEL would be vitrified, a process that should destroy complexing agents in TRU waste. Based on the preliminary data from INEL, the estimated amount of complexing agents due to RFETS waste stored at INEL could be reduced from that reported in Appendix B-4. A synopsis of the INEL information is reported in the third memorandum that estimates complexing agents in Transuranic Solidified Waste Forms Scheduled for Disposal in WIPP," provided in Appendix B-4. A synopsis of the INEL information is reported in the third memorandum that estimates complexing agents in Transuranic Solidified Waste Forms Scheduled for Disposal in WIPP," provided in Appendix B-5.

Table 3-2 provides a summary of the anticipated mass (in kilograms) of complexing agents in TRU waste reported by RFETS/INEL, LANL, and Hanford. The estimates in Table 3-2 include the anticipated reduction in mass of complexing agents reported from RFETS/INEL based on the preliminary data for proposed vitrification of waste at INEL (Appendix B-5). In addition to the mass of complexing agents reported in Table 3-2, ORNL has provided an estimate of total organic carbon (TOC) in their RH-TRU sludges (Table 3-3). ORNL does not have any analytical data to quantitatively estimate which organic chemicals are responsible for the TOC content of the sludges. However, ORNL has provided a list of chemicals, summarized in Table 3-3, that could contribute to the TOC value reported (see Table 1 in Appendix B-4). It is estimated that most of the TOC in the tanks is not associated with complexing agents, but that has not been verified at this time. As a conservatism, PA calculations can assume that any complexing agents listed in Table 3-3 could form the bulk of the TOC in the ORNL RH-TRU tanks.

| Compound | Low Estimate (kg) | Recommended Estimate (kg) | High Estimate (kg) |
|---|----------------------|------------------------------|-----------------------|
| Ascorbic Acid | 18 | 30 | 34 |
| Acetic Acid | 27 | 44 | 50 |
| Sodium Acetate | 141 | 282 | 333 |
| Citric Acid | 1110 | 1120 | 1130 |
| Sodium Citrate | 51 | 102 | 120 |
| Oxalic Acid | 13700 | 13700 | 13700 |
| EDTA | 3 | 6 | 7 |
| 8-Hydroxyquinoline | 6 | 12 | 14 |
| Tributyl Phosphate | 102 | 111 | 115 |
| 1,10 Phenanthroline | 0.03 | 0.06 | 0.07 |
| Dihexyl-n,n-diethyl carbamoylmethyl phosphonate | 9 | 18 | 22 |

Table 3-2. Estimates of Complexing Agents in Transuranic Waste from RFETS, INEL, LANL, and Hanford*

• Refer to Appendices B-4 and B-5 for methodology of calculated estimates.

3 - 12,

Table 3-3. Estimate of Total Organic Carbon (TOC) in ORNL Transuranic Sludge Tanks and Possible Complexing Agents that can Contribute to TOC.

Total Organic Carbon in ORNL Transuranic Sludge = 3691 kg

Possible Complexing Agents and Other Organic Compounds* in ORNL RH-TRU Sludges:

Acetic Acid Acetone Adogen-364-HP (~triluarylamine) Carbon tetrachloride Deodorized mineral spirits (Amsco) 2,5-di-tert-butylhydroquinone (DBHQ) Diethylbenzene (DEB) Diethylenetriaminepentaacetic acid (DPTA) Di (2-ethylhexyl) phosphoric acid (HDEHP) Di-isopropylbenzene (DIPB) Ethanol Ether Ethylenediaminetetraacetic acid (EDTA) 2-ethyl-1-hexanol a-hydroxyisobutyric acid Isopropanol Methanol n-dodecane n-paraffin (NPH) **Oxalic Acid** Thenoyitrifluoroacetone (TTA) Tributylphosphate (TBP) Trichloroethylene (TCE) Xylene

*Adapted from Table 1 in Appendix B-4.

3.3.2 Estimate of Nitrate, Sulfate, and Phosphate Content in Transuranic Solidified Wastes for Disposal in WIPP

Estimates of nitrate and sulfate in solidified TRU final waste forms were provided in the memorandum entitled "Preliminary Estimate for SNL/NM Performance Assessment Calculations of Nitrate, Sulfate, and Phosphate in Transuranic Solidified Wastes Destined for Disposal in WIPP," provided in Appendix B-6. In that memorandum, it is estimated that densities for the overall disposal inventory are as follows: 9.2 kilograms/cubic meter for nitrate and 3.6 kilograms/cubic meter for sulfate. No estimate of phosphate was reported due to lack of sufficient information.

3.3.3 Estimate of Cement in TRU Solidified Waste Forms for Disposal in WIPP

An estimate of cement (portland-based) in solidified TRU final waste forms was calculated in the memorandum entitled "Estimate of Cement Content in TRU Solidified Waste Forms Scheduled for Disposal in WIPP," provided in Appendix B-7. The estimated density of cement over the entire disposal inventory is 48.6 kilogram/cubic meter. This estimate includes both CH-TRU and RH-TRU final waste forms. The portland cement reported is both reacted and unreacted cement in the waste. There are no data available to estimate the percentage of reacted versus unreacted cement in the waste.

4. REFERENCES

Croff, A. G., 1983, "ORIGEN2: A Versatile Computer Code for Calculating the Nuclide Compositions and Characteristics of Nuclear Materials," *Nuclear Technology*, Vol. 62, pp. 335-352, November 1983.

Croff, A. G., 1980, A User's Manual for the ORIGEN2 Code, ORNL/TM-7175, Oak Ridge National Laboratory, July 1980.

DOE – See U.S. Department of Energy.

EPA – See U.S. Environmental Protection Agency.

SNL/NM - See Sandia National Laboratory/NM.

Public Law, 1992a, Public Law 102-386, 1992, Federal Facilities Compliance Act of 1992.

Public Law, 1992b, Public Law 102-579, 1992, Waste Isolation Pilot Plant Land Withdrawal Act.

Public Law, 1979, Public Law 96-164, 1979, National Security Programs, 93 Stat. 1259, 1265.

Public Law, 1954, U.S. Congress, Atomic Energy Act of 1954, Public Law 83-703, August 15, 1954.

U.S. Department of Energy, 1996, *Waste Acceptance Criteria for the WIPP*, DOE/WIPP-069, Revision 5, April 1996.

U.S. Department of Energy, 1995a, *DOE Waste Treatability Group Guidance*, DOE/LLW-217, Revision 0, January 1995.

U.S. Department of Energy, 1995b, Integrated Data Base for 1995: U.S. Spent Fuel and Radioactive Waste Inventories, Projections, and Characteristics, DOE/RW-0006, Revision 11, September 1995.

U.S. Department of Energy, 1995c, *Transuranic Waste Baseline Inventory Report*, DOE/CAO-95-1121, Revision 2, December 1995.

U.S. Department of Energy, 1990, *Final Supplement Environmental Impact Statement*, Volume 3, DOE/EIS-0026-FS, January 1990.

U.S. Department of Energy, 1988, *Radioactive Waste Management*, DOE Order 5820.2A, U.S. Department of Energy, Washington, D.C., September 26, 1988.

U.S. Department of Energy and State of New Mexico, 1981, "Agreement for Consultation and Cooperation Between the Department of Energy and the State of New Mexico on the Waste Isolation Pilot Plant," July 1, 1981 (dated April 18, 1988).

U.S. Environmental Protection Agency, 1996, *Criteria for the Certification and Recertification of the Waste isolation Pilot Plant's Compliance With the 40 CFR Part 191 Disposal Regulations,* Final Rule, 40 CFR 194, Federal Register, February 9, 1996.

U.S. Environmental Protection Agency, 1993, Environmental Radiation Protection Standards for the Management and Disposal of Spent Nuclear Fuel, High-Level and Transuranic Radioactive Wastes, 40 CFR 191, Final Rule, Federal Register, Vol. 58, Page 66398, December 20, 1993.

U.S. Environmental Protection Agency, 1980, *Listing of Hazardous Waste*, 40 Code of Federal Regulations, Part 261, May 19, 1980.

5. GLOSSARY

40 CFR Part 191, Protection of Environment. EPA: Environmental Radiation Protection Standards for Management and Disposal of Spent Nuclear Fuel, High-Level and Transuranic Radioactive Wastes – The EPA's environmental standards for the storage (Subpart A) and disposal (Subpart B) of spent nuclear fuel, and high-level and TRU radioactive wastes. This is the primary post-closure standard that applies to WIPP.

Anticipated Inventory – The sum of the stored and projected inventories, as defined in this document.

As-Generated Waste - The chemical and physical status of waste when it is generated.

Buried Waste – TRU waste buried in shallow trenches prior to the 1970 Atomic Energy Commission policy that required TRU waste to be retrievably stored.

Code of Federal Regulations (CFR) – (1) A codification of the general and permanent rules published in the **Federal Register** by the department and agencies of the federal government. The CFR is divided into 50 titles that represent broad areas subject to federal regulation. It is issued quarterly and revised annually. (2) All federal regulations in force are published annually in codified form in the CFR.

Contact-Handled (CH) TRU Waste – Packaged TRU wastes with an external surface dose rate of less than 200 mrem per hour.

Defense Waste – (1) Radioactive waste from any activity performed in whole or in part in support of DOE atomic energy defense activities; excludes waste under purview of the Nuclear Regulatory Commission or generated by the commercial nuclear power industry. (2) Nuclear waste derived mostly from the manufacture of nuclear weapons, weapons-related research programs, the operation of naval reactors, and the decontamination of nuclear weapons production facilities.

Department of Energy Site – A DOE-owned or -controlled tract used for DOE operations. Either a tract owned by DOE or a tract leased or otherwise made available to the federal government under terms that afford to DOE rights of access and control substantially equal to those that DOE would possess if it were the holder of the fee (or pertinent interest therein) as agent of and on behalf of the government. One or more DOE operations/program activities are carried out within the boundaries of the described tract.

Disposal – Emplacement of waste in a manner that assures isolation from the biosphere for the foreseeable future with no intent of retrieval and that requires deliberate action to regain access to the waste. For example, disposal of wastes in a mined geologic repository occurs when all of the shafts to the repository area are backfilled and sealed.

Disposal Inventory – The inventory volume defined for WIPP emplacement to be used for PA calculations is the "disposal inventory." The LWA defines the total amount of TRU waste allowed in the WIPP as 6,200,000 cubic feet (approximately 176,000 cubic meters) (Public Law, 1992b). The "Agreement for Consultation and Cooperation" (C&C Agreement) limits the RH-TRU inventory to 250,000 cubic feet (approximately 7,080 cubic meters) (DOE and State of New

5 - 1

Mexico, 1981). Therefore by difference, the CH-TRU inventory is limited to 5,950,000 cubic feet (approximately 168,500 cubic meters).

Final Waste Form – Consists of a series of WMCs that for PA purposes have similar physical and chemical properties.

Integrated Data Base (IDB) – The latest version of the IDB, the *Integrated Data Base for [1995]:* U.S. Spent Fuel and Radioactive Waste Inventories, Projections, and Characteristics (DOE, 1995b).

Land Withdrawal Act - The 1992 legislation passed by the U.S. Congress withdrawing the surface land and underlying minerals at the WIPP site from public use, transferring the property from the DOI to the DOE, and enabling the start of the WIPP Test Phase. The LWA sets prerequisites to be met before the start of the Test Phase, such as the repromulgation by EPA of 40 CFR 191 and the concurrence of EPA with the Test Phase Plan (Public Law, 1992b).

Mixed TRU Waste – TRU waste that contains both radioactive and hazardous components as defined by the Atomic Energy Act and the RCRA as codified in 40 CFR Parts 261.3 (EPA, 1980).

Newly Generated Wastes - See Projected Inventory.

Performance Assessment (PA) – (1) A systematic analysis of the potential risks posed by waste management systems to the public and environment and a comparison of those risks to established performance objectives. (2) An analysis that (a) identifies the processes and events that might affect the disposal system, (b) examines the effects of these processes and events on the performance of the disposal system, and (c) estimates the cumulative releases of radionuclides, considering the associated uncertainties, caused by all significant processes and events. These estimates shall be incorporated into an overall probability distribution of cumulative release to the extent practicable. (3) A term used to denote all activities (qualitative and quantitative) carried out to (a) determine the long-term ability of a site/facility to effectively isolate the waste and ensure the long-term health and safety of the public and (b) provide the basis for demonstrating regulatory compliance.

Projected Inventory – That part of the inventory that has not been generated but is estimated to be generated at some time in the future by the TRU waste generator/storage sites. The estimated timeframe may vary, but is usually between 20 and 30 years. "Newly generated waste" also is sometimes used as a synonym for the projected inventory.

Radioactive – Term used to refer to an unstable atomic nucleus that decays with the spontaneous emission of ionizing radiation (also see "radionuclide").

Radionuclide – (1) A species of atom having an unstable nucleus, that is subject to spontaneous decay or disintegration and usually accompanied by the emission of ionizing radiation. (2) Any nuclide that emits radiation. A nuclide is a species of atom characterized by the constitution of its nucleus and hence by the number of protons, the number of neutron, and the energy content.

Remote-Handled (RH) TRU Waste – Packaged TRU wastes with an external surface dose rate equal to or exceeding 200 mrem per hour.

Resource Conservation and Recovery Act (RCRA) – (1) Establishes a system for controlling hazardous waste from generation to disposal. (2) A Federal law passed in 1976, and amended under the HSWA of 1984, that established a structure to track and regulate hazardous wastes from the time of generation to disposal. The law requires safe and secure procedures to be used in treating, handling, transporting, storing, and disposing of hazardous substances. RCRA is designed to prevent new uncontrolled hazardous waste sites. The law also regulates the disposal of solid waste that may not be considered hazardous. (3) Specifically, Subtitle D of RCRA governs the management of solid waste. (Note: 40 CFR Parts 260-272 are the regulations for complying with RCRA with respect to hazardous waste and hazardous waste treatment, storage, and disposal facilities.)

Scaling – The process for adjusting the anticipated inventory to equal the maximum authorized disposal inventory of the WIPP repository for the purposes of WIPP performance assessment modeling.

Stored Inventory – That part of the TRU waste inventory currently in retrievable storage as of the time of the last data call for inventory information. Retrievably stored waste includes waste stored in buildings or in berms with earthen cover since 1970 and does not include any waste that was buried prior to 1970. Stored inventory can be in the "as-generated" form or "final waste form."

Transuranic – Pertaining to elements that have atomic numbers greater than 92, including neptunium, plutonium, americium, and curium; all are radioactive, are not naturally occurring, and are members of the actinide group.

Transuranic (TRU) Waste - (1) Waste containing alpha-emitting radionuclides with an atomic number greater than 92 and half-lives greater than 20 years, at concentrations of TRU isotopes greater than 100 nanocuries per gram of waste. This core definition appears in modified form in various relevant documents as follows: (a) For purposes of management, DOE Order 5820.2A: (i) considers TRU waste, as defined above, "without regard to source or form" (The proposed revision to the Order [DOE Order 5820.2A Major Issues for Revision, May 6, 1992] contemplates removing this clause); (ii) allows head of field elements to determine that wastes containing other alpha-emitting radionuclides must be managed as TRU waste; and (iii) adds "at time of assay," implying both that the classification of a waste as TRU waste is to be made based on an assay, and that such classification can be superseded only by another assay. (b) For purposes of setting standards for management and disposal, 40 CFR 191.02(i) adds "except for: (i) high-level wastes; (ii) wastes that the DOE has determined, with the concurrence of the EPA Administrator, do not need the degree of isolation required by this part; or (iii) wastes that the Nuclear Regulatory Commission has approved for disposal on a case-by-case basis in accordance with 10 CFR 61." (2) Waste materials contaminated with U-233 (and its daughter products), with certain isotopes of plutonium, or with other nuclides with atomic numbers greater than 92. In order to be classified as TRU waste, the long-lived alpha activity from subject isotopes must exceed 100 nanocuries per gram of waste material independent of the level of beta-gamma activity. These wastes are produced primarily from reprocessing spent fuel and from the use of plutonium in the fabrication of nuclear weapons. (3) Wastes that are contaminated with radioactive elements heavier than uranium, thus the name trans-(or beyond) uranic.



Waste Acceptance Criteria (WAC) – The criteria used to determine if waste packages are acceptable. For the purposes of this document, WAC refers to WIPP WAC.

Waste Form - The physical form of the waste such as sludges, combustibles, metals, etc.

TRU Waste Sites – The 8 major DOE facilities and several smaller sites throughout the U.S. that generate and store TRU waste.

Waste Isolation Pilot Plant (WIPP) – (1) The project authorized under Section 213 of the DOE National Security and Military Applications of Nuclear Energy Authorization Act of 1980 (Public Law, 1979) to demonstrate the safe and environmentally sound disposal of radioactive waste materials generated by atomic energy defense activities. (2) A research and development facility located near Carlsbad, New Mexico to be used to demonstrate a practical, long-term solution to a complex problem: the safe disposal in deep geologic repositories of TRU waste resulting from DOE activities.

Waste Material Parameter -- A waste material that occurs in TRU waste that is an input parameter into one (or more) current PA model(s) or is required to adequately describe the waste form.

Waste Matrix Code (WMC) – A DOE-developed coding system for grouping waste streams that have similar matrix constituents, especially for treatment objectives. This coding system allows waste streams within the DOE TRU waste system that have similar physical and chemical waste form properties to be categorized together. WMCs also have been called "waste treatability codes" in other DOE documents. An example of a WMC for "heterogeneous waste" is 5400 (DOE, 1995a).

Waste Stream – A flow of waste materials with specific definable characteristics that remain the same throughout the life of the process generating the waste stream.

Waste Stream Profile – A description of a CH-TRU or RH-TRU waste stream destined for shipment to and disposal in WIPP, if authorized under permits and certifications by appropriate regulatory agencies for disposal in the WIPP repository. The waste stream profile is presented in tabular format and is intended to provide a summary of the important information about a particular waste stream.

WIPP Waste Profile – Represents a summary of TRU waste at all DOE TRU waste generator/storage sites that have an identical Final Waste Form.

APPENDIX A



APPENDIX A - 1

Sandia National Laboratories

Managed and Operated by Sandis Corporation a Excented Marun Company Albuquerque, New Mexico 87185-1328

| date : | November 6, 1995 |
|--------|--|
| to : | R. L. Bisping, DOE/CAO, WORC 3, MS-560 |
| | Journ Change |
| from : | L. C. Sanchez, Olg 6342, MS-1328 (505)848-0685 |

subject : CH and RH-TRU Waste Parameters Potentially Important in WIPP PA

A) Requested PA Data From TWBIR

Below you will find an updated list of waste material parameters that have been identified as being potentially important to the performance analysis of the WIPP repository. It is requested that these parameters be be supplied in Rev. 2 of the Transuranic Waste Baseline Inventory Report (TWBIR). Itemized below you will find the two categories of requested waste parameter data.

1) Non-radioactive Materials

The non-radioactive materials are those which influence gas generation potential and those that are needed for mechanical models which predict waste consolidation and shear strength properties. The list of the non-radioactive materials is shown in Table 1.

2) Radionuclide

At this time there are no new requests for additional radionuclide inventory data beyond those previously reported in Rev. 1 of the WTWBIR. If there are significant inventory increases in radionuclides due to special circumstances (such as inclusion of residues to the TRU inventory), sufficient footnote explanations should be supplied.



| Waste Parameter | Input V Current | Input Variable in Input Variable in Input Variable in Current PA Models PA Models | | Input Variable in Possible |
|---|--------------------|---|-------------|-------------------------------|
| ; ; | Gas Generation | Mecnanical Characteristics | Development | PA Models |
| Iron-Based Metals and Allovs | x | X | х | Х |
| Aluminum- Based Metals and Allovs (a) | | X | x | |
| Other Metals | | x | | ; |
| Other Inorganics | | x | | ? |
| Cellulosics | x | <u>x</u> | x | x |
| Plastics | <u>½ (b)</u> | <u>x</u> | X (d) | <u>x</u> |
| Rubbers | <u>特(b)</u> | <u>x</u> | X (e) | x |
| Solidified Inorganics | | x | x | x |
| Solidified Organics Matrix | | x | x | x |
| Soils (c) | | X | ? . | ? |

Justification of TWBIR Nonradioactive Waste Parameters. Table 1.

(a) Future model for PA does not include aluminum.(b) Only one-half of material is assumed to generate gas.

(c) May impact colloids.

(d) As is.

(e) Percentage of material to generate gas is unknown at the present time.

B) Special Request Non-PA Items

Also wanted at this time is additional information for several waste material characteristics. Although these characteristics have not been identified as waste material parameters to be used for WIPP PA, they are needed for non-PA scoping calculations to assess their influence on PA. Since these items are not currently PA parameters, inventory estimates of these characteristics as "additional information" in the TWBIR or supplied outside of the TWBIR via written correspondence. Below you will find an itemized list of these special request items.

1) Non-radioactive Materials

Additional information is needed on the five waste material characteristics (see Table 2): 1) vitrified wastes, 2) nitrates (NO_5^-) , 3) sulfates (SO_4^{2-}) , 4) phosphorus, and 5) cement. Of these waste parameters, the last four are needed for the gas generation modeling. The nitrates and the sulfates are involved in the denitrification and sulfate reduction processes which breakup the cellulosics, while the phosphorus is a nutrient for biodecay of cellulosics. The estimate of the mass quantities of cement in the waste inventory should include both the cement that is contained in the waste as cement itself (due to D&D activities, etc.,) and the cement found in various sludges. Cement consumes CO_2 due to its content of $Ca(OH)_2$. The estimates for this non-radioactive waste constituent need only be "best estimates" at this present time so that non-PA scoping calculations can be made to determine their importance on overall repository performance. (Do not generate upper-bound estimates that are overly conservative.)

2) Residues

"Best estimates" are needed for residues, in addition to those already identified at the Rocky Flats Plant (RFP), that have the possibility of being changed from a resource category to a TRU waste category.

3) Organic Ligands (Chelating Agents)

"Best estimates", from currently available information, are needed for major water-soluble organic ligands which are under consideration for the actinide source term (see Table 3). If it is not possible to obtain data from major waste generating sites then supply guidance on how a first-order estimate may be made (from existing information such as process knowledge etc.,) so that non-PA scoping calculations can be performed to identify if the presence of these ligands would have any significant impacts. (Do not generate estimates that are overly conservative.) Requested data is for final form "process-level" quantities used in production only for the key sites. If information on the "process-level" values does not exist at the key sites, then "laboratory-scale" values should be used in the requested assessment of the inventory. Should it be determined that more detailed information on organic ligands will be needed, you will be given a specific written request at a future time. This effort should be performed in parallel with the TWBIR. Technical data should be supplied in memorandum form by the end of February 1996 with supporting documentation by the end of March 1996.

| Waste Pa rameter | Input Variable in Current PA Models | | Input Variable in PA Models | Input Variable in Possible |
|--|---|---|---|---|
| | Gas Generation | Mechanicai Characteristics | Development | Pature PA Models |
| Vitrified (b) | | X | <u>,</u> | 2 |
| Nitrates (NO 1) | X (c) | : | <u> </u> | ? |
| Sulfates (SO 1) | X (c) | \ ; | x | ? |
| Phosphorus | X (c) | | X | ? |
| Cement (d) | x | | x | ? |
| (a) Information of for assessme "best estima (b) New waste p anticipated i | on these addition on tof their impo te" level. arameter corresp n the future. | al waste materials a mance. These waste onding to treatment | re needed for non-PA characteristics can be , identified by some of | scoping calculations reported at the the sizes, to be |

l l

(d) Any concrete or cement (including dry portland cement) that contains calcium oxide.

M A1-4

| Table 3. | Justification of Special Request For Info On Organic Complexing Agents. (a) |
|---|--|
| Ligand (b) | Discussion (c) |
| 1) Total Compiexants | The most valuable information at this time is a "best estimate" of the i total amount of water soluble complexing agents (ligands) in the TRU waste matrix. |
| 2) Citrate | Preliminary information indicates that citrate (citric acid) may be the largest used ligand at TRU waste generating sites. Hence, inventory quantities are very important. |
| 3) Lactate | This is an important ligand that is produced by bacteria as part of its own metabolism. What is requested here is a "best estimate" of the quantity of lactate that actually exists in the TRU waste matrix (not just an initial amount supplied as part of a waste stream). However, if this information cannot be developed, then supply information on the initial amount. |
| 4) Oxalate | This is an important ligand that is produced by bacteria as part of its own metabolism. What is requested here is a "best estimate" of the quantity of oxalate that actually exists in the TRU waste matrix (not just an initial amount supplied as part of a waste stream). However, if this information cannot be developed, then supply information on the initial amount. |
| 5) EDTA | This ligand (ethylenediaminetetraacetic acid) is also of major impor- tance due to its common use as a cleaning solvent. |
| (a) Information on the for assessment of for the actinide so (b) These items are raised as the degradation of de In cases where no quantities. | se additional waste materials are needed for non-PA scoping calculations their importance. The presence of these complexing agents are important burce term, with respect to increasing the solubility of radionuclides. Inked in the order of their importance in the actinide source term, vailable information that TRU was a generation sites may have on the cay rates of ligands in current (and expected) waste matrixes if possible. Information is available, supply guidance on estimating first-order |

LCS:6741:lcs/(95-2082)

Copy to: P.E. Drez [Drez Environmental Associates] D. Bretzke [Science Applications International Corporation] S. Chakraborti [Science Applications International Corporation] MS-1320, C.F. Novak [Dept. 6119] MS-1328, H. Jow [Dept. 6741]

MS-1328. M.S. Tierney {Dept. 6741} MS-1328. D.R. Anderson {Dept. 6749} MS-1328. D.R. Anderson {Dept. 6749} MS-1328. M.E. Feweil {Dept. 6749} MS-1328. J.D. Schreiber {Dept. 6749} MS-1328. P. Vaughn {Dept. 6749} MS-1341. L.H. Brush {Dept. 6748} MS-1341. B.M. Butcher {Dept. 6748} MS-1341. A.C. Peterson {Dept. 6748} MS-1341. L.J. Storz {Dept. 6748} MS-1341. A. Reiser {Dept. 6747} MS-1341. R.F. Weiner {Dept. 6747} MS-1341. R.F. Weiner {Dept. 6747} MS-1328. Day File {Dept. 6741} MS-1328. L.C. Sanchez {Dept. 6741} File - SWCF-A WBS 1.1.6.2; PA: PBWAC - WIPP ACTIVITY



APPENDIX A-2



Sandia National Laboratories

Managed and Operated by Sandia Corporation a Lockhaed Martin Corporation Albuquerque, New Mexico \$7185-1328

date : January 11, 1996

S. Chakraborti [Science Applications International Corporation] to : C. Sanchez, Org 6741, MS-1328, PH-(505)848-0685, Fax-848-0705

subject : Information Needed from TWBIR (Rev. 2/Addendum)

I have read Paul Drez's memo [Ref. DEA-1] about the Rev. 2 of the TWBIR [Ref. BIR-1]. When updated values are available, please send me a memo with the WIPP-scale values (CH & RH waste material parameters – Tables 3-2 & 3-3 and CH & RH disposal radionuclide inventory data – Table 3-4). [Note – because the anticipated volume of RH waste is much greater than the WIPP disposal volume, the proper volume that should be used to determine the average waste material parameters should be a "truncated volume", i.e., the truncated volume is equal to the existing stored waste plus only the necessary amount of projected waste necessary to reach the WIPP disposal volume limit.] When regenerating Table 3-4, please add extra columns which also display the "total curies" (in addition to the data displaying the curie volumetric densities) for both CH & RH radionuclides. The volumes to be used for these conversions are: 1) 6.25+06 cu.ft = 175,584. cu.m. for CH-TRU waste and 2) 0.25E+06 cu.ft. = 7,080. cu.m. [the unit conversions for volume were done with the factor 1.0 cu.ft. = 2.832E-02 cu.m. taken from Ref. SNL-1].

105 1-12-5E

A second request, which should be documented in a separate memo, is that CH & RH activity loading tables be generated on a per-waste stream basis. The format for the data should look as close to that shown in Table 1 below. It is also needed that the information be made into an ASCII file and placed on a 3.5" diskette (IBM formatted). There are three versions of this table that are needed: 1) values corresponding to stored waste only, 2) values corresponding to projected waste only, and 3) values corresponding to WIPP disposal volume [Note - remember to use the truncated volumes for the RH waste].

REFERENCES

[DEA-1]

Memo from: P. Drez (Drez Environmental Associates, DEA) to: L.C. Sanchez (Sandia National Laboratories), subject: "BIR Error", dated: January 7, 1996.

[BIR-1]

DOE (U.S. Department of Energy); Transuranic Waste Baseline Inventory Report; DOE/CAO-95-1121; Revision 2; printed December 1995.

[SNL-1]

Sandia WIPP Project. 1992. Preliminary Performance Assessment for the Waste Isolation Pilot Plant, December 1992. Volume 3: Model Parameters (SAND92-0700/3), section: Conversion Tables For SI and Common English Units, Table 5, pg. Conversion Tables - 4. SAND92-0700/3. Albuquerque, NM: Sandia National Laboratories.

Exceptional Service in the National Interest

| Table 1.RadionuclideActivityLoadingTable(to be used for human intrusion calculations) | | | | | | |
|---|--|--|---|--|---|--|
| TRU Type | Site ID | Waste Stream ID | Volume of Waste Stream | Curie (<u>curies</u> vol | Loading [curies] | |
| CH CH CH CH CH CH CH CH CH CH | LANL LANL LANL LANL LANL LANL LANL LANL | LA-?001 LA-?002 LA-?003 LA-?004 LA-?005 LA-?006 + RF-?001 +. | $\begin{array}{c} \chi\chi\chi\chi\chi\chi \\ \chi\chi\chi\chi\chi\chi \\ \chi\chi\chi\chi\chi\chi \\ \chi\chi\chi\chi\chi\chi \\ \chi\chi\chi\chi\chi\chi$ | X.XXE-KK X.XXE-KK X.XXE-KK X.XXE-KK X.XXE-KK X.XXE-KK ↓ X.XXE-KK ↓ | X.XXE+KK X.XXE+KK X.XXE+KK X.XXE+KK X.XXE+KK X.XXE+KK ∑ X.XXE+KK X.XXE+KK ∑ X.XXE+KK ↓ | |
| Сн | Total | | $\sum XXXXXX.X$ | | ∑ X.XXE+KK | |
| RH RH RH RH RH RH RH RH | LANL LANL LANL LANL LANL LANL LANL LANL | LA-?001 LA-?002 LA-?003 LA-?004 LA-?005 LA-?006 + | $\sum_{\mathbf{X} \mathbf{X} \mathbf{X} \mathbf{X} \mathbf{X} \mathbf{X} \mathbf{X} \mathbf{X}$ | X.XXE-KK X.XXE-KK X.XXE-KK X.XXE-KK X.XXE-KK X.XXE-KK X.XXE-KK | X.XXE+KK X.XXE+KK X.XXE+KK X.XXE+KK X.XXE+KK X.XXE+KK ↓ ∑ X.XXE+KK | |
| RH RH RH | RFETS RFETS RFETS | RF-?00] ↓ - | $\sum_{XXXXXX} \sum_{XXXXXX}$ | X.XXE-KK | XXXE+KK XXXE+KK | |
| RH | Total | - | $\sum XXXXXXXX$ | - | ↓ ∑ X.XXE+KK | |

LCS:6741:lcs/(96-2096)

•. •• •• • Copy to: MS-1328, H. Jow [Dept. 6741] MS-1328, R.P. Anderson [Dept. 6749] MS-1328, Day File [Dept. 6741] MS-1328, L.C. Sanchez [Dept. 6741] File - SWCF-A WBS 1.1.6.2;PA;PBWAC - WIPP ACTIVITY APPENDIX A-3

, .

M

,

Sandia National Laboratories

naged and Operated by Sandia Corporation a Lockbeed Martin Corporation Albuquerque, New Mexico 87185-1328

date : January 30, 1996

S. Chalcraborti [Science Applications International Corporation] to: MS-1328, TH-(505)848-0685, Fax-848-0705 L. C. Sanchez, Org 6741. from :

subject : Information Needed from TWBIR (Rev. 2/Addendum)

With regards to the two requests previously made (Ref. LCS-1), the first is no longer needed and an update is needed for the second.

Since the data in the TWBIR (Ref. BIR-1) for projected waste material parameters and radionuclide inventory is based on data for stored waste (Ref. SC-1), the first request for data values to be volume averaged using truncated volume is not necessary (i.e., it would yield the same values).

For the second request from Ref. LCS-1, it has been identified that not all the radionuclide data in the TWBIR are incorporated in the radionuclide activity loading tables which are used for the human intrusion calculations (Refs. SNL-1 & JG-1). Instead, an abbreviated list of 21 radionuclides is all that should be used to generate the curie loading table (see Table 1 of Ref. LCS-1). The list of the 21 radionuclides (for both CH and RH) are shown in Table 1 below (this list is based on Table I of Appendix of Ref. EPA-1). Also, since the projected waste data is based on stored data, values generated are needed only for WIPP disposal volumes (data separated for stored and projected data would have yielded the same values).

REFERENCES

[BIR-1]

DOE (U.S. Department of Energy); Transuranic Waste Baseline Inventory Report; DOE/CAO-95-1121; Revision 2; printed December 1995.

[EPA-1]

"Environmental Standards for the Management and Disposal of Spent Nuclear Fuel, High-Level and Transuranic Radionactive Waste: Final Rule," 40CFR191, Federal Register, 50, 38066 (1985).

[JG-1]Communications with J. Garner [Piru Assoc., SNL/Dept 6749], date: January 30, 1996.

[LCS-1]

Memo from: L.C. Sanchez (Dept. 6741) to: S. Chakraborti (Science Applications International Corporation), subject: "Information Needed from TWBIR (Rev. 2/Addendum)", dated: January 11, 1996.

[SC-1]Communications with S. Chakraborti [Science Applications International Corporation], date: January 25, 1996.



[SNL-1]

Sandia WIPP Project. 1992. Preliminary Performance Assessment for the Waste Isolation Pilot Plant, December 1992. (SAND92-0700),

| Table 1.RadionuclidesThatShouldBeUsedToGenerateCurieLoading | | | | | |
|---|---|--|--|--|--|
| | Radionuclide | | | | |
| 1 2 3 4 5 6 7 8 9 10 11 11 12 13 14 15 16 17 18 19 20 | Am-241 Cm-248 Ci-137 Np-237 Pa-231 Pb-210 Pu-238 Pu-238 Pu-239 Pu-240 Pu-242 Pu-244 Ra-226 Sr-90 Th-229 Th-229 Th-230 Th-232 U-233 U-234 U-235 U-236 | | | | |

LCS:6741:lcs/(96-2098)

Copy to: MS-1328, H. Jow [Dept. 6741] MS-1328, R.P. Anderson [Dept. 6749] ⁷MS-1328, Day File [Dept. 6741] MS-1328, L.C. Sanchez [Dept. 6741] File - SWCF-A WBS 1.1.6.2;PA;PBWAC - WIPP ACTIVITY



APPENDIX B



APPENDIX B - 1

.

.

111

.

.

Department of Energy

United States Government

Carlsbad Area Office Carlsbad, New Mexico 88221

memorandum

DATE: June 4, 1996 REPLY TO CAO:NTP:RLB:96-1174

ATTN OF:

SUBJECT: Revised Radionucide Data in Support of the Compliance Certification Application

TO:

Les E. Shephard. Director, Nuclear Waste Management Programs Center, SNL/NM

Please find attached the revised WIPP disposal radionuclide inventory which was previously transmitted to your staff for their use. This inventory has been recalculated on the basis of new radionuclide information recently available from four TRU waste sites: the Hanford site (Hanford), the Oak Ridge National Laboratory (ORNL), the Rocky Flats Environmental Technology Site (RFETS), and the Savannah River Site (SRS). The revised WIPP disposal radionuclide inventory is provided in Attachment A in a format similar to Table.3-4 of Revision 2 of the Transuranic Waste Baseline Inventory Report (TWBIR).

The values in Attachment A were originally based on the extrapolation of the results of preliminary radionuclide decay calculations that were completed by Sandia National Laboratories (SNL) staff on April 8. These preliminary calculations have recently completed the formal quality assurance/quality control (QA/QC) review process by the SNL QA/QC group, and an approved version of these calculations was obtained on Tuesday, April 17. The QA/QC review process produced some changes in the preliminary values, and these changes have been incorporated in Attachment A.

Since the WIPP Performance Assessment (PA) group at SNL required the revised data as soon as possible in support of the Compliance Certification Application (CCA), Attachment A is being supplied as the most current update until the publication of Revision 3 and should be used by the WIPP PA in support of the CCA. As agreed with the SNL WIPP (PA) staff during the videoconference meeting on March 3, 1996, the revised data shown in Attachment A are based on the final waste form volumes published in Revision 2 of the TWBIR. The information in Attachment A will be included in the TWBIR, Rev. 3. as well as that previously supplied on complexing agents. cement content, and nitrate/sulfate/phosphate content, which will be included as an appendix to the TWBIR.

In summary, the revised data in Attachment A incorporates the effect of the following information received from four sites during the past two months:

• Corrections to the values for Cf-252, Cm-244, and Cm-245 reported in earlier Hanford submittals for the IDB.





Les E. Shephard

- Preliminary sludge sampling data from ORNL for the RH-TRU sludges showing the distribution of different uranum isotopes in the sludge: this enabled the redistribution of the uranum curies from previous Oak Ridge IDB submittals and corrected the previously high estimates of U-235.
- Break-up of radionuclide data for SRS waste between on-site and off-site waste (i.e., waste from Los Alamos and Mound that was shipped to SRS for storage in the early 1970s); this enabled more realistic extrapolation of the amount of Pu-238 and Pu-239 in SRS waste.

A description of the step-by-step methodology used to incorporate the new information from the four sites and to develop the revised inventory is provided in Attachment B.

If you have any questions concerning the enclosed information, please contact Mr. Russ Bisping of my staff at (505) 234-7446.

Don Watkins Manager National TRU Program

Attachments

cc w/attachments: R. Bisping, CAO G. Basabilvazo, CAO J. Mewhinney, CAO S. Chakraborti, CTAC P. Drez, CTAC J. Harvill, CTAC R. Anderson, SNL L. Sanchez, SNL M. Chu, SNL M. Marietta, SNL


| Nuclide | CH-TRU Waste (Ci/m ³) | RH-TRU Waste (Ci/m ³) | CH-TRU Waste (Total Curies ²) | RH-TRU Waste (Total Curies ²) | |
|---------|--------------------------------------|--------------------------------------|--|--|--|
| Ac225 | 1.71 E-05 | 1.66E-05 | 2.88E+00 | 1.17E-01 | |
| Ac227 | 3.61E-06 | 1.07E-07 | 6.08E-01 | 7.57E-04 | |
| Ac228 | 4.43E-06 | 1.10E-05 | 7.47E-01 | 7.77E-02 | |
| Ag109m | 9.32E-05 | NR | 1.57E+01 | NR | |
| Ag110 | 4,19E-14 | 2.46E-13 | 7.07E-09 | 1.74E-09 | |
| Ag110m | 3.15E-12 | 1.85E-11 | 5.31E-07 | 1.31E-07 | |
| Am241 | 2.62E+00 | 8.42E-01 | 4.42E+05 | 5.96E+03 | |
| Am242 | 1.04E-05 | NR | 1.75E+00 | NR | |
| Am242m | 1.04E-05 | NR | 1.75E+00 | NR | |
| Am243 | 1.93E-04 | 3.23E-08 | 3.26E+01 | 2.28E-04 | |
| Am245 | 7.89E-15 | 4.06E-20 | 1.33E-09 | 2.87E-16 | |
| At217 | 1.71E-05 | 1.66E-05 | 2.88E+00 | 1.17E-01 | |
| Ba137m | 4.53E-02 | 2.89E+01 | 7.63E+03 | 2.04E+05 | |
| Bi210 | 1.52E-05 | 1.01E-09 | 2.55E+00 | 7.16E-06 | |
| Bi211 | 3.61E-06 | 1.07E-07 | 6.09E-01 | 7.58E-04 | |
| Bi212 | 1.61E-04 | 1.04E-05 | 2.71E+01 | 7.36E-02 | |
| Bi213 | 1.71E-05 | 1.66E-05 | 2.88E+00 | 1.17E-01 | |
| Bi214 | 6.91E-05 | 5.05E-09 | 1.16E+01 | 3.58E-05 | |
| Bk249 | 5.44E-10 | 2.80E-15 | 9.16E-05 | 1.98E-11 | |
| Bk250 | 2.59E-16 | NR | 4.37E-11 | NR | |
| C14 | 6.43E-05 | 2.90E-04 | 1.08E+01 | 2.05E+00 | |

WIPP Disposal Radionuclide Inventory for the CCA¹

NR = Not reported by sites.

¹Decayed to December 1995.

²Total curies estimated by assuming a volume of 168,500 cubic meters for CH-TRU waste and 7,080 cubic meters for RH-TRU waste.



WIPP Disposal Radionuclide Inventory for the CCA (continued)

| Nuclide | CH-TRU Waste (Ci/m ³) | RH-TRU Waste (Ci/m ³) | CH-TRU Waste (Total Curies ²) | RH-TRU Waste (Total Curies ²) |
|---------|--------------------------------------|--------------------------------------|--|--|
| Cd109 | 9.31E-05 | NR | 1.57E+01 | NR |
| Cd113m | 1.08E-11 | 7.71E-11 | 1.82E-06 | 5.46E-07 |
| Ce144 | 3.71E-07 | 7.24E-04 | 6.26E-02 | 5.13E+00 |
| Cf249 | 3.81E-07 | 6.31E-07 | 6.42E-02 | 4.47E-03 |
| Cf250 | 1.965-06 | NR | 3.30E-01 | NR |
| Cf251 | 2.24E-08 | NR | 3.78E-03 | NR |
| Cf252_ | 1.42E-03 | 1.82E-04 | 2.39E+02 | 1.29E+00 |
| Cm242 | 6.76E-06 | NR | 1.14E+00 | NR |
| Cm243 | 1.61E-05 | 6.99E-03 | 2.72E+00 | 4.95E+01 |
| Cm244 | 1.87E-01 | 4.45E-02 | 3.15E+04 | 3.15E+02 |
| Cm245 | 6.81E-08 | 2.07E-10 | 1.15E-02 | 1.46E-06 |
| Cm246 | 6.06E-07 | NR | 1.02E-01 | NR |
| Cm247 | 1.91E-14 | NR | 3.21E-09 | NR |
| Cm248 | 5.31E-07 | 2.89E-08 | 8.95E-02 | 2.05E-04 |
| Co58 | 1.81E-18 | 1.75E-15 | 3.05E-13 | 1.24E-11 |
| Co60 | 3.83E-04 | 1.47E+00 | 6.46E+01 | 1.04E+04 |
| Cr51 | NR | 4.29E-10 | NR | 3.04E-06 |
| Cs134 | 7.97E-08 | 2.60E-03 | 1.34E-02 | 1.84E+01 |
| Cs135 | 2.98E-09 | 1.66E-08 | 5.02E-04 | 1.17E-04 |
| Cs137 | 4.78E-02 | 3.05E+01 | 8.06E+03 | 2.16E + 05 |

NR = Not reported by sites.

¹Decayed to December 1995.

²Total curies estimated by assuming a volume of 168,500 cubic meters for CH-TRU waste and 7,080 cubic meters for RH-TRU waste.

| Nuclide | CH-TRU Waste (Ci/m ³) | RH-TRU Waste (Ci/m ³) | CH-TRU Waste (Total Curies ²) | RH-TRU Waste (Total Curies ²) |
|---------|--------------------------------------|--------------------------------------|--|--|
| Es254_ | 2.51E-16 | NR | 4.24E-11 | NR |
| Eu 150 | 2.08E-10 | NR | 3.51E-05 | NR |
| Eu152 | 7.46E-06 | 1.73E-01 | 1.26E+00 | 1.22E+03 |
| Eu 154 | 6.80E-06 | 8.34E-02 | 1.15E+00 | 5.91E+02 |
| Eu155 | 5.62E-06 | 1.67E-02 | 9.46E-01 | 1.18E+02 |
| Fe55_ | 1.13E-10 | 2.38E-05 | 1.91E-05 | 1.69E-01 |
| Fe59 | 1.57E-12 | NR | 2.64E-07 | NR |
| Fr221 | 1.71E-05 | 1.66E-05 | 2.88E+00 | 1.17E-01 |
| Fr223 | 4.98E-08 | 1.48E-09 | 8.39E-03 | 1.04E-05 |
| нз | 5.16E-06 | 9.33E-06 | 8.69E-01 | 6.60E-02 |
| 1129 | 4.18E-12 | NR | 7.05E-07 | NR |
| Kr85 | 1.20E-06 | 2.37E-04 | 2.02E-01 | 1.68E+00 |
| Mn54 | 5.05E-09 | 3.32E-06 | 8.51E-04 | 2.35E-02 |
| Nb95 | 1.51E-14 | 9.45E-05 | 2.54E-09 | 6.69E-01 |
| Nb95m | 5.04E-17 | 3.17E-07 | 8.50E-12 | 2.24E-03 |
| Ni59 | 4.47E-08 | NR | 7.52E-03 | NR |
| Ni63 | 5.46E-06 | 1.40E-04 | 9.19E-01 | 9.88E-01 |
| Np237 | 3.33E-04 | 4.02E-04 | 5.61E+01 | 2.85E+00 |
| Np238 | 5.20E-08 | NR | 8.77E-03 | NR |
| Nn239 | 1.93E-04 | 3.23E-08 | 3.26E+01 | 2.28E-04 |

WIPP Disposal Radionuclide Inventory for the CCA (continued)

NR = Not reported by sites.

¹Decayed to December 1995.

²Total curies estimated by assuming a volume of 168,500 cubic meters for CH-TRU waste and 7,080 cubic meters for RH-TRU waste.

31-5

WIPP Disposal Radionuclide Inventory for the CCA (continued)

| Nuclide | CH-TRU Waste (Ci/m ³) | RH-TRU Waste (Ci/m ³) | CH-TRU Waste (Total Curies ²) | RH-TRU Waste (Total Curies ²) |
|---------|--------------------------------------|--------------------------------------|--|--|
| Np240m | 8.91E-12 | 3.12E-15 | 1.50E-06 | 2.21E-11 |
| Pa231 | 2.67E-06 | 2.70E-07 | 4.51E-01 | 1.91E-03 |
| Pa233 | 3.33E-04 | 4.02E-04 | 5.61E+01 | 2.85E+00 |
| Pa234 | 3.05E-07 | 1.92E-06 | 5.14E-02 | 1.36E-02 |
| Pa234m | 2.35E-04 | 1.48E-03 | 3.96E+01 | 1.05E+01 |
| Рь209 | 1.71E-05 | 1.66E-05 | 2.88E+00 | 1.17E-01 |
| РЬ210 | 1.52E-05 | 1.01E-09 | 2.55E+00 | 7.16E-06 |
| Р6211 | 3.61E-06 | 1.07E-07 | 6.09E-01 | 7.58E-04 |
| Pb212 | 1.61E-04 | 1.04E-05 | 2.71E+01 | 7.36E-02 |
| Pb214 | 6.91E-05 | 5.05E-09 | 1.16E+01 | 3.58E-05 |
| Pd107 | 4.40E-10 | 2.45E-09 | 7.41E-05 | 1.73E-05 |
| Pm147 | 4.67E-05 | 1.52E-03 | 7.87E+00 | 1.07E+01 |
| Po210 | 1.52E-05 | 1.01E-09 | 2.55E+00 | 7.16E-06 |
| Po211 | 1.01E-08 | 3.00E-10 | 1.71E-03 | 2.12E-06 |
| Po212 | 1.03E-04 | 6.66E-06 | 1.73E+01 | 4.72E-02 |
| Po213 | 1.67E-05 | 1.62E-05 | 2.82E+00 | 1.15E-01 |
| Po214 | 6.91E-05 | 5.05E-09 | 1.16E+01 | 3.57E-05 |
| Po215 | 3.61E-06 | 1.07E-07 | 6.09E-01 | 7.58E-04 |
| Po216 | 1.61E-04 | 1.04E-05 | 2.71E+01 | 7.36E-02 |
| Po218 | 6.91E-05 | 5.05E-09 | 1.16E+01 | 3.58E-05 |

NR = Not reported by sites.

¹Decayed to December 1995.

²Total curies estimated by assuming a volume of 168,500 cubic meters for CH-TRU waste and 7,080 cubic meters for RH-TRU waste.



| | WIPP Disposal Rad | dionuclide Invento | ory for the CCA (co | ontinued) |
|---------|--------------------------------------|--------------------------------------|--|--|
| Nuclide | CH-TRU Waste (Ci/m ³) | RH-TRU Waste (Ci/m ³) | CH-TRU Waste (Total Curies ²) | RH-TRU Waste (Total Curies ²) |
| Pr144 | 3.67E-07 | 7.16E-04 | 6.18E-02 | 5.07E+00 |
| Pu236 | 6.16E-08 | NR | 1.04E-02 | NR |
| Pu238 | 1.55E+01 | 2.05E-01 | 2.61E+06 | 1.45E+03 |
| Pu239 | 4.66E+00 | 1.45E+00 | 7.85E+05 | 1.03E+04 |
| Pu240 | 1.25E+00 | 7.15E-01 | 2.10E+05 | 5.07E+03 |
| Pu241 | 1.37E+01 | 2.00E+01 | 2.31E+06 | 1.42E+05 |
| Pu242 | 6.96E-03 | 2.11E-05 | 1.17E+03 | 1.50E-01 |
| Pu243 | 1.91E-14 | NR | 3.21E-09 | NR |
| Pu244 | 8.92E-12 | 3.12E-15 | 1.50E-06 | 2.21E-11 |
| Ra223 | 3.61E-06 | 1.07E-07 | 6.09E-01 | 7.58E-04 |
| Ra224 | 1.61E-04 | 1.04E-05 | 2.71E+01 | 7.36E-02 |
| Ra225 | 1.71E-05 | 1.66E-05 | 2.88E+00 | 1.17E-01 |
| Ra226 | 6.91E-05 | 5.05E-09 | 1.16E+01 | 3.58E-05 |
| Ra228 | 4.43E-06 | 1.10E-05 | 7.47E-01 | 7.77E-02 |
| Rh106 | 1.72E-07 | 1.54E-03 | 2.90E-02 | 1.09E+01 |
| Rn219 | 3.61E-06 | 1.07E-07 | 6.09E-01 | 7.58E-04 |
| Rn220 | 1.61E-04 | 1.04E-05 | 2.71E+01 | 7.36E-02 |
| Rn222 | 6.91E-05 | 5.05E-09 | 1.16E+01 | 3.58E-05 |
| Ru106 | 1.72E-07 | 1.54E-03 | 2.90E-02 | 1.09E+01 |

NR = Not reported by sites.

Sb125

7.17E-07

¹Decayed to December 1995.

²Total curies estimated by assuming a volume of 168,500 cubic meters for CH-TRU waste and 7,080 cubic meters for RH-TRU waste.

1.21E-01

2.67E-04

1.89E+00

| WIFF DISPUSAL AUDITUCITUE INVENTORY TOL THE COM (CONTINUED | WIPP | Disposal | Radionuclide | Inventory | for the | e CCA | (continued |
|--|------|----------|--------------|-----------|---------|-------|------------|
|--|------|----------|--------------|-----------|---------|-------|------------|

| Nuclide | CH-TRU Waste (Ci/m ³) | RH-TRU Waste (Ci/m ³) | CH-TRU Waste (Total Curies ²) | RH-TRU Waste (Total Curies ²) |
|-------------------|--------------------------------------|--------------------------------------|--|--|
| Sb126 | 8.02E-10 | 4.46E-09 | 1.35E-04 | 3.16E-05 |
| Sb126m | 5.73E-09 | 3.18E-08 | 9.65E-04 | 2.25E-04 |
| Se79 | 2.58E-09 | 1.44E-08 | 4.35E-04 | 1.02E-04 |
| Sm151 | 8.72E-06 | 5.05E-05 | 1.47E+00 | 3.57E-01 |
| <u>Sn119m</u> | 2.46E-11 | 1.35E-10 | 4.14E-06 | 9.59E-07 |
| Sn121m | 1.58E-07 | 9.45E-07 | 2.66E-02 | 6.69E-03 |
| Sn126 | 5.73E-09 | 3.18E-08 | 9.65E-04 | 2.25E-04 |
| Sr 9 0 | 4.07E-02 | 2.95E+01 | 6.85E+03 | 2.09E+05 |
| Ta182 | NR | 5.95E-12 | NR | 4.21E-08 |
| Tc 99 | 1.49E-04 | 8.26E-07 | 2.52E+01 | 5.85E-03 |
| Te125m | 1.75E-07 | 6.57E-05 | 2.95E-02 | 4.65E-01 |
| Te127 | 7.72E-13 | 2.41E-13 | 1.30E-07 | 1.71E-09 |
| Te127m | 7.88E-13 | 2.47E-13 | 1.33E-07 | 1.75E-09 |
| Th227 | 3.56E-06 | 1.06E-07 | 6.01E-01 | 7.47E-04 |
| Th228 | 1.61E-04 | 1.04E-05 | 2.71E+01 | 7.36E-02 |
| Th229 | 1.71E-05 | 1.66E-05 | 2.88E+00 | 1.17E-01 |
| Th230 | 4.78E-07 | 1.07E-06 | 8.06E-02 | 7.56E-03 |
| Th231 | 7.59E-05 | 6.53E-04 | 1.28E+01 | 4.63E+00 |
| Th232 | 5.42E-06 | 1.31E-05 | 9.13E-01 | 9.25E-02 |
| Th234 | 2.35E-04 | 1.48E-03 | 3.96E+01 | 1.05E+01 |

NR = Not reported by sites.

¹Decayed to December 1995.

²Total curies estimated by assuming a volume of 168,500 cubic meters for CH-TRU waste and 7,080 cubic meters for RH-TRU waste.

| Nuclide | CH-TRU Waste (Ci/m ³) | RH-TRU Waste (Ci/m ³) | CH-TRU Waste (Total Curies ²) | RH-TRU Waste (Total Curies ²) |
|---------|--------------------------------------|--------------------------------------|--|--|
| TI207 | 3.61E-06 | 1.07E-07 | 6.07E-01 | 7.56E-04 |
| T1208 | 5.77E-05 | 3.74E-06 | 9.73E+00 | 2.65E-02 |
| TI209 | 3.69E-07 | 3.58E-07 | 6.22E-02 | 2.53E-03 |
| U232 | 1.53E-04 | NR | 2.58E+01 | NR |
| U233 | 1.0 <u>6E-02</u> | 2.23E-02 | 1.79E+03 | 1.58E+02 |
| U234 | 2.76E-03 | 6.03E-03 | 4.65E+02 | 4.27E+01 |
| U235 | 7.59E-05 | 6.53E-04 | 1.28E+01 | 4.63E+00 |
| U236 | 1.98E-06 | 1.37E-05 | 3.33E-01 | 9.68E-02 |
| · U237 | 3.36E-04 | 4.91E-04 | 5.66E+01 | 3.48E+00 |
| U238 | 2.35E-04 | 1.48E-03 | 3.96E+01 | 1.05E+01 |
| U240 | 8.91E-12 | 3.12E-15 | 1.50E-06 | 2.21E-11 |
| Y90 | 4.07E-02 | 2.95E+01 | 6.85E+03 | 2.09E+05 |
| Zr93 | 3.34E-08 | 1.86E-07 | 5.63E-03 | 1.32E-03 |
| Zr95 | 6.80E-15 | 4.27E-05 | 1.15E-09 | 3.02E-01 |
| TOTALS | 3.81E+01 | 1.43E+02 | 6.42E+06 | 1.02E+06 |

WIPP Disposal Radionuclide Inventory for the CCA (continued)

NR = Not reported by sites.

¹Decayed to December 1995.

²Total curies estimated by assuming a volume of 168,500 cubic meters for CH-TRU waste and 7,080 cubic meters for RH-TRU waste.

31-9

ATTACHMENT - B

This attachment summarizes the major changes to the undecayed radionuclide data based on the new information obtained from four sites since the publication of Rev. 2 of the TWBIR. It also summarizes the methodology used to develop the revised WIPP disposal radionuclide inventories shown in Attachment A.

Major Changes in Data

The major changes to the undecayed radionuclide data from the four TRU waste sites (Hanford. Oak Ridge, Rocky Flats, and Savannah River) are summarized below for each site:

- <u>Changes to the Hanford Data</u> There were a few errors in the undecayed curies reported by the Hanford site for Cf-252, Cm-244, and Cm-245 in their previous IDB site submittals for CH-TRU waste. The corrected estimates of yearly activity for these radionuclides that were provided by the Hanford site have been used for the revised radionuclide inventory calculations. The previous and revised undecayed activity values are shown in Table B-1.
- Changes to the Oak Ridge Data In previous IDB submittals, Oak Ridge reported a very conservative (high) inventory for U-235 in the Oak Ridge RH-TRU waste due to the absence of any sampling data. Recently available mass spectrometry analytical data for the evaporator feed tank sludges at Oak Ridge have provided new distributions of the different uranium isotopes in the RH-TRU sludges showing that the primary uranium isotope by mass is U-238 (not U-235). Since the original IDB data are reported in terms of curies (i.e., not on a mass basis), the TWBIR team used the mass spectrometry data to develop new yearly estimates of activities for each uranium isotope. The previous and revised undecayed activities for uranium isotopes in Oak Ridge RH-TRU waste are shown in Table B-2.
- <u>Changes to the RFETS residues data</u> The RFETS residues were not included in any of the previous IDB submittals because they were not categorized as waste. Therefore, no break-ups were available for the yearly undecayed activity contributed by each radionuclide in the residues and consequently, no radionuclide decay calculations could be performed for the residues in Rev. 2 of the TWBIR. Based on recent estimates provided by RFETS, it was possible to divide the total undecayed curies for each radionuclide present in the residues into yearly activities. The yearly break-up of undecayed curies from each of these radionuclides is shown in Table B-3.
- <u>Changes to the SRS data</u> In previous IDB submittals, SRS had reported the total yearly undecayed curies contributed by each radionuclide in SRS CH-TRU waste and therefore no information was available from the IDB regarding the contribution from off-site waste stored at SRS versus on-site waste that was generated at SRS. Based on recent information available from SRS regarding the on-site versus off-site break-up,



the TWBIR team has divided the total yearly undecayed activities reported in previous SRS IDB submittals into yearly undecayed activities from on-site and off-site waste. The original IDB data and the break-ups are shown in Table B-4.

These new estimates of undecayed radionuclide activities for the four sites and unchanged data for all other sites were provided to SNL staff to perform radionuclide activity decay calculations. The undecayed activity data were decayed by SNL staff to the end of 1995 using the code ORIGEN2. The new decayed radionuclide inventory received from SNL staff has been used to develop the revised WIPP disposal radionuclide inventory shown in Attachment A.

Summary of the Methodology

The methodology used for development of the revised radionuclide inventory is the same as that described in Section 3.6 on pages 3-27 through 3-29 of Revision 2 of the TWBIR with the following exceptions:

- Decayed curies have been used for the RFETS residues (instead of the undecayed curies used in Rev. 2 of the TWBIR)
- Unlike Rev. 2 of the TWBIR, the estimated concentration of U-235 in RH-TRU waste in Attachment A is well within transportation limits for Pu-239 FGE and therefore does not require any adjustments.
- The curies and volumes contributed by TRU waste generated off-site but stored at SRS have been excluded from the process of estimating radionuclide activities for SRS waste to be generated in the future. Only the data for waste that has been generated and stored at SRS since 1970 has been used for this estimation. The curies contributed by the off-site waste stored at SRS are added to the WIPP radionuclide inventory (in a manner similar to the RFETS residues) but they are not included in any data extrapolation for future SRS waste.



TABLEB - 4SAVANNAH RIVER SITE

| PREVIOUS IDB CURIES FOR STORED WASTE AT THE BAVANNAH RIVER SITE (ON-SITE + OFF-SITE WASTE) | | | | | | | | | | | | | |
|--|--------------|-------------|----------|----------|----------|----------|----------|----------|-----------|----------|----------|-----------------------|----------|
| | 1970 | 1971 | 1972 | 1973 | 1974 | 1975 | 1776 | 1977 | 1978 | 1979 | 1980 | 1981 | 1982 |
| Am241 | 0.00 + 300.0 | 0.00E+00 | 4.325-01 | 5.28E+00 | 1.97E+01 | 1.89E+01 | 2.58E+01 | 4.00E+01 | 5.14E+01 | 7.062+01 | 7.14E+01 | 1.04E+02 | 9.08E+01 |
| Np237 | 0.00E+00 | 0.00E+00 | 1.335-03 | 1.865-01 | 2.895-01 | 4.016-01 | 2.725-01 | 3.376-01 | 4.646-01 | 1.04E+00 | 5.78E-01 | 5.24E-01 | 8.235-01 |
| Pu238 | 0.00£+00 | 2.08E+05 | 3.49E+04 | 1.462+03 | 3.67E+03 | 4.31E+03 | 0.83E+03 | 7.896+03 | 7.84E+03 | 2.49E+04 | 3,48E+04 | 3.51E+04 | 4.78E+04 |
| Pu239 | 0.00E+00 | 1.27E+02 | 2.37E+01 | 3.06E+01 | 1.138+92 | 1.146+02 | 1.496+02 | 2.30E+02 | 2.00E+02 | 1.632+02 | 3.926+02 | 6.76E+02 | 4.50E+02 |
| Pu240 | 0.00E+00 | #.#4E+01 | 1.17E+01 | 7.47E+00 | 2.732+01 | 2.78E+01 | 3.86E+01 | 6.67E+01 | 6.29E+01 | 4.20E+01 | B.89E+01 | 1.42E+02 | 1.15E+02 |
| Pu241 | 0.00E+00 | 4.502+03 | 7.752+02 | 2.96E+02 | 1.062+03 | 1.98E+04 | 2.21E+03 | 2.17E+03 | 2.44E+03 | 1.86E+03 | 4,06E+03 | 5.696+03 | 4.78E+03 |
| U234 | 0.00E+00 | 0.00€+00 | 0.00E+00 | 0.00E+00 | 2.11E-02 | 3.225-02 | 2.175-02 | 1.076-02 | 3.105-02 | 6.845-02 | 3.995-03 | 6.10 2 -03 | 0.00E+00 |
| U235 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.006+00 | 3.805-04 | 8.555-04 | 4.236-04 | 2.018-04 | 6.835-04 | 1.285-03 | 7.485-05 | 1.185-04 | 4.785-08 |
| U236 | 0.00E+00 | 0.00€+00 | 0.00€+00 | 0.00E+00 | 3.56E-03 | 6.435-03 | 3.005-03 | 1.816-03 | 5.245-03 | 1.165-02 | 6.72E-04 | 1.045-03 | 0.00E+00 |
| U238 | 0.005+00 | 0.002+00 | 0.00€+00 | 00+300.0 | 1.205-05 | 3.895-03 | 8.306-04 | 4.285-09 | 3.975-05 | 4.016-06 | 2.335-04 | 3.61E-08 | 3.695-06 |
| | | | | | | | | | | | | | |
| | 1983 | 1984 | 1985 | 1986 | 1987 | 1988 | 1989 | 1990 | 1991 | 1992 | 1993 | 1994 | TOTAL |
| Am241 | 4.08E+01 | 1.028+02 | 2.48E+02 | 3.37E+02 | 1.69E+02 | 5.46E+02 | 8.636+01 | 0.42E+01 | 1.72E+01 | 3.78E+00 | 7.276-01 | 5.005+00 | 2.11E+03 |
| Np237 | 2.345-01 | 1.772+00 | 2.786-02 | 1.465-02 | 7.545-02 | 3.755-02 | 3,385-02 | 1.388+00 | 8.335-02 | 1.885-03 | 0.00E+00 | 5.875-03 | 8.58E+00 |
| Pu238 | 4.44E+04 | 1.33E+04 | 2.16E+04 | 8.846+03 | 1.44E+04 | 5.60E+03 | 1.73E+03 | 3.005+03 | 2.918+03 | 1.40€+03 | 4.702+03 | 1.78E+04 | 6.67E+06 |
| Pu239 | 2.23E+02 | 4.828+02 | 1.30E+03 | 1.88E+03 | 8.68E+02 | 8.80E+02 | 4.80E+02 | 3.566+02 | \$.50E+01 | 1.006+01 | 6.84E+00 | 3.852+01 | 9.28E+03 |
| Pu244 | 0.04E+01 | 1.16E+02 | 3.252+02 | 4.456+02 | 2.085+02 | 2.04E+02 | 1.14E+02 | 8.478+01 | 2.328+01 | 2.61E+00 | 2,46E+00 | 1.17E+01 | 2.29E+03 |
| Pu241 | 2.70E+03 | 4.51E+03 | 1.24E+04 | 1.69E+04 | 7.86E+03 | 7.75E+03 | 4.308+03 | 3.225+03 | 9.01E+02 | 1.126+02 | 1.386+02 | 0.11E+02 | 1.11E+06 |
| U234 | 1.658-02 | 0.000+00 | 7.505-03 | 1.848-02 | 1.226-02 | 0.146-03 | 2.278-04 | 7.002-04 | 0.002+00 | 2.265-04 | 2.126-02 | 1.805-02 | 3.005-01 |
| U235 | 3.135-04 | 9.945-05 | 1.425-04 | 3.478-04 | 2.295-04 | 1.728-04 | 4.265-06 | 1.302-06 | 4.325-00 | 4.235-06 | 4.195-04 | 3.436-04 | 6.746-03 |
| U236 | 2.785-03 | 0.000+300.0 | 1.275-03 | 3.115-03 | 2.065-03 | 1.646-03 | 3.835-00 | 1.185-04 | 0.00E+00 | 3.805-05 | 3.685-05 | 3.216-03 | 4.705-02 |
| U238 | 2.716-04 | 7.725-04 | 7.506-05 | 1.105-05 | 7.146-08 | 6.70E-06 | 1.335-07 | 3.406-05 | 3.385-07 | 1.325-07 | 4.116-05 | 4.216-04 | 5.70E-03 |

| REVISED | UNDECAY | D CURIES | FOR STOP | ED WAST | E AT THE | AVANNAL | I RIVER ST | TE (ON-SIT | E WASTE) | | | | 1 |
|---------|------------|-------------|------------|------------|----------|-------------------------|---------------------------------------|------------|----------|----------|----------|----------|----------|
| · | 1970 | 1971 | 1972 | 1973 | 1974 | 1975 | 1976 | 1977 | 1978 | 1979 | 1980 | 1981 | 1982 |
| Am241 | 0.000 + 00 | 0.002+00 | 4.325-01 | 6.28E+00 | 1.87E+01 | 1.88E+01 / | 2.585+01 | 4.00E+01 | 6.14E+01 | 7.056+01 | 7,14E+01 | 1.04E+02 | 9.08E+01 |
| Np237 | 0.000 + 00 | 0.000+300.0 | 1.335-03 | 1.865-01 | 2.885-01 | (4.01E-01 ¹ | 2.728-01 | 3.375-01 | 4.545-01 | 1.04E+00 | 6.795-01 | 5.24E-01 | 8.235-01 |
| Pu238 | 0.000 + 00 | 0.006+00 | 4.00E+01 | 1.46E+03 | 3.67E+03 | 4.31E+03 | 6.83E+03 | 7.896+03 | 7.848+03 | 2.49E+04 | 3.48E+04 | 3.51E+04 | 4.79E+04 |
| Pu239 | 0.00E+00 / | 0.00E+00 | 2.48E+00 | 3.06E+01 | 1.13E+02 | 1.14E+02 | 1.40E+02 | 2.306+02 | 2.80E+02 | 1.83E+02 | 3.92E+02 | 5.75E+02 | 4.502+02 |
| Pu240 | 0.00E+00 | 0.00€+00 | 6.655-01 | 7.47E+00 | 2.73E+01 | 2.78E+01 | 3.052+01 | 6.572+01 | 6.29E+01 | 4.285+01 | 9.89E+01 | 1.428+02 | 1.168+02 |
| Pu241 | 0.000 + 00 | 0.00E+00 | 2.15E+01 | 2.962+02 | 1.068+03 | 1.98E+04 | 2.212+03 | 2.178+03 | 2.448+03 | 1.868+03 | 4.062+03 | 5.00E+03 | 4.78E+03 |
| U234 | 0.00€+00 | 0.002+00 | 0.00E+00 | 0.000 + 00 | 2.116-02 | 3.228-02 | 2.178-02 | 1.078-02 | 3.105-02 | 6.845-02 | 3.885-03 | 6.18£-03 | 0.002+00 |
| U235 | 0.00+300.0 | 0.00E+00 | 0.000000 | 0.00E+00 | 3.905-04 | 6.555-04 | 4.238-04 | 2.016-04 | 6.835-04 | 1.285-03 | 7.485-05 | 1.165-04 | 4.785-08 |
| U236 | 0.000 + 00 | 0.00E+00 | 0.000 + 00 | 0.00E+00 | 3.505-03 | 6.435-03 | 3.005-03 | 1.816-03 | 6.246-03 | 1.156-02 | 6.725-04 | 1.048-03 | 0.008+00 |
| U238 | 0.008+00 | 0.00E+00 | 0.005+00 | 0.000 + 00 | 1.205-05 | 3.885-03 | 8,305-04 | 6.285-00 | 3.975-05 | 4.015-05 | 2.335-06 | 3.61E-06 | 3.895-06 |
| | | | | | | | · · · · · · · · · · · · · · · · · · · | | | | | | |
| | 1983 | 1984 | 1985 | 1986 | 1987 | 1795 | 1989 | 1990 | 1991 | 1972 | 1993 | 1994 | TOTAL |
| Am241 | 3.84E+01 | 1.02E+02 | 2.48E+02 | 3.37E+02 | 1.59E+02 | 5.48E+02 | 8.43E+01 | 6.42E+01 | 1.72E+01 | 3.79E+00 | 2.525-01 | 5.80E+00 | 2.11E+07 |
| Np237 | 2.315-01 | 1,77E+00 | 2.785-02 | 1.446-02 | 7.545-02 | 3.755-02 | 3.385-02 | 1.386+00 | 0.335-02 | 1.885-03 | 0.006+00 | 6.876-03 | 8.585+0 |

| | 789 | | 1765 | | | 795 | | | 1991 | | | 1774 | |
|-------|----------|----------|----------|----------|----------|----------|----------|----------|----------|----------|----------|----------|----------|
| Am241 | 3.84E+01 | 1.02E+02 | 2.48E+02 | 3.37E+02 | 1.59E+02 | 5.46E+02 | 8.43E+01 | 6.42E+01 | 1.72E+01 | 3.79E+00 | 2.525-01 | 5.90E+00 | 2.11E+03 |
| Np237 | 2.316-01 | 1,77E+00 | 2.785-02 | 1.466-02 | 7.545-02 | 3.755-02 | 3,385-02 | 1.388+00 | 8.335-02 | 1.885-03 | 0.006+00 | 6.876-03 | 8.582+00 |
| Pu238 | 4.44E+04 | 1.33E+04 | 2.18E+04 | 8.84E+03 | 1.448+04 | 5.50E+03 | 1.73E+03 | 3.005+03 | 2.01E+03 | 1.406+03 | 4.00E+03 | 1.76E+04 | 3.14E+06 |
| Pu239 | 2.168+02 | 4.82E+02 | 1.38E+03 | 1.88E+03 | 8.68E+02 | 8.80E+02 | 4.90E+02 | 3.502+02 | 9.59E+01 | 1.00E+01 | 4.278+00 | 3.05E+01 | 8.13E+03 |
| Pu240 | 5.878+01 | 1.16E+02 | 3.268+02 | 4.45E+02 | 2.08E+02 | 2.04E+02 | 1.148+02 | 8.47E+01 | 2.32E+01 | 2.61E+00 | 1.83E+00 | 1.17E+01 | 2.21E+03 |
| Pu241 | 2.63E+03 | 4.61E+03 | 1.24E+04 | 1.892+04 | 7.98E+03 | 7.768+03 | 4,308+03 | 3.222+03 | 9.01E+02 | 1.12E+02 | 1.14E+02 | 6.11E+02 | 1.06E+05 |
| 10234 | 1.646-02 | 0.00£+00 | 7.50E-03 | 1.845-02 | 1.226-02 | 9.14E-03 | 2.278-04 | 7.065-04 | 0.005+00 | 2.265-04 | 2.10E-02 | 1.90E-02 | 3.005-01 |
| jU235 | 3.116-04 | 8.84E-08 | 1.426-04 | 3.478-04 | 2.295-04 | 1.728-04 | 4.265-06 | 1.388-05 | 4.326-00 | 4.235-06 | 4.14E-04 | 3.635-04 | 6.73E-03 |
| U236 | 2.762-03 | 0.006+00 | 1.278-03 | 3.115-03 | 2.055-03 | 1.646-03 | 3.835-08 | 1.185-04 | 0.00E+00 | 3.805-05 | 4.205-08 | 3.216-03 | 4.695-02 |
| U238 | 2.71E-04 | 7.725-06 | 7.605-06 | 1.105-05 | 7.148-06 | 6.70E-08 | 1.335-07 | 3.405-05 | 3.365-07 | 1.325-07 | 7.09E-07 | 4.218-04 | 5.00E-0J |
| | | | | | | | | | | | | | |

| REVISED | UNDECAYE | D CURIES | FOR STOR | ED WASTE | AT THE | AVANNAH | RIVER ST | TE (OFF-61 | TE WASTE | ONLY) | | | |
|---------|------------|----------|--------------|---------------|----------|----------|------------|------------|------------|----------|----------|------------|-----------|
| | 1976 | 1971 | 1972 | 1973 | | 1975 | 1976 | 1977 | 1978 | 1979 | 1980 | 1981 | 1982 |
| Am241 | 0.00E+00 | 0.00E+00 | Q.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.002+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00€+00 | 0.00E+00 | 0.002+00 |
| Np237 | 0.005+00 | 0.00E+00 | 0.002+00 | 0.000 + 300.0 | 0.00E+00 | 0.00€+00 | 0.00E+00 | 0.00€+00 | 0.00€+00 | 0.00E+00 | 0.00€+00 | 0.00€+00 | 0.00E+00 |
| Pu238 | 0.00E+00 | 2.08E+06 | 3,49E+04 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00€+00 | 0.00E+00 | 0.00£+00 | 0.005+00 |
| Pu239 | 0.00E+00 | 1.278+02 | 2.12E+01 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00€+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00£+00 i | 0.00E+00 |
| Pu240 | 0.00E+00 | 8.84E+01 | 1.118+01 | 0.00€+00 | 0.00E+00 | 0.002+00 | 0.002+00 | 0.002+00 | 0.00E+00 | 0.002+00 | 0.002+00 | 0.005+00 | 0.00E+00 |
| Pu241 | 0.00E+00 | 4.50E+03 | 7.54E+02 | 0.005+00 | 0.00E+00 | 0.00E+00 | 0.00€+00 | 0.00E+00 | 0.005+00 | 0.90E+00 | 0.00€+00 | 0.006+00 | 0.00E+00 |
| U234 | 0.00€+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.006+00 | 0.00E+00 | 0.00€+00 | 0.00E+00 |
| U235 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.005+00 | 0.008+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.005+00 | 0.00E+00 | 0.002+00 | 0.00E+00 | 0.00E+00 |
| U236 | 0.00E+00 | 0.002+00 | 0.005+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.006+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| U238 | 0.00E + 00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.002+00 | 0.002+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00£+00 | 0.000 +00 |
| | | | | | | | | | | | _ | | |
| | 1983 | 1984 | 1985 | 1986 | 1987 | 1988 | 1989 | 1990 | 1991 | 1992 | 1993 | 1994 | TOTAL |
| Am241 | 1.40E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.002+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.000 + 00 | 0.00E+00 | 4.76E-01 | 0.00E+00 | 1.87E+00 |
| Np237 | 3.03E-03 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.002+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00€+00 | 0.000000 | 3.035-03 |
| Pu238 | 4.376-01 | 0.00E+00 | 0.00E+00 | 0.005+00 | 0.00E+00 | 0.00E+00 | 2.57E-01 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 7.31E+00 | 0.00E+00 | 2.43E+05 |
| Pu239 | 7.372+00 | 0.00E+00 | 0.005+00 | 0.002+00 | 0.00E+00 | 0.00E+00 | 1.57E-04 | 0.00E+00 | 0.00E+00 | 0.002+00 | 2.87E+00 | 0.00E+00 | 1.58E+02 |
| Pu240 | 1.74E+00 | 0.002+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | \$.20E-05 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 6.33E-01 | 0.00E+00 | 7.99E+01 |
| Pu241 | 6.67E+01 | 0.008+00 | 0.00E+00 | 0.000+300.0 | 0.00E+00 | 0.00E+00 | 5.56E-03 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 2.39E+01 | 0.00E+00 | 6.34E+03 |
| U234 | 1.185-04 | 0.00E+00 | 0.00 + 300.0 | 0.001+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.002+00 | 2.18E-04 | 0.00E+00 | 3.375-04 |
| U235 | 2.23E-06 | 0.002+00 | 0.000 + 00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.002+00 | 0.00E+00 | 4.61E-06 | 0.00E+00 | 8.84E-08 |
| U236 | 2.008-06 | 0.00E+00 | 0.00E+00 | 0.000 + 00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 3.68E-05 | 0.00E+00 | 6.685-06 |
| U238 | 6.965-09 | 0.00E+00 | 1 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1 0.00E+00 | 0.00E+00 | 1 0.00E+00 | 0.005+00 | 4.04E-05 | 0.00E+00 | 4.04E-05 |

N

TABLE B-3

Rocky Flats Environmental Technology Site

| UNDECAY | ED YEARI | LY ACTIV | ITY DATA | FOR TH | E RFETS | RESIDUES | |
|---------|----------|----------|--------------|----------|----------|-----------|------------|
| | 1982 | 1983 | 1984 | 1985 | 1986 | 1987 | 1988 |
| Am-241 | 2.06E+04 | 2.22E+03 | 6.81E+03 | 1.56E+04 | 9.20E+03 | 7.81E+03 | 1.03E+04 |
| Pu-238 | 1.84E+03 | 1.77E+02 | 5.43E+02 | 1.24E+03 | 7.34E+02 | 6.23E+02 | 8.19E+02 |
| Pu-239 | 3.50E+04 | 3.77E+03 | 1.16E+04 | 2,64E+04 | 1.56E+04 | 1.33E+04 | 1.75E+04 |
| Pu-240 | 8.01E+03 | 8.64E+02 | 2.65E+03 | 6.05E+03 | 3.58E+03 | 3.04E+03 | 4.00E+03 |
| Pu-241 | 2.05E+05 | 2.21E+04 | 6.77E+04 | 1.55E+05 | 9.15E+04 | 7.77E+04 | 1.02E+05 |
| Pu-242 | 1.01E+00 | 1.09E-01 | 3.35E-01 | 7.65E-01 | 4.52E-01 | 3.84E-01 | 5.05E-01 |
| | | | | | | | |
| | 1989 | 1990 | 1 991 | 1992 | 1993 | 1994 | TOTALS |
| Am-241 | 1.74E+04 | 1.57E+04 | 9.386+02 | 1.04E+02 | 3.47E+01 | 1.81E+03 | 1.08E+05 |
| Pu-238 | 1.39E+03 | 1.25E+03 | 7.47E+01 | 8.30E+00 | 2.77E+00 | 1.44E+02 | 8.85E+03 |
| Pu-239 | 2.96E+04 | 2.67E+04 | 1.59E+03 | 1.77E+02 | 5.90E+01 | 3.07E+03 | 1.84E+05 |
| Pu-240 | 6.78E+03 | 6.10E+03 | 3.65E+02 | 4.05E+01 | 1.35E+01 | 7.02E+02 | 4.22E+04 |
| Pu-241 | 1.73E+05 | 1.56E+05 | 9.32E+03 | 1.04E+03 | 3.45E+02 | 1.80E+04 | 1.08E+06 |
| In. 2/2 | 0.575.01 | 7 795 61 | 4 015 00 | E 195 02 | 1 715 02 | 0 000 110 | E 222 . AA |



Ì

TABLE B - 2Oak Ridge National Laboratory

| PREVIOU | S VALUES (|)F URANI | IM ISOTOF | 'ES IN THE | IDB (CUR | (ES) | | | ···· | | | | 1 |
|---------|------------|-----------------|------------------|------------|-----------|----------|----------|----------|----------|----------|----------|----------|----------|
| | 1970 | 1971 | 1972 | 1973 | 1974 | 1975 | 1976 | 1977 | 1978 | 1979 | 1980 | 1981 | 1982 |
| U232 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00/ | 0.00E+00 |
| U233 | 1.25E+00 | 1.25E+00 | 1.25E+00 | 1.25E+00 | 1.56E+00 | 1.75E+00 | 1.29E+00 | 3.36E+00 | 1.25E+00 | 0.00E+00 | 1.00E·01 | 0.00E+00 | 1.00E+00 |
| U234 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.008+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| U235 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 2.01E-04 | 2.84E-04 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| U238 | 3.74E-01 | 3.74E-01 | 3.74E·01 | 3.74E-01 | 3.74E-D1 | 3.74E-01 | 3.74E-01 | 3.74E-01 | 3.74E-01 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| | | | | | | | | | | | | | |
| | 1983 | 1984 | 1985 | 1986 | 1987 | 1988 | 1989 | 1990 | 1991 | 1992 | 1993 | 1994 | TOTAL |
| U232 | 0.00E+00 | 0.00E+00 | 9.28E-02 | 9.28E-02 | 0.28E-02 | 9.28E-02 | 9.28E-01 |
| U233 | 0.00E+00 | 0.00E+00 | 5.37E+00 | 5.37E+00 | 5.37E+00 | 5.37E+00 | 5.37E+00 | 5.37E+00 | 5.37E+00 | 5.37E+00 | 5.37E+00 | 5.37E+00 | 8.90E+01 |
| U234 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.008+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| U235 | 0.00E+D0 | 0.00E+00 | 1.75E+01 | 1.75E+01 | 1.76E+01 | 1.75E+01 | 1.75E+02 |
| U238 | 0.00E+00 | 0.00E+00 | 0.00£+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 3.37E+00 |

}

B1-14

| REVISED | UNDECAY | ED ACTIVI | TY FOR E | ACH URAN | NIUM ISOT | OPE (CUR | IES) | | | | | | 1 |
|---------|----------|-----------|----------|----------|-----------|----------|----------|----------|----------|------------|----------|----------|----------|
| | 1970 | 1971 | 1972 | 1973 | 1974 | 1975 | 1976 | 1977 | 1978 | 1979 | 1980 | 1981 | 1982 |
| U233 | 5.58E+00 | 5.58E+00 | 5.58E+00 | 5.58E+00 | 5.78E+00 | 5.88E+00 | 5.60E+00 | 6.86E+00 | 5.58E+00 | 0.00E+00 | 6.09E-02 | 0.00E+00 | 6.08E-01 |
| U234 | 1.25E·01 | 1.25E-01 | 1.25E-01 | 1.25E·01 | 1.25E·01 | 1.25E-01 | 1.25E-01 | 1.25E·01 | 1.25E·01 | 0.00E+00 | 4.58E-07 | 0.00E+00 | 4.55E-06 |
| U235 | 5.69E-03 | 5.69E 03 | 5.69E-03 | 5.69E-03 | 5.69E·03 | 5.89E-03 | 5.69E-03 | 6.69E·03 | 5.69E·03 | 0.00E+00 | 2.07E-08 | 0.00E+00 | 2.07E-07 |
| U236 | 3.45E-03 | 3.45E-03 | 3.45E·03 | 3.45E-03 | 3.45E-03 | 3.45E-03 | 3.45E-03 | 3.45E-03 | 3.45E-03 | 0.00E+00 | 1.26E-08 | 0.00E+00 | 1.28E·07 |
| U238 | 3.73E-01 | 3.73E-01 | 3.73E-01 | 3.73E·01 | 3.73E-01 | 3.73E-01 | 3.73E-01 | 3.73E·01 | 3.73E-01 | 0.00+300.0 | 1.36E-06 | 0.00E+00 | 1.38E-05 |
| | | | | | | | | | | | | | |
| | 1983 | 1984 | 1985 | 1986 | 1987 | 1988 | 1989 | 1990 | 1991 | 1992 | 1993 | 1994 | TOTAL |
| U233 | 0.00E+00 | 0.00E+00 | 3.83E+01 | 3.83E+01 | 3.83E+01 | 3.83E+01 | 3.83E+01 | 3.83E+01 | 3.83E+01 | 3.83E+01 | 3.83E+01 | 3.83E+01 | 4.36E+02 |
| U234 | 0.00E+00 | 0.00E+00 | 9.10E-01 | 9.10E-01 | 9.10E-01 | 9.10E·01 | 9.10E-01 | 9.10E-01 | 9.10E-01 | 9.10E-01 | 9.10E-01 | 9.10E-01 | 1.02E+01 |
| U235 | 0.00E+00 | 0.00E+00 | 5.01E-02 | 5.01E-02 | 5.01E-02 | 5.01E-02 | 5.01E-02 | 5.01E·02 | 5.01E·02 | 5.01E-02 | 5.01E-02 | 5.01E-02 | 5.53E-01 |
| U236 | 0.00E+00 | 0.00E+00 | 2.51E-02 | 2.51E-02 | 2.51E-02 | 2.51E-02 | 2.51E-02 | 2.51E-02 | 2.51E-02 | 2.51E-02 | 2.51E-02 | 2.51E 02 | 2.82E-01 |
| U238 | 0.00E+00 | 0.00E+00 | 2.71E+00 | 2.71E+00 | 2.71E+00 | 2.71E+00 | 2.71E+00 | 2.71E+00 | 2.71E+00 | 2.71E+00 | 2.71E+00 | 2.71E+00 | 3.05E+01 |

TABLE B - 1

Hanford Site

| PREVIOU | S UNDECA | YED CUR | IES FOR | Cf-252, Cm | -244, and (| m-245 IN | CH-TRU W | ASTE AT | THE HAN | FORD SITI | 2 | | |
|---------|----------|----------|------------|------------|-------------|----------|----------|----------|----------|-----------|----------|----------|----------|
| | 1970 | 1971 | 1972 | 1973 | 1974 | 1975 | 1976 | 1977 | 1978 | 1979 | 1980 | 1981 | 1982 |
| Cf252 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00€+00 | 0.00E+00 | 0.00E+00 | 1.07E+03 |
| Cm244 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.006+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| Cm245 | 0.00E+00 | 0.00E+00 | 0.00E + 00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 3.42E-01 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 7.54E+00 |

| | 1983 | 1984 | 1985 | 1986 | 1987 | 1988 | 1989 | 1990 | 1991 | 1992 | 1993 | 1994 | TOTAL |
|-------|----------|----------|----------|----------|----------|----------|----------|----------|----------|----------|----------|----------|----------|
| Cf252 | 0.00E+00 | 1.07E+03 |
| Cm244 | 0.00E+00 | 7.62E-01 | 6.72E-03 | 7.58E+01 | 0.00E+00 | 0.00E+00 | 7.66E+01 |
| Cm245 | 3.59E-01 | 1.71E-03 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 8.54E+00 | 0.00E+00 | 0.00E+00 | 0.008+00 | 0.00E+00 | 0.00E+00 | 1.68E+01 |

B1-15

| REVISED | UNDECAY | ED CURI | ES FOR C | f-252, Cm-2 | 44, and Cn | n-245 IN C | H-TRU W | ASTE AT | THE HAN | FORD SIT | E | | |
|---------|----------|----------|----------|-------------|------------|------------|----------|----------|----------|----------|----------|----------|----------|
| | 1970 | 1971 | 1972 | 1973 | 1974 | 1975 | 1976 | 1977 | 1978 | 1979 | 1980 | 1981 | 1982 |
| Cf252 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.08E-03 |
| Cm244 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.62E+02 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 3.72E+02 |
| Cm245 | 0.008+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00€+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |

| Г | | 1983 | 1984 | 1985 | 1986 | 1987 | 1988 | 1989 | 1990 | 1991 | 1992 | 1993 | 1994 | TOTAL |
|----|------|----------|----------|----------|------------|----------|----------|----------|----------|----------|----------|----------|----------|----------|
| Cf | 252 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.08E-03 |
| Cn | n244 | 1.70E+02 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 4.04E+03 | 7.62E-01 | 6.72E-03 | 7.58€+01 | D.DOE+00 | 0.00E+00 | 4.82E+03 |
| Сп | n245 | 0.00E+00 | 1.71E-03 | 0.00E+00 | 0.00E + 00 | 0.00E+00 | 1.71E-03 |

APPENDIX B - 2

_





Department of Energy

Carlsbad Area Office P. O. Box 3090 Carlsbad, New Mexico 88221

June 12, 1996

To: Dr. Les E. Shephard, Director, SNL

Subject: Preliminary Activities for Selected Radionuclides for CH-TRU Waste Streams

The following information from the Transuranic (TRU) Waste Baseline Inventory Report (TWBIR) team was requested during a meeting with SNL representatives on April 23, 1996. The TWBIR team was requested to calculate the radionuclide activity (total curies) for seven radionuclides (Am-241, Cm-244, Pu-238, Pu-239, Pu-240, Pu-241, and U-234) on a waste stream basis for contact-handled (CH)-TRU waste to be disposed of at the WIPP.

During this meeting, it was agreed that since the radionuclide data used by SNL WIPP PA were based on the site-level radionuclide data from the Integrated Data Base (IDB), the waste stream radionuclide data in curies per cubic meter provided by the DOE sites in Revision 2 of the Transuranic Waste Baseline Inventory Report (TWBIR) would be normalized to the extent necessary for consistency with the IDB data. This letter summarizes the methodology for normalization of the waste stream radionuclide data from the TWBIR Rev. 2 and subsequent scale-up of the normalized data to obtain estimates of the total curies of each of the seven selected radionuclides on a waste stream basis. The results of these calculations are presented in Table 1. Please note that the results in Table 1 are not directly obtainable from the TWBIR database; but all of the data in Table 1 are derived from TWBIR Rev. 2 on the basis of the methodology and assumptions discussed later in this memorandum.

Methodology for Normalization of the Waste Stream Radionuclide Data

The waste stream radionuclide data provided by the sites in TWBIR Rev. 2 were first normalized to be consistent with the site-wide values reported for CH-TRU waste in the IDB using the following step-by-step approach:

Extraction of Volume and Activity Data from the TWBIR Rev. 2 Database -For each CH-TRU waste stream, the stored and projected final waste form volumes as well as activities in curies per cubic meter (Ci/m³) reported by the sites for the seven selected radionuclides were obtained from the database. All RH-TRU waste streams, non-WIPP waste streams, and waste streams for which no data were reported by the site were excluded.



CAO:NTP:RLB 96-1199

Dr. Les E. Shephard, SNL - 2 -

- Estimation of Undecayed Total Activity for Each Radionuclide at Each Site -The Ci/m³ value reported for each radionuclide for each waste stream was multiplied by the stored waste volume to obtain the total undecayed activity of each radionuclide for each waste stream. Next, the total undecayed activity for a given radionuclide (e.g., Pu-238) for all waste streams at a given site were added together to obtain the total undecayed activity for each radionuclide at each site.
- <u>Comparison with IDB Values and Normalization</u> The total undecayed activity estimated above for a given radionuclide at a given site were compared with the values reported for the same radionuclide by the same site in their IDB submittal. Based on this comparison, a normalization factor (NF) was developed for each radionuclide at each site as follows:

NF = <u>Total curies reported by the site in the IDB</u> Total curies estimated from TWBIR Rev. 2 waste stream data

The NFs calculated in this fashion are shown in Table 2. The total activity for each radionuclide for each waste stream was then multiplied by the normalization factor to obtain the total normalized undecayed stored curies on a waste stream basis.

• Estimation of Decayed Activities - For each radionuclide at each site, a ratio of the activity decayed to the end of 1995 to the undecayed activity for each of seven selected radionuclides was calculated based on the ORIGEN2 activity decay calculations performed by SNL staff in support of the development of the WIPP disposal radionuclide inventory for the Compliance Certification Application (CCA). The total normalized undecayed stored curies were then multiplied by this calculated ratio to estimate the decayed curies of each radionuclide that are present in the stored volume of each waste stream. Subsequently, the curies from the stored volume were multiplied by the ratio of the projected to the stored volume to obtain the estimated curies for the projected volume of each waste stream.

Methodology for Scale-up of Waste Stream Decayed Activity to WIPP Repository Volume

This step involves scale-up of the estimated decayed activity for each radionuclide present in the stored volume of each waste stream to the WIPP disposal volume for CH-TRU waste, which is 168,500 m³. Since the total WIPP activity for CH-TRU waste for each radionuclide has already been estimated in an earlier memorandum prepared in support of the CCA, it was assumed that the total WIPP activity in curies for each of the seven radionuclides would be equal, for the sake of consistency, to the values reported in the earlier memorandum. For each

CAO:NTP:RLB 96-1199

radionuclide, a scale-up factor for activity was calculated as follows:

SF, = <u>Total WIPP Activity from CCA memo - Total Estimated Activity for Stored Volume (all waste streams)</u> Total Estimated Activity for Projected Volume (for all waste streams)

These SF₂'s are shown in Table 3. The estimated activity in curies for the projected volume for each radionuclide for each waste stream was then multiplied by the appropriate scale-up factor derived above, and the result added to the corresponding estimated stored activity in curies to obtain the "Scaled Curies" at a WIPP level for the waste stream. These are the values reported in Table 1.

Methodology for Scale-up of Waste Stream Volumes to WIPP Repository Volume

The summation of the total stored and projected volumes for all CH-TRU waste streams is less than the WIPP disposal capacity for CH-TRU waste (i.e., 168,500 m³). However, since the WIPP PA modeling is based on the effect of a full repository (i.e., 168,500 m³ for CH-TRU waste), it is necessary to scale-up the total volume of each waste stream in order to be consistent with the WIPP PA assumptions. This step involves the scale-up of the total volume of each waste stream to the WIPP disposal capacity for CH-TRU waste. A scale-up factor for volume (common to all waste streams) was calculated as follows:

SF_x = <u>WIPP Capacity for CH-TRU Waste (168,500 m³) - Total Stored Volume (all waste streams)</u> Total Projected Volume (for all waste streams)

This factor is shown in Table 4. The projected volume for each waste stream was then multiplied by the scale-up factor derived above, and the result added to the corresponding stored volume to obtain the "Scaled Volume" at a WIPP level for each waste stream. These are the values reported in Table 1.

If you have any questions concerning the enclosed information, please contact Mr. Russ Bisping of my staff at (505) 234-7446.

Manager National TRU Program



Enclosures

Dr. Les E. Shephard, SNL

- 4 -

cc w/enclosures: M. McFadden, CAO R. Bisping, CAO S. Chakraborti, CTAC / P. Drez, DEA J. Harvill, CTAC R. Anderson, SNL L. Sanchez, SNL M. Chu, SNL M. Marietta, SNL



TABLE - 1

SCALED VOLUME AND ACTIVITIES FOR SELECTED RADIONUCLIDES FOR EACH WASTE STREAM

| | Weste | Scaled | SCALED T | OTAL CURIES OF | EACH RADIONUC | LIDE FOR EACH | WASTE STREAM | | |
|------|--------------|-------------|---------------|----------------|-----------------|-----------------|---------------|---------------|--------------|
| SITE | Stream ID# | Volume (m3) | Scaled Am-241 | Scaled Cm-244 | Scaled Pu-238 | Scaled Pu-239 | Scaled Pu-240 | Scaled Pu-241 | Scaled U-234 |
| IN | IN-W139.627 | 12.27 | 2.84E+01 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W146.699 | 2.29 | 8.24E-01 | 4.91E+02 | 7.98E-01 | 6.40E-01 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| | IN-W157.144 | 49.92 | 8.51E+00 | 0.00E+00 | 1.52E+00 | 4.22E+01 | 9.31E+00 | 1.74E+02 | 0.00E+00 |
| IN | IN-W157.906 | 163.70 | 2.79E+01 | 0.00E+00 | 5.00E+00 | 1.38E+02 | 3.05E+01 | 5.69E+02 | 0.00E+00 |
| ГN | IN-W157.907 | 9.36 | 3.19E+00 | 0.00E+00 | 5.71E-01 | 1.58E+01 | 3.49E+00 | 6.51E+01 | 0.00E+00 |
| IN | IN-W159.1072 | 0.68 | 0.00E+00 | 0.00E+00 | 5.05E+02 | 3.67E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| | IN-W159.119 | 0.21 | 0.00E+00 | 0.00E+00 | 5.15E+01 | 3.74E-01 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W159.120 | 0.42 | 0.00E+00 | 0.00E+00 | 6.17E+02 | 4.49E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| | IN-W161.231 | 97.55 | 5.22E+00 | 0.00E+00 | 1.31E+01 | 3.63E+02 | 8.02E+01 | 1.49E+03 | 0.00E+00 |
| in | IN-W161.806 | 15.79 | 8.44E-01 | 0.00E+00 | 2.12E+00 | 5.88E+01 | 1.30E+01 | 2.42E+02 | 0.00E+00 |
| IN | IN-W163.1007 | 0.68 | 0.00E+00 | 0.00E+00 | 5.11E-01 | 1.42E+01 | 3.13E+00 | 5.83E+01 | 0.00E+00 |
| โเพ | IN-W163.234 | 0.42 | 0.00E+00 | 0.00E+00 | 6.25E-01 | 1.73E+01 | 3.82E+00 | 7.13E+01 | 0.00E+00 |
| IN T | IN-W164.1060 | 1.66 | 0.00E+00 | 0.00E+00 | 2.38E-02 | 6.60E-01 | 1.46E-01 | 2.72E+00 | 0.00E+00 |
| IN | IN-W164.153 | 0.89 | 0.00E+00 | 0.00E+00 | 1.27E-02 | <u>3.52E-01</u> | 7.78E-02 | 1,45E+00 | 0.00E+00 |
| ี้เพ | IN-W166.151 | 16.00 | 4.80E-01 | 0.00E+00 | 2.08E+00 | 5.78E+01 | 1.27E+01 | 2.38E+02 | 0.00E+00 |
| IN | IN-W166.928 | 56.78 | I.70E+00 | 0.00E+00 | 7.40E+00 | 2.05E+02 | 4.53E+01 | 8.44E+02 | 0.00E+00 |
| IN | IN-W167.149 | 36.68 | 1.72E+00 | 0.00E+00 | 1.05E+00 | 2.90E+01 | 6.41E+00 | 1.19E+02 | 0.00E+00 |
| IN | IN-W167.926 | 131.46 | 6.16E+00 | 0.00E+00 | 3.75E+00 | 1.04E+02 | 2.30E+01 | 4,28E+02 | 0.00E+00 |
| IN | IN-W169.191 | 4267.12 | 1.79E+03 | 0.00E+00 | 8.48E+01 | 2.35E+03 | 5.19E+02 | 9.67E+03 | 0.00E+00 |
| IN | IN-W169.192 | 14.56 | 6.12E+02 | 0.00E+00 | 2.89E+01 | \$.02E+02 | 1.77E+02 | 3,30E+03 | 0.00E+00 |
| IN | IN-W169.985 | 41.79 | 1.76E+01 | 0.00E+00 | 8.31E-01 | 2.30E+01 | 5.08E+00 | 9,47E+01 | 0.00E+00 |
| IN | IN-W170.189 | 0.68 | 3.88E+00 | 0.00E+00 | 0.00E+00 | 1.29E+01 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W170.938 | 0.42 | 2.37E+00 | 0.00E+00 | 0.00E+00 | 7.91E+00 | 0.00E+00 | 0,00E+00 | 0.002+00 |
| ÎN | IN-W171.184 | 3.54 | 1.57E+00 | 0.00E+00 | 0.00E+00 | 1.67E+01 | 0.00E+00 |],10E+02 | 0.000000 |
| IN | IN-W171.801 | 0.68 | 3.01E-01 | 0.00E+00 | 0.00E+00 | 3.21E+00 | 0.00E+00 | 2,23E+U1 | 0.000000 |
| IN | IN-W174.1082 | 30.37 | 0.00E+00 | 0.00E+00 | 4.35E+02 | 2.84E-UI | 5.308-01 | 0,002+00 | 0.000000 |
| IN | IN-W174.154 | 134.32 | 0.00E+00 | 0.00E+00 | 1.92E+03 | 1.26E+00 | 2.43E+00 | 0.006700 | 0.000000 |
| IN | IN-W177.1083 | 141.02 | 0.00E+00 | 0.00E+00 | 2.32E+03 | 6.71E-01 | 4.00E-03 | 1.88E-UI | 0.002+00 |
| IN | IN-W177.156 | 39.23 | 0.00E+00 | 0.00E+00 | 6.44E+02 | 1.87E-01 | 1.11E-03 | 3.226-02 | 0.000000 |
| IN | IN-W179.1084 | 4.58 | 0.00£+00 | 0.00E+00 | 2.99E+01 | 3.05E-04 | 1.57E-04 | 1.648-02 | 0.002+00 |
| IN | IN-W179.158 | 1.51 | 0.00E+00 | 0.00E+00 | 9.88E+00 | 1.67E-04 | 8.48E-03 | 3.4[E-03 | 0.002+00 |
| IN | IN-W181.162 | 9.57 | 0.00E+00 | 0.00E+00 | 1.08E-01 | 2.99E+00 | 0.01E-UI | 1,236401 | 0.000000 |
| IN | IN-W186.187 | 2695.26 | 2.00E+02 | 0.00E+00 | 5.50E+01 | 1.536+03 | 3.376402 | 6,1/ETU3 | 0.00E+00 |
| IN | IN-W187.1094 | 0.68 | 0.00E+00 | 0.00E+00 | 6.84E-02 | 1.89E+00 | 4.18E-01 | 7,798,400 | 0.002+00 |
| IN | IN-W187.121 | 0.21 | 0.00E+00 | 0.00E+00 | 4.18E-02 | 1.16E+00 | 2.36E-01 | 4,//E+UU | 0.002+00 |
| IN | IN-W188.1093 | 1.04 | 0.00E+00 | 0.00E+00 | 4.26E-02 | 1.156400 | 1.000-01 | 1.63CTV0 | |
| IN | IN-W188,160 | 0.68 | 0.00E+00 | 0.00E+00 | 2.78E-02 | 7.726-01 | 1.708-01 | 3,1/E100 | 0.002+00 |
| IN | IN-W189,1048 | 4.99 | 0.00E+00 | 0.00E+00 | 1.368-01 | 3.77E+00 | B.J32-01 | 1,JJETUI | 0.000 100 |
| IN | IN-W189.131 | 1.72 | 0.00E+00 | 0.00E+00 | 4.69E-02 | 1.3UE+00 | 2.8/E-UL | 3,336400 | |
| IN | IN-W197.196 | 2.29 | 2.09E+02 | 0.00E+00 | <u>5.19E+00</u> | 1.44E+02 | 3.18E+01 | 3.926+02 | 0.008+00 |

TABLE - 1 SCALED VOLUME AND ACTIVITIES FOR SELECTED RADIONUCLIDES FOR EACH WASTE STREAM

| | Waste | Scated | SCALED T | OTAL CURIES OF | EACH RADIONU | CLIDE FOR EACH | WASTE STREAM | | |
|------|--------------|-------------|---------------|-----------------------|---------------|----------------|---------------|---------------|--------------|
| SITE | Stream 1D# | Volume (m3) | Scaled Am-241 | Scaled Cm-244 | Scaled Pu-238 | Scaled Pu-239 | Scaled Pa-240 | Scaled Pu-241 | Scaled U-234 |
| IN | IN-W197.802 | 510.22 | 4.67E+02 | 0.00E+00 | 1.16E+01 | 3.21E+02 | 7.08E+01 | 1.32E+03 | 0.00E+00 |
| IN | IN-W197.803 | 45.23 | 4,14E+01 | 0.00E+00 | 1.03E+00 | 2.85E+01 | 6.28E+00 | 1.17E+02 | 0.00E+00 |
| เพิ่ | IN-W198.202 | 119.60 | 2.16E+02 | 0.00E+00 | 3.53E+00 | 9.78E+01 | 2.16E+01 | 4.02E+02 | 0.00E+00 |
| IN | IN-W198.203 | 0.21 | 3.75E+01 | 0.00E+00 | 6.14E-01 | 1.70E+01 | 3.75E+00 | 7.00E+01 | 0.00E+00 |
| IN | IN-W198.804 | 32.82 | 5.92E+01 | 0.00E+00 | 9.69E-01 | 2.68E+01 | 5.93E+00 | 1.10E+02 | 0.00E+00 |
| IN | IN-W199.1039 | 0.89 | 0,00E+00 | 0.00E+00 | 1.10E-01 | 3.04E+00 | 6.70E-01 | 1.25E+01 | 0.00E+00 |
| IN | IN-W199.209 | 0.21 | 0.00E+00 | 0.00E+00 | 2.57E+00 | 7.11E+01 | 1.57E+01 | 2.92E+02 | 0.00E+00 |
| IN | IN-W202.1092 | 0.89 | 0.00E+00 | 0.00E+00 | 7.20E-03 | 2.00E-01 | 4.40E-02 | 8.21E-01 | 0.00E+00 |
| IN | IN-W202.224 | 109.62 | 0.00E+00 | 0.00E+00 | 8.88E-01 | 2.46E+01 | 5.43E+00 | 1.01E+02 | 0.00E+00 |
| IN | IN-W203.1081 | 0.68 | 1.28E-01 | 0.00E+00 | 5.78E-01 | 1.38E-02 | 6.54E-03 | 1.69E-03 | 0.00E+00 |
| IN | IN-W203.210 | 73.22 | 1.37E+01 | 0.00E+00 | 6.22E+01 | 1.48E+00 | 7.04E-01 | 1.\$2E-01 | 0.00E+00 |
| เท | IN-W203.211 | 3.33 | 1.25E+00 | 0.00E+00 | 5.66E+00 | 1.35E-01 | 6.40E-02 | 1.65E-02 | 0.00E+00 |
| IN | IN-W203.212 | 0.21 | 1.30E-02 | 0.00E+00 | 5.89E-02 | 1.41E-03 | 6.67E-04 | 1.72E-04 | 0.00E+00 |
| IN | IN-W204.215 | 0.89 | 7.56E+00 | 0.00E+00 | 7.68E+00 | 1.22E-02 | 4.01E-03 | 1.80E-01 | 0.00E+00 |
| IN | IN-W204.216 | 1.66 | 1.42E+01 | 0.00E+00 | 1.44E+01 | 2.29E-02 | 7.52E-03 | 3.36E-01 | 0.00E+00 |
| IN | IN-W204.217 | 0.21 | 5.90E-01 | 0.00E+00 | 5.99E-01 | 9.55E-04 | 3.13E-04 | 1.40E-02 | 0.00E+00 |
| IN | IN-W205.1086 | 0.83 | 0.00E+00 | 0.00E+00 | 1.49E-03 | 4.12E-02 | 9.09E-03 | 1.69E-01 | 0.00E+00 |
| ÎN | IN-W205.1087 | 0.21 | 0.00E+00 | 0.00E+00 | 3.72E-02 | 1.03E+00 | 2.27E-01 | 4.24E+00 | 0.00E+00 |
| IN | IN-W205.220 | 0.68 | 0.00E+00 | 0.00E+00 | 1.22E-03 | 3.37E-02 | 7.43E-03 | 1.39E-01 | 0.00E+00 |
| IN | IN-W206.935 | 10.89 | 4.82E-01 | 0.00E+00 | 5.41E-01 | 1.50E+01 | 3.31E+00 | 6.17E+01 | 0.00E+00 |
| IN | IN-W206.936 | 22.46 | 1.66E+01 | 0.00E+00 | 1.86E+01 | 5.15E+02 | 1.14E+02 | 2.12E+03 | 0.00E+00 |
| IN | IN-W207.238 | 0.21 | 0.00E+00 | 0.00E+00 | 1.65E+00 | 4.56E+01 | 1.01E+01 | 1.88E+02 | 0.00E+00 |
| IN | IN-W207.980 | 0.89 | 0.00E+00 | 0.00E+00 | 4.22E-01 | 1.17E+01 | 2.58E+00 | 4.81E+01 | 0.00E+00 |
| IN | IN-W207.981 | 0.42 | 0.00E+00 | 0.00E+00 | 1.97E-01 | 5.47E+00 | 1.21E+00 | 2.25E+01 | 0.00E+00 |
| IN | IN-W208.242 | 1.46 | 2.13E+01 | 0.00E+00 | 2.31E+00 | 6.41E+01 | 1.42E+01 | 2.64E+02 | 0.00E+00 |
| IN | IN-W208.988 | 2.34 | 2.06E+00 | 0.00E+00 | 2.24E-01 | 6.20E+00 | 1.37E+00 | 2.55E+01 | 0.00E+00 |
| IN | IN-W209.244 | 3.12 | 6.70E-01 | 0.00E+00 | 1.10E+01 | 3.06E+02 | 6.76E+01 | 1.26E+03 | 0.00E+00 |
| IN | IN-W209.994 | 10.27 | 1.32E-01 | 0.00E+00 | 2.18E+00 | 6.04E+01 | 1.33E+01 | 2.49E+02 | 0.00E+00 |
| IN | IN-W210.1001 | 1.10 | 0.00E+00 | 0.00E+00 | 8.83E-02 | 2.45E+00 | 5.40E-01 | 1.01E+01 | 0.00E+00 |
| IN | IN-W210.247 | 0.21 | 0.00E+00 | 0.00E+00 | 2.79E-01 | 7.74E+00 | 1.71E+00 | 3.18E+01 | 0.00E+00 |
| IN | 1N-W211.1009 | 98.47 | 8.53E+01 | 0.00E+00 | 3.64E+01 | 1.01E+03 | 2.23E+02 | 4.15E+03 | 0.00E+00 |
| IN | IN-W211.249 | 22.46 | 3.24E+02 | 0.00E+00 | 1.38E+02 | 3.83E+03 | 8.46E+02 | 1.58E+04 | 0.00E+00 |
| IN | IN-W212.1058 | 3.44 | 1.03E-01 | 0.00E+00 | 4,75E-02 | 1.32E+00 | 2.90E-01 | 5.41E+00 | 0.00E+00 |
| IN | IN-W212.251 | 150.59 | 7.50E+01 | 0.00E+00 | 3.47E+01 | 9.60E+02 | 2.12E+02 | 3.95E+03 | 0.00E+00 |
| IN | IN-W213.1069 | 1.93 | 0.00E+00 | 0.00E+00 | 1.01E+03 | 5.96E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W213.252 | 0.42 | 0.00E+00 | 0.00E+00 | 3.62E+03 | 2.14E+01 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W213.253 | 0.21 | 0.00E+00 | 0.00E+00 | 3.62E+01 | 2.14E-01 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W214.1075 | 0.62 | 0.00E+00 | 0.00E+00 | 4.51E+02 | 3.93E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W214.755 | 0.68 | 0.00E+00 | 0.00E+00 | 4.92E+02 | 4.29E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W214.756 | 0.21 | 001300.0 | 0.00E+00 | 5.01E+01 | 4.36E-01 | 0.0012+00 | 0.00E100 | 0.0012+00 |

Page 2

}

| TABLE - 1 |
|---|
| SCALED VOLUME AND ACTIVITIES FOR SELECTED RADIONUCLIDES FOR EACH WASTE STREAM |

| | Waste | Scaled | SCALED T | OTAL CURIES OF | EACH RADIONUC | LIDE FOR EACH | WASTE STREAM | | |
|------|--------------|-------------|---------------|----------------|---------------|---------------|-----------------------|----------------------|--------------|
| SITE | Stream 1D# | Volume (m3) | Scaled Am-241 | Scaled Cm-244 | Scaled Pu-238 | Scaled Pu-239 | Scaled Pu-240 | Scaled Pu-241 | Scaled U-234 |
| IN | IN-W216.875 | 1478.88 | 4.26E+04 | 0.00E+00 | 5.67E+01 | 1.57E+03 | 3.47E+02 | 6.46E+03 | 0.00E+00 |
| IN | IN-W216.98 | 555.65 | 1.60E+04 | 0.00E+00 | 2.13E+01 | 5.90E+02 | 1.30E+02 | 2.43E+03 | 0.00E+00 |
| IN | IN-W216.99 | 255.01 | 1.47E+04 | 0.00E+00 | 1.95E+01 | 5.42E+02 | 1.20E+02 | 2.23E+03 | 0.00E+00 |
| IN | IN-W218.109 | 183.87 | 3.17E+02 | 0.00E+00 | 1.91E+00 | 5.30E+01 | 1.17E+01 | 2.18E+02 | 0.00E+00 |
| IN | IN-W218.909 | 101.91 | 8.77E+01 | 0.00E+00 | 5.30E-01 | 1.47E+01 | 3.24E+00 | 6.04E+01 | 0.00E+00 |
| IN | IN-W220.114 | 122.80 | 8.39E+02 | 0.00E+00 | 2.49E+00 | 7.42E+01 | 1.59E+01 | 2.84E+02 | 0.00E+00 |
| | IN-W220.925 | 443,04 | 3.03E+03 | 0.00E+00 | 8.98E+00 | 2.68E+02 | 5.74E+01 | 1.03E+03 | 0.00E+00 |
| | IN-W221.113 | 11.65 | 0.00E+00 | 0.00E+00 | 6.71E-01 | 1.86E+01 | 4.10E+00 | 7.65E+01 | 0.00E+00 |
| IN | IN-W221.927 | 3.65 | 0.00E+00 | 0.00E+00 | 2.10E-01 | 5.82E+00 | 1.29E+00 | 2.40E+01 | 0.00E+00 |
| IN T | IN-W222.116 | 24.75 | 3.71E-01 | 0.00E+00 | 7.19E+00 | 1.99E+02 | 4.40E+01 | 8.20E+02 | 0.000000 |
| IN | IN-W222.117 | 39.10 | 1.17E+00 | 0.00E+00 | 2.27E+01 | 6.30E+02 | 1.39E+02 | 2.398+03 | 0.000000 |
| IN | IN-W222.965 | 10.61 | 1.59E-01 | 0.00E+00 | 3.08E+00 | 8.54E+01 | 1.89E+01 | 3.52E+02 | 0.002+00 |
| IN | IN-W225.127 | 21.63 | 9.85E-02 | 0.00E+00 | 1.80E-01 | 4.98E+00 | 1.10E+00 | 2.058+01 | 0.002+00 |
| IN | IN-W225.800 | 1.10 | 4.99E-03 | 0.00E+00 | 9.11E-03 | 2.53E-01 | 5.37E-02 | 1.048+00 | 0.002100 |
| IN | IN-W228.101 | 287.33 | 1.18E+02 | 0.00E+00 | 8.74E-01 | 2.42E+01 | 5.33E+00 | 9,902+UI | 0.002100 |
| IN | IN-W228.102 | 198.85 | 1.63E+02 | 0.00E+00 | 1.21E+00 | 3.35E+01 | 7.40E+00 | 1.386+02 | 0.000+00 |
| IN | IN-W228.103 | 31.82 | 4.36E+00 | 0.00E+00 | 3.23E-02 | 8.94E-D1 | 1.972-01 | 3.885+00 | 0.000000 |
| | IN-W228.883 | 608.82 | 2.50E+02 | 0.00E+00 | 1.85E+00 | 5.13E+01 | 1.13E+01 | 2.112+02 | 0.000700 |
| IN | IN-W230.229 | 4.27 | 2.41E-02 | 0.00E+00 | 1.19E+00 | 3.31E+01 | 7.31E+00 | 1.308+02 | 0.0000+00 |
| IN | IN-W230.940 | 14.77 | 8.32E-02 | 0.00E+00 | 4.13E+00 | 1.14E+02 | 2.33E+01 | 9./12+02 | |
| IN | IN-W240.272 | 167.65 | 6.48E+01 | 0.00E+00 | 1.32E+01 | 3.67E+02 | 8, IUE+UI | 1.318403 | 0.002400 |
| IN | IN-W240.931 | 1.93 | 7.46E-01 | 0.00E+00 | 1.52E-01 | 4.22E+00 | 7.37E-01 | 1./46401 | 0.005+00 |
| IN | IN-W243.274 | 174.30 | 2.95E+01 | 0.00E+00 | 1.24E+01 | 3.43E+02 | 7.388+01 | 1.416403 | 0.002100 |
| IN | IN-W243.275 | 7.28 | 4.93E+00 | 0.00E+00 | 2.07E+00 | 3.73E+01 | 1.2/E+U1 | 1 735-102 | 0.002400 |
| IN | IN-W243.808 | 46.06 | 7.79E+00 | 0.00E+00 | 3.27E+00 | 9.07E+01 | 2.000101 | 5.736102 | 0.000.000 |
| IN | IN-W245.1034 | 0.21 | 5.63E-03 | 0.00E+00 | 5.94E-02 | 1.63E+00 | 3.032-01 | 6.776400 6.11E-02 | 0.002+00 |
| IN | IN-W245.301 | 37.51 | 5.08E-01 | 0.00E+00 | 5.36E+00 | 1.486402 | 3.48ETVI | 112102 | 0.002100 |
| IN | IN-W245.302 | 133.74 | 1.81E+00 | 0.00E+00 | 1.91E+01 | 3.296402 | 1.1/2704 | 1365-00 | 0.002100 |
| IN | IN-W247.1038 | 0.21 | 2.39E-03 | 0.00E+00 | 2.86E-02 | 1.94E-01 | 7 215-01 | 1 162 10 | 0.002100 |
| IN | IN-W247.523 | 173.68 | 9.96E-01 | 0.00E+00 | 1.20E+01 | 3.512402 | 1.316701 | 1.306703 | 0.002+00 |
| IN | IN-W247.810 | 27.51 | 1.58E-01 | 0.00E+00 | 1.89E+00 | J.43E+UI | 1.106701 | 1.100702 | D 005+00 |
| IN | IN-W249.1071 | 2.29 | 0.00E+00 | 0.00E+00 | 1.28E+03 | Y.U2E+00 | 0.000000 | 0.000-00 | D 005+00 |
| IN | IN-W249.527 | 1.10 | 0.00E+00 | 0.00E+00 | 6.15E+02 | 4.346+00 | 0.002100 | 0.000700 | 0.002100 |
| IN | IN-W249.528 | 0.21 | 0.00E+00 | 0.00E+00 | J.89E+01 | 4.13C-UL | 101300.0 | 2 (15+0) | D 00E+00 |
| IN | IN-W250.259 | 14.07 | 1.25E-02 | 0.00E+00 | J.U9E400 | 8.30CTVI | 6 86F101 | 1 285+01 | 0.00E+00 |
| IN | IN-W250.941 | 50.96 | 4.54E-02 | 0.00E+00 | 1.1/201 | 1 205-02 | 1075101 | < 77E+03 | 0.00E+00 |
| IN | IN-W252.1000 | 0.21 | 5.95E+00 | 0.00E+00 | 3.018400 | 1.376404 | 1 746403 | 1 746-01 | 0.002+00 |
| IN | IN-W252.283 | 117.73 | 3.37E+01 | 0.00E+00 | 2.592401 | 2 105:01 | 1.146702 4 \$4E101 | 9.075403 | 0.00E+00 |
| IN | IN-W252.811 | 32.82 | 9.39E+00 | 0.00E+00 | 7.912400 | 2.19C+U2 | 1.016401 | 1 476107 | 0.000.000 |
| IN | IN-W254.1044 | 0.21 | 0.00E+00 | 0.00E+00 | 3.002+00 | 8.JIC+UI | 1.036701 | 3.746.702 | 1 |

-

| _ | Waste | Scaled | SCALED T | OTAL CURIES OF | EACH RADIONUC | LIDE FOR EACH | WASTE STREAM | | |
|--------------|--------------|-------------|---------------|----------------|----------------------|---------------|---------------|----------------------|--------------|
| SITE | Stream ID# | Volume (m3) | Scaled Am-241 | Scaled Cm-244 | Scaled Pu-238 | Scaled Pu-239 | Scaled Pu-240 | Scaled Pu-241 | Scaled U-234 |
| IN | IN-W254.289 | 2.34 | 0.00E+00 | 0.00E+00 | 3.38E-01 | 9.36E+00 | 2.07E+00 | 3.85E+01 | 0.00E+00 |
| ĪN | IN-W254.290 | 7.28 | 0.00E+00 | 0,00E+00 | 1.05E+00 | 2.91E+01 | 6.42E+00 | 1.20E+02 | 0.00E+00 |
| IN | IN-W256.1062 | 20.59 | 1.22E+00 | 0.00E+00 | 1.91E+03 | 1.33E+01 | 2.78E+01 | 0.00E+00 | 0.00E+00 |
| IN | IN-W256.295 | 5.99 | 3.56E-01 | 0.00E+00 | 5.55E+02 | 3.87E+00 | 8.08E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W257.558 | 0.21 | 0.00E+00 | 0.00E+00 | 1.14E-02 | 3.16E-01 | 6.97E-02 | 1.30E+00 | 0.00E+00 |
| IN | IN-W257.947 | 0.68 | 0.00E+00 | 0.00E+00 | 1.86E-02 | 5.17E-01 | 1.14E-01 | 2.13E+00 | 0.00E+00 |
| IN | IN-W259.552 | 10.06 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 2.27E+00 | 2.44E-01 | 0.00E+00 | 0.00E+00 |
| IN | IN-W259.920 | 2.50 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 3.75E+00 | 4.04E-01 | 0.00E+00 | 0.001+00 |
| IN | IN-W263.520 | 14.35 | 0.00E+00 | 0.00E+00 | 1.98E+01 | 8.99E-01 | 1.39E-03 | 8.89E-02 | 0.0012+00 |
| IN | IN-W265.516 | 7.92 | 8.49E-02 | 0.00E+00 | 2.70E-01 | 7,49E+00 | 1.65E+00 | 3.08E+01 | 0.002+00 |
| IN | IN-W265.517 | 0.62 | 6.69E-01 | 0.00E+00 | 2.13E+00 | 5.90E+01 | 1.30E+01 | 2.43E+02 | 0.002.00 |
| IN | IN-W267.1005 | 1.10 | 0.00E+00 | 0.00E+00 | 1.57E+00 | 4.35E+01 | 9.59E+00 | 1.792+02 | 0.002100 |
| IN | IN-W267.514 | 1.25 | 0.00E+00 | 0.00E+00 | 3.57E+00 | 9.89E+01 | 2.186+01 | 4.072+02 | 0.002700 |
| IN | IN-W269.510 | 5.99 | 3.80E+01 | 0.00E+00 | 3.77E+01 | 3.24E+02 | 3,206+01 | 5.35E-UI | 0.002100 |
| IN | IN-W269.535 | 20.80 | 1.32E+02 | 0.00E+00 | 1.31E+02 | 1.12E+03 | 1.132+02 | 1.712100 0.00E100 | 0.002100 |
| IN | IN-W271.532 | 0.89 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.338+01 | 2.972401 | 0.002+00 | 0.005+00 |
| IN | IN-W271.533 | 0.21 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.04E+00 | 4.33E+00 | P AE 101 | 0.005100 |
| IN | IN-W272.504 | 0.89 | 0.00E+00 | 0.00E+00 | 7.06E-01 | 1.yoE+UI | 4.346100 | 1 415-02 | 0.005.00 |
| IN | IN-W272.974 | 1.66 | 0.00E+00 | 0.00E+00 | 1.32E+00 | 3.00E+U1 | 5.USETUU | 1.515704 | 0.000100 |
| IN | IN-W275.502 | 1.72 | 1.03E-01 | 0.00E+00 | 2.68E-01 | 7.442+00 | 1.012100 | 3.002701 | 0.005+00 |
| IN | IN-W275.967 | 5.20 | 3.13E-01 | 0.00E+00 | 8.11E-01 | 2.236401 | 4.702100 | 1 765-01 | 0.002100 |
| IN | IN-W276.500 | 86.75 | 1.39E+01 | 0.00E+00 | 1.11E+01 | 3.0/12+02 | 0. /06+01 | 1.20E703 | 0.005.00 |
| IN | IN-W276.966 | 313.46 | 5.04E+01 | 0.00E+00 | 4.00E+01 | 1.11E+U3 | 2.476702 | 4.505-05 | 0.005+00 |
| IN | IN-W278.1090 | 0.89 | 0.00E+00 | 0.00E+00 | 5.70E-03 | 1.362-01 | 3.47E-04 | 1.01F+01 | 0.005+00 |
| IN | IN-W278.495 | 4.16 | 0.00E+00 | 0.00E+00 | 8.90E-02 | 1.4/2400 | 3.445.01 | 1 30F+01 | 0.000.100 |
| IN | IN-W280.1066 | 28.50 | 2.91E-01 | 0.00E+00 | 1.81E+04 | 1 476401 | \$ 02F.07 | 1 #2E+00 | 0.00E+00 |
| IN | IN-W280.448 | 8.34 | 8.52E-02 | 0.00E+00 | 7.3UE+U3 | 1.4/CTUL | 4 07F_04 | 1.17E-02 | 0.00E+00 |
| IN | IN-W280.449 | 0.21 | 7.08E-04 | 0.008+00 | 4.41E7U | 2 165-01 | 1.09E.02 | 6.99E-01 | 0.00E+00 |
| IN | IN-W281.487 | 317.82 | 0.00E+00 | 0.002+00 | 9.365103 | A 745400 | 2 15E-01 | 1.37E-01 | 0.00E+00 |
| IN | IN-W281.488 | 0.62 | 0.00E+00 | 0.000100 | 10.786704 (A1E.A) | 1 56E+00 | 3.44E-01 | 6.42E+00 | 0.00E+00 |
| IN | IN-W283.481 | 0.21 | 0.00E+00 | 0.002100 | 1 845 01 | \$ 11E+00 | L.13E+00 | 2.10E+01 | 0.00E+00 |
| IN | IN-W283.534 | 0.68 | 0,002+00 | 0.000100 | 1 885.01 | 5 20E+00 | 1.15E+00 | 2.14E+01 | 0.00E+00 |
| IN | IN-W283.963 | | 00+300.0 | 0.005100 | 0.005+00 | 1.66E+01 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W285.471 | 63.02 | 0.000000 | 0.000000 | 0.000.00 | 6 19F.01 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W285.815 | 2.34 | 0.00E+00 | | 0.000100 | 5.04E401 | 5.84E+02 | 3.80E+01 | 0.00E+00 |
| IN | IN-W287.460 | 211.95 | 4.086-01 | 0.000100 | 0.000100 | 1 186+02 | 0.005+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W289.466 | 25.38 | 1,51E+01 | U.UUETUU | 0.000100 | 1 36F_01 | 193F.01 | 0.00E+00 | 0.00E+00 |
| IN | IN-W291.454 | 0.68 | 3.952-01 | 0.000100 | 0.000.00 | 7 91F+01 | 1.27E+02 | 0.00E+00 | 0.00E+00 |
| IN | IN-W291.455 | 1.46 | 8.45E+01 | 0.000100 | 0.000100 | 1 275+07 | \$ \$1E+07 | 0.00E+00 | 0.00E+00 |
| IN | IN-W291.456 | 634.40 | 3.68E+02 | 0.002+00 | | L | 1 | | |

TABLE - 1 SCALED VOLUME AND ACTIVITIES FOR SELECTED RADIONUCLIDES FOR EACH WASTE STREAM

Paye 4

| TABLE - 1 |
|---|
| SCALED VOLUME AND ACTIVITIES FOR SELECTED RADIONUCLIDES FOR EACH WASTE STREAM |

| | Waste | Scaled | SCALED T | OTAL CURIES OF | EACH RADIONUC | LIDE FOR EACH | WASTE STREAM | | |
|------|--------------|-------------|---------------|----------------|---------------|---------------|---------------|---------------|--------------|
| SITE | Stream 10# | Volume (m3) | Scaled Am-241 | Scaled Cm-244 | Scaled Pu-138 | Scaled Pu-239 | Scaled Pu-240 | Scaled Pu-241 | Scaled U-234 |
| IN | IN-W294.1057 | 0.42 | 1.16E-01 | 0.00E+00 | 1.45E-01 | 4.03E+00 | 8.88E-01 | 1.66E+01 | 0.00E+00 |
| IN I | IN-W294.342 | 406.85 | 3.40E+01 | 0.00E+00 | 4.26E+01 | 1.18E+03 | 2.61E+02 | 4.86E+03 | 0.00E+00 |
| N | IN-W294.814 | 33.50 | 2.80E+00 | 0.00E+00 | 3.51E+00 | 9.73E+01 | 2.15E+01 | 4.00E+02 | 0.00E+00 |
| IN | IN-W296.327 | 3450.30 | 9.73E+01 | 0.00E+00 | 8.37E+01 | 2.32E+03 | 5,12E+02 | 9.54E+03 | 0.00E+00 |
| IN | IN-W296.329 | 520.21 | 4.89E+01 | 0.00E+00 | 4.20E+01 | 1.17E+03 | 2.57E+02 | 4.79E+03 | 0.00E+00 |
| IN | IN-W296.813 | 47.99 | 1.35E+00 | 0.00E+00 | 1.16E+00 | 3.22E+01 | 7.12E+00 | 1.33E+02 | 0.00E+00 |
| IN | IN-W298.317 | 54.70 | 7.31E+01 | 0.00E+00 | 2.19E+01 | 6.08E+02 | 1.34E+02 | 2.50E+03 | 0.00E+00 |
| IN | IN-W298.812 | 15.37 | 2.05E+01 | 0.00E+00 | 6.16E+00 | 1.7IE+02 | 3.77E+01 | 7.03E+02 | 0.00E+00 |
| IN | IN-W298.979 | 0.42 | L.85E+00 | 0.00E+00 | 5.56E-01 | 1.54E+01 | 3.40E+00 | 6.34E+01 | 0.00E+00 |
| IN | IN-W300.308 | 1509.46 | 2.05E+02 | 0.00E+00 | 8.83E+01 | 2.45E+03 | 5.40E+02 | 1.01E+04 | 0.00E+00 |
| IN | IN-W300.930 | 4.69 | 6.36E-01 | 0.00E+00 | 2.74E-01 | 7.60E+00 | 1.68E+00 | 3.13E+01 | 0.00E+00 |
| IN | IN-W302.299 | 23.45 | 2.05E+01 | 0.00E+00 | 0.00E+00 | 3.08E+00 | 0.00E+00 | 0.00E+00 | 0.001+00 |
| IN | IN-W302.913 | 84.86 | 7.43E+01 | 0.00E+00 | 0.00E+00 | 1.11E+01 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W304.860 | 8.75 | 0.00E+00 | 0.00E+00 | 4.77E+02 | 2,49E+00 | 5,13E-01 | 9.79E-01 | 0.001:+00 |
| IN | IN-W304.861 | 59.07 | 0.00E+00 | 0.00E+00 | 3.22E+03 | 1.68E+01 | 3.46E+00 | 6.61E+00 | 0.008.+00 |
| IN | IN-W305,1068 | 37.44 | 0.00E+00 | 0.00E+00 | 3.61E+03 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.002+00 |
| IN | IN-W305.828 | 10.68 | 0.00E+00 | 0.00E+00 | 1.03E+03 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.002+00 |
| IN | IN-W308.618 | 503.57 | 3.17E+03 | 0.00E+00 | 1.53E+02 | L.13E+03 | 2.53E+01 | 4.71E+01 | 0.005+00 |
| IN | IN-W308.816 | 864.91 | 8.18E+02 | 0.00E+00 | 3.95E+01 | 2.92E+02 | 6.52E+00 | 1.21E+02 | 0.000000 |
| IN | IN-W309.609 | 108.58 | 1.25E+01 | 0.00E+00 | 3.00E+00 | 8.31E+01 | 1.83E+01 | 3.42E+U2 | 0.002+00 |
| IN | IN-W309.610 | 352.77 | 2.03E+01 | 0.00E+00 | 4.87E+00 | 1.35E+02 | 2.98E+01 | 3.35E+02 | 0.002100 |
| IN | IN-W311.1013 | 5,41 | 6.81E+02 | 0.00E+00 | 6.51E+00 | 1.\$0E+02 | 3.98E+01 | 7.42E+02 | 0.000000 |
| IN | IN-W311.604 | 1.72 | 2.17E+02 | 0.00E+00 | 2.07E+00 | 5.74E+01 | 1.27E+01 | 2.306+02 | 0.000+00 |
| IN | IN-W312.602 | 1.10 | 0.00E+00 | 0.00E+00 | 1.78E+00 | 4.92E+01 | 1.09E+01 | 2.02E+02 | 0.0000+00 |
| IN | IN-W312.942 | 2.70 | 0.00E+00 | 0.00E+00 | 4.38E+00 | 1.21E+02 | 2.68E+01 | 4.992+02 | 0.000000 |
| IN | IN-W314.1017 | 1.04 | 9.73Ę-02 | 0.00E+00 | 1.46E+00 | 4.03E+01 | 8.90E+00 | 1.00E+02 | 0.00000 |
| IN | IN-W314.606 | 0.68 | 6.37E-02 | 0.00E+00 | 9.52E-01 | 2.64E+01 | 5.82E+00 | 1.076+04 | 0.005400 |
| IN | IN-W315.601 | 0.42 | 2.99E+01 | 0.00E+00 | 1.14E-02 | 3.16E-UI | 0.972-02 | 1.30CTU | 0.005+00 |
| IN | IN-W317.1028 | 0.21 | 1.26E+00 | 0.00E+00 | 1.44E-01 | 3.99E+00 | 5.5UC-UI | 1.046401 | 0.000.+00 |
| IN | IN-W317.757 | 39.10 | 1.19E+02 | 0.00E+00 | 1.35E+01 | 3,736+02 | 8.48CTUI | 4 \$45407 | 0.005+00 |
| IN | IN-W317.758 | 11.51 | 3.50E+01 | 0.00E+00 | 3.98E+00 | 1.100+02 | 2.446701 | 1.416402 | 0.0000+00 |
| IN | IN-W319.583 | 0.21 | 0.00E+00 | 0.00E+00 | 5.24E+US | 3.436+02 | 7.576401 | 4.625401 | 0.000.00 |
| IN | IN-W319.584 | 0.68 | 0.00E+00 | 0.00000 | 4.05E-01 | 1.120+01 | 2.46ETUU | 1 705+02 | 0.002+00 |
| IN | IN-W321.1023 | 1.30 | 0.00E+00 | 0.00E+00 | 1.5/E+00 | 4.336401 | 9.002100 | 2 \$55+03 | 0.00E+00 |
| IN | IN-W321.578 | 0.21 | 0.00E+00 | 0.00E+00 | 2,302401 | 0.792104 | 2 425+04 | 0.005+00 | 0.000000 |
| IN | IN-W322.851 | 0.89 | 0.00E+00 | 0.00E+00 | | 1.4UCTUI | A 125-00 | 0.002100 | 0.002.00 |
| IN | IN-W322.952 | 1.66 | 0.00E+00 | 0.00E+00 | 0.000000 | 2,24E+U1 | 4.336100 | 2 615-10 | 0.000.000 |
| IN | IN-W323.562 | 0.89 | 0.00€+00 | 0.00E+00 | 1.8ZE+00 | 3.452-01 | 0.002100 | A MEAN | 0.000,700 |
| IN | IN-W325.1076 | 0.42 | 0.00E+00 | 0.00E+00 | 1.27E+01 | 0.002+00 | 0.002+00 | 0.002100 | 0.000-00 |
| IN | IN-W325.679 | 0.68 | 0.00E+00 | 0.00E+00 | 2.07E+01 | 0.002+00 | 0.002100 | 0.006700 | 0.002700 |

)

TABLE - 1 SCALED VOLUME AND ACTIVITIES FOR SELECTED RADIONUCLIDES FOR EACH WASTE STREAM

| | Waste | Scaled | SCALED T | OTAL CURIES OF | EACH RADIONU | CLIDE FOR EACH | WASTE STREAM | | |
|------|--------------|-------------|---------------|----------------|---------------|----------------|---------------|---------------|--------------|
| SITE | Stream ID# | Volume (m3) | Scaled Am-241 | Scaled Cm-244 | Scaled Pu-238 | Scaled Pu-239 | Scaled Pu-240 | Scaled Pu-241 | Scaled U-234 |
| IN | IN-W327.1085 | 3.54 | 0.00E+00 | 0.00E+00 | 7.43E+01 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W327.735 | 1.30 | 0.00E+00 | 0.00E+00 | 2.74E+01 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W329.681 | 0.89 | 0.00E+00 | 0.00E+00 | 1.02E+02 | 4.37E-02 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W329.682 | 0.21 | 0.00E+00 | 0.00E+00 | 1.60E+02 | 6.82E-02 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W330.677 | 6.03 | 0.00E+00 | 0.00E+00 | 3.67E+02 | 2.88E-03 | 1.46E-03 | 9.35E-02 | 0.00E+00 |
| IN | IN-W330.678 | 1.93 | 0.00E+00 | 0.00E+00 | 1.17E+02 | 9.21E-04 | 4.68E-04 | 2.99E-02 | 0.00E+00 |
| IN | IN-W332.661 | 0.68 | 0.00E+00 | 0.00E+00 | 6.89E+00 | 4.88E-02 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W332.962 | 0.83 | 0.00E+00 | 0.00E+00 | 8.42E+00 | 5.97E-02 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W334.675 | 1.51 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.30E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| [N | IN-W334.961 | 4.58 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 3.93E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W336.660 | 4.16 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 5.68E-01 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W336.820 | 0.68 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 9.29E-02 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W338.657 | 0.89 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 3.83E-01 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W338.956 | 1.04 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 4.48E-01 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W319.655 | 2.14 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 2.17E+01 | 8.60E-02 | 0.00E+00 | 0.00E+00 |
| IN | IN-W339.955 | 7.07 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 7.19E+01 | 2.85E-01 | 0.00E+00 | 0.00E+00 |
| IN | IN-W341.671 | 0.21 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.80E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W341.954 | 0.68 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 5.89E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W342.652 | 0.68 | 5.65E+00 | 0.00E+00 | 0.00E+00 | 4.05E-02 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W342.953 | 0.42 | 3.45E+00 | 0.00E+00 | 0.00E+00 | 2.48E-02 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W345.669 | 14.35 | 9.51E+01 | 0.00E+00 | 2.25E+01 | 1.79E+01 | 1.10E+01 | 0.00E+00 | 0.00E+00 |
| IN | IN-W345.819 | 0.89 | 5.89E+00 | 0.00E+00 | 1.39E+00 | 1.11E+00 | 6.84E-01 | 0.00E+00 | 0.00E+00 |
| IN | IN-W347.646 | 51.79 | 2.06E+00 | 0.00E+00 | 0.00E+00 | 5.84E+01 | 1.04E+02 | 0.00E+00 | 0.00E+00 |
| IN | IN-W347.818 | 3.44 | 1,37E-01 | 0.00E+00 | 0.00E+00 | 3.88E+00 | 6.91E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W348.1012 | 2.34 | 3,28E-02 | 0.00E+00 | 3.19E+00 | 8.84E+01 | 1.95E+01 | 3.64E+02 | 0.00E+00 |
| IN | IN-W348.846 | 4.16 | 1,16E-01 | 0.00E+00 | 1.13E+01 | 3.14E+02 | 6.92E+01 | L.29E+03 | 0.00E+00 |
| IN | IN-W350.650 | 0.68 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 3.60E+01 | 1.07E+02 | 0.00E+00 | 0.00E+00 |
| IN | IN-W350.923 | 0.21 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.10E+01 | 3.27E+01 | 0.00E+00 | 0.00E+00 |
| IN | IN-W351.648 | 0.89 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.43E+00 | 4.79E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W351.922 | 1.25 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 2.01E+00 | 6.72E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W353.859 | 0.68 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 7.53E-02 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W353.917 | 0.21 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 2.30E-02 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W354.1016 | 0.21 | 0.00E+00 | 0.00E+00 | 3.99E-02 | I.11E+00 | 2.44E-01 | 4.55E+00 | 0.00E+00 |
| IN | IN-W354.858 | 0.68 | 0.00E+00 | 0.00E+00 | 1.31E-01 | 3.62E+00 | 7.98E-01 | 1.49E+01 | 0.00E+00 |
| IN | IN-W355.1015 | 1.04 | 0.00E+00 | 0.00E+00 | 1.01E+00 | 2.79E+01 | 6.16E+00 | 1.15E+02 | 0.00E+00 |
| IN | IN-W355.857 | 0.89 | 0.00E+00 | 0.00E+00 | \$.60E-01 | 2.38E+01 | 5.26E+00 | 9.81E+01 | 0.00E+00 |
| IN | IN-W356.1014 | 3.74 | 6.31E+01 | 0.00E+00 | 3.62E-01 | 1.00E+01 | 2.22E+00 | 4.13E+01 | 0.00E+00 |
| IN | IN-W356.856 | 1.30 | 2.20E+01 | 0.00E+00 | 1.26E-01 | 3.50E+00 | 7.72E-01 | 1.44E+01 | 0.00E+00 |
| IN | IN-W357.1022 | 0.68 | 0.00E+00 | 0.00E+00 | 9.89E-03 | 2.74E-01 | 6.05E-02 | 1.13E+00 | 0.00E+00 |
| IN | IN-W357.850 | 0.21 | 0.00E+00 | 0.00E+00 | 6.05E-03 | 1.68E-01 | 3.70E-02 | 6.89E-01 | 0.00E+00 |

 Ξ

TABLE - 1

SCALED VOLUME AND ACTIVITIES FOR SELECTED RADIONUCLIDES FOR EACH WASTE STREAM

| · | Waste | Scaled | SCALED T | OTAL CURIES OF | EACH RADIONUC | LIDE FOR EACH | WASTE STREAM | | |
|------|--------------|-------------|---------------|----------------|---------------|---------------|-----------------|---------------|--------------|
| SITE | Stream ID# | Volume (m3) | Scaled Am-241 | Scaled Cm-244 | Scaled Pu-238 | Scaled Pu-239 | Scaled Pu-240 | Scaled Pu-241 | Scaled U-234 |
| IN | IN-W358.854 | 0.89 | 0.00E+00 | 0.00E+00 | 5.56E+02 | 2.47E+00 | 4.62E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W358.855 | 3.33 | 0.00E+00 | 0.00E+00 | 2.08E+03 | 9.26E+00 | 1.73E+01 | 0.00E+00 | 0.00E+00 |
| IN | IN-W358.948 | 0.21 | 0.00E+00 | 0.00E+00 | 4.34E+02 | 1.93E+00 | 3.61E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W359.853 | 0.83 | 0.00E+00 | 0.00E+00 | 1.10E+02 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| IN | IN-W361.1021 | 1.51 | 1.10E-02 | 0.00E+00 | 7.65E-01 | 2.126+01 | 4.68E+00 | 8.73E+01 | 0.00E+00 |
| IN | IN-W361.849 | 2.08 | 3.04E-02 | 0.00E+00 | 2.10E+00 | 5.83E+01 | 1.29E+01 | 2.40E+02 | 0.00E+00 |
| IN | IN-W362.1020 | 5.37 | 0.00E+00 | 0.00E+00 | 8.63E+00 | 2.39E+02 | 5.28E+01 | 9.84E+02 | 0.00E+00 |
| IN | IN-W362.848 | 8.74 | 0.00E+00 | 0.00E+00 | 2.81E+01 | 7.78E+02 | 1.72E+02 | 3.20E+03 | 0.00E+00 |
| IN | IN-W363.1019 | 0.89 | 0.00E+00 | 0.00E+00 | 5.76E-01 | 1.60E+01 | 3.52E+00 | 6.57E+01 | 0.00E+00 |
| IN | IN-W363.847 | 1.04 | 0.00E+00 | 0.00E+00 | 1.35E+00 | 3.74E+01 | 8.25E+00 | 1.54E+02 | 0.00E+00 |
| IN | IN-W364.1011 | 0.89 | 0.00E+00 | 0.00E+00 | 1.43E+00 | 3.96E+01 | 8.74E+00 | 1.63E+02 | 0.00E+00 |
| IN | IN-W364.844 | 0.62 | 0.00E+00 | 0.00E+00 | 2.01E+00 | 5.56E+01 | 1.23E+01 | 2.29E+02 | 0.00E+00 |
| IN | IN-W365.1010 | 1.30 | 9.68E+01 | 0.00E+00 | 5.77E-01 | 1.60E+01 | 3.53E+00 | 6.58E+01 | 0.00E+00 |
| IN | IN-W365.842 | 1.04 | 2.57E+02 | 0.00E+00 | 1.53E+00 | 4.25E+01 | 9.38E+00 | 1.75E+02 | 0.00E+00 |
| IN | IN-W366.1004 | 2.08 | 3.52E-01 | 0.00E+00 | 5.01E-01 | 1.39E+01 | 3.06E+00 | 5.71E+01 | 0.00E+00 |
| IN | IN-W366.841 | 1.10 | 1.86E-01 | 0.00E+00 | 2.64E-01 | 7.31E+00 | 1.61E+00 | 3.01E+01 | 0.00E+00 |
| IN. | IN-W367.840 | 0.21 | 0.00E+00 | 0.00E+00 | 1.03E+01 | 2.85E+02 | 6.29E+01 | 1.17E+03 | 0.00E+00 |
| IN | IN-W367.973 | 4.69 | 0.00E+00 | 0.00E+00 | 2.32E+00 | 6.42E+01 | 1,42E+01 | 2.64E+02 | 0.00E+00 |
| IN | IN-W368.839 | 0.21 | 0.00E+00 | 0.00E+00 | 2.64E+00 | 7.31E+01 | 1.61E+01 | 3.01E+02 | 0.00E+00 |
| IN | IN-W368.971 | 1.10 | 0.00E+00 | 0.00E+00 | 1.39E-01 | 3.85E+00 | 8.50E-01 | 1.58E+01 | 0.00E+00 |
| IN | IN-W369.837 | 3.23 | 5.43E-01 | 0.00E+00 | 7.35E-01 | 2.04E+01 | 4.49E+00 | 8.38E+01 | 0.00E+00 |
| IN | IN-W369.970 | 9.98 | 1.68E+00 | 0.00E+00 | 2.27E+00 | 6.29E+01 | 1.39E+01 | 2.59E+02 | 0.00E+00 |
| IN | IN-W370.836 | 15.16 | 0.00E+00 | 0.00E+00 | 4.22E+00 | 1.17E+02 | 2.58E+01 | 4.81E+02 | 0.00E+00 |
| IN | IN-W370.929 | 53.46 | 0.00E+00 | 0.00E+00 | 1.49E+01 | 4.12E+02 | 9,10E+01 | 1.70E+03 | 0.00E+00 |
| IN | IN-W371.1018 | 0.21 | 1.16E+02 | 0.00E+00 | 3.23E-01 | \$.95E+00 | 1.98E+00 | 3.68E+01 | 0.00E+00 |
| IN | IN-W371.831 | 0.68 | 3.79E+02 | 0.00E+00 | 1.06E+00 | 2.93E+01 | 6.46E+00 | 1.20E+02 | 0.00E+00 |
| IN | IN-W373.1003 | 0.68 | 0.00E+00 | 0.00E+00 | 1.24E+00 | 3.43E+01 | 7.56E+00 | 1.41E+02 | 0.00E+00 |
| IN | IN-W373.830 | 0.21 | 0.00E+00 | 0.00E+00 | 7.56E-01 | 2.10E+01 | 4.63E+00 | 8.62E+01 | 0.00E+00 |
| IN | IN-W374.1091 | 2.08 | 0.00E+00 | 0.00E+00 | 5.32E-01 | 1.47E+01 | 3.25E+00 | 6.07E+01 | 0.00E+00 |
| IN | IN-W374.829 | 2.34 | 0.00E+00 | 0.00E+00 | 1.50E-01 | 4.15E+00 | <u>9.17E-01</u> | 1.71E+01 | 0.00E+00 |
| IN | IN-W375.1096 | 4.48 | 0.00E+00 | 0.00E+00 | 3.38E-02 | 9.38E-01 | 2.07E-01 | 3.86E+00 | 0.00E+00 |
| IN | IN-W375.827 | 7.90 | 0.00E+00 | 0.00E+00 | 1.19E-01 | 3.31E+00 | 7.30E-01 | 1.36E+01 | 0.00E+00 |
| LA | LA-M002 | 6706.45 | 7.02E+03 | 0.00E+00 | 2.06E+02 | 4.68E+03 | 0.00E+00 | L.12E-01 | 3.88E+01 |
| LA | LA-T001 | 3787.32 | 0.00E+00 | 8.14E-03 | 1.91E+03 | 1.33E+03 | 6.21E-01 | 1.09E+01 | U.00E+00 |
| LA | LA-T002 | 193.71 | 9.07E+01 | 0.00E+00 | 8.55E+00 | 4.33E+02 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| LA | LA-T004 | 12629.26 | 4.68E+01 | 4.25E+02 | 2.55E+05 | 1.17E+04 | 2.84E+01 | 6.03E+02 | 1.13E+02 |
| LA | LA-T005 | 8885.76 | 8.79E+01 | 8.19E+02 | 1.98E+05 | 4.64E+04 | 1.00E+02 | 1.71E+03 | 7.56E+01 |
| LA | LA-T006 | 543.32 | 8.38E+01 | 0.00E+00 | 3.15E+04 | 8.64E+02 | 2.37E+00 | 5.17E+01 | <u> </u> |
| LA | LA-T007 | 198.91 | 0.00E+00 | 0.00E+00 | 3.53E+02 | 1.71E+03 | 1.12E-01 | 1.87E+00 | 1.95E+00 |
| LA | LA-T008 | 302.83 | 3.61E-03 | 0.00E+00 | 3.53E+02 | 1.72E+02 | 2.01E-03 | 1.24E-01 | 0.00E+00 |

)

TABLE - 1 SCALED VOLUME AND ACTIVITIES FOR SELECTED RADIONUCLIDES FOR EACH WASTE STREAM

| | Waste | Scaled | SCALED T | OTAL CURIES OF | EACII RADIONUO | CLIDE FOR EACH | WASTE STREAM | | |
|---------|------------|-------------|---------------|----------------|----------------|-----------------|---------------|---------------|--------------|
| SITE | Stream ID# | Volume (m3) | Scaled Am-241 | Scaled Cm-244 | Scaled Pu-238 | Scaled Pu-239 | Scaled Pu-240 | Scaled Pu-241 | Scaled U-234 |
| LA | LA-T009 | 438.06 | 0.00E+00 | 0.00E+00 | 1.79E+01 | 5.24E+02 | 1.40E+00 | 4.28E+01 | 0.00E+00 |
| LA | LA-W001 | 3126.19 | 2.74E-03 | 0.00E+00 | 5.14E+03 | 2.42E+03 | 7.14E-01 | 1.13E+01 | 1.07E+01 |
| LA | LA-W003 | 4968.84 | 3.42E+02 | 0.00E+00 | 2.97E+02 | 3.30E+03 | 0.00E+00 | 3.42E-03 | 0.00E+00 |
| LA | LA-W004 | 4880.50 | 6.00E+01 | 0.00E+00 | 4.02E+04 | 3.04E+04 | 7.56E+01 | 1.26E+03 | 4.68E+01 |
| LA | LA-W005 | 4828.92 | 7.97E+01 | 0.00E+00 | 8.01E+03 | 1.90E+05 | 4.98E+02 | 8.81E+03 | 4.68E+01 |
| LÃ | LA-W006 | 6097.49 | 3.36E+04 | 0.00E+00 | 1.62E+04 | 6.31E+04 | 1.53E+02 | 2.72E+03 | 5.64E+01 |
| LA | LA-W009 | 1989.53 | 1.21E+03 | 0.00E+00 | 1.23E+00 | 1.19E+02 | 2.84E-01 | 4.49E+00 | 0.00E+00 |
| LA | LA-W066 | 1.89 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| LA | LA-W067 | 8.94 | 1.46E-01 | 4.61E+00 | 4.76E+00 | 3.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| LA | LA-W068 | 0.42 | 0.00E+00 | 0.00E+00 | 1.76E-01 | 6.11E-01 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| | LL-M001 | 119.39 | 1.94E+02 | 3.79E+02 | 3.16E+02 | 7.52E+01 | 5.27E+01 | 1.46E+03 | 0.00E+00 |
| lu | LL-T001 | 52.80 | 3.44E+01 | 0.00E+00 | 0.00E+00 | 3.34E+01 | 1.86E+01 | 5.01E+02 | 0.00E+00 |
| LL | LL-T002 | 3368.07 | 3.71E+03 | 0.00E+00 | 1.15E+03 | 2.41E+03 | 1.62E+03 | 4.54E+04 | 0.00E+00 |
| LL | LL-T003 | 917.30 | 8.32E+01 | 0.00E+00 | 7.50E+01 | 3.44E+01 | 3.79E+01 | 1.03E+03 | 0.00E+00 |
| LL. | LL-T004 | 20.54 | 3.59E+01 | 0.00E+00 | 1.04E+01 | 1.25E+01 | 1.61E+01 | 4.49E+02 | 0.00E+00 |
| II. | LL-T005 | 228.68 | 7.41E+01 | 9.85E+02 | 4.20E+01 | 1.67E+01 | 2.04E+01 | 5.65E+02 | 0.00E+00 |
| LL | LL-W018 | 176.59 | 1.13E+00 | 0.008+00 | 0.00E+00 | 4.40E-01 | L.67E+00 | 4.45E+01 | 0.00E+00 |
| LL | LL-W019 | 39.49 | 3.04E+01 | 0.00E+00 | 0.00E+00 | 9.15E+00 | 1.23E+01 | 3.40E+02 | 0.00E+00 |
| MD | MD-M001 | 0.42 | 0.00E+00 | 0.00E+00 | 4.26E-01 | 9.63E-03 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| MD | MD-T001 | 4.16 | 0.00E+00 | 0.00E+00 | 3.14E-01 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| MD | MD-T003 | 146.94 | 0.00E+00 | 0.00E+00 | 2.42E+01 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| MD | MD-T004 | 26.84 | 0.00E+00 | 0.00E+00 | \$.68E+02 | 7.72E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| MD | MD-T005 | 30.24 | 0.00E+00 | 0.00E+00 | 2.74E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| MD | MD-T006 | 58.59 | 0.00E+00 | 0.00E+00 | 1.97E+01 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| MD | MD-T007 | 23.89 | 0.00E+00 | 0.00E+00 | 1.93E+02 | 7.34E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| MD | MD-T008 | 3.74 | 0.00E+00 | 0.00E+00 | 6.40E+01 | 8.67E-02 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| MD | MD-T009 | 0.21 | 0.00E+00 | 0.00E+00 | 4.04E+00 | 1.35E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| MD | MD-T010 | 0.42 | 0.00E+00 | 0.00E+00 | 2.13E-01 | 4.82E-03 | 0.00E+00 | 0.006+00 | 0.00E+00 |
| MD | MD-T012 | 0.62 | 0.00E+00 | 0.00E+00 | 7.64E+00 | 4,87E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| MD | MD-W002 | 1.87 | 0.00E+00 | 0.00E+00 | 8.50E+00 | 3.38E-02 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| MD | MD-W003 | 1.66 | 0.00E+00 | 0.00E+00 | 9.28E+01 | 7.97E+00 | 0.00E+00 | 0.00E+00 | 0.00+300.0 |
| MD | MD-W017 | 1.46 | 0.00E+00 | 0.00E+00 | 2.43E+02 | 4.37E-01 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| NT | NT-W001 | 672.55 | 3.01E+02 | 2.57E+02 | 2.05E+02 | 2.81E+03 | 1.42E+01 | 1.67E+02 | 3.22E-02 |
| NT | NT-W021 | 5.67 | 0.00E+00 | 0.00E+00 | 1.43E+00 | <u>3.17E+01</u> | 5.33E+00 | \$.26E+01 | 0.00E+00 |
| OR | OR-W041 | 170.77 | 4.21E-01 | 0.00E+00 | 1.05E+00 | 4.91E+01 | 1.99E+01 | 1.76E+02 | 1.64E-01 |
| OR | OR-W044 | 2214.79 | 6.08E+00 | 3.45E+03 | 8.02E+02 | 7.09E+01 | 1.61E+03 | 1.30E+05 | 5.77E-02 |
| OR | OR-W045 | 5.41 | 0.00E+00 | 0.00E+00 | 5.09E+01 | 2.39E+02 | 3.38E+02 | 3.39E+03 | 0.00E+00 |
| OR | OR-W047 | 154.13 | \$.38E-01 | 3.32E+02 | 1.66E+02 | 1.32E+01 | 1.76E+01 | 1.56E+03 | 0.00E+00 |
| OR | OR-W048 | 15.18 | 0.00E+00 | 5.87E+01 | 0.00E+00 | 6.38E-05 | 0.00E+00 | 0.00E+00 | 0.00£+00 |
| | OR-W049 | 17.68 | 0.00E+00 | 0.00E+00 | 3.00E+03 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |

| TA | RI. | Е - | 1 |
|----|-----|-----|---|

SCALED VOLUME AND ACTIVITIES FOR SELECTED RADIONUCLIDES FOR EACH WASTE STREAM

| | Waste | Scaled | SCALED T | OTAL CURIES OF | EACH RADIONUC | LIDE FOR EACH | WASTE STREAM | | |
|------------|-------------|-------------|---------------|----------------|---------------|---------------|---------------|-----------------|--------------|
| STTE | Stream ID# | Volume (m3) | Scaled Am-241 | Scaled Cm-244 | Scaled Pu-238 | Scaled Pu-239 | Scaled Pu-240 | Scaled Pu-241 | Scaled U-234 |
| | OP-WOS3 | 435.76 | 1.61E+03 | 7.25E+00 | 1.97E-05 | 6.71E+02 | 7.15E+00 | \$.93E-01 | 1.55E+01 |
| 9F | RF.MT-0115 | 2645.01 | 1.75E+03 | 0.00E+00 | 0.00E+00 | 2.15E+04 | 2.96E+04 | 1.72E+05 | 0.00E+00 |
| <u> </u> | RF-11T-0368 | 19.85 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 2.90E+02 | 3.82E+02 | 2.43E+02 | 0.00E+00 |
| N1 | PE-MT-0418 | 104.79 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 5.66E+03 | 0.00E+00 |
| or | PF.MT.0491 | 176.40 | 1.17E+02 | 0.00E+00 | 0.00E+00 | 1.44E+03 | 1.98E+03 | L.15E+04 | 0.00E+00 |
| DC | RF.14T.0873 | 0.21 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 8.76E+00 | 6.32E+00 | 3,32E+01 | 0.00E+00 |
| DE | RF-MT0001 | 3.74 | 1.34E+02 | 0.00E+00 | 0.00E+00 | 1.60E+01 | 1.16E+01 | 6.06E+01 | 0.00E+00 |
| <u>p</u> e | RF-MT0003 | 0.62 | 2.92E+00 | 0.00E+00 | 0.00E+00 | 2.43E+00 | 1.77E+00 | 9,21E+01 | 0.00E+00 |
| DC | RF-MT0007 | 0.83 | 2.98E+01 | 0.00E+00 | 0.00E+00 | 3.56E+00 | 2.57E+00 | 1.35E+01 | 0.00E+00 |
| DE. | RF.MT0320 | 130.54 | 3.65E+03 | 0.00E+00 | 0.00E+00 | 6.93E+03 | 9.41E+03 | <u>5.46E+04</u> | 0.00E+00 |
| DE | RF-MT0321 | 55.93 | 1.16E+02 | 0.00E+00 | 0.00E+00 | 8.98E+01 | I.12E+02 | 6.46E+02 | 0.00E+00 |
| RF | RF-MT0339 | 934.74 | 2.06E+04 | 0.00E+00 | 0.00E+00 | 9.77E+03 | 1.30E+04 | 7,49E+04 | 0.002+00 |
| RF | RF-MT0374 | 1.25 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 6.27E+00 | 4.54E+00 | 2,37E+01 | 0.0000+00 |
| RF | RF-MT0375 | 0.21 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 2.45E-01 | 1.77E-01 | 9.97E-01 | 0.000000 |
| 8F | RF-MT0377 | 3.54 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.54E+02 | 1.11E402 | 5.81E+02 | 0.000000 |
| RF | RF-MT0440 | 637.99 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 2.06E+03 | 2.71E+03 | 1,38E+04 | 0.00000 |
| RF | RF-MT0442 | 1117.64 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 3.59E+03 | 4.74E+03 | 2.70E+U4 | 0.002+00 |
| RF | RF-MT0444 | 58.13 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 3.92E+01 | 5.04E+01 | 2.892402 | 0.002+00 |
| RF | RF-MT0480 | 1983.22 | 3.91E+04 | 0.00E+00 | 0.00E+00 | 9.97E+03 | 1.37E+04 | 7,93E+04 | 0.002+00 |
| RF | RF-MT0800 | 322.32 | 7.94E+03 | 0.00E+00 | 0.00E+00 | 6.09E+02 | 6.55E+02 | 3.692+03 | 0.00E+00 |
| RF | RF-MT0801 | 108.99 | 5.10E+02 | 0.00E+00 | 0.00E+00 | 4.25E+02 | 3.08E+02 | 1.012104 | 0.005+00 |
| RF | RF-MT0803 | 16.64 | 4.03E+02 | 0.00E+00 | 0.00E+00 | 3.00E+01 | 3.32E+UI | 1.85ETU4 | 0.002+00 |
| RF | RF-MT0807 | 348.08 | 8.61E+03 | 0.00E+00 | 0.00E+00 | 6.65E+02 | 7.10E+02 | 4.002103 | 0.000-+00 |
| RF | RF-MT0821 | 0.42 | 5.39E+00 | 0.00E+00 | 0.00E+00 | 3.27E+00 | 2.306+00 | 1.476701 | 0.002+00 |
| RF | RF-MT0831 | 1522.20 | 1.22E+04 | 0.00E+00 | 0.00E+00 | 3.83E+03 | J.U4E+U3 | 2.726704 | 0.002+00 |
| RF | RF-MT0832 | 2433.05 | 1.95E+04 | 0.00E+00 | 0.00E+00 | 0.12E+U3 | 8.036403 | 4.002704 | 0.00E+00 |
| RF | RF-MT0833 | 318.79 | 2.55E+03 | 0.00E+00 | 0.00E+00 | 7.986+02 | 0 506+03 | 5 496+01 | 0.00E+00 |
| RF | RF-MT0855 | 11.19 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 7.096+00 | 9.500+00 | 9.46E+07 | 0.002+00 |
| RF | RF-MT0856 | 35.91 | 0.00E+00 | 0.00E+00 | 0.002+00 | 1.132+02 | 4 442101 | 2 975-107 | 0.00E+00 |
| RF | RF-MT2116 | 2.08 | 1.27E+02 | 0.00E+00 | 0.00E+00 | 1.048101 | 3.30E101 | 7 725+05 | 2.03E-01 |
| RF-RES | RF-RESIDUES | 2800.00 | 1.19E+05 | 0.00E+00 | 8.U9E+U3 | 1.04010 | 1 935400 | 1015+01 | 0.00E+00 |
| RF | RF-T010 | 0.62 | 2.24E+01 | 0.00E+00 | 0.0000000 | 2.076700 | 0.055402 | 5 55F403 | 0.00E+00 |
| RF | RF-TT0300 | 44.48 | 0.00E+00 | 0.002100 | 0.000000 | 9.305104 | 6 70F+00 | 3.51E+01 | 0.00E+00 |
| RF | RF-TT0303 | 0.21 | 0.00E+00 | 0.00E+00 | | 3.400700 | \$ 11E+03 | 2 97E+04 | 0.00E+00 |
| RF | RF-TT0312 | 278.03 | 0.00E+00 | 0.00E+00 | | 9.615-07 | 1.08F+03 | 6.15E+03 | 0.00E+00 |
| RF | RF-TT0320 | - 29.29 | 0.00E+00 | 0.00E+00 | | 6 235-02 | £ 115+03 | 4.66E+04 | 0.00E+00 |
| RF | RF-TT0335 | 373.65 | 0.00E+00 | 0.00E+00 | | C 965103 | 1 705103 | 1016401 | 0 00E+00 |
| RF | RF-TT0338 | 40.53 | 0.00E+00 | 0.002+00 | 0.005.00 | 1 145-00 | 2 275+00 | 1.19E+01 | 0.00E+00 |
| RF | RF-TT0374 | 0.62 | 0.00E+00 | 0.00E+00 | | 1 845102 | 2 215+01 | 1.275+04 | 0.00E+00 |
| RF | RF-TT0376 | 91.34 | 0.00E+00 | 0.00E+00 | 0.002+00 | 1.646703 | | | 1 |

)

TABLE - 1 SCALED VOLUME AND ACTIVITIES FOR SELECTED RADIONUCLIDES FOR EACH WASTE STREAM

٠

| | Waste | Scaled | SCALED T | OTAL CURIES OF | EACH RADIONU | CLIDE FOR EACH | WASTE STREAM | | |
|------|-------------|-------------|---------------|----------------|-----------------|----------------|---------------|---------------|-----------------|
| SITE | Stream 11)# | Volume (m3) | Scaled Am-241 | Scaled Cm-244 | Scaled Pu-238 | Scaled Pu-239 | Scaled Pu-240 | Scaled Pu-241 | Scaled U-234 |
| RF | RF-TT0438 | 55.76 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 8.54E+02 | 9.83E+02 | 5.57E+03 | 0.00E+00 |
| RF | RF-TT0440 | 149.76 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 6.90E+02 | 8.99E+02 | 5.19E+03 | 0.00E+00 |
| RF | RF-TT0442 | 181.82 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.05E+03 | 1.18E+03 | 6.71E+03 | 0.00E+00 |
| RF | RF-TT0480 | 1446.53 | 1.53E+04 | 0.00E+00 | 0.00E+00 | \$.00E+03 | 6.38E+03 | 3.67E+04 | 0.00E+00 |
| RF | RF-TT0481 | 0.21 | 3.50E+00 | 0.00E+00 | 0.00E+00 | 2.14E+00 | 1.55E+00 | 8.10E+00 | 0.00E+00 |
| RF | RF-TT0490 | 186.97 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 3.61E+03 | 4.26E+03 | 2.42E+04 | 0.00E+00 |
| RF | RF-TT0491 | 16.02 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 8.10E+02 | 5.87E+02 | 3.06E+03 | 0.00E+00 |
| RF | RF-TT0802 | 179.15 | 6.22E+03 | 0.00E+00 | 0.00E+00 | 8.72E+01 | 1.13E+02 | 6.52E+02 | 0.00E+00 |
| RF | RF-TT0821 | 406.61 | 4.35E+03 | 0.00E+00 | 0.00E+00 | 9.61E+02 | 1.27E+03 | 7.33E+03 | 0.00E+00 |
| RF | RF-TT0823 | 159.51 | 6.69E+02 | 0.00E+00 | 0.00E+00 | 5.76E+01 | 7.43E+01 | 4.29E+02 | 0.00E+00 |
| RF | RF-TT0824 | 140.24 | 1.50E+03 | 0.00E+00 | 0.00E+00 | 5.01E+02 | 6.26E+02 | 3.59E+03 | 0.00E+00 |
| RF | RF-TT0825 | 550.34 | 5.91E+03 | 0.00E+00 | 0.00E+00 | 1.33E+03 | 1.72E+03 | 9.97E+03 | 0.00E+00 |
| RL | RL-TIOI | 567.94 | 0.00E+00 | 0.00E+00 | 2.02E+01 | 7.02E+02 | 1.64E+02 | 1.01E+03 | 3.30E-10 |
| RL | RL-T102 | 200.12 | 0.00E+00 | 0.00E+00 | 2.57E-04 | 8.96E-03 | 2.09E-03 | 1.28E-02 | 1.90E-06 |
| RL | RL-T103 | 99.63 | 0.00E+00 | 0.00E+00 | 1.08E+02 | 3.75E+03 | 8.75E+02 | 5.36E+03 | 0.00E+00 |
| RL | RL-TI04 | 4.99 | 0.00E+00 | 0.00E+00 | 3.67E-04 | 1.28E-02 | 2.99E-03 | 1.83E-02 | 5.39E-08 |
| RL | RL-T105 | 80.40 | 7.03E-02 | 0.00E+00 | 1.39E-01 | 4.85E+00 | 1.13E+00 | 6.92E+00 | 7.40E-05 |
| RL | RL-T106 | 8.11 | 0.00E+00 | 0.00E+00 | 1.36E-01 | 4.74E+00 | 1.11E+00 | 6.78E+00 | 0.00E+00 |
| RL | RL-T107 | 6156.09 | 2.03E+01 | 0.00E+00 | 8.00E+04 | 1.31E+04 | 3.05E+03 | 1.86E+04 | 1.39E+00 |
| RL | RL-T108 | 192.62 | 0.00E+00 | 0.00E+00 | 1.38E+01 | 7.45E+00 | 1.74E+00 | 1.06E+01 | 4.84E-05 |
| RL | RL-T109 | 19.72 | 3.76E-01 | 0.00E+00 | 2.84E-01 | 9.88E+00 | 2.31E+00 | 1.41E+01 | 3.85E-02 |
| RL | RL-TII0 | 494.03 | 1.42E+01 | 0.00E+00 | 5.42E+01 | 1.13E+03 | 2.65E+02 | 1.62E+03 | 2.25E+00 |
| RL | RL-TII2 | 137.74 | 3.12E+02 | 0.00E+00 | 2.29E+01 | 1.50E+02 | 3.50E+01 | 2.15E+02 | 1.22E+00 |
| RL | RL-TII3 | 42.80 | 0.00E+00 | 0.00E+00 | 4.42E-02 | 4.95E-01 | 1.16E-01 | 7.08E-01 | 0.00E+00 |
| RL | RL-TI14 | 19.58 | 0.00E+00 | 0.00E+00 | 2.16E+00 | 7.51E+01 | 1.75E+01 | 1.07E+02 | 0.00E+00 |
| RL | RL-TI15 | 1025.43 | 0.00E+00 | 0.00E+00 | 8.67E+00 | 3.04E+02 | 7.08E+01 | 4.34E+02 | 6.83E-01 |
| RL | RL-T116 | 11.02 | 0.00E+00 | 0.00E+00 | <u>3.55E+00</u> | 1.23E+02 | 2.88E+01 | 1.77E+02 | <u>9.29E-02</u> |
| RL. | RL-T118 | 261.96 | 1.95E+02 | 0.00E+00 | 2.83E+01 | 1.22E+02 | 2.85E+01 | 1.75E+02 | 1.38E+00 |
| RI. | R1-T120 | 133.81 | 0.001300.0 | 0.00E+00 | 6.54E-01 | 2.28E+01 | 5.32E+00 | 3.25E+01 | 9.33E-07 |
| RL | RL-1122 | 29.30 | 0.00E+00 | 0.00E+00 | 1.26E-01 | 4.35E+00 | 1.02E+00 | 6.23E+00 | 2.41E+00 |
| RL | RL-T123 | 0.62 | 0.00E+00 | 0.00E+00 | 3.68E-01 | 1.28E+01 | 3.00E+00 | 1.84E+01 | 9.86E-02 |
| RL | RL-T125 | 15.18 | 0.00E+00 | 0.00E+00 | 7.60E-06 | 2.64E-04 | 6.17E-05 | 3.81E-04 | 0.00E+00 |
| RL | RL-T127 | 283.60 | 1.66E+03 | 0.00E+00 | 2.29E+01 | 7.99E+02 | 1.86E+02 | I.14E+03 | 1.32E-01 |
| RL. | RL-T128 | 0.42 | 3.64E+00 | 0.00E+00 | 5.57E-07 | 1.94E-05 | 4.52E-06 | 2.77E-05 | 0.00E+00 |
| RL | RL-T129 | 28.75 | 0.00E+00 | 0.00E+00 | 1,06E+02 | 1.10E+01 | 2.55E+00 | 1.56E+01 | 1.27E-02 |
| RL | RL-T130 | 0.21 | 0.00E+00 | 0.00E+00 | 6.69E-04 | 2.34E-02 | 5.45E-03 | 3.33E-02 | 1.37E-04 |
| RL | RL-T131 | 30.16 | 5.20E+01 | 0.00E+00 | 6.54E-01 | 2.28E+01 | 5.30E+00 | 3.25E+01 | 1.36E-02 |
| RL | RL-T132 | 28.70 | 0.00E+00 | 0.00E+00 | 6.45E+01 | 2.25E+03 | 5.26E+02 | 3.21E+03 | 4.05E-01 |
| RL. | RL-T133 | 0.21 | 0.00E+00 | 0.00E+00 | 5.41E-02 | 1.89E+00 | 4.40E-01 | 2.69E+00 | 0.00E+00 |
| RL | RI-T134 | 0.21 | 0.00E+00 | 0.00E+00 | 2.79E-03 | 9.72E-02 | 2.26E-02 | 1.39E-01 | 0.00E+00 |

2

)

| TABLE | - | 1 |
|-------|---|---|

SCALED VOLUME AND ACTIVITIES FOR SELECTED RADIONUCLIDES FOR EACH WASTE STREAM

| <u> </u> | Waste | Scaled | SCALED T | OTAL CURIES OF | EACH RADIONUC | LIDE FOR EACH | WASTE STREAM | | |
|----------|------------|-------------|---------------|----------------|---------------|-----------------|---------------|----------------------|--------------|
| SITE | Stream ID# | Volume (m3) | Scaled Am-241 | Scaled Cm-244 | Scaled Pu-238 | Scaled Pa-239 | Scaled Pu-240 | Scaled Pu-241 | Scaled U-234 |
| RL | RL-T135 | 0.42 | 0.00E+00 | 0.00E+00 | 1.30E-02 | 4.54E-01 | 1.06E-01 | 6.48E-01 | 6.86E-03 |
| RL | RL-T137 | 151.63 | 1.03E+03 | 0.00E+00 | 1.64E+01 | 5.71E+02 | 1.33E+02 | \$.15E+02 | 1.03E-02 |
| RL | RL-T140 | 138.11 | 5.19E+02 | 0.00E+00 | 3.93E+00 | 1.36E+02 | 3.19E+01 | 1.95E+02 | 4.34E+01 |
| RL | RL-T143 | 403.71 | 0.00E+00 | 0.00E+00 | 1.56E+00 | 5.41E+01 | 1.26E+01 | 7.75E+01 | 6.37E-02 |
| RL | RL-TI45 | 711.19 | 0.00E+00 | 0.00E+00 | 4.42E+00 | 1.54E+02 | 3.59E+01 | 2.20E+02 | 1.48E-01 |
| RL | RL-W277 | 0.60 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| RL | RL-W278 | 0.42 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| RL | RL-W279 | 6.93 | 0.00E+00 | 0.00E+00 | 0.00E+00 | <u>3.00E+00</u> | 6.95E-01 | 4.26E+00 | 0.00E+00 |
| RL | RL-W280 | 0.21 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 9.02E-02 | 2.09E-02 | 1.28E-01 | 0.00E+00 |
| RL | RL-W281 | 0.37 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| RL | RL-W282 | 0.33 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| RL | RL-W283 | 11.65 | 1.46E+02 | 0.00E+00 | 0.00E+00 | 9.35E-02 | 0.00E+00 | 1.77E-01 | 0.00£+00 |
| RL | RL-W284 | 0.42 | 5.23E+00 | 0.00E+00 | 0.00E+00 | 3.34E-03 | 0.00E+00 | 6.32E-03 | 0.00E+00 |
| RL | RL-W285 | 1.21 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.27E+01 | 2.99E+00 | 1.90E+01 | 0.001.100 |
| RL | RL-W286 | 0.21 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 9.02E-02 | 2.092-02 | 1.28E-01 | 0.00E+00 |
| RL | RL-W287 | 0.42 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 4.38E+00 | 1.03E+00 | 6.54E+00 | 0.002+00 |
| RL | RL-W288 | 1.04 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.10E+01 | 2.58E+00 | 1.032+01 | 0.000000 |
| RL | RL-W289 | 2.08 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 2.19E+01 | 5.15E+00 | 3.1/6+01 | 0.000+00 |
| RL | RL-W290 | 2.29 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 2.41E+01 | 3.67E+00 | 3.396+01 | 0.002.100 |
| RL | RL-W291 | 7.98 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 8.40E+01 | 1.976+01 | 1.232+02 | 0.002+00 |
| RL | RL-W292 | 0.21 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 2.19E+00 | 5.15E-01 | 3.172400 | 0.002.00 |
| RL | RL-W293 | 1.25 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.31E+01 | 3.096+00 | [.90E+U] | 0.000000 |
| RL | RL-W294 | 1.04 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.10E+01 | 2.38E+UU | 1.03E+VI | 0.002.000 |
| RL | RL-W295 | 1.87 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.97E+01 | 4.642+00 | 1.94E+U1 | 0.002100 |
| RL | RL-W296 | 3.16 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 3.33E+01 | 7.832,400 | 4.9/E+U | 0.002100 |
| RL | RL-W297 | 1.66 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.75E+01 | 4.126+00 | 2.01E+U1 | 0.00E+00 |
| RL | RL-W298 | 19.34 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.83E+02 | 4.30E+01 | 4.86ETUA | 0.002100 |
| RL | RL-W299 | 0.62 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 8.15E+00 | 1.912+00 | 1.102+01 | 0.002+00 |
| RL | R1W300 | 0.42 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 3.43E+00 | 1.2/6+00 | 7.70E+UU | 0.006+00 |
| RL | RL-W301 | 0.62 | 0.0012100 | 0.00E+00 | 0.00E+00 | L.42E+UI | 3.316+00 | 2.03ETUI | 0.005+00 |
| RL | R1,-W302 | 0.42 | | 0.00E100 | 0.00E+00 | 1.245.00 | | 0.000,700 | 0.00E+00 |
| RL | RL-W303 | 0.21 | 0.00E+00 | 0.00E+00 | U.UUE+00 | 1.246700 | 1.766-01 | 0.002700 | 0.002.100 |
| RL | RL-W304 | 2.51 | 1.410+00 | 0.00E+00 | 0.001:100 | 2.025-01 | 1.496-01 | 8.000-01 | 0.000.000 |
| RL | RL-W305 | 57.01 | 1.44E+01 | 0.00E+00 | | 2.705701 | 1.175101 | 2 476101 | 0.001.100 |
| RL | RIW306 | 15.94 | 4.07E+00 | 0.00+300.0 | | 7.78ETUU | J.348700 | A 405+00 | 0.005+00 |
| RL. | RL-W307 | ll.89 | 7.69E-01 | 0.00E+00 | 0.002+00 | 4.840100 | 0.04C-U[| 1 045100 | 0.005100 |
| RL. | R1-W308 | 1.79 | 5.09E-01 | 0.00E100 | 0.002+00 | 1.436400 | 7.410-01 | 3.036700 4.04E n1 | 0.000 +00 |
| RL | RL-W309 | 0.21 | 8.46E-02 | 0.00E+00 | 0.0012+00 | J.11E-01 | 1.JIE-02 | 7.735-01 | 0.006100 |
| RL | RL-W310 | 1.60 | 4.58E-01 | 0.00E+00 | 0.002+00 | 1.036400 | J.//E-UI | 2.702700 | 0.005100 |
| RL | RL-W311 | 90.93 | 2.32E+01 | 0.00£+00 | 0.00E+00 | 4.346101 | 1.676+01 | 1.412.402 | 0,002+00 |

TABLE - 1 SCALED VOLUME AND ACTIVITIES FOR SELECTED RADIONUCLIDES FOR EACH WASTE STREAM

| | Waste | Scaled | SCALED T | OTAL CURIES OF | EACH RADIONU | CLIDE FOR EACH | WASTE STREAM | | |
|------|------------|-------------|---------------|----------------|---------------|----------------|------------------|---------------|--------------|
| SITE | Stream ID# | Volume (m3) | Scaled Am-241 | Scaled Cm-244 | Scaled Pu-238 | Scaled Pu-239 | Scaled Pu-240 | Scaled Pu-241 | Scaled U-234 |
| RL | RI_W312 | 58.59 | 1.48E+01 | 0.00E+00 | 0.00E+00 | 2.85E+01 | 1.21E+01 | 9.01E+01 | 0.00E+00 |
| RL. | RL-W313 | 114.07 | 2.97E+01 | 0.00E+00 | 0.00E+00 | 6.05E+01 | 2.42E+01 | 1.80E+02 | 0.00E+00 |
| RL | RL-W314 | 117.18 | 2.97E+01 | 0.00E+00 | 0.00E+00 | 5.70E+01 | 2.41E+01 | 1.80E+02 | 0.00E+00 |
| RL | RL-W315 | 3.16 | 8.48E-01 | 0.00E+00 | 0.00E+00 | 1.85E+00 | 6.96E-01 | 5.12E+00 | 0.00E+00 |
| RL | RL-W316 | 0.21 | 8.46E-02 | 0.00E+00 | 0.00E+00 | 3.11E-01 | 7.31E-02 | 4.95E-01 | 0.00E+00 |
| RL | RL-W317 | 16.15 | 4.16E+00 | 0.00E+00 | 0.00E+00 | 8.29E+00 | 3.39E+00 | 2.52E+01 | 0.00E+00 |
| RL | RL-W318 | 56.60 | 1.43E+01 | 0.00E+00 | 0.00E+00 | 2.70E+01 | 1.16E+01 | 8.65E+01 | 0.00E+00 |
| RL | RL-W319 | 7.56 | 3.08E+00 | 0.00E+00 | 0.00E+00 | 1.13E+01 | 2.66E+00 | L.80E+01 | 0.00E+00 |
| RL | RL-W320 | 56.60 | 1.43E+01 | 0.00E+00 | 0.00E+00 | 2.70E+01 | 1.16E+01 | 8.65E+01 | 0.00E+00 |
| RL | RL-W321 | 0.21 | 8.46E-02 | 0.00E+00 | 0.00E+00 | 3.11E-01 | 7.31E-02 | 4.95E-01 | 0.00E+00 |
| RL | RL-W322 | 15.94 | 4.07E+00 | 0.00E+00 | 0.00E+00 | 7.98E+00 | 3.32E+00 | 2.47E+01 | 0.00E+00 |
| RL | RL-W323 | 14.36 | 3.65E+00 | 0.00E+00 | 0.00E+00 | 7.05E+00 | 2.97E+00 | 2.21E+01 | 0.00E+00 |
| RL | RL-W324 | 3.78 | L.54E+00 | 0.00E+00 | 0.00E+00 | 5.64E+00 | 1.33E+00 | 8.99E+00 | 0.00E+00 |
| RL | RL-W325 | 8.66 | 2.21E+00 | 0.00E+00 | 0.00E+00 | 4.30E+00 | 1.80E+00 | 1.34E+01 | 0.00E+00 |
| RL | RL-W326 | 56.80 | 1.43E+01 | 0.00E+00 | 0.00E+00 | 2.73E+01 | 1.17E+01 | 8.70E+01 | 0.00E+00 |
| RL | RL-W327 | 789.89 | 2.06E+02 | 0.00E+00 | 0.00E+00 | 4.21E+02 | 1.68E+02 | 1.25E+03 | 0.00E+00 |
| RL | RL-W328 | 3.78 | L.54E+00 | 0.00E+00 | 0.00E+00 | 5.64E+00 | 1.33E+00 | 8.99E+00 | 0.00E+00 |
| RL | RL-W329 | 57.01 | 1.44E+01 | 0.00E+00 | 0.00E+00 | 2.76E+01 | 1.17E+01 | 8.75E+01 | 0.00E+00 |
| RL | RL-W330 | 281.70 | 7.47E+01 | 0.00E+00 | 0.00E+00 | 1.59E+02 | 6.12E+01 | 4.52E+02 | 0.00E+00 |
| RL | RL-W331 | 721.16 | 1.86E+02 | 0.00E+00 | 0.00E+00 | 3.75E+02 | 1.52E+02 | 1.13E+03 | 0.00E+00 |
| RL | RL-W332 | 0.20 | 8.14E-02 | 0.00E+00 | 0.00E+00 | 2.99E-01 | 7.03 E-02 | 4.76E-01 | 0.00E+00 |
| RL | RL-W333 | 17.73 | 4.58E+00 | 0.00E+00 | 0.00E+00 | 9.21E+00 | 3.74E+00 | 2.77E+01 | 0.00E+00 |
| RL | RL-W334 | 0.21 | 8.46E-02 | 0.00E+00 | 0.00E+00 | 3.11E-01 | 7.31E-02 | 4.95E-01 | 0.00E+00 |
| RL | RL-W335 | 2.10 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.18E-01 | 1.75E-02 | 0.00E+00 | 0.00E+00 |
| RL | RL-W336 | 0.42 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| RL | RL-W338 | 0.21 | 9.52E-02 | 0.00E+00 | 0.00E+00 | 3.34E-03 | 1.74E-03 | 1.50E-02 | 0.00E+00 |
| RL | RL-W339 | 0.42 | 1.90E-01 | 0.00E+00 | 0.00E+00 | 6.68E-03 | 3.48E-03 | 3.00E-02 | 0.00E+00 |
| RL | RL-W340 | 0.21 | 9.52E-02 | 0.00E+00 | 0.00E+00 | 3.34E-03 | 1.74E-03 | 1.50E-02 | 0.00E+00 |
| RL | RL-W341 | 0.21 | 8.46E-02 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.98E-03 | 0.00E+00 |
| RL | RL-W342 | 0.83 | 3.39E-01 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 7.90E-03 | 0.00E+00 |
| RL | RL-W343 | 0.62 | 2.54E-01 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 5.93E-03 | 0.00E+00 |
| RL | RL-W344 | 0.21 | 6.35E-02 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.98E-03 | 0.00E+00 |
| RL | RL-W345 | 8.95 | 2.54E+00 | 0.00E+00 | 0.00E+00 | 6.17E+00 | 2.10E+00 | 1.53E+01 | 0.00E+00 |
| RL | RL-W346 | 0.42 | I.61E+00 | 0.00E+00 | 0.00E+00 | 2.04E-01 | 5.57E-02 | 1.84E-01 | 0.00E+00 |
| RL | RL-W347 | 0.21 | 8.04E-01 | 0.00E+00 | 0.00E+00 | 1.02E-01 | 2.78E-02 | 9.21E-02 | 0.00E+00 |
| RL | RL-W348 | 0.21 | 8.04E-01 | 0.00E+00 | 0.00E+00 | 1.02E-01 | 2.78E-02 | 9.21E-02 | 0.00E+00 |
| RL | RL-W349 | 0.21 | 8.04E-01 | 0.00E+00 | 0.00E+00 | 1.02E-01 | 2.78E-02 | 9.21E-02 | 0.00E+00 |
| RL | RL-W350 | 0.21 | 8.04E-01 | 0.00E+00 | 0.00E+00 | 1.02E-01 | 2.78E-02 | 9.21E-02 | 0.00E+00 |
| RL | RL-W351 | 0.21 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| RL | RL-W352 | 0.21 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |

TABLE - 1

SCALED VOLUME AND ACTIVITIES FOR SELECTED RADIONUCLIDES FOR EACH WASTE STREAM

| | Waste | Scaled | SCALED T | OTAL CURIES OF | EACH RADIONUC | LIDE FOR EACH | WASTE STREAM | | |
|-----------|------------|-------------|----------------------|----------------|---------------|---------------|---------------|---------------|--------------|
| SITE | Stream ID# | Volume (m3) | Scaled Am-241 | Scaled Cm-244 | Scaled Pu-238 | Scaled Pu-239 | Scaled Pu-240 | Scaled Pu-241 | Scaled U-234 |
| B1 | RI-W353 | 0.83 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 3.22E+00 | 7.52E-01 | 4.60E+00 | 0.00E+00 |
| 81 | RL-W354 | 0.21 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 8.05E-01 | 1.88E-01 | 1.15E+00 | 0.00E+00 |
| | RL-W355 | 2.08 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 8.05E+00 | 1.88E+00 | 1.15E+01 | 0.00E+00 |
| | RL-W356 | 1.25 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 4.83E+00 | 1.13E+00 | 6.90E+00 | 0.00E+00 |
| | RL-W357 | 0.21 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| RI | RLW358 | 2.50 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 5.81E-01 | 1.25E-01 | 8.25E-01 | 0.00E+00 |
| RI | RI-W359 | 16.64 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 3.87E+00 | 8.35E-01 | 5.50E+00 | 0.00E+00 |
| RL. | RL-W360 | 4.78 | 0.00E+00 | 0.00E+00 | 0.00E+00 | I.11E+00 | 2.40E-01 | 1.58E+00 | 0.00E+00 |
| RI. | RL-W361 | 0.62 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.45E-01 | 3.13E-02 | 2.06E-01 | 0.00E+00 |
| RL | RL-W362 | 16.64 | 5.72E-01 | 0.00E+00 | 0.00E+00 | 3.21E+01 | 1.19E+01 | 8.92E+01 | 0.00E+00 |
| RI. | RL-W363 | 1.58 | 5.30E-02 | 0.00E+00 | 0.00E+00 | 2.83E+00 | 1.09E+00 | \$.28E+00 | 0.00E+00 |
| RL. | RL-W364 | 11.69 | 4.03E-01 | 0.00E+00 | 0.00E+00 | 2.27E+01 | 8.34E+00 | 6.28E+01 | 0.001.+00 |
| R1. | RL-W365 | 64.04 | 2.22E+00 | 0.00E+00 | 0.00E+00 | 1.26E+02 | 4.59E+01 | 3.45E+02 | 0.00E+00 |
| RL. | RL-W366 | 6.95 | 2.44E-01 | 0.00E+00 | 0.00E+00 | 1.42E+01 | 5.06E+00 | 3.79E+01 | 0.006+00 |
| RL | RL-W367 | 16.64 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 4.28E+00 | 1.53E+00 | 1.01E+01 | 0.0000+00 |
| RL | RL-W368 | 4.74 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.13E+00 | 4.226-01 | 2.81E+00 | 0.002+00 |
| RL | RL-W369 | 161.21 | 6.78E+01 | 0.00E+00 | 0.00E+00 | 2.28E+02 | 8.09E+01 | 3.40E+02 | 0.00E+00 |
| RL | RL-W370 | 0.42 | 2.54E-01 | 0.00E+00 | 0.00E+00 | I.32E+00 | 3.178-01 | 1.9/6+00 | 0.005+00 |
| RL | RL-W371 | 21.17 | 9.30E+00 | 0.00E+00 | 0.00E+00 | 3.36E+01 | 1.120+01 | 1.075.00 | 0.002+00 |
| RL | RL-W372 | 0.42 | 2.54E-01 | 0.00E+00 | 0.00E+00 | 1.32E+00 | J.1/E-UL | 1.976+00 | 0.000-100 |
| RL | RL-W373 | 88.45 | 4.33E+00 | 0.00E+00 | 0.00E+00 | 4.636+00 | 1.426700 | 7.176+03 | 0.00E+00 |
| RL | RL-W374 | 2800.78 | 8.11E+02 | 0.00E+00 | 0.002+00 | 2.012+03 | 1.000103 | 1.372.03 | 0.00E+00 |
| RL | RL-W375 | 272.44 | 7.96E+01 | 0.00E+00 | 0.002+00 | 2.012+02 | 1.036102 | 9 56F+07 | 0.00E+00 |
| RL | RL-W376 | 367.78 | 1.05E+02 | 0.00E+00 | 0.00E+00 | 5.26E+U2 | 7.58E+02 | 1 83F+04 | 0.00E+00 |
| RL | RL-W377 | 7029 61 | 2.01E+03 | 0.00E+00 | 0.002+00 | 2 795+02 | 1 155+02 | \$ 00E+02 | 0.00E+00 |
| RL | RL-W378 | 306.06 | 8.81E+01 | 0.00E+00 | 0.00000 | 4.77ETV2 | 1 325.01 | 8 14E-01 | 0.00E+00 |
| RL | RL-W379 | 0.21 | 9.52E-02 | 0.00E+00 | 0.00E+00 | 5.63E-01 | 1.32E-01 | R 34E-01 | 0.00E+00 |
| RL | RL-W380 | 0.21 | 9.52E-02 | 0.00E+00 | 0.005.00 | 1 435402 | 6 07E+01 | 4.22E+02 | 0.00E+00 |
| RL | RL-W381 | 162.79 | 4.64E+01 | 0.002+00 | 0.002100 | 3 785+07 | 1 59E+02 | 1.10E+03 | 0.00E+00 |
| RL | RL-W382 | 423.84 | 1.21E+02 | 0.006+00 | 0.002+00 | 2 56E+01 | 6.01E+00 | 3.79E+01 | 0.00E+00 |
| RL | RL-W383 | 9.45 | 4.33E+UU | 0.00000 | 0.000000 | 461E-01 | L.15E-01 | 6.38E-01 | 0.00E+00 |
| RL | RL-W384 | 0.62 | 1.81E+00 | 0.000000 | 0.000000 | 2 67E+01 | 7.87E+00 | 4.48E+01 | 0.00E+00 |
| <u>rl</u> | RL-W385 | 12.23 | 1.352+01 | 0.005+00 | 0.002+00 | 1.19E+00 | 3.13E-01 | 1.74E+00 | 0.00E+00 |
| RL | RL-W386 | 0.42 | 3.292-01 | 0.002+00 | 0.002700 | 5.31E+00 | 1.69E+00 | 9.73E+00 | 0.00E+00 |
| RL | RL-W387 | 2.83 | 2.916+00 | 0.00E100 | 0.0000+00 | 5.13E+01 | 1.44E+01 | \$.09E+01 | 0.00E+00 |
| RL | RL-W388 | 20.83 | 1.43E+UI | 0.00E100 | 0.000.00 | 5.94E-01 | 1.57E-01 | 8.70E-01 | 0.00E+00 |
| RL | RL-W389 | 0.2 | 4.04E-UI 7.03E.01 | 0.002700 | 0.005+00 | 1.78E+00 | 4.70E-01 | 2.61E+00 | 0.00E+00 |
| RL | RL-W390 | 0.62 | 1.73E-01 | 0.00ET00 | 0.002+00 | L.19E+00 | 3.13E-01 | 1.74E+00 | 0.00E+00 |
| RL | RL-W391 | 0.42 | 3.492-01 | 0.002100 | 0.002100 | 3.01F-03 | 1.74E-03 | 8.30E-03 | 0.00E+00 |
| 101 | RE-W392 | 0.21 | 1 0.00E+00 | 0.008400 | 1.002.100 | 1 | 1 | | L |

Ì

| | Waste | Scaled | SCALED T | OTAL CURIES OF | EACH RADIONUC | LIDE FOR EACH | WASTE STREAM | | |
|------|----------------|-------------|---------------|----------------|---------------|---------------|-----------------|-----------------|--------------|
| SITE | Stream 1D# | Votume (m3) | Scaled Am-241 | Scaled Cm-244 | Scaled Pu-238 | Scaled Pu-239 | Scaled Pu-240 | Scaled Pu-241 | Scaled U-234 |
| RL | RI-W393 | 67.21 | 4.13E+02 | 0.00E+00 | 0.00E+00 | 3.98E+01 | 2.55E+01 | 1.83E+02 | 0.00E+00 |
| RI | RL-W194 | 49.81 | 3.03E+02 | 0.00E+00 | 0.00E+00 | 2.\$5E+01 | 1.86E+01 | 1.34E+02 | 0.00E+00 |
| RI | RI-W195 | 174.45 | 1.11E+03 | 0.00E+00 | 0.00E+00 | 1.14E+02 | 6.\$7E+01 | 4.89E+02 | 0.00£+00 |
| RI | RL-W396 | 0.21 | 2.00E+00 | 0.00E+00 | 0.00E+00 | 3.47E-01 | 1.30E-01 | 8.55E-01 | 0.00E+00 |
| 81. | RL-W397 | 55.72 | 3,39E+02 | 0.00E+00 | 0.00E+00 | 3.19E+01 | 2.09E+01 | 1.50E+02 | 0.00E+00 |
| RI. | RL-W398 | 0.21 | 2.00E+00 | 0.00E+00 | 0.00E+00 | 3.47E-01 | 1.30E-01 | 8.55E-01 | 0.00E+00 |
| RI. | RL-W399 | 23.55 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 9.16E+01 | 8.94E+01 | 1.42E+03 | 0.00E+00 |
| RI | R1_W400 | 15.31 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 6.02E+01 | 5.83E+01 | 9.28E+02 | 0.00E+00 |
| RI. | RL-W401 | 214.86 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 8.24E+02 | 8.12E+02 | 1.29E+04 | 0.00E+00 |
| RL | RL-W402 | 14.98 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 4.08E+01 | <u>1.62E+01</u> | 1.08E+02 | 0.00E+00 |
| RI | RL-W403 | 0.62 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 4.76E+00 | <u>1.11E+00</u> | 6.79E+00 | 0.00E+00 |
| RI | RL-W404 | 15.81 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 4.71E+01 | 1.76E+01 | <u>1.17E+02</u> | 0.00E+00 |
| RL | RL-W405 | 0.21 | 9.10E+00 | 0.00E+00 | 0.00E+00 | 9.02E-02 | 2.09E-02 | 1.29E-01 | 0.00E+00 |
| RL | RL-W406 | 0.42 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 7.01E-02 | 1.74E-02 | 9.80E-02 | 0.00E+00 |
| SR | T001-221F-HET | 11492.34 | 9,51E+03 | 0.00E+00 | 7.17E+05 | 2.78E+04 | 5.56E+03 | 1.66E+03 | 0.00E+00 |
| SR | T001-221F-MET | 490.50 | 3.98E+02 | 0.00E+00 | 2.99E+04 | 1.11E+03 | 2.32E+02 | 6.95E+03 | 0.00E+00 |
| SR | T001-221F-VIT | 954.27 | 4.95E+02 | 4.68E+03 | 3.71E+04 | 1.33E+02 | 2.88E+02 | 8.66E+03 | 0.00E+00 |
| SR | T001-221H-HET | 6572.31 | 5.25E+03 | 0.00E+00 | 3.93E+05 | 1.41E+04 | 3.05E+03 | 9.18E+04 | 0.00E+00 |
| SR | T001-221H-MET | 95.38 | 7.54E+01 | 0.00E+00 | 5.64E+03 | 1.97E+02 | 4.38E+01 | 1.32E+03 | 0.002+00 |
| SR | T001-221H-VIT | 3192.47 | 1.64E+03 | 1.57E+04 | L.23E+05 | 4.33E+02 | 9.53E+02 | 2.87E+04 | 0.00E+00 |
| SR | T001-235F-HET | 1517.71 | 1.28E+03 | 0.00E+00 | 9.65E+04 | 3.85E+03 | 7.48E+02 | 2.22E+04 | 0.002+00 |
| SR | T001-235F-VIT | 566.20 | 2.90E+02 | 2.79E+03 | 2.17E+04 | 7.54E+01 | 1.68E+02 | 5.07E+03 | 0.008+00 |
| SR | T001-772F-HET | 104.88 | 9.72E+01 | 0.00E+00 | 7.46E+03 | 3.51E+02 | 5.78E+01 | [.68E+03 | 0.00E+00 |
| SR | T001-772F-VIT | 50.24 | 2.57E+01 | 2.47E+02 | 1.92E+03 | 6.71E+00 | 1.49E+01 | 4.51E+02 | 0.002+00 |
| SR | T001-773A-CLAS | 4.58 | 5.92E+00 | 0.00E+00 | 4.73E+02 | 3.11E+01 | 3.66E+00 | 9.97E+UI | 0.002+00 |
| SR | T001-773A-HET | 1721.93 | 1,36E+03 | 0.00E+00 | 1.02E+05 | 3.60E+03 | 7.93E+02 | 2.39E+04 | 0.000000 |
| SR | T001-773A-MET | 210.01 | 1.65E+02 | 0.00E+00 | 1.24E+04 | 4.28E+02 | 9.598401 | 2.90E+03 | 0.002100 |
| SR | T001-773A-VIT | 100.37 | 5,14E+01 | 4.94E+02 | 3.84E+03 | 1.34E+01 | 2.98E+01 | 9.002+02 | 0.002+00 |
| SR | T003-773A-HET | 45.94 | 0.00E+00 | 0.00E+00 | 5.98E+00 | 0.00E+00 | 0.00E+00 | 0.0012+00 | 0.002+00 |
| SR | T003-773A-VIT | 0.21 | 1.75E-01 | 7.85E-01 | 1.40E+01 | 9.22E-02 | 1.086-01 | 2.94E+00 | 0.002100 |
| SR | W006-773A-VIT | 0.52 | 1.09E-02 | 0.00E+00 | 0.00E+00 | 2.36E+02 | 0.006+00 | U.UUE+00 | 0.005+00 |
| SR | W027-221F-HET | 265.62 | 3.44E+02 | 0.00E+00 | 2.75E+04 | 1.\$0E+03 | 2.[JE+02 | 3.79E+03 | 0.000+00 |
| SR | W027-221F-MET | 1.69 | 2.45E+00 | 0.00E+00 | 1.95E+02 | 1.28E+01 | 1.51£+00 | 4.12E+UL | 0.002100 |
| ŜR | W027-221F-VIT | 33.18 | 2.79E+01 | 1.25E+02 | 2.23E+03 | 1.47E+01 | 1.73E+01 | 4.70E+02 | 0.002+00 |
| SR | W027-221H-HET | 125.42 | 1.62E+02 | 0.00E+00 | 1.30E+04 | 8.52E+02 | 1.00E+02 | 2.736+03 | 0.002100 |
| SR | W027-221H-MET | 1.89 | 2.45E+00 | 0.00E+00 | 1.95E+02 | 1.28E+01 | 1.316+00 | 4.12B+UI | 0.005+00 |
| SR | W027-221H-VIT | 25.88 | 2.18E+01 | 9.77E+01 | 1.74E+03 | 1.15E+01 | 1.33E+01 | J.60E+02 | 0.000+00 |
| SR | W027-235F-HET | 34.74 | 4.50E+01 | 0.00E+00 | 3.59E+03 | 2.36E+02 | 2.78E+01 | 7.576+02 | 0.002.00 |
| en | W027-231E-MET | 1.89 | 2.45E+00 | 0.00E+00 | L 1.95E+02 | 1.28E+01 | 1.516+00 | 4.126401 | 0.006+00 |

TABLE - 1 SCALED VOLUME AND ACTIVITIES FOR SELECTED RADIONUCLIDES FOR EACH WASTE STREAM

B2-18

SR SR

W027-235F-MET

W027-235F-VIT

Page 14

1.11E+03

6.26E+01

7.35E+00

8.63E+00

0.00E+00

2.35E+02

1.39E+01

1.89

16.59

| TABLE - 1 | |
|--|---------------------------------------|
| SCALED VOLUME AND ACTIVITIES FOR SELECTED RADION | NUCLIDES FOR EACH WASTE STREAM |

)

| | Waste | Waste | Scaled | SCALED T | OTAL CURIES OF | EACH RADIONU | CLIDE FOR EACH | WASTE STREAM | | |
|--------|---------------|-------------|---------------|---------------|-----------------------|---------------|----------------|---------------|--------------|--|
| SITE | Stream 1D# | Volume (m3) | Scaled Am-241 | Scaled Cm-244 | Scaled Pu-238 | Scaled Pu-239 | Scaled Pu-240 | Scaled Pu-241 | Scaled U-234 | |
| SR | W027-772F-HET | 515.42 | 6.67E+02 | 0.00E+00 | 5.33E+04 | 3.50E+03 | 4.12E+02 | 1.12E+04 | 0.00E+00 | |
| SR | W027-772F-MET | 32.13 | 4.16E+01 | 0.00E+00 | 3.32E+03 | 2.18E+02 | 2.57E+01 | 7.00E+02 | 0.00E+00 | |
| SR | W027-772F-VIT | 10.62 | 8.93E+00 | 4.01E+01 | 7.13E+02 | 4.70E+00 | 5.52E+00 | 1.50E+02 | 0.00E+00 | |
| SR | W027-773A-HET | 331.14 | 4.29E+02 | 0.00E+00 | 3.42E+04 | 2.25E+03 | 2.65E+02 | 7.22E+03 | 0.00E+00 | |
| SR | W027-773A-MET | 7.56 | 9.78E+00 | 0.00E+00 | 7.81E+02 | 5.13E+01 | 6.05E+00 | 1.65E+02 | 0.00E+00 | |
| SR | W027-773A-VIT | 17.25 | 1.45E+01 | 6.51E+01 | 1.16E+03 | 7.64E+00 | 8.97E+00 | 2.44E+02 | 0.00E+00 | |
| SR-OFF | W027-999-HET | 27.66 | 6.85E+01 | 0.00E+00 | 1.15E+05 | 7.87E+01 | 4,56E+01 | 9.88E+02 | 0.00E+00 | |
| SR-OFF | W027-999-VIT | 31.85 | 5.12E+01 | 0.00E+00 | 8.61E+04 | 5,91E+00 | 3.41E+01 | 7.38E+02 | 0.00E+00 | |
| SR-OFF | W053-773A-VIT | 0.52 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 7.36E+01 | 0.00E+00 | 0.00E+00 | 0.00E+00 | |
| TOTALS | | 168500.00 | 4.42E+05 | 3.15E+04 | 2.61E+06 | 7.85E+05 | 2.10E+05 | 2.31E+06 | 4.65E+02 | |

<u>ج</u> _

)

Table 2NORMALIZATION FACTORS (NF)

TOTAL CURIES ESTIMATED FROM BIR REV. 2 WASTE STREAM DATA

| | UN | DECAYED | STORED C | URIES OF | EACH RA | DIONUCL | DE |
|----------|----------|----------|----------|-----------|----------|----------|----------|
| SITE | Am241 | Cm244 | Pu238 | Pu239 | Pu240 | Pu241 | U234 |
| AE Total | 3.90E+01 | 0.00E+00 | 7.45E-05 | 2.14E+01 | 0.00E+00 | 1.12E+01 | 0.00E+00 |
| AL Total | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| AW Total | 6.97E+00 | 0.00E+00 | 0.00E+00 | 5.54E-01 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| BT Total | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| ET Total | 2.52E-02 | 0.00E+00 | 2.02E-02 | 1.34E-01 | 3.36E-02 | 8.40E-01 | 3.36E-04 |
| IN Total | 8.11E+04 | 9.29E-01 | 6.34E+04 | 4.35E+04 | 1.10E+04 | 2.38E+05 | 0.00E+00 |
| LA Total | 3.12E+04 | 2.29E+02 | 1.36E+05 | 1.86E+04 | 4.10E+03 | 6.97E+04 | 1.54E-01 |
| LL Total | 1.43E+02 | 8.06E+01 | 4.18E+01 | 1.71E+02 | 7.96E+01 | 2.44E+03 | 0.00E+00 |
| MC Total | 1.55E-01 | 0.00E+00 | 0.00E+00 | 6.07E-02 | 0.00E+00 | 2.77E-01 | 0.00E+00 |
| MD Total | 0.00E+00 | 0.00E+00 | 2.44E+03 | 3.84E+01 | 5.36E+02 | 0.00E+00 | 0.00E+00 |
| NT Total | 3.01E+02 | 4.16E+00 | 1.49E+02 | .2.81E+03 | 2.61E+01 | 5.25E+02 | 5.00E-03 |
| OR Total | 1.10E+03 | 4.51E+00 | 3.55E+02 | 1.58E+01 | 1.82E+01 | 1.75E+03 | 1.87E+00 |
| RF Total | 6.22E+02 | 0.00E+00 | 0.00E+00 | 1.20E+03 | 2.76E+02 | 9.07E+03 | 0.00E+00 |
| RL Total | 9.30E+02 | 0.00E+00 | 1.03E+05 | 3.27E+04 | 7.35E+03 | 1.99E+05 | 3.25E+01 |
| SA Total | 1.35E+00 | 4.33E+00 | 0.00E+00 | 2.70E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| SR Totai | 7.66E+02 | 1.69E+01 | 2.13E+05 | 1.72E+04 | 8.76E+02 | 4.26E+04 | 0.00E+00 |
| SR-OFF | 1.34E+01 | 3.31E+00 | 3.73E+03 | 7.12E+02 | 1,53E+01 | 7.45E+02 | 0.00E+00 |

| | TOTAL U | NDECAYE | D CURIES | REPORTE | ED BY TH | E SITE IN | THE IDB |
|----------|----------|----------|-----------------|----------|----------|-----------|----------|
| SITE | Am241 | Cm244 | Pu238 | Pu239 | Pu240 | Pu241 | U234 |
| ARCO | 0.00E+00 | 0.00E+00 | 3.73E+02 | 0.00E+00 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| ARMY | 0.00E+00 | 0.00E+00 | 0.00E+00 | 1.80E+01 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| ETEC | 4.54E-01 | 0.00E+00 | 1.16E-01 | 1.79E+00 | 6.12E-01 | 8.29E+00 | 0.00E+00 |
| HANF | 3.76E+03 | 4.82E+03 | 9.06E+04 | 2.63E+04 | 6.15E+03 | 7.08E+04 | 5.01E+01 |
| INEL | 8.79E+04 | 1.13E+03 | 6.75E+04 | 4.01E+04 | 9.83E+03 | 2.88E+05 | 3.36E+00 |
| LANL | 8.69E+03 | 2.23E+02 | 1.31E+05 | 7.69E+04 | 1.00E+02 | 1.70E+03 | 0.00E+00 |
| LBL | | | | | | | |
| LLNL | 1.33E+02 | 7.44E+01 | 7.75E+01 | 1.58E+02 | 6.44E+01 | 1.97E+03 | 2.78E-03 |
| MOUND | 0.00E+00 | 0.00E+00 | 1.68E+03 | 2.98E+01 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| MURR | 3.24E-01 | 0.00E+00 | 0.00E+00 | 2.46E-02 | 0.00E+00 | 6.63E-03 | 0.00E+00 |
| NEVADA | 2.86E+02 | 3.54E+02 | 2.16E+02 | 2.76E+03 | 1.84E+01 | 3.31E+02 | 5.00E-03 |
| ORNL | 6.19E+02 | 2.26E+03 | 3.98E+03 | 1.01E+03 | 9.44E+02 | 7.84E+04 | 1.55E+01 |
| PAD | | | | | | | |
| PANTEX | 0.00E+00 | 0.00E+00 | 0.00E+00 | 5.55E-02 | 0.00E+00 | 0.00E+00 | 0.00E+00 |
| RFETS | 1.06E+04 | 0.00E+00 | 3.56E+02 | 9.98E+03 | 7.22E+03 | 6.58E+04 | 0.00E+00 |
| RF-RES | | | | | | | |
| SRS-ON | 2.11E+03 | 1.16E+03 | 3.14E+05 | 9.13E+03 | 2.21E+03 | 1.06E+05 | 3.00E-01 |
| SR-OFF | 1.87E+00 | 0.00E+00 | 2.43E+05 | 1.58E+02 | 7.99E+01 | 5.34E+03 | 3.37E-04 |
| SR-TOTAL | 2.11E+03 | 1.16E+03 | 5.57E+05 | 9.29E+03 | 2.29E+03 | 1.11E+05 | 3.00E-01 |



Page 1

Table 2 (continued) NORMALIZATION FACTORS (NF)

| CALCULATION OF IDB/BIR RATIOS (NF) | | | | | | | | | | | |
|------------------------------------|----------|----------|----------|----------|----------|----------|----------|--|--|--|--|
| | | | | | | | | | | | |
| SITE | Am241 | Cm244 | Pu238 | Pu239 | Pu240 | Pu241 | U234 | | | | |
| RL | 4.04E+00 | NC | 8.81E-01 | 8.03E-01 | 8.37E-01 | 3.56E-01 | 1.54E+00 | | | | |
| IN | 1.08E+00 | 1.22E+03 | 1.06E+00 | 9.22E-01 | 8.97E-01 | 1.21E+00 | NC | | | | |
| LA | 2.79E-01 | 9.74E-01 | 9.65E-01 | 4.13E+00 | 2.44E-02 | 2.44E-02 | 0.00E+00 | | | | |
| LL | 9.31E-01 | 9.23E-01 | 1.85E+00 | 9.26E-01 | 8.09E-01 | 8.05E-01 | NC | | | | |
| MD | NC | NC | 6.90E-01 | 7.77E-01 | 0.00E+00 | NC | NC | | | | |
| NT | 9.49E-01 | 8.51E+01 | 1.45E+00 | 9.83E-01 | 7.07E-01 | 6.31E-01 | 1.00E+00 | | | | |
| OR | 5.61E-01 | 5.02E+02 | 1.12E+01 | 6.38E+01 | 5.18E+01 | 4.47E+01 | 8.25E+00 | | | | |
| RF | 1.71E+01 | NC | NC | 8.29E+00 | 2.62E+01 | 7.26E+00 | NC | | | | |
| RF-RES | | | | | | | | | | | |
| SR | 2.75E+00 | 6.86E+01 | 1.47E+00 | 5.30E-01 | 2.52E+00 | 2.49E+00 | NC | | | | |
| SR-OFF | 1.40E-01 | 0.00E+00 | 6.51E+01 | 2.22E-01 | 5.21E+00 | 7.18E+00 | NC | | | | |
| SR-TOTAL | 2.76E+00 | 6.86E+01 | 2.61E+00 | 5.40E-01 | 2.62E+00 | 2.61E+00 | NC | | | | |

Γ

NOTE:

NC ---> Cannot Be Calculated Due to Data Discrepancy

Page 2

B2-21

Table 3RADIONUCLIDE SCALING FACTORS (SFa)

| TOTAL | ESTIMATEI | D ACTIVITY | FOR STOR | ED VOLUM | E (Without S | caie-up) |
|---------------|------------------|---------------|---------------|---------------|---------------|--------------|
| Stored Am-241 | Stored Cm-244 | Stored Pu-238 | Stored Pu-239 | Stored Pu-240 | Stored Pu-241 | Stored U-234 |
| 2.40E+05 | 2.61E+03 | 7.55E+05 | 3.60E+05 | 6.88E+04 | 1.08E+06 | 7.54E+01 |

| TOTAL E | STIMATED | ACTIVITY F | OR PROJEC | TED VOLUI | ME (Without | Scale-up) |
|--------------|--------------|--------------|--------------|--------------|--------------|-------------|
| Proj. Am-241 | Proj. Cm-244 | Proj. Pu-238 | Proj. Pu-239 | Proj. Pu-240 | Proj. Pu-241 | Proj. U-234 |
| 5.05E+04 | 3.35E+03 | 4.94E+05 | 2.16E+05 | 3.75E+04 | 2.96E+05 | 3.64E+00 |

| | TOTAL WIPP | ACTIVITIE | S (Based on | CCA Radion | uclide Table) | |
|----------|------------|------------------|-------------|---------------|----------------|----------|
| Am-241 | Cm-244 | Pu-238 | Pu-239 | Pu-240 | Pu-2 41 | U-234 |
| 4.42E+05 | 3.15E+04 | 2.61E+06 | 7.85E+05 | 2.10E+05 | 2.31E+06 | 4.65E+02 |

| | CALCULATI | ED SCALING | FACTOR FC | R EACH NU | CLIDE | |
|--------|-----------|------------|---------------|---------------|--------|--------|
| Am-241 | Cm-244 | Pu-238 | Pu-239 | Pu-240 | Pu-241 | U-234 |
| 4.01 | 8.61 | 3.75 | 1.97 | 3.76 | 4.17 | 106.94 |
Table 4VOLUME SCALING FACTOR (SF.)

 WIPP CAPACITY FOR CH-TRU WASTE

 168500

 TOTAL STORED VOLUME FOR ALL WASTE STREAMS

 58533.25

 TOTAL PROJ. VOLUME FOR ALL WASTE STREAMS WITH RAD DATA

 16865.15

| VOLUME | SCALING | FACTOR | (SF.) | |
|--------|---------|--------|-------|------|
| | | | | 6.52 |

Note: (168500 - 58533.25) / 16865.15 = 6.52



•

.

.

•

.

.

United States Government

Carlsbad Area Office Carlsbad, New Mexico 88221

DATE: MAR 1 5 1996

ATTN OF: CAO:NTP:RLB 96-0687

SUBJECT: Preliminary Estimate of Complexing Agents in TRU Solidified Waste Forms Scheduled for Disposal in WIPP

TO:

Les E.Shephard, Director, SNL/NM

memorandum

Attached is a copy of the report containing the preliminary estimates of complexing agents in transuranic (TRU) solidified waste forms scheduled for disposal in the Waste Isolation Pilot Plant (WIPP). This information was requested from the Transuranic (TRU) Waste Baseline Inventory Report (TWBIR) team in support of the Performance Assessment (PA) being conducted by Sandia National Laboratory (SNL). Information has been received from the Rocky Flats Environmental Technology Site (RFETS), the Los Alamos National Laboratory (LANL), and the Oak Ridge National Laboratory (ORNL) on potential complexing agents in their solidified waste forms.

The original scope of this request was to ask the TRU waste generator/storage sites about potential "aqueous-soluble chelating agents" in their solidified waste forms. As this subject was researched, two things were realized. First, in lieu of the term "chelating agent," the term "complexing agent" should be used. "Chelating agents" are a subset of "complexing agents" and as such a more complete assessment would cover the presence of potential "complexing agents." Secondly, it was recognized that "aqueous-soluble" is a relative concept in that essentially everything is "aqueous-soluble" at some concentration level. Therefore, the data provided here are for all complexing agents reported by the sites. These data will allow SNL personnel to determine the cutoff of solubility where certain compounds are no longer considered to be of interest for PA calculations.

The final report at the end of March will contain the necessary attached documentation, references, and elaborated text summaries.

If you have any questions concerning the attached information, please contact Mr. Russ Bisping of my staff at (505) 234-7446.

Manager National TRU Program





B3-1

cc w/attachment: K. Hunter, CAO M. McFadden, CAO R. Bisping, CAO P. Drez, CTAC J. Harvill, CTAC L. Sanchez, SNL M. Chu, SNL M. Marietta, SNL

(M)

- 2 -

B3-2

Complexing Agents Site Summaries

ORNL

ORNL has provided a list of organic compounds which contain some aqueous-soluble compounds that are apparent complexing agents. A copy of the list of all compounds reported by ORNL to the BIR team is attached for completeness (Table 1). The list in Table 1 is from an ORNL report on low-level waste, but the same compounds are anticipated to occur in the TRU waste based on process history. ORNL cannot quantify these compounds in their solidified wastes, but have provided an estimate of Total Organic Carbon (TOC) for each TRU waste tank (Table 2). The sum of the TOC from all the transuranic RH-TRU tanks is approximately 3691 kg. It is anticipated that most of the TOC in the tanks is not associated with complexing agents, but that has not been verified at this time. As a conservatism, SNL/NM can assume that any complexing agents listed in Table 1 could form the bulk of the TOC in the ORNL RH-TRU tanks.

LANL

Los Alamos National Laboratory has provided estimates of four complexing agents that are anticipated to occur in their TRU solidified waste streams and as materials used in decontamination and spill clean-up operations (that would occur with the debris wastes). The quantities of these compounds are listed in Table 3.

RFETS/INEL

The information provided by RFETS will also be used to estimate the amount of complexing agents in the RFETS retrievable waste (post 1970) at Idaho National Engineering Laboratory (INEL). Attached is a listing of chemicals from RFETS that was provided to the BIR team as a basis for potential complexing agents in TRU waste scheduled for shipment to and disposal in WIPP. This same list was originally put together as part of the documentation requested by the State of Nevada to document that less than 1% "complexing" agents occur in RFETS solidified low-level "saltcrete" waste that would be shipped to NTS for disposal.

The list was provided as a yearly estimate of complexing agents used on site at RFETS. It is conservative to assume that all of these complexing agents would reside in the TRU waste. Based on the authors understanding at this time, the inventory of RFETS complexing agents is across the entire site, so this should include material expected to occur in the debris wastes (this will be verified for the final version of this memo). The mass of complexing agents reported in Table 3 for RFETS results from multiplying the yearly estimates (in kilograms) by 20 years of production at RFETS (1970-1989), which includes RFETS waste in storage at INEL.

1

| | - |
|---|--------------|
| Chemical | Approximate |
| | Annual Usage |
| | |
| Acetic acid | m² |
| Acetone | 100 L |
| Adogen-364-HP (~triluaryiamine) | 100 L |
| Carbon tetrachloride | m |
| Deodorized mineral spirits (Amsco) | 1000 L |
| 2,5-di-tert-butylhydroquinone (DBHQ) | m |
| Diethylbenzene (DEB) | 800 L |
| Diethylenetriaminepentaacetic acid (DPTA) | m |
| Di (2-ethylhexyl) phosphoric acid (HDEHP) | 200 L |
| Di-isopropylbenzene (DIPB) | 100 L |
| Ethanol | 100 L |
| Ether | m |
| Ethylenediaminetetraacetic acid (EDTA) | m |
| 2-ethyl-i-hexanol | m |
| α-hydroxyisobutyric acid | m |
| Isopropanol | m |
| Methanol | m |
| n-dodecane | m — |
| n-paraffin (NPH) | m |
| Oxalic acid | m |
| Thenoyltrifluoroacetone (TTĂ) | m |
| Tributylphosphate (TBP) | m |
| Trichloroethylene (TCE) | m |
| Xylene | m |
| | |

Table 1.Organic chemicals used regularly in the TPP (7920) and TURF (7930) and
subsequently discharged to the ORNL LLLW system

*m = minimal usage: ≤ 10 kg/year or $\leq L/year$. Bates, 1988



| TRU Tanks | Tank No. | Volume (m3) | Mass (kg) | TOC (mg/kg) | TOC (kg) |
|----------------------------|--------------|---------------|-----------|-------------|----------|
| | | ļ | | | |
| INACTIVE TANKS | | | | | |
| North Tank Room | W_03 | 5.2 | \$670 | 5300 | 30.05 |
| | W-04 | 18.2 | 24527 | 200 | 4.91 |
| | | | | | |
| South Tank Farm | W-07 | 37.5 | 45715 | 1300 | 59.43 |
| | W-08 | 11.4 | 14080 | 8400 | 118.27 |
| | W-09 | 0.8 | 833 | 2900 | 2.42 |
| | W-10 | 28 | 31650 | 4900 | 155.09 |
| Old Hydrofracture Facility | T-01 | 3 | 4845 | 18600 | 90.12 |
| | T-02 | 4.6 | 7328 | 28000 | 205.18 |
| | T-03 | 7.7 | 14,829 | 9140 | 135.54 |
| | T-04 | 5 | (6442 | 4620 | 28.84 |
| | T-09 | 1.9 | 2967 | 7620 | 22.61 |
| | | | | | |
| | | | | | |
| ACTIVE TANKS | | | III. | | |
| | | | | > | |
| Evaporator Facility | C-2 | 100 | 63253 | 3281 | 209.50 |
| | W-21 | | \$6524 | 6480 | 249.64 |
| | W-22 _ | | 60939 | 22.1 | 1.35 |
| | W-23 | A 2012 | 89818 | 4120 | 370.05 |
| | | | ľ | | |
| MVST | W PAR | 52 | 72861 | 2940 | 214.21 |
| | | 90.7 | 126911 | 2330 | 295.70 |
| | THE CO | 59.2 | 82930 | 6220 | 515.82 |
| | AND N | 69.1 | 96707 | 3135 | 303.18 |
| | W-24 | 16.5 | 23051 | 2500 | 57.63 |
| | W-29 | 46.4 | 64913 | 3531 | 229.21 |
| | W-30 | 46 | 64383 | 3531 | 227.34 |
| | W- 31 | 26.3 | 36828 | 4470 | 164.62 |
| | <u> </u> | ļ | | Total TOC | |
| | | | <u> </u> | 10th IUC | |

Table 2. ORNL Total Organic Carbon Estimates

•

•

Table 3. RF/INEL and LANL Complexing Chemicals Estimate

| Compound | RF Mass (kg) | LANL Mass (kg) | Total Mass (kg) |
|---|--------------|----------------|-----------------|
| Ascorbic Acid | 90 | 7 | 9. |
| Acetic Acid | 132 | 10 | 14: |
| Sodium Acetate | 1110 | | 1110 |
| Citric Acid | 90 | 1100.5 | 119 |
| Sedium Citrate | 400 | | 40 |
| Ozalic Acid | 90 | 13706 | 1379 |
| EDTA | 23 | | 2 |
| -Hydrozyquinoline | 46 | | 44 |
| Tributyi Phospanis | 74 | | 7 |
| 1,19 romantareune | 0.24 | | |
| Disksyi-s,s-distayicardasbeyi- | 72 | | - T. |
| meen a la serie de la serie | | | |
| | | | ~ |



APPENDIX B - 4

.

.

United States Government

Department of Energy

~ memorandum

Carlsbad Area Office Carlsbad, New Mexico 88221

March 29, 1996

REPLY TO ATTN OF:

DATE:

NTP:DW:96-1111

SUBJECT:

Current Estimate of Complexing Agents in Transuranic Solidified Waste Forms Scheduled for Disposal in WIPP

TO:

Les E. Shephard, Director, SNL/NM

Attached is a copy of the report containing the preliminary estimates of complexing agents in transuranic (TRU) solidified waste forms scheduled for disposal in the Waste Isolation Pilot Plant (WIPP). This information was requested from the TRU Waste Baseline Inventory Report (TWBIR) team in support of the Performance Assessment (PA) being conducted by Sandia National Laboratory (SNL) and is based on input from the following TRU waste sites: Rocky Flats Environmental Technology Site (RFETS), Los Alamos National Laboratory (LANL), Oak Ridge National Laboratory (ORNL), Savannah River Site (SRS), Hanford Operations (Hanford), and Lawrence Livermore National Laboratory (LLNL).

The complexing agent inventories provided in this letter are in response to a Sandia National Laboratory (SNL) request for information from the U. S. Department of Energy (DOE) Carlsbad Area Office (CAO). A copy of the original request for this complexing agent information is contained in Appendix B of Revision 2 of the TWBIR (DOE/CAO-95-1121. December 1995). The documents attached represent the final information requested for this input to the Performance Assessment (PA) and satisfy the commitment on this subject contained in the March 15, 1996, memorandum (CAO:NTP:RLB 96-0687) to respond to SNL before the end of March. It should be specifically noted that all waste inventory volumes quoted are derived from Rev. 2 of the TWBIR.

Tables 1 and 2 provide a summary of Total Organic Carbon (TOC) in the remote-handled (RH)-TRU sludges from ORNL and a list of possible complexing agents that may contribute to the TOC in the sludges. Table 3 provides a summary of specific complexing agents that may be present in the TRU waste for SNL use.

Table 4 summarizes the volume of stored and projected TRU waste that contributes to the estimate of complexing agents in the waste. For contact handled (CH)-TRU waste, greater than 94% of TRU stored and projected final waste forms, greater than 98% of the Solidified Organic final waste forms, and greater than 92% of the Solidified Inorganic final waste forms contribute to the complexing agent estimate. For RH-TRU waste, greater than 86% of TRU stored and projected final waste forms, 100% of the Solidified Organic final waste forms, and greater than 86% of TRU stored and projected final waste forms, 100% of the Solidified Organic final waste forms, and 100% of the Solidified Inorganic final waste forms contribute to the complexing agent estimate.





Les E. Shephard

The attached site summary, tables, and background references contain greater detail about the basis for these estimates.

If you have any questions concerning the enclosed information, please contact Mr. Russ Bisping of my staff at (505) 234-7446.

Watkins

Don Watkins Manager National TRU Program

Attachment

cc w/attachment: R. Bisping, CAO G. Basabilvazo, CAO P. Drez, CTAC L. Sanchez, SNL M. Chu, SNL M. Marietta, SNL J. Harvill, CTAC



SITE SUMMARY

BACKGROUND

Information has been received from all sites that were requested to provide data on potential complexing agents in their solidified waste forms: Rocky Flats Environmental Technology Site (RFETS), Los Alamos National Laboratory (LANL), and Oak Ridge National Laboratory (ORNL). Several transuranic (TRU) waste sites which either generate no solidified waste forms or small quantities have also responded. A copy of the Carlsbad Area Office (CAO) memorandum requesting the complexing agent information from the sites is included (Attachment 1).

The term "complexing agent" is being used in lieu of "chelating agents" in this memo, since chelating agents usually have a certain structure (chelating comes from the Greek work "chele" for claw, as in a crab) and are considered a subset of complexing agents. That is, the acetate ion will "complex" with some metals and increase their solubility but does not have the structure that would label it as a chelating agent. A "commonly" known chelating agent is EDTA (ethylenediaminetetraacetic acid), which contains functional (acetate) anion groups arranged in parallel which resemble a "claw"-like structure for complexing the cations. EDTA has two claw structures at either end of the molecule.

The original scope of this task was to ask the TRU waste sites about "aqueous-soluble" complexing agents in their solidified waste forms. As this task was researched, the authors realized that the term "aqueous-soluble" is only a relative term, since everything is aqueous-soluble at some concentration level. Therefore, every potential chemical compound that has been reported from the TRU waste sites is included and the task of selecting aqueous-soluble compounds is left to the Sandia National Laboratory (SNL) personnel in charge of Performance Assessment (PA) calculations.

TRU WASTE SITE RESPONSES

Oak Ridge National Laboratory (ORNL)

ORNL has provided a list of organic compounds that contain some aqueous-soluble compounds that are apparent complexing agents. A copy of the list of all compounds reported by ORNL to the TRU Waste Baseline Inventory Report (TWBIR) team is attached for completeness (Table 1). The list in Table 1 is from an ORNL report on low-level waste (Kaiser, 1988), but the same compounds are anticipated to occur in the TRU waste based on process history (but not necessarily at the same concentrations). ORNL cannot quantify these compounds in their remote-handled (RH)-TRU solidified wastes, but have provided an estimate of Total Organic Carbon (TOC) for each RH-TRU waste tank (Table 2). The sum of the TOC from all the RH-TRU tanks is approximately 3691 kg. It is anticipated that most of the TOC in the tanks is not

Commanne Agents - Estimated Quantities March 27, 1996

Page 1 of 3

B4-3

associated with complexing agents, but that has not been verified at this time. As a conservatism, SNL can assume that any complexing agents listed in Table 1 could form the bulk of the TOC in the ORNL RH-TRU tanks.

Los Alamos National Laboratory (LANL)

LANL has provided estimates of four complexing agents that are anticipated to occur in their TRU solidified waste streams and as materials used in decontamination and spill clean-up operations (that would occur with the debris wastes) (Attachment 2). The quantities of these compounds are summarized in Table 3.

Rocky Flats Environmental Technology Site (RFETS/INEL)

The information provided by RFETS has been used to estimate the amount of complexing agents in the RFETS retrievable waste (post 1970) at Idaho National Engineering Laboratory (INEL). Attached is a listing of chemicals from RFETS that was provided to the TWBIR team as a basis for potential complexing agents in TRU waste scheduled for shipment to and disposal in WIPP (Table 3). This same list was originally put together as part of the documentation requested by the State of Nevada to document that less than 1% "complexing" agents occur in RFETS solidified low-level "saltcrete" waste that would be shipped to the Nevada Test Site (NTS) for disposal (Attachment 3).

The list was provided as a yearly estimate of complexing agents used on site at RFETS. It is conservative to assume that all of these complexing agents would reside in the TRU waste. The inventory of complexing agents is the best estimate for all TRU waste generated across the entire RFETS site, which includes debris wastes. The mass of complexing agents reported in Table 3 for RFETS are arrived at by multiplying the yearly estimates (in kilograms) by 20 years of production at RFETS (1970-1989), which includes RFETS waste in storage at INEL. The yearly estimates can be found in Attachment 3.

Savannah River Site (SRS)

The SRS has provided information (see letter included as Attachment 4) on three complexing agents used on site in connection with their operations: tributyl phosphate (TBP), tri-octyl phosphine oxide (TOPO), and tri-iso octylamine (TiOA). As discussed in the SRS letter, none of these compounds are expected to be found in SRS TRU waste.

Hanford Operations

Hanford Operations has provided a listing from their database of potential chemicals in their TRU waste. The only chemical that appears on the list that might act as a chelating agent in aqueous solutions and has a reportable quantity associated with the waste is tributyl phosphate (TBP). TBP is reported under three different spellings with a total of 92.5 kg. This value is

Page 2 of 3

B4-4

summarized in Table 3. The entire list of chemicals and the associated quantities (in kg) reported by Hanford are included in Attachment 5.

Lawrence Livermore National Laboratory (LLNL)

LLNL submitted the letter included as Attachment 6 which documents that no chelating agents occur in the LLNL TRU waste streams.

ESTIMATED VOLUME OF TRU WASTE INCLUDED IN COMPLEXING AGENT MEMO

Column 2 of Table 4 contains a list of the total TRU waste destined for disposal in WIPP (stored plus projected to 2022). Column 3 estimates the volume of waste from each major site that has contributed to the estimate of complexing agents in TRU waste. Columns 4 and 5 provide the same data for Solidified Organics and Solidified Inorganics final waste forms. The two rows labeled "PERCENTAGE" provide an estimate of the percentage of waste for which the TRU waste sites have provided data used in estimating the complexing agents in the waste. It should be specifically noted that all waste inventory volumes quoted are derived from Rev. 2 of the TWBIR (DOE, 1995).

REFERENCES

Kaiser, L. L., 1988, "ORNL Inactive Waste Tanks Sampling and Analysis Plan," ORNL/RAP/LTR-88/24, April 29, 1988, Oak Ridge National Laboratory, Oak Ridge, Tennessee.

U. S. Department of Energy, 1995, "Transuranic Waste Baseline Inventory Report (Revision 2)," DOE/CAO-95-1121, December 1995. Carlsbad, New Mexico.



| Chemical | Approximate |
|---|--------------|
| | Annual Usage |
| | |
| Acetic acid | m* |
| Acetone | 100 L |
| Adogen-364-HP (~triluarylamine) | 100 L |
| Carbon tetrachloride | m |
| Deodorized mineral spirits (Amsco) | 1000 L |
| 2.5-di-tert-butylhydroquinone (DBHQ) | m |
| Diethylbenzene (DEB) | 800 L |
| Diethylenetriaminepentaacetic acid (DPTA) | m |
| Di (2-ethylhexyl) phosphoric acid (HDEHP) | 200 L |
| Di-isopropylbenzene (DIPB) | 100 L |
| Ethanol | 100 L |
| Ether | m |
| Ethylenediaminetetraacetic acid (EDTA) | m |
| 2-ethyl-l-hexanol | m |
| α-hydroxyisobutyric acid | m |
| Isopropanol | m |
| Methanoi | m |
| n-dodecane | m |
| n-paraffin (NPH) | m |
| Oxalic acid | m |
| Thenoyitrifluoroacetone (TTA) | m |
| Tributylphosphate (TBP) | m |
| Trichloroethylene (TCE) | m |
| Xylene | m |

Table 1. Organic Chemicals Used Regularly in the TPP (7920) and TURF (7930) and Subsequently Discharged to the ORNL LLLW System

 $m = minimal usage: \le 10 \text{ kg/year or } \le 10 \text{ kg/year}$. Bates, 1988



| TRUTANKS | TANK NO.I | VOLUME (m ³) | MASS (kg) | TOC (mg/kg) | TOC (kg) |
|----------------------------|-----------|--------------------------|-----------|-------------|----------|
| | | | | | |
| INACTIVE TANKS | | | | | |
| | W-03 | 5 3 | 5670 | 5300 | 30.05 |
| North Lank Farm | W-04 | 18.2 | 24527 | 200 | 4.91 |
| | W_07 | 37 5 | 45715 | 1300 | 59.43 |
| South Tank Farm | W-08 | 11 4 | 14080 | 8400 | 118.27 |
| | w.00 | | 833 | 2900 | 2.42 |
| | W-10 | 28 | 31650 | 4900 | 155.09 |
| Old Hudenfragture Facility | τ-01 | | 4845 | 18600 | 90.12 |
| Old Hydrotracture racinty | T-07 | 4.6 | 7328 | 28000 | 205.18 |
| | T-03 | 7.1 | 14829 | 9140 | 135.54 |
| | T-04 | | 6242 | 4620 | 28.84 |
| | T-0 | 1.9 | 2967 | 7620 | 22.61 |
| ACTIVE TANKS | | | | | |
| | | 15 | 6 63853 | 3781 | 209.50 |
| Evaporator facility | U- | 41 43. 11 77 | 5 3852 | 6480 | 249.64 |
| | W-2 | $\frac{1}{27}$ | 5 60030 | | 1.35 |
| | W-2 | 2 43. 3 64 | 2 8081 | RI 4120 | 370.05 |
| | | | | | |
| MUST | W-2 | 4 | 7286 | 1 2940 | 214.21 |
| | W-2 | 5 90 | .7 12691 | 1 2330 | 295.70 |
| | W-2 | 59 | .2 8293 | 0 6220 | 515.82 |
| | W-2 | 69 | .1 9670 | 7 3135 | 303.18 |
| | W-: | 16 | .5 2305 | 1 2500 | 57.63 |
| | w- | 29 46 | .4 6491 | 3 3531 | 229.2 |
| | W- | 30 | 46 6438 | 3 3531 | 227.3 |
| | w- | 31 26 | 3682 | 28 4470 | 164.6 |
| | | | | Total TOC | 3690.6 |

Table 2. ORNL Total Organic Carbon Estimates



| COMPOUND | RF MASS (kg) ⁽¹⁾ | LANL MASS (kg) | HANFORD MASS (kg) ⁽⁵⁾ | TOTAL MASS (kg) |
|---|-----------------------------|----------------|----------------------------------|-----------------|
| | | | | |
| Ascorbic Acid | 90 | 7 | , | 97 |
| Acetic Acid | 132 | 10 | | 142 |
| Sødium Acetate | 1110 | | | 1110 |
| Citric Acid | 90 | 1100.5 | 1 | 1190.5 |
| odium Citrate | 400 | | | 400 |
| Dxalic Acid | 90 | 13706 | 1 | 13796 |
| EDTA | 23 | | | 23 |
| Hydroxyquinoline | 46 | Í | | 46 |
| ributly Phosphate | 74 | | 92.5 | 166.5 |
| , 10 Phenanthroline | 0.24 | | | 0.24 |
| ihexyl-n,n-diethylcarbamoyl- aethylphosphonate | 72 | | | 72 |

Table 3. RF/INEL and LANL Complexing Chemicals Estimate

⁽¹⁾ Letter from W.F. Weston to E.S. Goldberg, No. 89-RF-3055, dated September 1, 1989 (Attachment 3)
 ⁽²⁾ Memorandum from C.L. Foxx to P. Drez dated March 12, 1996 (Attachment 2)
 ⁽³⁾ Memorandum from F.M. Coony and M.R. Kerns to L.C. Sanchez through S. Lott, dated January 25, 1996 (Attachment 5)

memorandum

Carisbad Area Office Carisbad, New Mexico 88221

| DATE: | UN | 5 | 1996 |
|----------------------|-----------|-----|---------------------|
| AEPLY TO LITH OF: | CAO:? | VTP | <u>RT B 96-0605</u> |

sualeer: Additional Transurance (TRU) Waste Data Request for Sandia National Laboratories' Waste isolation Pilot Plan (WIPP) Performance Assessment

to: Distribution

We have been informed by representatives from Sandia National Laboratories (SNL) working on WIPP Performance Assessment (PA) that they require more information on certain TRU waste-related parameters in order to assess their influence on WIPP PA (see attached copy of relevant pages form SNL memo).

Data for most of these parameters have already been received from the sues either through responses to the Baseline inventory Report (BIR), Revision 2, questionnaire or by discussions with sue representatives. However, since the request from SNL for data on water soluble organic ligands (i.e., chelating agents) was not received in time for inclusion in the BIR Rev. 2 data call, WIPP PA still needs data for this parameter. As per the SNL memo, the data are needed by the end of February 1996, and therefore it is being addressed through this request separately from the upcoming BIR Rev. 3 data call.

As documented in the SNL memo, WIPP PA would like to have "best estimates" that are realistic and not overry conservative. Consequently, all sites that have existing data on chelating agents present in their waste are requested to submit the best available information to the BIR technical staff by February 26, 1996. The details on the name of the information being requested by WIPP PA are being provided in Table 3 of the autommut.

A representative from SNL WIPP PA will be available at the upcoming BIR, Revision 3, Data Call Meeting to be held in Concord, California, on January 10, 1996. We anticipate that a brief presentation will be made at this meeting by WIPP PA staff explaining the importance of the data followed by any questions from site representatives. If you have any questions/clarifications regarding this matter, please be ready to discuss these at the upcoming meeting in Concord with the SNL WIPP PA representative.

Thank you for your continued cooperation.

Russ Bisping

Waste Certification Manager

Attaciunent



B4-9

 Table 4.
 Calculation of Amount of Waste Covered

| | | Accounted For in | | |
|-----------------------|------------------|---------------------------|-------------------|-------------------|
| Major Sites | Totai TRU | Complexing Agent Estimate | Solidif. Org | Solidif. Inorg. |
| | <u>(m³)</u> | (m ³) | (m ³) | (m ³) |
| | | | | 1 |
| CH-TRU ⁽¹⁾ | | | | |
| RL ⁽⁷⁾ | 45515.43 | 45515.43 | 0 | 23.39 |
| INEL | 28606.74 | 25657.4 | 789.67 | 3349.6 |
| LLNL ⁽³⁾ | 941.13 | 941.13 | 0 | 20.18 |
| LANL (4) | 18405.15 | 18405.15 | 30.58 | 6922.02 |
| NTS (S) | 627.91 | 627.91 | 0 | 5.67 |
| ORNL (9) | 1560.42 | 0 | 0 | 0 |
| RFETS (7) | 5107.92 | 5107.92 | 140.93 | 1423.01 |
| SRS 🔊 | 9 648. 15 | 9648.15 | 0 | 1369.8 |
| | | | | |
| Total Major Sites | 110412.85 | 105903.09 | 961.18 | 13113.67 |
| Total CH-TRU | 111721.43 | 111721.43 | 980 | 14108.51 |
| PERCENTAGE (10) | | 94.79% | 98.08% | 92.95% |
| (1) | | | | |
| RH-TRU ⁽¹⁾ | | | - | |
| RL ⁽⁷⁾ | 21729.35 | 21729.35 | 0 | 0 |
| INEL ²⁾ | 220.72 | 196.98 | 3.56 | 65.27 |
| LANL (*) | 193.13 | 193.13 | 0. | 0 |
| ORNL (9) | 2915.64 | 1243.33 | 0 | 1243.33 |
| | | | 1 | |
| Total Major Sites | 25058.84 | 23362.79 | 3.56 | 1308.6 |
| Total RH-TRU | 26930.88 | 26930.88 | 3.56 | 1 1308.6 |
| PERCENTAGE (19) | | 86.75% | 100.00% | 100.00% |

(1) Table 4-3 to 4-23, Rev. 2 TWBIR

⁽²⁾ Non RFETS Waste Subtracted

⁽³⁾ Letter from K. Hainebach to J. Teak dated March 7, 1996 (Attachment 6)

⁽⁴⁾ Memorandum from C.L. Foxx to P. Drez dated March 12, 1996 (Attachment 2)

(5) NTS waste is derived from LLNL only, see (4)

- ⁽⁶⁾ ORNL was only asked to estimate complexing agents in solidified RH-TRU waste per DOE memorandum dated January 5, 1996 (Attachment 1)
- ⁽⁷⁾ Letter from W.F. Weston to E.S. Goldberg, Letter No. 89-RF-3055, dated September 1, 1989 (Attachment 3)

(1) Letter from J. D'Amelio to J. Teak. SWE-SWE-96-0106, dated February 28, 1996 (Attachment 4)

(9) Memorandum from F.M. Coony and M.R. Kerns to L.C. Sanchez through S. Lott, dated January 25, 1996 (Attachment 5)

⁽¹⁰⁾ Volume percentage of total TRU waste, Solidified Organics, and Solidified Inorganics accounted for in complexing agent memorandum.

B) Special Request Non-PA Items

Also wanted at this time is additional information for several waste material inaracteristics. Although these characteristics have not been identified as waste material parameters to be used for WIPP PA, they are needed for non-PA scoping calculations to assess their influence on PA. Since these items are not currently PA parameters, inventory estimates of these characteristics as "additional information" in the TWBIR or supplied outside of the TWBIR via written correspondence. Below you will find an itemized list of these special request items.

1) Non-redioactive Materials

Additional information is sensed on the five wasta material characteristics (see Table 2): 1) visities wasta, 2) nitrans (NO_3^-) , 3) suffices (SO_2^+) , 4) phosphorum, and 5) content. Of these waste parameters, the last four are nonzed for the gas generates modeling. The nutrans and the sulfates are involves in the desirrificance and sulfate reduction processes which breakup the cellulours, while the phospherus is a nutrient for biodecay of cellulosics. The estimate of the mass quantities of coment in the waste inventory should include both the comment that is commined in the waste as coment itself (due to D&D activities, etc...) and the centent found in various studges. Coment consumes CO_2 due to its consent of $Cd(OH)_2$. The estimates for this non-radioactive waste constituent asset only be "best estimates" at this present time so that non-PA acoping calculations can be made to determine their importance on overall repository performance. (Do not generate upper-bound estimates that are overly conservative.)

2) Residnes

"Best estimates" are nonner for residues, in addition to those already identified at the Rocky Flats Plant (RFP), that have the possibility of being changes from a resource category to a TRU waste category.

3) Organic Ligands (Cheiating Agams)

"Best estimates", from currently available information, are needed for major water-soluble organic ligands which are under consideration for the actinida source term (and Table 3). If it is not possible to obtain data from mater wants generating sites then supply guidance on how a first-order estimate may be made (from existing information such as process knowledge etc.) so that non-PA scoping calculations can be performed to identify if the presence of these ligends would have any significant impacts. (Do not generate commans that are overly conservative.) Requested data is for final form "process-level" quantities used in production only for the key sites. If information on the "process-level" values does not exist at the key sites, then "laboratory-actia" values should be used in the requested assessment of the inventory. Should it be determined that more detailed information on organic ligands will be needed, you will be given a specific written request at a future time. This effort should be performed in parallel with the TWBIR. Technical data should be supplied in memoraneum form by the end of February 1996 with supporting documentation by the end of Maren 1996.



B4-11

| Table 3. | Justification of Special Request For Info On Organic Complexing Agents. (a) |
|---|--|
| Ligans (b) | Discustion (c) |
| 1) Totai Complexatu | The most valuable informance at this time is a "best estimate" of the s total amount of water soluble complexing agents (liganes) in the s TRU waste matrix. |
| 2) Cirae | Pretiminary information indicates that citrate (citric acid) may be the s largest used ligand at TRU wasts generating sitts. Hence, inventory quastities are very important. |
| 3) Licinie | This is an important light that is preduced by bacteria as part of its is own methodism. What is requested here is a "best estimate" of the quantity of lactant that actually exists in the TRU wasts manna (not just an initial amount supplied as part of a wasts atream). However, i if this information cannot be developed, then supply information on the initial amount. |
| 4) Ozaiate | This is an important ligand that is produced by bacteria as part of its i own metabolism. What is requested here is a "best estimate" of the quantity of oxaiate that actually exists in the TRU waste matrix (not i just an initial amount supplied as part of a waste stream). However, i if this information cannot be developed, then supply information on a the initial amount. |
| 5) EDTA | This light (chylenediamineterracetic acid) is also of major impor- cance due to its common use as a cleaning solvent. |
| (a) Informance on the for assessment of for the actinide a (b) These items are no (c) Also supply any a degradation of d in cases where no quantities. | ene additional waste materials are needed for non-PA scoping calculations f their importance. The presence of these complexing agents are important source term, with respect to increasing the solubility of radionuclides. anked in the order of their importance in the actinide source term, available information that TRU waste generation sites may have on the actay rates of ligands in current (and expected) waste matrizes if possible, no information is available, supply guidance on estimating first-order |

LCS:5741:1cz/(95-2082)

Copy to: P.E. Drez [Drez Environmental Associates] D. Bretzke (Science Applications International Corporation) S. Chakraborti (Science Applications International Corporation) MS-1320, C.F. Novak (Dept. 6119) MS-1323, H. Jow (Dept. 6741)



Distribution

Rich Nevarez, AL Tom Baillieul, BCL, CH Joseph Ginnani, NV Gary Riner, OR Regina Sarier, RF Rav Lang, CH Frank Schmaitz, OH Bruce LeBrun, LAAO Roy Kearns, OAK Ruay Guercia, RL Dale Ormend, SR Jerry Wells, ID JAN 5 1338

-

Los Alamos

memorandum

Waste Management and Environmental Complemence NM1+7 MS E501 Touris Paul Drez, Drez Environ Assoc. Thur James J Balkey, NMT-7, MS E501999

C. L. FORX, NMT-7, MS E501 CX From/MS: 7-2328/ 7-9201 Phone/FAX: NMT-7-WM/EC-96-035 Symbul March 12, 1996 Data:

SUBJECT: CHELATING AGENTS IN LANL WASTE

I am certain that I have not captured all chelating agents, but I believe that I have identified and quantified roughly the important materials. The chelators are found in three waste streams: 1) Cemented evaporator bottoms from TA-55

- Cemented sludge from the TA-50 Pretreatment Plant and dewatered sludge from the TA-50 Liquid Waste Treatment Plant
- 3) Combustible waste from TA-55

.

The three streams are summarized below.

It should be noted that waste generation data and analyses exist over the time frame of 1980 through 1995 or shorter intervals to support the estimated values. In some cases, quantitative data is almost nonexistent and the results are qualitative at best. Like Rocky Flats, plutonium processing at LANL attempts to avoid chelating agents which can interfere with recovery operations. From your list of compounds of interest, I am unaware of any significant usage of lactate or EDTA, so they have been eliminated from detailed consideration. I have added ascorbate which has been used as a reducing agent in HCl solutions, but not in nitric acid which attacks and decomposes ascorbate. One of the above streams is not an immobilized stream, but I believe that it is an important contributor of a soluble chelating agent in the form of citrate. If this information is extraneous to your purposes, just ignore it.

Cemented evaporator bottoms from TA-55. The evaporator bottoms are derived from nitric acid solutions some of which (27%) contain oxalate resulting from the precipitation of plutonium oxalate. Because of the pervasive usage of oxalate, it is contained at lower concentrations even in those solutions that do not arise from filtering an oxalate precipitate. Those numbers are based on analytical results. In addition the drums contain on the average, 3.2 liters of analytical solution residues. Those solutions contribute a negligible additional quantity of oxalate and small quantities of ascorbate, citrate and acetate. We have semi-quantitative values from the analytical organization for those chelators, based on the quantities used in the analytical processes that give rise to the residues. We know that 28 liters of solution went into a drum of cemented waste on the average from 1980 through June of 1988. Since that time, the average has been 43 liters of solution. In addition we have information regarding the number of drums generated from May, 1987 through April, 1995. The drum numbers and alternate cemented forms

MAR-10-1996 08:00



F.02

Paul Drez NMT-7-WM/EC-96-035

•

for the remaining years are esumated. The totals based on those data and estimates are shown here. oxalate 1600 kg 90.09/83.09 = 1632ascorbate 7 kg $176.19/175.19 = 7.09 \approx 7$ citrate 0.5 kg $172.19/191.19 = 0.5 \approx 0.5$ acetate 10 kg $60.05/57.05 \approx 10$

2

Cemented sludge from the TA-50 Pretreatment Plant and dewatered sludge from the TA-50 Liquid Waste Treatment Plant. Based on experience at the liquid waste treatment plant with upsets in the treatment process due to the presence of chelators in the waste stream, it has been assumed that TA-55 is the only significant source of chelating agents in the sludge generated at that facility. Three waste lines carry liquids from TA-55 to TA-50. The industrial waste line is thought to be reasonably free of chelating agents. The evaporator distillate in the process acid waste line is unlikely to contain significant quantities of chelators because the distillation process creates a sharp reduction in the content of nonvolatile solution species.

The process caustic waste line solution is dominated by oxalate filtrates in hydrochloric acid that have been subjected to caustic treatment and filtration. Under the conditions of that treatment the oxalate and ascorbate (used historically) are soluble and follow the solution to TA-50 for a ferrofloculation treatment. The solution is used to neutralize the nitric acid distillate. Because there is an excess of nitric acid, the neutralization is completed with the addition of stock sodium hydroxide. I have assumed that the short term excess of nitric acid decomposes the ascorbate leaving only the oxalate. I have estimated the oxalate concentration in the hydroxide filtrate at 0.075 moles/liter. If this mumber drives the calculation then we should sample the solution in the caustic holding tank at TA-50 and get a representative value.

Volumes of caustic solution generated by TA-55 were available for the years 1983 and 1986 through 1992. Volumes for all other years were estimated. I am assuming that the oxalate will appear in the sludges due to the low solubility of calcium oxalate and because the floculations have relatively high concentrations of calcium. In addition magnesium and aluminum oxalates are insoluble in a caustic environment. The oxalate precipitates will be found in the cemented sludge, whenever generated, and in the dewatered sludge from the early and middle 80's. These oxalates will also be found in the cement-filled corrugated metal pipe (CMP) waste stream generated at DP site when plutonium operations were located there. The total of oxalate in those waste streams is 11,800 kg. = /2070

Combustible waste from TA-SS. The combustible waste stream contains rags that were used in decontamination and spill clean-up operations. In spill clean-up the rags from the first pass are nearly always TRU waste as measured on our MEGAS assay instrument. The rags are dampened with a solution labeled "versene". Versene is a name for EDTA. In the very early days of the laboratory versene solution may have contained EDTA, but it had been changed to sodium citrate solution by the time I arrived in 1969. Drums of combustible waste do not usually contain only decontamination rags and often contain no

- -

Paul Drez NMT-7-WM/EC-96-035

such rags. However our waste management personnel apparently used a unique identifier over about a four year period (1987 to 1991) for the decontamination rags. Each item also had a net disposal weight associated with it. Thus I was able to get a handle on the weight of decon rags generated in that time frame. The rags were discarded not dripping but distinctly damp. I dampened some cheesecloth, weighing before and after, to estimate the weight of solution contained in the rags. Knowing the weight of solution and the concentration of the citrate, I was able to calculate a weight of citrate in the discarded rags. In May, 1991 the usage of citrate for decontamination was restricted to certain matrices. I was able to locate records for versene solution preparation from 1989 into early 1991 and then again for the past year so I could understand usage before and after 1991. From that I have estimated that the citrate containing years. With that information, I have estimated that the citrate contained in the combustible waste stream from 1971 to 2033 will be 1100 kg.

Cy: Andy Montoya, NMT-7, MS E501 NMT-7 File



ALLACHMENT 3

- . -

· · **- -** - .

CRRES, CONTROL UTGOING LTR. NO.

| · | == |
|------------------|--|
| | |
| | |
| CIST. | 13 |
| HINE OLL | |
| 13 C.P | |
| AOT AL | |
| TZ ER | |
| D. R.C. | |
| EA EM | |
| IR. J.E. | |
| • WA | |
| ettuae. | |
| | |
| DOER LM | · · · · |
| MCN. W.M. | Ī |
| TON, W.F. | X |
| 41AK. 8.D. | L i |
| ICLEA. | |
| | |
| | |
| | |
| | 1 |
| THER DH | |
| HNAL G.L | <u> </u> |
| 7ERA, D.W. | · · · |
| MAN, L.K. | |
| | <u> </u> |
| 1.2 | _ |
| -AN, R.B. | |
| JANN RL | · · · |
| <u>-G. O.N.</u> | |
| CENEURG GE. | |
| -4.EY. K.B. | <u> </u> |
| N.E.R. | $\underline{\times}\underline{\times}$ |
| <u>A.L</u> | |
| <u>. (R. H.L</u> | |
| SOUEZ, R.N. | |
| | |

| HES. CONTROL 1 11 |
|---------------------------------------|
| TRACT ADMIN. |
| The second |
| THE REAL PROPERTY. |
| |
| STOT OF AX |
| <u> </u> |
| · · · · · · · · · · · · · · · · · · · |
| |
| |
| |
| |
| <u>_</u> |
| |
| SIFEATION |
| ACCIENCE SALAR |
| |
| FOENTIAL |
| AET L |
| |
| |
| cielan |
| - / / 100 |
| |
| <u> </u> |
| |
| |
| |
| |
| |
| |
| · |
| LODGOVALS |
| |
| <u>v / v ~ / /</u> |
| |
| 1 |
| / |

Rocky Fists Plant Aarospace Operations Rockwell International Corporation P.O. Box 464 Golden, Colorado 80402-0464 (303) 966-7000 Contractor to U.S. Department of Energy

SEP 0 1 '999

Edward S. Goldberg Acting Area Manager, RFO

Attn: Mark Van Der Puy

APPLICATION TO SHIP SALTCRETE

Attached is a copy of the re-formatted Application to Ship Waste for saltcrete. This application addresses all the comments from the Nevada Operations Office document attached to your letter 1245-RF-89.

Please refer any questions regarding the attached application to E.L. D'Amico at (303) 966-5362 or P.M. Arnold at FTS 320-2056.

W.F. Weston, Director Plutonium Operations

Orig. and 3 cc - E.S. Goldberg Enc.



89-RF-3055

| Reference to | Table 8 (continued) Bocuments/Results Outling the General Waste Form Cr | ng Compliance Hiteria |
|------------------|--|--|
| | | Boxes." specifies Waste Operations personnel to visually inspect for and remove any excessive particulate from each stored saltcrete box. |
| Gases | Not Applicable | Saltcrete is not a gaseous waste and does not contain radioactive gases. |
| Stabilization | WO-5004 | As described in WO-5004. "Waste Treatment Spray Dryer and Saltcrete Process." cement is added to the salt waste stream to immobilize the particulate, solidify the liquids and moderate oxidizing characteristics. |
| Etiologic Agents | Not Applicable | Saltcrete does not contain pathogens, infectious wastes or other etiologic agents. |
| Chelating Agents | Quantity and type of complexing agents used per year at Rocky Flats: Ascorbic Acid: 4.5 kg Acetic Acid: 55.5 kg Citric Acid: 4.5 kg Sodium Citrate: 20.0 kg Oxalic Acid: 4.5 kg | Between 5/15/87 and 5/7/88, 917 triwall boxes of saltcrete were produced. The estimated saltcrete generation for any given year is between 1200 to 1600 triwalls. The average net weight of one triwall box of saltcrete is approximately 1600 pounds. Total weight of saltcrete produced between 5/15/87 and 5/7/88 is 917 boxes * 1600 pounds * 1 kg/2.2 pounds = 6.67*10 ⁵ kg. As a worst case, if it is assumed |
| | EUIA: 1.15 Kg | complexing agents are |

Page 20 August 1989

Table 2 (continued) Reference Documents/Results Outlining Compliance to the General Waste Form Criteria 3-Hvaroxyauinoline: disposed of with the 2.3 ka saltcrete, then, 106.36/6.67*10°=1.59*10** Tributyi Phosphate: is the weight fraction of 3.7 kg 1,10 Phenanthroline: the complexing agents with respect to the saltcrete. 0.012 ka Therefore, Rocky Flats' dihexyl-n.ndiethylcarbamoyl total yearly usage of complexing agents amounts methylphosphonate: to only 0.0159 weight 3.6 kg percent of the total saltcrete production Total: 106.36 kg between 5/15/87 and 5/7/88. This extremely conservative estimate is weil under the NTS limit of 1 weight percent. Not Applicable Saltcrete does not meet GCD Waste any of the guidelines to be identified as a GCD' waste. Not Applicable Saltcrete is not a bulk Bulk LLW LLW.

4. Additional Mixed Waste Form Criteria

Table 9 references the documents (procedures, specifications, etc.) or test/analysis results that specify compliance to the Additional Mixed Waste Form Criteria outlined in Section 2.2.2 of NVO-325.

Table 9 Reference Documents/Results Outlining Compliance to the Additional Mixed Waste Form Criteria

| | Compliance | |
|---------------|----------------------|--|
| Criterion | Documents or Results | Comments |
| Treated Waste | Not Applicaple | Saltcrete is a treated waste that meets the land |
| | | disposal restrictions and |

Page 21 August 1989

ATTACHMENT 4



P 1 Box 615 Auxen SC 29802

February 28, 1996

SWE-SWE-96-0106 F/WSWE/XXX/ARNR Response Required: N/A Key Words: TRU Waste Record Retention: Permanent

Jim Teak Advanced Sciences. Incorporated 6739 Academy Road. N. E. Albuquerque, New Mexico 87106-3345

Dear Mr. Teak:

FY96 TRANSURANIC WASTE BASELINE INVENTORY REPORT (TWBIR): RESPONSE TO THE TWBIR MEETING MINUTES REGARDING CHELATING AGENTS AND CONCRETE STABILIZATION (U)

The Savannah River Site (SRS) has reviewed its waste practices to determine whether chelating agents are present in retrievably stored TRU waste. SRS also has reviewed these practices to determine whether concrete has been used to solidify/stabilize TRU waste. These reviews revealed that SRS TRU waste steams do not currently contain chelating agents/complexants nor has SRS used concrete to solidify/stabilize TRU waste.

The Separations processes and the analytical/research laboratories at SRS have used chelating agents in the separation of plutonium from irradiated uranium and other materials. For example, tri-butyl phosphate (TBP) is the complexing agent used in SRS's PUREX process and many other laboratory processes. Also, agents such as tri-octyl phosphine oxide (TOPO) and tri-iso octylamine (TIOA) have been used or investigated through the years. However, none of these chelating agents/complexants has entered SRS TRU waste. The complexants are dissolved in organic solvents for use as liquid/liquid extractants in the separation process. These solvents are recycled until depleted and then discarded to SRS's solvent waste tanks in the Waste Disposal Facility. This means that SRS organic liquid streams have not entered the production lines (e.g., HB and FB-Lines) where most of SRS TRU waste is generated. Further, a small amount of liquid TBP containing TRU nuclides is generated by SRS laboratories. This laboratory waste is discarded to liquid waste streams, which are eventually disposed in SRS's High Levei Waste Tanks. So, none of these liquid streams that contain complexants have entered SRS solid TRU waste streams.

SRS has not used concrete to solidify/stabilize TRU waste. The processes that generate slurries, which require stabilization, do not contain TRU radionuclides (e.g., plating of depieted uranium). For other processes that generate slurries, the waste is disposed in SRS's High Level Waste Tanks. Even the Low Level Waste (LLW) sludge generated by SRS's Effluent Treatment Facility (ETF) is disposed in the High Level Waste Tanks and is eventually

B4-20



) Teak SWE-SWE-96-0106 Page 2

ted to SRS's Saltstone Facility or the Detense Waste Processing Facility (DWPF). Finally, SRS loes not expect to generate TRU waste containing chelating agents nor anticipate using concrete to solidify/stabilize IRU waste in the near-tuture.

Please direct your questions to L. Williams (803) 557-6759.

Sincerety.

Joseph A. D'Ametio TRU Engineering Manager

JAD:lw

cc: A. Gibbs. 724-21E W. T. Goldston. 705-3C F. H. Gunneis. 705-3C S. J. Mackmuil. 703-A S. J. Mentrup. 724-21E D. Ormond. 703-A L. Williams. 705-3C Records Management. 705-3C SWE Files. 705-3C



ATTACHMENT 5

To: L.C. Sanchez, SNL

January 25, 1996

Thru: Shella Lott, CTAC **X M.C.** From: F. M. Coony and M. R. Kems, Hanford Site

RE: Additional TRU Waste Data Request for Sandia National Laboratories' Waste Isolation Pilot. Plan Performance Assessment

References: 1) Memorandum, Russ Elsping, COE/CAO to Distribution, same subject, dated. January 5, 1998.

2) Trip Report, F. M. Coony to K. L. Hisdek, January 15, 1996

The Reference 1 memo requests additional data on waste soluble organic ligands (i.e. chalating agents) from the generating sites by February 25, 1996.

Henford's approach for responding to the additional data request is presented in the Reference 2 trip report. The first item of this approach is to provide SNL, through CTAC, a list of all hexardous constituents, and their quantities, that have been reported in solid TRU waste at Hantord since 1987, the date of the By-Product Rule.

The fist of hazardous constituents and their quantities, from Hanford's record container tracking system, are presented in Table 2. The chemical names have been truncated to 30 characters. Hanford can provide complete names if needed. In some cases, the constituent is listed more than once because the constituent is spelled differently in the container tracking system. A quantity of 0.00 kg means typically that the constituent has been identified solely because it is a listed hazardous waste under RCRA. In these cases, the quantity is either absent or minimat.

Please evaluate the list of constituents, and indicate, in the space provided for each constituent, if the constituent is a soluble organic sigend. The suggested nomenciature is the following:

- N/A (meaning not sciuble organic legend) _
- C (meaning citrate
- L (meaning lactate)
- OX (meaning excision)
- EDTA (meaning sinvienediaminetetrascedid acids

Please indicate any other relevant information by footnotes.

To meet the requested due date, please provide a response to me (by fax) no later than February 5, 1998. Please copy CTAC on the response.

If you have any questions, please contact Mike Coony at 509-378-9774 or Mark Kerns at 509-372-2383.

| | CZAM | | | | |
|--|---------|--|--|--|--|
| Table 7 Questiline of Vice | | | | | |
| HEADOLING CONTRACTOR CONTRACTOR CONStituents | | | | | |
| | | | | | |
| | 0.001 | | | | |
| ACETONE | 0.021 | | | | |
| ACETONE | 0.001 | | | | |
| AGU | 0.14 | | | | |
| ALUMINUM NITRATE | 0.101 | | | | |
| ALUMINUM NITRATE MONOHYDRATE | 3.601 | | | | |
| AMERCOAT 234 | 0.051 | | | | |
| AMMONIUM CHLORIDE | 0.01 | | | | |
| ARSENIC | 0.021 | | | | |
| ASBESTOS | 27.001 | | | | |
| BARIUM | 1.851 | | | | |
| BERYLLIUM | 0.17 | | | | |
| BIS(2-ETHYLHEXYL)PHTHALATE | 0.82 | | | | |
| BISPHENOL A REBIN | 0.54 | | | | |
| BUTYL ALCOHOL | 0.41 | | | | |
| BUTYL GLYCIDYL ETHER | 0.11 | | | | |
| CADMIUM | 99.17 | | | | |
| CADMIUM HYDROXIDE | 0.101 | | | | |
| CALCIUM | 0.83 | | | | |
| CHLOROFLUOROPHOSPHATE | | | | | |
| CALCIUM HTDROXIDE | 0.06) | | | | |
| | 57.68 | | | | |
| CARBONIEIRACHLORIDE | 95.901 | | | | |
| CHLOROFORM | 0.001 | | | | |
| CHROMIUM | 14.52 | | | | |
| | 0.001 | | | | |
| COPPER SULFATE | 0.381 | | | | |
| | 0.00 | | | | |
| CUPROUS CYANIDE | 0.211 | | | | |
| CYANIDE SOLUTIONS | 0.211 | | | | |
| CTCLDHEXANE | 0.001 | | | | |
| DI(Z-ETHYLHEXYL)PHTHALATE | 0.08 | | | | |
| DFOCTTL PHTHALATE | 0.401 | | | | |
| DIOCTYL PHTHALATE | 0.201 | | | | |
| ETVANOL | 8.47 | | | | |
| | 0.20 | | | | |
| FERRIC NITRATE | 4.38 | | | | |
| | 0.21 | | | | |
| | 0.101 | | | | |
| | 328.201 | | | | |
| | 0.071 | | | | |

 $\mathbf{x} \in \tilde{\mathbf{x}}$

1. D. :

۰.,

| | | White an up to an and |
|----------------------------------|----------|-----------------------|
| KEROSINE | 0.00 | مسيد به دون |
| LEAD | 8.915.59 | |
| LEAD ACID | 0.27 | |
| EAD CHROMATE | 26.97 | |
| LEAD CHROMATE CXIDE | 1.54 | |
| LEAD CHROMATE CHLORIN PARAFFIN I | 1.33 | |
| LEAD CHROMATES | 0.05 | |
| LEAD SHIFLDING | 5.597.50 | |
| LIGHT AROMATIC NAPHTHA | 0.30 | |
| MERCURY | 1.51 | |
| MERCURY METAL | 0.00 | |
| METHYL ETHYL KETONE | 0.00 | |
| METHYL ISOBUTYL KETONE | 0.00 | |
| METHYLENE CHLORIDE | 8.03 | |
| NICKEL HYDROXIDE | 0.10 | |
| NITRIC AGID | 1.21 | |
| OIL | 0.00 | |
| PC8 | 130.13 | |
| PHOSPHORIC ACID | 0.33 | |
| PHTHAUC ACID BENZYL BUTYL EST | 0.00 | |
| PHTHALIC ACID BISC-ETHYLHEXYL | 0.00 | |
| PHTHALIC ACID, BISC-ETHYLHEXY | 0.05 | ii |
| POTASSIUM CYANIDE | 0.21 | |
| POTASSIUM FLUORIDE | 0.00 | |
| POTASSIUM HYDROXIDE | 5.80 | |
| RESIDUAL TANK FARM CORE SAMPLE | 0.8 | |
| SELENIUM | 1.1 | 0 |
| SILVER | 0.0 | |
| 500IUM | 0.1 | 31 |
| SODIUM CYANIDE | 0.2 | 1 |
| SODIUM FLUORIDE | 1.0 | 81 |
| SODIUM HYDROXIDE | 24.3 | 71 |
| SODIUM NITRATE | 173.0 | 01 |
| SODIUM SULFATE | 3.9 | 21 |
| STRIPCOAT | 34.0 | 181 |
| SULFAMIC ACID | 0.0 | 4 |
| SULFURIC ACID | 1. | 131 |
| THENOYL TRIFLUOROACETONE | 0.0 | 101 |
| TRI BUTYL PHOSPHATE | 49.3 | 10 |
| TRIBUTYL PHOSPHATE | 43. | 131 |
| TRIBUTYLPHOSPHATE | 0.0 | 071 |

MASS (KG)



•

MASS

| MARSHE WINER KOOM ODENIO | | |
|--------------------------|--|--|
| 3.261 | | |
| 001 | | |
| 1.011 | | |
| , 0.001 | | |
| 0.21 | | |
| 4.231 | | |
| | | |

.



B4-25



Lawrence Livermore National Laboratory

WASTE CERTIFICATION PROGRAM WCP96-055

March 7, 1996

Jim Teak Advanced Sciences Incorporated 6739 Academy Road NE Albuquerque, NM 87109

Dear Jim,

This is in response to the CAO request concerning the presence of organic ligands (chelating agents) in TRU waste. I have consulted with Joe Magana, a chemist working in LLNL's Plutonium Facility. He tells me that there are no chelating agents in LLNL's TRU waste.

Sincerely yours,

Ken Hamlad

Kem Hainebach, Ph. D. Waste Certification Engineer Environmental Protection Department

KH:lh c: Robert Fischer

-





APPENDIX B - 5

 (k_i)

_
memorandum

Carlsbad Area Office Carlsbad, New Mexico 88221

DATE: June 26, 1996

ATTN OF: CAO:NTP:DW 96-1528

SUBJECT: Revision of Current Estimate of Complexing Agents in Transuranic Solidified Waste Forms Scheduled for Disposal in WIPP

TO:

Dr. Les E. Shephard, Director, Nuclear Waste Management Programs Center, SNL

The mass of potential complexing agents in transuranic (TRU) waste generated at the Rocky Flats Environmental Technology Site (RFETS) and currently stored at RFETS and Idaho National Engineering Laboratory (INEL) was previously estimated in our March 29, 1996 memorandum, CAO:NTP:DW 96-1111, (Subject: "Current Estimate of Complexing Agents in Transuranic Solidified Waste Forms Scheduled for Disposal in WIPP"). Per our May 3, 1996 discussion, this information has been revised based on assumed or anticipated activities to be performed on the waste prior to final waste form generation.

The assumed or anticipated activities upon which these revisions were made are based on the preliminary submittal by INEL for Revision 3 of the TRU Waste Baseline Inventory Report (TWBIR). From this submittal, a very high percentage of INEL waste will be thermally treated and most complexing agents should therefore be destroyed by the treatment. A methodology is presented for estimating the amount of complexing agents that will be destroyed by the proposed thermal treatment at INEL. Using Ethylene Diamine Tetraaccetic Acid (EDTA) as an example, the original estimate of 23 kg in RFETS waste (stored at INEL and RFETS) has been reduced to a recommended value of 5.9 kg with a high range estimate of 6.9 kg and a low range estimate of 2.9 kg. All other complexing agents reported from RFETS (including that in storage at INEL) in the previous letter should also be reduced by the same methodology.

The original inventory estimates provided in the above referenced letter were based on the following information contained in the original transmittal:

- Estimates provided by the TRU waste sites on the amount of anticipated complexing agents in TRU waste which are summarized in Tables 1, 2, and 3 from TRU waste site memoranda in Attachments 1 through 6.
- Volumes from Revision 2 of the Transuranic Waste Baseline Inventory Report (TWBIR) used in Table 4.

In Revision 2 of the TWBIR, the volumes used for waste stored at the INEL were assumed to be unprocessed through any type of treatment (i.e., thermai) that would destroy potential



complexing agents. There was a small percentage of RFETS waste (~33%) stored at INEL scheduled for processing by thermal treatment in the TWBIR, Revision 2. Because these percentages of waste scheduled for thermal treatment were low, no credit was assumed in the original letter for the destruction of potential complexing agents occurring in RFETS TRU waste stored at INEL. This assumption also provided a conservative estimate of the potential complexing agents in TRU waste.

However, the INEL preliminary submittal received for Revision 3 of the TWBIR contains a much higher percentage of waste that will be processed thermally prior to shipment to WIPP for disposal. This much higher percentage of RFETS TRU waste that will be thermally processed will make a significant impact on the calculated amounts of potential complexing agents in TRU waste.

As stated in the original letter, most of the complexing agents were expected in the solidified waste forms, particularly in the solidified inorganic waste forms, since Sandia National Laboratory/New Mexico (SNL/NM) was only requesting information on "aqueous-soluble" complexing agents.

The RFETS estimate (Attachment 3 of the original letter) included all known sources (as of the time frame of the RFETS memo) of complexing agents regardless of what waste forms the chemicals occurred in the waste. Discussions with RFETS indicate the most likely occurrences of complexing agents in the waste would be:

Solidified Lab Waste> Solidified Inorganic Sludges> Debris Wastes

Based on the above relative occurrence for complexing agents, three estimates of the effects of extensive planned thermal treatment of RFETS waste at INEL can be made to modify the mass of chelating agents estimated in the original letter.

Tables AD-1, AD-2, and AD-3 summarize the calculations of the amount of decrease of complexing agents for RFETS in storage at INEL using EDTA as an example:

ASSUMPTIONS

- As stated in the original letter, RFETS was in production for 20 years (1971-1990) during which retrievably stored (post 1970) production waste would have been generated. Buried waste is not part of the WIPP inventory in the TWBIR.
- RFETS stopped shipments of waste to INEL initially in October 1988, then shipped additional quantities of waste from March to August 1989.

Dr. Les E. Shephard - 3 - June 26, 1996

- Assuming that RFETS essentially caught up on their backlog of waste during the second shipping period and a modest lag of 2 months from date of closure to actual shipping, effectively provides the beginning of July 1989 as the date for TRU waste accumulation at RFETS.
- Therefore, it is assumed that 18 months (1.5 years) of production waste still exists at RFETS in storage and 18.5 years of post 1970 production waste is in storage at INEL.

CALCULATIONS

As shown in Table AD-1 (for Solidified Lab waste - Content Codes 004 and 113), using EDTA as an example:

- 347.7 m³ of CH-TRU waste is in storage at INEL.
- 280.1 m³ will be vitrified, and
- 67.5 m³ will be set aside for direct shipment to WIPP (including 0.33 m³ for macroencapsulation)
- Therefore, 80.58% will be vitrified
- RFETS provided an EDTA generation rate of 1.15 kg/year (Attachment 3 of Original Complexing Agent Memo)
- 1.15 kg/year x 18.5 years = 21.3 kg EDTA at INEL in storage
- 1.15 kg/year x 18.5 years generation in storage at INEL x 80.58% vitrification of waste = 17.1 kg of EDTA destroyed by vitrification
- Therefore, 4.1 kg of EDTA (21.3 minus 17.1 kg) will be left in the untreated waste at INEL scheduled for shipment and disposal in WIPP
- The total EDTA in RFETS waste (both in storage at INEL and RFETS) = 4.1 kg (untreated waste at INEL) + 1.15 kg/year x 1.5 years (in storage at RFETS) = 5.9 kg

Since Content Codes 004 and 113 are the waste forms most likely to have the complexing agents, 5.9 kg of EDTA is the <u>RECOMMENDED VALUE</u> for performance assessment.



Dr. Les E. Shephard

- 4 -

June 26, 1996

Using similar methodology in Tables AD-2 and AD-3, estimates of EDTA (after treatment at INEL) are 6.9 kg (assuming the distribution of treatment for all inorganic solidified waste forms - 75.68% treated) and 2.9 kg (assuming the distribution of treatment for all RFETS waste in storage at INEL - 94.44% treated).

The value of 5.9 kg of EDTA is the recommended value, since Content Codes 004 and 113 are the waste forms expected to contain the majority of the complexing agents. The other two values, 6.9 kg for inorganic solidified waste and 2.9 kg for all treated RFETS waste, should be considered lower and upper bounds on this analysis. In particular, the 2.9 kg is a nonconservative estimate because INEL is planning to vitrify almost all their debris waste, particularly the organic debris waste, which may contain some EDTA from wipeup of spills, but is expected to be the least contributor to the overall complexing agents in the waste.

All other complexing agents from RFETS should be reduced by the same percentages for those values reported in Table 3 of the original complexing agent letter.

If you have any questions concerning the attached information, please contact Mr. Russ Bisping of my staff at (505) 234-7446.

Don Watkins Manager National TRU Program

Attachment

cc w/attachment: R. Bisping, CAO S. Chakraborti, CTAC J. Harvill, CTAC P. Drez, DEA R. Anderson, SNL L. Sanchez, SNL M. Chu, SNL M. Marietta, SNL





DETAILS OF EDTA CALCULATIONS (BASIS: ROCKY FLATS WASTE AT INEL WITH IDCs 004 AND 113)

| | | | | | | UNPROCESSED WASTE VOLUMES (m ³) | | | | | |
|---------|-------------|-----|-----------|--------|--------|---|----------------|-----------|-------|-------|--|
| FFCA_ID | WS_ID | СС | Total Vol | CH Vol | RH Vol | CH_Direct Ship | RH_Direct Ship | Vitrified | Amalg | Macro | |
| IN-W157 | ID-RFO-004T | 4 | 226.8 | 226.8 | 0.0 | 54.3 | 0.0 | 172.3 | 0.0 | 0.2 | |
| IN-W195 | ID-RFO-113 | 113 | 2.5 | 2.5 | 0.0 | 0.0 | 0.0 | 2.5 | 0.0 | 0.0 | |
| IN-W221 | ID-RFO-113T | 113 | 14.4 | 14.4 | 0.0 | 12.9 | 0.0 | 1.5 | 0.0 | 0.0 | |
| IN-W229 | ID-RFO-004 | 4 | 103.9 | 103.9 | 0.0 | 0.0 | 0.0 | 103.8 | 0.0 | 0.1 | |
| | | | 347.7 | 347.7 | 0.0 | 67.2 | 0.0 | 280.1 | 0.0 | 0.3 | |

| TOTAL EDTA IN RF WASTE AT INEL> | 21.3 kg |
|--|---------|
| PERCENT VITRIFIED> | 80.6% |
| AMOUNT VITRIFIED (80.58% of 21.3 kg)> | 17.1 kg |
| AMOUNT IN UNTREATED INEL WASTE> | 4.1 kg |
| TOTAL EDTA IN RF WASTE AT RF> (1.15 kg/yr for 1.5 years) | 1.7 kg |
| NEW EDTA ESTIMATE> | 5.9 ka |



DETAILS OF REVISED EDTA CALCULATIONS (BASIS: ALL ROCKY FLATS SLUDGES AT INEL)

| | | | | | | UNPROCE | SSED WASTE V | OLUMES (m | ·) | |
|---------|--------------|-----|-----------|--------|--------|-----------------|----------------|-----------|-------|-------|
| FFCA_ID | WS_tD | СС | Total Vol | CH Vol | RH Vol | CII_Direct Ship | RH_Direct Ship | Vitrified | Amalg | Macro |
| IN-W216 | ID-RFO-001T | 1 | 2531.8 | 2531.8 | 0.0 | 775.3 | 0.0 | 1741.6 | 0.0 | 14.9 |
| IN-W190 | ID-RFO-001 | 11 | 58.9 | 58.9 | 0.0 | 0.0 | 0.0 | 58.6 | 00 | 0.3 |
| IN-W221 | ID-RFO-113T | 113 | 14.4 | 14.4 | 0.0 | 12.9 | 0.0 | 1.5 | 00 | 00 |
| IN-W195 | ID-RFO-113 | 113 | 2.5 | 2.5 | 0.0 | 0.0 | 0.0 | 2.5 | 00 | 00 |
| IN-W228 | ID-RFO-002T | 2 | 1296.8 | 1296.8 | 0.0 | 15.3 | 0.0 | 1260.9 | 12.4 | 82 |
| IN-W191 | ID-RFO-002 | 2 | 342.4 | 342.4 | 0.0 | 0.0 | 0.0 | 336.9 | 3.3 | 2.2 |
| IN-W157 | ID-RFO-004T | 4 | 226.8 | 226.8 | 0.0 | 54.3 | 00 | 172.3 | 00 | 02 |
| IN-W229 | 1D-RFO-004 | 4 | 103.9 | 103.9 | 0.0 | 0.0 | 0.0 | 103.8 | 0.0 | 01 |
| IN-W218 | ID-RFO-007T | 7 | 461.5 | 461.5 | 0.0 | 461.5 | 0.0 | 0.0 | 00 | 00 |
| IN-W192 | ID-RFO-007 | 7 | 464.3 | 464.3 | 0.0 | 0.0 | 0.0 | 464.3 | 0.0 | 0.0 |
| IN-X001 | ID-RFO-095N | 95 | 4.9 | 4.9 | 0.0 | 0.0 | 0.0 | 4.9 | 00 | 0.0 |
| IN-W375 | ID-RFO-995TN | 995 | 19.3 | 19.3 | 0.0 | 0.0 | 0.0 | 19.3 | 0.0 | 0.0 |
| IN-X002 | ID-RFO-995N | 995 | 68.8 | 68.8 | 0.0 | 0.0 | 0.0 | 68.8 | 0.0 | 0.0 |
| | TOTALS | | 5596.4 | 5596.4 | 0.0 | 1319.3 | 0.0 | 4235.5 | 15.7 | 25.9 |

| Ξ |
|----|
| Ċл |
| ் |
| |

| TOTAL EDTA IN RF WASTE AT INEL> | 21.3 kg |
|---------------------------------------|---------|
| (1.15 kg/yr for 18.5 ycars) | |
| PERCENT VITRIFIED> | 75.7% |
| AMOUNT VITRIFIED (75.68% of 21.3 kg)> | 16.1 kg |
| AMOUNT IN UNTREATED INEL WASTE> | 5.2 kg |
| TOTAL EDTA IN RF WASTE AT RF> | 1.7 kg |
| (1.15 kg/yr for 1.5 years) | |
| | |

NEW EDTA ESTIMATE -----> 6.

6.9 kg

)



DETAILS OF EDTA CALCULATIONS (BASIS. ALL ROCKY FLATS WASTE AT INEL)

| | | | | | | UNPROCE | SSED WASTE V | DLUMES (m | 3) | |
|---------------|--------------|---------------|---------------|---------------|-----------|-----------------|-----------------------|---------------|-------|----------|
| FFCA_ID | WS_ID | CC | Total Vol | CH Vol | RH Vol | CII_Direct Ship | RH_Direct Ship | Vitrified | Amalg | Macro |
| IN-W307 | ID-RFO-000 | <u>l</u> | 136.7 | 136.7 | 0.0 | 0.0 | 0.0 | 135.8 | 0.0 | 1.0 |
| IN-W108 | ID-RFO-000T | 0 | 4139.7 | 4139.7 | 0.0 | 0.0 | 0.0 | | 0 | 29.0 |
| IN-W216 | ID-RFO-001T | I_I_ | 2531.8 | 2531.8 | 0.0 | 775.3 | 0.0 | 1741.6 | 00 | 14.9 |
| IN-W190 | ID-RFO-001 | 1 | 58.9 | 58.9 | 0.0 | 0.0 | 0.0 | 58.6 | 0.0 | 0.3 |
| IN-W167 | ID-RFO-112T | 112 | 164.1 | 164.1 | 0.0 | 120.2 | 00 | | 0.0 | 0 |
| IN-W168 | ID-RFO-112 | 112 | 51 | 51 | 0.0 | 00 | 00 | 51 | 0.0 | 0 |
| IN-W221 | ID-RFO-113T | 113 | 14.4 | 14.4 | 0.0 | 12.9 | 00 | 1.5 | 0.0 | 0 |
| IN-W195 | ID-RFO-113 | 113 | 2.5 | 2.5 | 0.0 | 0.0 | Q.Q | 2.5 | 0.0 | 0_0 |
| IN-W166 | ID-REO-114T | 111 | 70.8 | 70.8 | 0.0 | 56.2 | 0 | 14.6 | 0.0 | 0.0 |
| IN-W165 | ID-RFO-114 | 114 | 4.0 | 1.0 | 0.0 | 0.0 | <u>0.0</u> | _ | 00 | <u> </u> |
| IN-W370 | ID-REO-115TN | 115 | 67.2 | 67.2 | 0.0 | 40.7 | 0.0 | 26.5 | 0.0 | 0.0 |
| IN-X006 | ID-REO-115N | 115 | | | 0.0 | 0.0 | 0.0 | 1 .1 | 0 | 0 |
| IN-W186 | ID-REQ-116T | 116 | 2696.6 | 2696.6 | 0.0 | 06 | | 2696.0 | 0.0 | 0 |
| N-W185 | ID-RFO-116 | 116 | 371.1 | | 0.0 | 0.0 | 0.0 | | 0.0 | 00 |
| IN-W300 | ID-RFO-117T | 117 | 1520.2 | 1520.2 | 0.0 | 1 <u>1.8</u> | 0.0 | 193.2 | 0 | 12.2 |
| IN-W299 | ID-REO-117 | 112 | 147.5 | 147.5 | 0.0 | 0.0 | 0.0 | 146.4 | 0.0 | 2 |
| IN-W240 | ID-REO-118T | 8 | 174.6 | 174.6 | 0.0 | 7.8 | 0.0 | 163.3 | 00 | |
| IN-W241 | ID-REO-118 | 118 | 6.4 | 6.4 | 0 | <u> </u> | 0.0 | 6.2 | 00 | 0.1 |
| IN-W206 | ID-RFO-119T | 112 | | | 0.0 | | 0 | | 0.0 | 00 |
| IN-W232 | ID-RFO-119 | _112_ | 69.2 | 69.2 | 0 | 00 | 00 | 69.2 | 0.0 | 0.0 |
| IN-W230 | ID-RFO-122T | 122 | 18.2 | 18.2 | 0.0 | 10.0 | 0 | 8.3 | 00 | 0 |
| IN-W231 | ID-BEO-122 | 122 | 12.3 | 12.3 | 0.0 | 00 | ····· | 2.3 | 0 | 0 |
| IN-W250 | ID-BEO-123T | 123 | 63.8 | 63.8 | 0.0 | | 0 | 20.2 | 00 | 6.5 |
| N-W251 | ID-RFO-123 | 123 | 2.3 | 2.3 | 0.0 | 00 | 00 | 2 | 00 | 0.2 |
| N-W312 | ID-BEO-124TN | 124 | 1.2 | 3.2 | 0.0 | 21 | <u> </u> | 0.8 | 0.0 | 0 |
| N-W228 | ID-RFO-002T | _2_ | 1296.8 | 1296.8 | 001 | 15.3 | | 1260.9 | 12.41 | 82 |
| N-W191 | ID-REO-002 | 2 | 342.4 | 342.4 | 0 | 00} | | 336.9 | | 22 |
| N-W282 | ID-REO-241 | _241_ | 24.2 | 24.2 | 0 | 0 | 00 | 24.1 | 0 | 0.0 |
| N-W281 | ID-REO-241T | 241 | | | 0.0 | 00 | <u> </u> | | 0.0 | 0.0 |
| N-W196 | LD-REO-290 | _290 | 02 | 02 | 0 | 0.0 | 00 | 02 | 0.0 | 0 |
| N-W222 | ID-RFO-292T | 292 | | 110.5 | 0 | <u> </u> | 00 | 68.3 | 001 | 0.0 |
| N-W215 | ID-BEO-292 | 292 | 19 | 19 | 0_0 | | 0.0 | | 001 | 0.0 |
| N-W309 | ID-REO-003T | _1_[| 569.4 | 562.4 | <u>00</u> | | <u> </u> | | 001 | 0 |
| N-W310 | ID-REO-001 | _1_1 | 1001.9 | | <u></u> | | | | 001 | 00 |
| N-W276 | D-BEO-300T | 300 | | | <u></u> | <u></u> | | | 001 | 0 |
| <u>N-W274</u> | D-RFO-300 | .300 | 18.4[| <u>+¥</u> t | <u></u> | <u> </u> | | <u> </u> | | |
| N-W275 1 | D-BEO-301T | <u>-101 I</u> | 6.4 | 6.4 | 0.0 | | 0.01 | 551 | | 0.0 |



DETAILS OF EDTA CALCULATIONS (BASIS: ALL ROCKY FLATS WASTE AT INEL)

| FFCA_ID | WS_ID | CC | Total Vol | CH Vol | RH Vol | CH_Direct Ship | RH_Direct Ship | Vitrified | Amalg | Macro |
|---------|--------------|-------|------------|----------------|----------|----------------|----------------|--------------|---|-------------|
| IN-W273 | ID-RFO-301 | 301 | 1.3 | 1.3 | 0.0 | 0.0 | 0.0 | 1.3 | 0.0 | 0.0 |
| IN-W184 | ID-RFO-302 | 302 | 55.4 | 55.4 | 0.0 | 0.0 | 00 | 49.8 | 00 | 5.5 |
| IN-W225 | ID-RFO-302T | 302 | 22.2 | 22.2 | 0.0 | 0.0 | 0.0 | 20.0 | 0.0 | 22 |
| IN-W369 | ID-RFO-303TN | 303 | 12.3 | 12.3 | 0.0 | | 0.0 | 3.2 | 0.0 | 0.0 |
| IN-W368 | ID-RFO-310TN | 310 | 34 | 3.4 | 0.0 | 0.2 | 00 | 1.2 | 0.0 | 0.0 |
| IN-X007 | ID-RFO-310N | 310 | 0.2 | 0.2 | 0.0 | 0.0 | 0.0 | 0.2 | 0.0 | 0.0 |
| IN.W167 | ID-REO-111TN | 311 | 44 | 4.4 | 0.0 | 0.0 | 0.0 | 4.4 | 0.0 | 0.0 |
| IN-W272 | ID-RFO-312T | 312 | 19 | 1.9 | 0.0 | 1.9 | 0.0 | 0.0 | 0.0 | 0.0 |
| IN-W298 | ID-RFO-320T | 320 | 74.6 | 74.6 | 0.0 | 21.4 | 0.0 | 51.7 | 0.0 | 15 |
| IN-W297 | ID-RFO-320 | 320 | 28.6 | 28.6 | 0.0 | 00 | 00 | 28.0 | 00 | _0.6 |
| IN-W207 | ID-RFO-328T | 328 | 1.5 | 1.5 | 0.0 | <u> </u> | | 1.5 | 0 | 0.0 |
| IN-W233 | ID-RFO-328 | 328 | 0.2 | 0.2 | 0_ | 0.0 | Q.Q | 0.2 | 0 | 0 |
| IN-W169 | ID-REO-330T | 330 | 5774.6 | 5774.6 | 0.0 | 18.7 | 0_ | 5756.0 | 0 | 00 |
| IN-W158 | ID-RFO-330 | 330 | 3150.6 | 3150.6 | 0.0 | 00 | 00 | | 00 | 0.0 |
| IN-W208 | ID-REO-335T | 335 | 26.2 | 26.2 | 0.0 | 25 | 0.0 | 23.7 | 0 | 0 |
| IN-W234 | ID-REO-335 | _335_ | 16.5 | 16.5 | 0.0 | 00 | 0.0 | 6.5 | 0 | 0 |
| IN-W127 | ID-RFO-336T | _336_ | 778.3 | 778.3 | 0.0 | 20.4 | 00 | <u>758.0</u> | 0 | <u> </u> |
| IN-W160 | ID-REO-316 | 336 | 1452.4 | 1452.4 | 0.0 | 0_ | | | 0 | 0 |
| IN-W198 | ID-RFO-337T | 337 | 170.4 | 170.4 | 0.0 | 37.5 | 0.0 | 132.9 | 0_0 | 0 |
| IN-W217 | ID-RFO-337 | 332 | 352.9 | | 00 | 00 | 00 | 352.9 | 0 | 0.0 |
| IN-W209 | ID-RFO-338T | _138_ | 60.2 | 60.2 | 0.0 | 3.4 | 00 | 56.8 | 0.0 | 0.0 |
| IN-W235 | ID-REO-338 | 338 | 240.7 | 240.7 | 0.0 | <u> </u> | 00 | 240.7 | 0.0 | 0 |
| IN-W252 | ID-REO-339T | 132 | 160.2 | 160.2 | 0.0 | 13.4 | 0.0 | 0.0 | 0 | 146_9 |
| IN-W253 | ID-RFO-119 | 339 | 4.9 | 4.9 | 0.0 | 00 | 0_ | 03 | 0_0 | |
| IN-W210 | ID-RFO-360T | 360 | 14 | 3.4 | 0 | 00 | 0_ | | 0 | 0.0 |
| IN-W237 | ID-RFO-160 | 360 | 50.4 | | 0.0 | 00 | 0.0 | | 0 | 0.0 |
| IN-W373 | ID-REO-361TN | _361 | 0.2 | 0.2 | 0.0 | 00 | 00 | 0.2 | 0 | 0.0 |
| IN-W366 | ID-REO-370TN | _170 | 2.5 | 2.5 | 0.0 | 00 | 00 | 2.5 | 0 | 0 |
| IN-X008 | ID-REO-170N | _170 | 4.9 | 4.9 | 0.0 | 00 | 00 | 4.9 | 0 | 00 |
| IN-W161 | ID-REO-371T | _371 | 111.4 | 111.4 | 0_0 | 16.7 | 00 | 94.6 | 0 | |
| IN-W162 | ID-REO 171 | 371 | 183.5 | 183.5 | 0.0 | 00 | <u> </u> | | 0 | 00 |
| IN-W266 | ID-REO-172N | _322 | 0.8 | 0.8 | 0.0 | 00 | 00 | 0.8 | | 0 |
| IN-W267 | ID-REO-372TN | _372 | <u>3.0</u> | | 0 | | <u> </u> | 30 | 0 | 0 |
| IN-W265 | ID-REO-374T | _174 | | | 00 | | 00 | 43.6 | 0 | 00 |
| IN-W264 | ID-REO-374 | _174 | | <u>168_0</u> [| <u> </u> | <u>00</u> | <u> </u> | | 0 | 0.0 |
| IN-W163 | ID-REO-375T | 375 | <u> </u> | <u> </u> | <u></u> | <u> </u> | 00 | <u>0.8</u> | 00 | <u> </u> |
| IN-W223 | ID-REO-375 | 375 | 3.2 | | 0 | | 00 | | 0 | 0.0 |
| IN:W211 | ID-REO-376T | 176 | 460.2 | 460.2 | 0 | 215.4 | 001 | 211.8 | | 0 |
| IN-W238 | ID-RFO-376 | _176_ | 94.7 | | <u> </u> | <u>₽₽</u> | <u>00</u> | 94.2 | 9.0] | 0_01 |
| IN-W365 | ID-REO-391TN | _191 | <u> </u> | 4.7 | 0.0 | 00 | | | 0.0 | 0.0[|
| N-W1<1 | ID-RFO-392TN | 392 | | | <u> </u> | 0.01 | 0_0 | | <u> 0.0 </u> | <u>60</u>] |
|) | | | | |) | Page 2 | | | |) |

Page 2

.



DETAILS OF EDTA CALCULATIONS (BASIS: ALL ROCKY FLATS WASTE AT INEL)

| FFCA_ID | WS_ID | CC | Total Vol | CH Vol | RH Vol | CH_Direct Ship | RH_Direct Ship | Vitrified | Amalg | Macro |
|---------|--------------|---------|-----------|--------|----------|-----------------------|----------------|-----------|----------|--------|
| IN-W348 | ID-REO-393TN | 393 | 10.0 | 10.0 | 0.0 | 3.8 | 0.0 | 61 | 0.0 | _00 |
| IN-W157 | ID-RFO-004T | | 226.8 | 226.8 | | 54.3 | 0.0 | 172.3 | 0.0 | 02 |
| IN-W229 | ID-RFO-004 | 4 | 103.9 | 103.9 | 0.0 | 0.0 | 0.0 | 103.8 | 0.0 | _ 0 1 |
| IN-W311 | ID-REO-409T | 409 | 66 | 6.6 | 0.0 | 2.3 | 0.0 | 4.2 | 0.0 | 0.0 |
| IN-W356 | ID-RFO-410TN | 410 | 4.7 | 4.7 | 0.0 | 0.0 | 00 | 4.7 | 0.0 | 0.0 |
| IN-W355 | ID-RFO-411TN | 411 | 13 | 1.3 | 0.0 | 00 | 0 | 1.3 | 0.0 | _ 0.0_ |
| IN-W151 | ID-RFO-412TN | 412 | 0.2 | 0.2 | 0.0 | 0.0 | 0.0 | 0.2 | 0.0 | 0.0 |
| IN-W314 | ID-RFO-414T | 414 | 11 | | 0.0 | 0.0 | 00 | 1 | 0 | 00 |
| IN-W371 | ID-RFO-416TN | 416 | 0.2 | 0.2 | 0.0 | 0.0 | 00 | 0.2 | 0 | 00 |
| IN-W363 | ID-RFO-420TN | 420 | 2.3 | 2.3 | <u> </u> | 0.0 | 0_ | | 00 | 00 |
| IN-W362 | ID-RFO-J21TN | 421 | 21.4 | 21.4 | 0.0 | 0.0 | | 21.4 | 0.0 | |
| IN-W361 | ID-RFO-422TN | 422 | 5.1 | 5.1 | 0.0 | <u> </u> | 0.0 | 5.1 | <u> </u> | 00 |
| IN-W357 | ID-RFO-425TN | 425 | 0.4 | 0.4 | 0.0 | 0 | 00 | 0.4 | 0.0 | 0.0 |
| IN-X002 | ID-REO-425N | 425 | 1.1 | 13 | 00 | 0.0 | 0.0 | | 00 | 0 |
| IN-W320 | ID-RFO-430 | 430 | 19 | 1_2 | <u> </u> | 00 | 0.0 | | 0.0 | 0 |
| IN-W321 | ID-RFO-430T | 430 | 4.2 | 4.2 | 0.0 | 0_ | 00 | <u> </u> | 0.0 | 0 |
| IN-W318 | ID-REO-431 | _431_ | 0.4 | 0.4 | 0.0 | 0 | 00 | | 0.0 | 0 |
| IN-W319 | ID-RFO-411T | 411 | 0.8 | 0.8 | 0.0 | 0.0 | 0.0 | 0.8 | 0.0 | 00 |
| IN-W317 | ID-RFO-432T | 432 | 51.5 | 515 | 0.0 | 12.9 | | | 0 | 0.0 |
| IN-W316 | ID-RFO-432 | 432 | 8.9 | 8.9 | 0.0 | 0_ | 00 | 8.9 | 00 | 00 |
| IN-W243 | ID-RFO-440T | 440 | 247.7 | 247.7 | 0.0 | 56.2 | 00 | 191.5 | 0 | 0 |
| IN-W242 | 1D-RFQ-440 | _440_ | 95.4 | 95.4 | 0.0 | | 00 | 95.4 | 0.0 | 0.0 |
| IN-W244 | ID-RFO-441 | 441 | 164.7 | 164.7 | 0.0 | 00 | 00 | 164.7 | 0 | 00 |
| IN-W245 | ID-RFO-441T | 441 | 169.0 | | 0.0 | 0_ | 00 | 69.0 | 0 | 0 |
| IN-W247 | (D-REO-442T | 442 | 199.5 | 199.5 | 0.0 | 79.3 | 00 | 120.2 | 0 | 0 |
| IN-W248 | 1D-REO-442 | 442 | 138.4 | 138.4 | 0.0 | 0.0 | 00 | | 00 | 00 |
| IN-W199 | 1D-RFO-460T | _460_ | 13 | 1.3 | 0 | 0.0 | 00 | 13 | | 0 |
| IN-W254 | ID-RFO-463T | 463 | 10.2 | 10.2 | 0.0 | 0.0 | 0.0 | 0.6 | 0 | 95 |
| IN-W255 | ID-RFQ-461 | 463 | | | 00 | 00 | 0.0 | 01 | 0_0 | |
| IN-W183 | ID-RFO-464 | 464 | 38 | 3.8 | 0.0 | 00 | 00 | 3.1 | 0.0 | 0.8 |
| IN-W182 | 1D-REO-464T | 464 | 6.1 | 6.1 | 00 | 00 | 00 | 4.9 | 0.0 | |
| IN-W296 | ID-BEO-180T | 480 | 5243.4 | 5243.4 | 0 | | 0.0 | | 0.0 | 26.2 |
| IN-W295 | ID-REO-480 | 480 | 6688.0 | 6688.0 | 0.0 | 0 | 001 | 6654.6 | 0.0 | |
| N-W294 | ID-BEO-481T | 481 | 443.2 | 443.2 | | <u></u> <u></u> | 00 | 428.3 | 0_ | _35_ |
| IN-W293 | ID-8FO-181 | 481 | 164.3 | 164.3 | <u> </u> | <u> </u> | <u>µ</u> Q[| 163.1 | 00 | |
| IN-W212 | ID-RFO-190T | 490 | 2512.4 | 2512.4 | 00 | <u>i</u> , <u>+</u> [| <u> </u> | 2509.0 | 0_ | |
| IN-W219 | ID-REQ-190 | 490 | 873.4 | 873.4 | <u> </u> | 00 | <u> </u> | 873.4 | 00 | 0 |
| N-W313 | ID-REO-005 | <u></u> | 13.6 | 13.6 | 00 | 00 | | 13.6 | 0 | 0 |
| IN-W315 | ID-BEO-005T | -1-1- | | 0.6 | -0.0 | 0.0 | <u> </u> | | 0 | |
| N-W218 | ID-RFO-007T | _1_ | 461.5 | 461.5 | 001 | 461.5 | !! Q | 0 | 0 | 00 |
| N-W192 | ID-RFO-007 | _1 | 464.3 | 464.3 | 0 | | <u>0.0</u> I | 464.3 | 0.0 | 0.0 |



DETAILS OF EDTA CALCULATIONS (BASIS: ALL ROCKY FLATS WASTE AT INEL)

| FFCA_ID | WS_ID | CC | Total Vol | CH Vol | RH Vol | CH_Direct Ship | RH_Direct Ship | Vitrified | Amalg | Macro |
|---------|--------------|------|-----------|----------|--------|----------------|----------------|------------|----------|----------|
| IN-W164 | ID-RFO-700T | 700 | 19 | 1.9 | 0.0 | 0.6 | 0.0 | 1.3 | 0.0 | 0.0 |
| IN-W270 | ID-RFO-090 | 90 | 28.6 | 28.6 | 0.0 | 0.0 | 0.0 | 28.6 | 0.0 | 0_0 |
| IN-W205 | ID-REO-900T | 900 | 0.8 | 0.8 | 00 | 0.1 | 00 | 0.0 | 0.0 | 0.1 |
| IN-W227 | ID-REO-900 | 900 | 92.4 | 92.4 | 0.0 | 0_0 | 0.0 | 92.4 | 0.0 | 0.0 |
| IN-X001 | ID-REO-095N | 95 | 4.9 | 4.9 | 0 | <u> </u> | 0.0 | 4.9 | 00 | 0.0 |
| IN-W277 | ID-REO-950 | 950 | 1065.0 | 1065.0 | 00 | 0.0 | 0.0 | 1006.6 | 0.0 | 58.4 |
| IN-W278 | ID-REO-950T | 950 | 14.0 | 14.0 | 0.0 | 0.0 | 0.0 | 13.2 | 0 | 0.8 |
| IN-W374 | ID-RFO-960TN | 960 | 9.8 | 9.8 | 0.0 | 0.2 | 0.0 | 9.5 | 0.0 | |
| IN-X003 | ID-REO-960N | 960 | 681.4 | 681.4 | 0.0 | 00 | 0.0 | 681.4 | <u> </u> | 0 |
| IN-W202 | ID-REO-970T | 970 | 109.9 | 109.9 | 0 | 0.0 | 00 | 109.9 | 0 | 0_0 |
| IN-W224 | ID-RFO-970 | 970 | 91.3 | 91.3 | 0 | 00 | 0.0 | 91.3 | 00 | 0_0 |
| IN-W180 | ID-REO-976 | 976 | 63.8 | 63.8 | 0.0 | <u> </u> | 0.0 | 63.8 | 0 | 0 |
| IN-W188 | ID-RFO-976T | 976 | | | 0.0 | 0.0 | 0.0 | <u>].l</u> | 0.0 | 0_0 |
| IN-W181 | ID-REO-978T | 978 | 9.5 | 9.5 | 0 | 0_0 | 00 | 9.5 | 0.0 | 0.0 |
| IN-W182 | ID-REO-978 | 978 | 25.4 | 25.4 | 0 | 0_0 | 0.0 | 25.4 | 0.0 | 0.0 |
| IN-W187 | ID-REO-980T | 980 | 0.2 | 0.2 | 0 | 0.0 | 0.0 | 0.2 | <u> </u> | 0.0 |
| IN-W261 | ID-RFO-990 | 990 | 99.6 | <u> </u> | 0.0 | 0 | 00 | 99.6 | 0.0 | 0_0 |
| IN-W375 | ID-RFO-995TN | 995 | <u> </u> | <u> </u> | Q.0 | 0.0 | 00 | 19.3 | 0_0 | 0.0 |
| IN-X002 | ID-RFQ-995N | 995 | 68.8 | 68.8 | 0.0 | 00 | 0.0 | 68.8 | 0.0 | 0.0 |
| IN-W306 | ID-RFO-9999T | 9999 | 4492.5 | 4489.3 | 3.2 | 0.0 | | 4354.5 | 0.0 | 134.8 |
| IN-W352 | ID-RFO-9999 | 9999 | 2993.7 | 2991.5 | | 00 | 21 | 2901_7 | 00 | <u> </u> |
| | TOTALS | | 58402.2 | 58396.9 | 5,3 | 2626.5 | 5.3 | 55152.7 | 15.7 | 601.9 |

| TOTAL EDTA IN RF WASTE AT INEL> | 21.3 kg |
|--|---------|
| (1.15 kg/yr for 18.5 ycars) | 01.194 |
| PERCENT VITRIFIED | 24.470 |
| AMOUNT VITRIFIED (94.44% of 21.3 kg)> | 20.1 kg |
| AMOUNT IN UNTREATED INEL WASTE> | 1.2 kg |
| TOTAL EDTA IN RF WASTE AT RF> (1.15 kg/yr for 1.5 ycars) | 1.7 kg |
| NEW EDTA ESTIMATE> | 2.9 kg |

B5-10

APPENDIX B - 6

*

memorandum

Carlsbad Area Office Carlsbad, New Mexico 88221

DATE: FEB 2 0 1996

ATTNOF: NTP:DW:96-0655

SUBJECT: Preliminary Estimate for SNL/NM Performance Assessment Calculations of Nitrate, Sulfate, and Phosphate Content in Transuranic Solidified Wastes Destined for Disposal in WIPP

TO:

Dr. Les Shephard, SNL/NM

Attached is a copy of the report containing the preliminary estimates for the nitrate, sulfate, and phosphate contents in solidified transuranic (TRU) wastes destined for the Waste Isolation Pilot Plant (WIPP). This information was requested by your staff from the Transuranic (TRU) Waste Baseline Inventory Report (TWBIR) team in support of the Performance Assessment efforts.

Briefly, the enclosed document provides estimates of the average density and total mass of nitrate and sulfate in TRU waste to be disposed of at the WIPP. These values have been estimated based on data obtained from the TRU waste generator/storage sites during the TWBIR preparation process. From these data, the average densities scaled over the entire WIPP disposal inventory are 9.2 kg/m^3 for nitrate and 3.6 kg/m^3 for sulfate. The total masses scaled over the entire WIPP disposal inventory are 9.2 kg/m^3 for nitrate and 3.6 kg/m^3 for sulfate. The total masses scaled over the entire WIPP disposal inventory are 1.6E+06 kg for nitrate and 6.3E+05 kg for sulfate. These densities and masses are for combined CH and RH TRU waste inventories. No value for phosphate has been proposed due to the lack of sufficient information. Trace quantities of inorganic phosphate might be expected in some of the sludges and solidification agents, but no supporting analytical data are available to support a specific value. This is discussed in the enclosed report.

If you have any questions concerning the attached information, please contact Mr. Russ Bisping of my staff at (505) 234-7446.

Kund Dugin

Don Watkins Manager National TRU Program

B6-1

Attachment

L. Shephard

cc w/enclosure: J. Mewhinney, CAO R. Bisping, CAO P. Drez, CTAC J. Harvill, CTAC L. Sanchez, SNL M. Chu, SNL M. Marietta, SNL

B6-2

2

Preliminary Estimates of Nitrate, Sulfate, and Phosphate Content in Transuranic Solidified Wastes

I. INTRODUCTION

This report provides preliminary estimates of the amount of nitrate, sulfate, and phosphate expected to be in the transuranic (TRU) inventory that will be transported to and disposal of at the Waste Isolation Pilot Plant (WIPP) (Appendix B: DOE, 1995). Tables 1 and 2 of this report provide the volumetric basis for the nitrate and sulphate estimates, and Tables 3, 4, and 5 provide the calculational methodology. No quantifiable sources of phosphate have been identified in the Inorganic Solidified final waste forms at present. Trace quantities might be expected in some of the sludges and solidification agents, but no data currently exist to support this.

II. BACKGROUND

These PRELIMINARY estimates are made based on the following:

- Values presented are those expected for the final waste forms to be disposed of at WIPP.
- Information has been requested from sites based on Solidified Inorganic and Solidified Organic waste forms only, and is the best available data from the TRU waste generator/storage sites:
 - The main source of nitrate is anticipated to be from the Solidified Inorganic waste forms, which in most cases, are sludges produced from the neutralization/ solidification of nitric acid-based solutions used at the TRU waste generator/storage sites. Nitrates are very soluble in aqueous solutions and generally do not produce precipitates in the sludges. The nitrates are generally thought to be present as ions sorbed on precipitates or as interstitial solution trapped in the precipitated sludges prior to solidification.

Minor amounts of nitrate, as evaporites, are anticipated in the debris waste forms that will be acceptable for WIPP disposal, but insufficient data are available to estimate the amount of such TRU waste at this time.

The main sources of sulfates are anticipated to be: 1) chemicals (e.g. iron sulfates) added to the inorganic solutions at the time of flocculation and precipitation of sludges. and 2) the use of Envirostone [a gypsum (CaSO₄) based solidification material] for solidification of inorganic and/or organic solutions/ sludges at some TRU waste generator/storage sites. No quantifiable sources of phosphate have been identified in the Solidified Inorganic final waste forms at present. Trace quantities might be expected in some of the sludges and solidification agents, but no supporting analytical data are available. The quantities of inorganic phosphate are anticipated to be low in inorganic sludges based on process histories at TRU waste sites.

Analytical data in Attachment 2 provide only "less than 0.0025" weight percent values for phosphate, which are similar to the 0.001 weight percent estimate provided by LANL in Attachment 1. These values are too low to make any reliable estimate of phosphate in TRU waste, but indicate that the quantities will be very small, compared with the nitrate and sulfate values reported. The phosphate value of "40%" reported on page A2-7 is an analytical error. Based on process knowledge and the lack of cations to support such a large value of phosphate in that particular analysis, no such value is possible.

III. GENERAL VOLUME CALCULATIONS

A. <u>Nitrate</u>

1. Nitrate Assumptions

The amount of nitrate is estimated on the basis of the volumes of Solidified Inorganics, which are calculated as explained below:

- Table 1 lists (in Column 2) the final waste form volumes of Solidified Inorganics for Contact-Handled (CH) TRU and Remote Handled (RH) TRU from Figures 3-9 and 3-16 of Revision 2 of the TWBIR (DOE, 1995) for the anticipated WIPP inventory (stored plus projected volumes until 2022).
- Footnotes in Columns 3 and 4 indicate why certain volumes of waste have been eliminated from further consideration in the calculations:



- Footnote 1 eliminates those volumes of chemically precipitated Solidified Inorganics for which no nitrate estimates in the waste are available. An estimate of the nitrate contribution from these Solidified Inorganics will be accounted for in the scaling process.
- Footnote 2 eliminates the volume of Solidified Inorganics from SRS from further consideration because it is a "vitrified" waste form which should not contain any significant amount of nitrates due to the thermal treatment proposed for that waste form.

- Footnote 3 eliminates from further consideration those volumes of
 Solidified Inorganics which represent non-precipitated particulates (e.g., incinerator ash, graphite fines, etc.) which have been cemented to meet the WIPP WAC; nitrates are not expected to be present in these particulates.
- Rocky Flats Environmental Technology Site (RFETS) and Los Alamos National Laboratory (LANL) have provided analytical data/estimates for nitrate in Solidified Inorganics. The RFETS data has been used also for the RFETS waste stored at INEL.
- 2. Nitrate Mass Calculations

Table 3 contains in Column 1 a list of those waste streams that contain the volume of waste from each TRU waste generator/storage site listed in Column 4 of Table 1. The additional data provided are:

- Column 2 lists the Item Description Codes (IDCs) for waste streams produced at RFETS and/or stored at INEL. The RF111 designation is for Content Code 111 from RFETS, where the IDC is not specified.
- Column 3 lists the stored + projected volume for each waste stream.
- Column 4 lists the sum of the waste material parameters (WMP) for each waste stream from the individual Waste Stream Profiles in Revision 2 of the TWBIR. Exceptions to this rule are listed in footnotes in Table 3.
- Column 5 lists the mass of the waste for each waste stream which is the product of multiplying Columns 3 and 4.
- Column 6 lists the values of nitrate used for each waste stream. The sources of the these values are:
 - For RFETS, the nitrate values are from Appendix I of Revision 2 of the TWBIR. The 8% values for IDC 001 has also been applied to IDCs 002 and 007 at both RFETS and INEL. All these IDCs represent "older" methods of solidification where the sludges contain portland cement mainly as a sorbent interlayered with sludge which did not contain diatomaceous earth (see Clements, 1982 for drawings).

Baseline Inventory Report Data, February 1996 Nitrates, Sulfates, and Phosphates

The 4% value listed in Appendix I of the TWBIR for IDC 807 represents a "newer" method of solidification where diatomaceous earth is used as a vacuum filtration agent and portland cement is mixed with the resulting sludge to form a "monolithic" solidified final waste form. The dilution with diatomaceous earth and additional portland cement lowers the overall nitrate value of the final waste form.

- For waste stream IN-W315.601, Clements (1982) indicates that the waste stream is made up of approximately 60% NaNO3 and 30% KNO3 (assumed weight percents). This calculates as 62% nitrate.
 - Attachment 1 represents a memo from LANL that provides estimates for nitrates in the waste streams. Note that the Envirostone process only accounts for a small percentage of stored volume for 3 of the waste streams. The values quoted in Column 6 are based on the small percentage of Envirostone solidification agent in the overall waste streams.
- Column 7 represents the mass of nitrates in kg which is the product of multiplying Columns 5 and 6.
- B. Sulfate

1. Sulfate Assumptions

- To determine the amount of solidified wastes that need to be considered for calculating the sulfate content of the WIPP inventory (Table 2), the volume of Solidified Organics must be added to the volume of Solidified Inorganics from Table 1:
 - The Solidified Organics from Figures 3-10 and 3-17 of Revision 2 of the TWBIR (DOE, 1995) have been added to Table 1 (above) to produce Table 2
 - LANL has used an Envirostone (gypsum-based) process for solidification of inorganic sludges in the past (approximately 9% of 4888 m³ in storage at LANL) but plan to eliminate the process in the future and only use portland-based cement for solidification (as was used in the past prior to usage of the Envirostone)



Baseline Inventory Report Data, February 1996 Nitrates, Sulfates, and Phosphates

- Since the mid 1980's, RFETS has used an Envirostone solidification process for their organic sludges. Therefore, some of their waste in storage and projected contain large amounts of sulfate, as well as some Solidified Organics in storage at INEL.
- LLNL is the only other TRU waste site known to be using Envirostone for the solidification of organic liquids/sludges (approximately 7 m³ stored/projected).

2. Sulfate Mass Calculations

The sulfate calculations presents in Table 4 follow the same format as the nitrate calculations in Table 3. The origin of the values used for sulfate in the RFETS, INEL, LLNL, and LANL waste streams are summarized below:

• RFETS/INEL

- The 0.11% sulfate value is an average of the three analyses marked "7412 Sludge" in Attachment 2 which are applied to IDCs 001 and 002, and at half that value for IDCs 800 and 803 (as explained in the nitrate section).
- The sulfate value of 0.02% is derived from the Attachment 2 analysis marked "374 Waste Sludge - Dried Sludge". This value is used for IDC 007 and at half value for IDC 807.
- The sulfate value (25.1%) for the Envirostone solidification of organic sludges (IDC 801) is derived from an average value in Attachment 3, which represents guidelines for mixing constituents together for IDC 801 and IDC 700 (at INEL only in storage).

• LANL

The values for sulfate quoted in Column 7 are derived from data provided in Attachment 1. As with the nitrate calculations, the percentage of waste in each waste stream solidified by Envirostone versus portland cement is used to calculated the overall sulfate value for each waste stream.

• LLNL



No value for sulfate was requested from LLNL for their one Solidified Organic waste stream. The same value for Envirostone-solidified waste at RFETS (25.1%) was assumed for the LLNL waste stream.

5

IV. SUMMARY CALCULATIONS

Table 5 presents the summary calculations for determining the density (kg/m^3) of nitrate and sulfate in the overall WIPP inventory and scaling of the density to take into account those chemically precipitated waste streams for which data was not available. SNL/NM should use the scaled densities for their calculations. The last column in Table 5 provides the estimated mass of nitrate and sulfate if the design capacity of WIPP for CH-TRU and RH-TRU are fully utilized based on the scaled densities for nitrate and sulfate.

V. REFERENCES

Clements, 1982, "Content Code Assessments for INEL Contact-Handled Stored Transuranic Wastes," WM-F1-82-021, Idaho Falls, Idaho.

U. S. Department of Energy, 1995, "Transuranic Waste Baseline Inventory Report (Revision 2)," DOE/CAO-95-1121, Carlsbad, New Mexico.



TABLE 1. TRU VOLUMES FOR NITRATE CALCULATIONS (SOLIDIFIED INORGANICS ONLY)

| TRU WASTE SITE | TOTAL VOLUME (STORED + PROJECTED) (m ²) | VOLUMES WITH NITRATE DATA OR WITH PARTICULATES (m ³) | VOLUMES OF SLUDGES WITH NITRATE DATA (m ²) |
|-------------------|--|---|---|
| Hanford (CH) | 23.39 | (TO BE SCALED) | (TO BE SCALED) ¹ |
| ANL-E (CH) | 5.20 | (TO BE SCALED) | (TO BE SCALED) ¹ |
| NTS (CH) | 5.67 | (TO BE SCALED) | (TO BE SCALED) ¹ |
| SRS (CH) | 1369.8 | 1369.8 | 2 |
| RFETS (CH) | 1423.01 | 1389.52 | 229.63 ³ |
| INEL (CH) | 4344.44 | 3900.39 | 3598.84 ³ |
| Mound (CH) | 6.03 | (TO BE SCALED)' | (TO BE SCALED) |
| LANL(CH) | 6922.02 | 6922.02 | 6922.02 |
| AL (CH) | 0.42 | (TO BE SCALED) | (TO BE SCALED) ¹ |
| LLNL (CH) | 20.18 | (TO BE SCALED) ¹ | (TO BE SCALED) ¹ |
| CH TOTAL | 14120.15 | 13581.73 | 10750.49 |
| | | | |
| ORNL (RH) | 1243.33 | (TO BE SCALED) ¹ | (TO BE SCALED) ¹ |
| INEL (RH) | 65.27 | 65.27 | 65.27 |
| ANL-E (RH) | 30.26 | (TO BE SCALED) | (TO BE SCALED) ¹ |
| RH TOTAL | 1338.86 | 65.27 | 65.27 |
| | | | |
| TRU TOTAL | 15459.01 | 13647.0 | 10815.76 |

Eliminates those volumes of chemically precipitated solidified inorganics for which no nitrate estimates in the waste are available. An estimate of the nitrate contribution from these solidified inorganics will be accounted for in the scaling process.

Eliminates the volume of Solidified Inorganics from SRS from further consideration because it is a "vitrified" waste form which should not contain any significant amount of nitrates due to the thermal treatment proposed for that waste form. 3

Eliminates from further consideration those volumes of Solidified Inorganics which represent non-precipitated particulates (e.g., incinerator ash, graphite fines, etc.) which have been comented to meet the WIPP WAC and nitrates are not expected to be present in the particulates.

1

2

| TRU WASTE SITE | FINAL WASTE FORM | TOTAL VOLUME (m ²) | VOLUME WITH SULFATE DATA (៣ ²) | |
|----------------|------------------|--------------------------------------|--|--|
| Hanford (CH) | Solídiť. Inorg. | 23.39 | (TO BE SCALED) | |
| ANL-E (CH) | Solidif. Inorg. | 5.20 | (TO BE SCALED) ¹ | |
| NTS (CH) | Solidif. Inorg. | 5.67 | (TO BE SCALED) | |
| SRS (CH) | Solidif. Inorg. | 1369.8 | (TO BE SCALED) | |
| RFETS (CH) | Solidif. Inorg. | 1423.01 | 229.63 | |
| INEL (CH) | Solidif. Inorg. | 4344.44 | 3598.42 | |
| Mound (CH) | Solidif. Inorg. | 6.03 | (T BE SCALED) ¹ | |
| LANL (CH) | Solidif. Inorg. | 6922.02 | 6922.02 | |
| AL (CH) | Solidif. Inorg. | 0.42 | (TO BE SCALED) | |
| LLNL (CH) | Solidif. Inorg. | 20.18 | (TO BE SCALED) ¹ | |
| RFETS (CH) | Solidif. Org. | 140.93 | 108.99 | |
| Hanford (CH) | Solidif. Org. | 76.13 | (TO BE SCALED) ¹ | |
| LANL (CH) | Solidif. Org. | 30.58 | (TO BE SCALED) ⁴ | |
| INEL (CH) | Solidif. Org. | 789.67 | 2.55 | |
| ANL-E (CH) | Solidit. Org. | 0.21 | (TO BE SCALED) ¹ | |
| LLNL (CH) | Solidit. Org. | 6.86 | 6.86 | |
| CH TOTAL | | 15164.53 | 10868.93 | |
| | | | | |
| ORNL (RH) | Solidif. Inorg. | 1243.33 | (TO BE SCALED) ¹ | |
| INEL (RH) | Solidif. Inorg. | 65.27 | 65.27 | |
| ANL-E (RH) | Solidif. Inorg. | 30.26 | (TO BE SCALED) ¹ | |
| INEL (RH) | Solidif. Org. | 3.56 | (TO BE SCALED) ¹ | |
| RH TOTAL | | 1342.42 | 65.27 | |
| | | | | |
| TRU TOTAL | | 16506.95 | 10933.74 | |

TABLE 2. TRU VOLUMES FOR SULFATE CALCULATIONS

No sulfate data available from these sites for any waste streams.



1

| Waste Stream | DCs | Volume | Sum WMP | Mass Waste | % Nitrate | Nitrate |
|---------------|-------------|----------|---------|---------------|-----------|-----------|
| | | (m3) | (kg/m3) | (kg) | (weight%) | (kg) |
| RF-MT0001 | 001 | 3.74 | 781.9 | 2924.31 | 8 | 233,94 |
| RF-MT0800 | 800 | 104.42 | 775.2 | 80946.38 | 4 | 3237.86 |
| RF-MT0803 | 803 | 4.99 | 635.2 | 3169.65 | 4 | 126.79 |
| RF-MT0807 | 807 | 115.02 | 819.6 | 94270.39 | 4 | 3770.82 |
| RF-T010 | 800/803/807 | 0.62 | 796.1 | 493.58 | 4 | 19.74 |
| TOTAL RFETS | | 228.79 | | 181804.31 | | 7389.14 |
| | | | | | | |
| IN-W216.875 | 001/002 | 1478.88 | 819.6 | 1212090.05 | 8 | 96967.20 |
| IN-W216.877 | 001/002 | 43.91 | 571.4 | 25090.17 | 8 | 2007.21 |
| IN-W216.98 | 001/002 | 555.65 | 726.6 | 403735.29 | 8 | 32298.82 |
| IN-W218.909* | 007 | 101.91 | 544.3 | 55469.61 | 8 | 4437.57 |
| IN-W220.114 | RF111 | 122.80 | 725.6 | 89103.68 | 4 | 3564.15 |
| IN-W220.925 | RF111 | 443.04 | 819.6 | 363115.58 | 4 | 14524.62 |
| IN-W228.101 | 002 | 287.33 | 317.3 | 91169.81 | 8 | 7293.58 |
| IN-W228.883 | 002 | 608.82 | 358.0 | 217957.56 | 8 | 17436.60 |
| EN-W228.886 | 002 | 21.36 | 249.6 | 5331.46 | 8 | 426.52 |
| IN-W315.601++ | 005 | 0.42 | 664.0 | 278.88 | 62 | 172.91 |
| TOTAL INEL | | 3664.12 | | 2463342.09 | | 179129.19 |
| LA-M002 | | 3606.81 | 1296.0 | 4674425.76 | 8.8 | 411349.47 |
| LA-T006 | | 86.53 | 1004.8 | 86945.34 | 8.8 | 7651.19 |
| LA-W003 | 1 | 1836.58 | 1339.3 | 2459731.59 | 8.7 | 213996.65 |
| LA-W006 | <u> </u> | 1392.10 | 1004.8 | 1398782.08 | 8.7 | 121694.04 |
| TOTAL LANL | | 6922.02 | | 8619884.78 | | 754691.35 |
| TOTAL TRU | | 10814.93 | | 1 11265031.18 | | 941209.68 |

TABLE 3 : NITRATE CALCULATION

* INEL did not report waste material parameters for this waste stream. The value for this IDC at RFETS was assumed.

** This waste stream was reported in Clements (1983) to be 60% NaNO3 and 30% KNO3. The weight of the waste for this IDC was used from Clements (1983), since no value was quoted in Revision 2 of the TWBIR.

.

| Waste Stream | IDCs | Waste | Volume | Sum WMP | Mass Waste | Seulfate. | Sulfate |
|---------------|---|---------------|------------|------------------|-------------------|-----------|-----------|
| | | Form | (m3) | (kg/m3) | (Lg) | (wights) | (kg) |
| RF-MT0001 | 001 | Sol. Inorg. | 3.74 | 781.9 | 2924.31 | 0.11 | 3.22 |
| RF-MT007 | 007 | Sol. inorg. | 0.83 | 544.3 | 452.86 | 0.02 | 0.09 |
| RF-MT0800 | 800 | Sol. inorg. | 104.42 | 775.2 | 80946.38 | 0.055 | 44.52 |
| RF-MT0801 | 801 | Sol. Org. | 108.99 | 877.1 | 95595.13 | 25.1 | 23994.38 |
| RF-MT0503 | 803 | Sol. Inorg. | 4.99 | 635.2 | 3169.65 | 0.055 | 1.74 |
| RF-MT0807 | 807 | Soi. Inorg. | 115.02 | 819.6 | 94270.39 | 0.01 | 9.43 |
| RF-TOIO | 800/803/807 | Sol. Inorg. | 0.62 | 796 .1 | 493.58 | 0.055 | 0.27 |
| TOTAL RFETS | Į — — — — — — — — — — — — — — — — — — — | | 338.61 | | 277852.30 | | 24053.65 |
| IN-W164.1060* | 700 | Sol. Org. | 1.66 | 877.1 | 1455.99 | 25.1 | 365.45 |
| IN-W164.153* | 700 | Sol. Org. | 0.89 | 877.1 | 780.62 | 25.1 | 195.94 |
| EN-W216.875 | 001/002 | Sol. Inorg. | 1478.88 | 819.6 | 1212090.05 | 0.11 | 1333.30 |
| IN-W216.877 | 001/002 | Sol. Inorg. | 43.91 | 571.4 | 25090.17 | 0.11 | 27.60 |
| EN-W216.98 | 001/002 | Soi. Inorg. | 555.65 | 726.6 | 403735.29 | 0.11 | 444.11 |
| IN-W218.909* | 007 | Sol. Inorg. | 101.91 | 544.3 | 55469.61 | 0.02 | 11.09 |
| IN-W220.114 | RF111 | Sol. Inorg. | 122.80 | 725.6 | 89103.68 | 0.055 | 49.01 |
| IN-W229.925 | RFILI | Sol. Inorg. | 443.04 | 819.6 | 363115.58 | 0.055 | 199.71 |
| IN-W228.191 | 002 | Sol. Inorg. | 287.33 | 317.3 | 91169.81 | 0.11 | 100.29 |
| IN-W228.883 | 002 | . Soi. Inorg. | 608.82 | 358.0 | 21 7957.56 | 0.11 | 239.75 |
| IN-W228.886 | 002 | Sol. Inorg. | 21.36 | 249.6 | 5331.46 | 0.11 | 5.86 |
| TOTAL INEL | ļ | | 3666.25 | | 2465299.82 | | 2972.11 |
| LA-M082 | | Sol. Inorg. | 3606.81 | 1296.0 | 4674425.76 | 1.4 | 65441.96 |
| LA-T006 | | Sol. Inorg. | 86.53 | 1004.8 | 86945.34 | 1.7 | 1478.07 |
| LA-W083 | | Soi. Inorg. | 1836.58 | 1339.3 | 2459731.59 | 5.5 | 135285.24 |
| LA-W006 | | Sol. Inorg. | 1392.10 | 1004.8 | 1398782.08 | 8.1 | 113301.35 |
| TOTAL LANL | ļ | | 6922.02 | | 3619884.78 | St | 315506.62 |
| LL-W019** | | Soi. Org | .36 | <u>,268.0</u> | 1838.48 | 25.1 | 461.46 |
| TOTAL LLNL | Į | | 6.86 | | 1838.48 | SI | 461.40 |
| TOTAL TRU | + | <u>+</u> | 1 10933.74 | 1 | 11364875.38 | 3 | 342993.8- |

TABLE 4 : SULFATE CALCULATION

* INEL did not report waste material parameters for this waste stream. The value for this IDC at RFETS was assumed.

** Sulfate value for LLNL Solidified Organics was assumed to be the same as for RFETS Solidified Organics (IDC 801).



TABLE 5. NITRATE/SULFATE DENSITY CALCULATIONS

| Constituent | Volume Solidified Waste (m3) | Mass Solidified Waste (kg) | Mass Constituent (kg) | Auticipated Waste Volume (m3) | WIPP Average Density of Constituents (kg/m3) | % Sindge Used in Calculations (%) | WIPP Average Scaled Density of Constituents (kg/m3) | Total Mass of Constituent for WIPP Design Capacity (kg) |
|--------------------|---------------------------------------|-------------------------------------|-----------------------------|--|---|--|--|--|
| Footnotes | 1 | 2 | 3 | 4 | 5 | 6 | 7 | 8 |
| Nitrate Sulfate | 10815.76 10933.74 | 1 1265484 1 1364875 | 941245.9 342993.8 | 1.19E+05 1.19E+05 | 7.91 2.88 | 85.6 80 | 9.24 3.60 | 1.62E + 06 6.33E + 05 |

L "Total TRU" Volumes for Tables 3 and 4.

2. "Total TRU" Mass from Tables 3 and 4.

3. "Total TRU" Nitrate/Sulfate from Tables 3 and 4.

4. Anticipated Volume of CH-and RH-TRU Waste (stored + projected to 2022) from Table 3-1 in Rev. 2 of TWBIR. RH-TRU anticipated volume is limited to 7080 m3, the design capacity of WIPP.

5. "Mass of Constituent" column divided by "Anticipated Waste Volume" column.

6. Calculated from Table 1 "Total TRU" data. Nitrate = subtract 10815.76 from 13647 to yield particulate waste (2831.24). Subtract 2831.24 from 15459.01 to get total chemically precipitated waste (12627.77). Divide 10815.76 by 12627.77 and multiply by 100%. Suffate is calculated in a similar manner.

7. Divide "Density of Constituent" by "% Sludge Used in Calculations."

8. Multiply "Scaled Density of Constituent" by 175,600 m3 (design capacity of WIPP).

TELEPHONE CONFERENCE SUMMARY

Parties: Paul Drez, DEA/CTAC Davis Christenson, LANL

For Solidified Inorganics waste stream LA-T006; LA-W003; LA-W006; and LA-M002 assume the following composition for final waste form:

Envirostone-based solidified waste forms:

Nitrate 8.2% Sulfate 38.5% Phosphate 0.001%

Portland Cement-based solidified waste forms:

Nitrate 8.8% Sulfate 1.4% Phosphate 0.001%

LA-M002 has only used portland cement; the other three have use portland cement until 1985 and then Envirostone:

| | Store | d Wasted | Projected | Waste |
|---------|----------|-------------|-----------|-------------|
| WS# | Portland | Envirostone | Portland | Envirostone |
| LA-T006 | 84.5% | 15.5% | 100% | 0\$ |
| LA-W006 | 54.65% | 45.35% | 100% | 0\$ |
| LA-W003 | 84.5% | 15.5% | 100% | 0\$ |
| LA-MOO2 | 100% | 0% | 100% | 0\$ |

A1 + 1

| | D61472-01 | | LAB | RATORY SAMPLA 7412 Slud | FRESULTS | | | DATE | 04/10/80 PAGE 1 |
|-----|---------------------------------------|------|-----------------------------------|--|----------------------------------|---------------------|-------------|-------------------------------|------------------------------------|
| - | SAMPLE-18 ENTRY DATE COMPLETION | DATE | 00-009395 11-01-79 04-10-80 | | DJC N Accou Build CINSS | UMBE NT C Ing | R HARGED | 9763800 8037 559 559 | 0 |
| _ | CUSTOMER | | P. T. GOD | | | | | | |
| - | | • | # ATOMEC ABS | DRPTION SPECT | ROMETRY R | ESUL | TS | | |
| •, | CA | | 86512. | PPK(W) | FE | | 6159 | 7. | PPMENT |
| | GÅ | < | 50. | PPMEWI | ĸ | | 616 | 2. | PPH(¥) |
| | NA | | 655 01 . | PPR(V) | 2 I | | 365 | Q_ | PPH(W) |
| | | 4 | + PLUTONIUN | CHEMISTRY LAB | ORATGRY R | FSUL | TS | | |
| - | C1 (-) | | B-16 | 76.93 | 65 | | | 0.74 | •/ • • |
| | 5/-1 | | 57. | 000.01 | UC3- | | 4 | 1 0 | 4(#/ |
| | NDa | | 6.7 | TTN: | | | 0 | 1.0 | 4587 |
| - | 504 | | 0.085 | 2(8) | 664 | | | 0.0027 | 4. 4 # 3 - |
| - | | | SEMI-QUANT | ATIVE EMISSIO | N SPEC RE | SUL1 | 15 | | |
| | AG | | 59. | PPK(¥) | AL. | | 2000 | 0. | PPH(W) |
| - | AS | < | 50. | 55H(K) | B | | 10 | 0. | PPM(V) |
| | BA | | 130- | PP#(¥) | BE | | 5 | Ð. | ₽₽4(¥) |
| | BI | < | 70 e | PPM(¥) | CA | > | 20000 | 0. | \$ \$ # { k } |
| - | ED | < . | 1000- | · # # M (¥) | CE | < | 50 | 0. | PPH(W) - |
| | C0 | Ś | 53. | PPH(¥) | CR | | 50 | 0. | ₽₽#(¥) |
| | 22 | < | 1093- | PP# (V) | CU | _ | 400 | 6. | \$\$4(#) |
| ·• | FE | | 50000 - | PPR(V) | GE | < | 1 | .0. | PP4(¥) |
| | HG | Ś | 10- | ***** | ĸ | | 4000 | 0. | PPH(W) |
| | LI | K | - 1000- | 644(¥) | MG | | 1000 | 10. | 66146 |
| - | 7.N | | 200 - | PPRIEJ | P D | | 50 | ж. | PPM(W)- |
| | | | | PPMLUI | ~ ~ ~ | | 2 | | PPH(X) |
| | NI | | 2393- | PPTIL¥1 | ۳ ۵۳ | | 100 | 10. | 55×(8) |
| | P 8 | < C | 7J 4 63 | PPR(¥) | KE | < | 70 | 70 - | PP5(W)- |
| | 26 | < c | 30. | PPR(W) | 21 | | 10000 | | PPP(¥) |
| | 24 | C C | 10- 5-7 | "BOMENT | 74 | | 1001 | 30. | BBW(R) |
| - | 14 | > | 503 · | | 12 | ς. | 11 | | ***[¥]- |
| | 1 M T 1 | ~ | 50J. | ************************************** | 11 | , | フリーニン | ມນ. ພາກ | ****** |
| | 1 L. 10 | > | 70J• 5 | 909/41 | U u | | 10 | UT | 564155 222(9) |
| . • | * 2 M | 2 | 532- | PPMINS | 79 | Ì | 101 | 50. | 577181- 952(v) |
| | 211 | | | | 4.5 | • | | ~~= | FF7187 |
| | | | ** RADIOCHER | ISTRY LABORAT | ORY RESUL | T 5 | | | |

AM

0,0000317 G/G

A2 - 1

-

| | ATTACH | MENT 2 | | |
|-----------|--------------------------|------------------|-----------|----------|
| 061472-01 | LABORATORY SAM | PLE RESULTS | DATE | 04/10/50 |
| SAMPLE-ID | 00-008395 | | | PAGE 2 |
| | ** RADIUCHENISTRY LABORA | TORY RESULTS | CONTINUED | - |
| PU | 0.0000223 G/G | U | C-0017 | G/G |
| | TUA | HORIZED SIGNATUR | F | |

11000

-__

Hewel 8

_

...

.

A2 - 2

| | 261472-01 | | L A 8 | CRATURY SAXPLI 7412 Study | E RESULTS | | | OBTE | 04/10/80 PAGE 1 |
|----|---------------------------------------|----|--------------------------------------|------------------------------|-------------------------|---------------------|-------------|------------------------|-------------------------|
| | SAMPLE-ID Entry date Completion | DA | 00-008396 11-01-79 TE 04-10-80 | | DJC N Accou Build | UMBE NT C Ing | R HARGED | 9703809 8037 559 | 0 |
| | CUSTOMER | | P. T. GOI | ESATBOIS | LLAJJ | | | 2261 | |
| | همین بندی بند من مردی ور <u>می</u> . | | ** AT381C 685 | DEPTION SPECT | | FSUR | | | |
| | | | 104687 | | | | | - | |
| | | | TA4301* | PP={\} | + E | | 4/91 | 2- | PPATWI |
| | GA | ۲. | 7U | | A G | | 958 | 1. | PPM(W) |
| •• | NA | | 102093* | PERENT | 51 | | 15 | ē., | PF4(W) |
| | | | ** PLUTONIUN | CHEMISTRY LAB | | ESUL | TS. | | |
| - | CI (-) | | 0.15 | 2144 | CC3+ | | | 0 74 | 7691 |
| | F(-) | | 101. | PPMENS | H2D | | 5 | 5.0 | 2(2) |
| - | ND3 | | 3.0 | ZCYI | 904 | ٢ | | 0-0025 | 7/81 |
| | 504 | | 0.096 | 2(¥) | | | | | |
| • | | | ** SEMI-QUAN | TATIVE EMISSIG | IN SPEC RE | SUL | | | |
| | A.G. | | 19. | PPN(W) | AL | | 1000 | 0. | PPM(W) |
| | 2 A | < | 50. | PPM(¥) | B | | 10 | Ċ. | PPM(W) |
| | BA | | 502. | PPM(W) | BE | | 100 | 0. | PPM(W) |
| | BI | < | 50. | PPH(¥) | CA | > | 20000 | 0. | PPMEWS |
| - | CD | < | 1950. | PPHENI | 53 | < | 50 | 6. | PPM(¥)- |
| | CO | < | 57. | 72#(¥) | CR | | 50 | 0_ | PPM(¥) |
| | CS | < | 1000. | PPM(¥) | CU | | 50 | č. | PPM(X) |
| - | FE | | 50000. | PPMEND | .GE | < | 1 | e. | PP#{¥} |
| | HG | < | 19. | PPMEWS | K | | 4000 | 0. | PPK(W) |
| | LI | < | 1000. | PPR(¥) | RG | | 5000 | 0. | bbW(A) |
| - | PIN | | 503- | PPM(¥) | #0 | | ZC | 10. | PPM(W) |
| | NA | | 50000. | 222 CB3 | NB | < | 5 | -0- | PPM(¥) |
| | NI | | 1000. | PPMEWS | P | < | 100 | 10. | PPM(V) |
| • | P B | | 5J. | PPMENS | 9U | Κ. | 10 | | PPH(W) |
| | RB | < | 593. | PPMCW3 | 25 | < c | | 50_ | PPN(¥} |
| | 51 | | 100009. | ***** | | Ś | 1 | .U. | 446(X) |
| • | 2 R | | 10203. | 2224(¥) | 1 A 7 | Ś | |)U. | PPM(V) |
| | 15 | ς | 100. | PPS(W) | 4)* ** | Ś | うい | /V ↓ | ****** |
| | 11 | | 20J. 60) | 777 181 888/55 | 1 L V | | 21 | /V • 6 | |
| | U 1 | ~ | 787. 1344 | 66644.001 5544.001 | 7 7 k: | | £. | 2. | 22223 22223 22223 |
| | W 7 D | | e) Taqa• | 777 (W) 990761 | 21 | | 21 | JÛ ● | T P P L W L |
| | 2 K | | 79.0 | rr 1907 | | | | | |



.

A2 - 3

.

LABORATORY SAMPLE RESULTS

. -

-008396

** RADIOCHEMISTRY LABORATORY RESULTS

| AH | 0-00000546 | G/ G |
|----|------------|------|
| PU | 9520000.0 | G/ G |

U

0.000195 G/G

AUTHORIZED SIGNATURE

Acuters

| 5 | A | 4 | P | 1 | F | - 1 | D | 00 | • |
|----|-----|---|----|---|----|-----|----|----|---|
| ٤. | ÷4, | ч | T. | 5 | τ. | - 1 | υ. | | |

061472-01

B6-18

A2 - 4

| | 061472-01 · | | i.≰ | BORATORY SAMPLE 7412 Sludy | RESULTS E | | DATE | U4/10/80 PAGE 1 |
|----------|---------------------------------------|-------|-------------------------------------|-------------------------------|----------------------------------|---------------------|---|--------------------|
| - | SAMPLE-IO Entry date Completion | TAC I | 07-00839 11-01-79 12 03-10-80 | 7 | DJD N ACEDU Quild CIBSS | UMBE NT C ING | R 9703800 Harged 8037 559 SSPI | D |
| | CUSTOMER | | P. T. 50 | DESAIBOIS | | | | |
| | | | ** ATOMIC A | SORPTION SPECT | ROMETRY R | ESUL | TS | |
| | CA | | 121661- | PPHCWT | FE | | 49286. | |
| | G 🛦 | < | 50. | PPM(W) | MG | | 18377. | PPM(W) |
| - | ħ A | | 190179. | PPMENE | 21 | | 217. | ₽₽Ħ(¥) |
| | | | | CHEHISTRY LAB | ORATORY P | ESUL | TS | |
| | | | 1 6 | | 6 6 7 7 | | | |
| | | | 163 | 4(¥) Doment | 113= | | 0.59 | 7.1 W |
| | r (-) | | 1434 | ****** | H20 | | 50.2 | Z(W) |
| •• | NU3 | | 7.4 | 2(8) | 문민족 | < | 0.0025 | Z (¥) |
| | 204 | | 0.14 | 2183 | | | | |
| | | | ** SENI-QUAN | ITATIVE EMISSIO | N SPEC RI | ESULT | 2 | |
| | AG | | 40000. | PPMCWS | AL. | | 10000- | \$P\$(V) |
| • | A S | < | 50. | PDM(W) | . 8 | | 100. | PPM(V)- |
| | BA | | 5). | PPMCWI | 88 | | 1000. | PPM(W) |
| | BI | < | 50. | PPH (W) | CA | | 200000. | PPHEWS |
| - | CD | < | 1000. | PPMEWI | C E | < | 500_ | PPM(V)- |
| <u> </u> | C 0 | < | 5). | PPN(W) | CR | | 508 | PPHINI |
| | 23 | < | 1000. | PPM(W) | 2ม | | 1000. | PPM(V) |
| - | FE | | 50000. | PPMENA | GE | < | 10. | PPM(Y)- |
| | HG | < | 10. | PPMCWI | ĸ | - | 40000 | PPH(¥) |
| | LI | < | 1000. | PPHCWI | MG | | 100000. | PPH(V) |
| • | NN | | 132. | PPMENS | HC | | 200. | PPH(H)- |
| | NΔ | | 60003. | PPM (W) | NB | < | 50. | PPM(¥) |
| | NT | | 520. | PPMCW1 | P | < | 1000. | PPH(Y) |
| - | PB | | 50. | PPHEWE | PU | < | 100. | PPH(W)- |
| | RB | < | 533. | PPMEND | 25 | < | 50. | PPH(V) |
| | S 1 | | 100000. | PPK(W) | SN | < | 10. | PPM(W) |
| - | SR | | 10000. | PPMCWI | TA | < | 50. | PPH(W)- |
| | ŤE | < | 102. | PPM(W) | TH | < | 500. | PPMENT |
| | TI | - | 305. | PPMEWS | TL | < | 500. | PPR(¥) |
| | ป | < | 503. | PPM(W) | v | k | 5. | PPM(H). |
| | Ÿ | č | 1002. | PPMCWI | ZN | < | 500. | PPMINI |
| | ŽR | Ż | 5). | PPMIWI | | - | | |
| _ | • ·· | - | | • | | | | |



A2 - 5

.

LABORATORY SAMPLE RESULTS

PATE 04/10/RC PAGE 2

_

061472-01

SAMPLE-10 00-008397

** RADIOCHENISTRY LABORATORY RESULTS

0.000628

0.00000481 6/6

AM PU

MG/G G/G

0.000561 G/G

AUTHORIZED SIGNATURE

U

Hunes

A2 - 6

| Rockwell International There is a series of the state o | | | ALIA | |
|---|---|--|---|--|
| Except for the Grand Start Grand Start S | Rockwell Internation | nal | | ANALYTICAL REPOR |
| Account No. C.T. Hewitt 372 Account No. C.T. Hewitt 372 Account No. Approved Ap | Energy Systems Groun | | | |
| Control Course start C.T. Hewitt 374 Account No. 171 Date 7-24-81 Lab. No. M81-1109 File Remorted by Approved $(1.6.1.11)$ Hiller hubble Description 74 Waste Sludge - Dried studge characterization of the 374 waste sludge was requested. The malysis of a composited sample is given. All results are in 2. Ca 11 Mg 3.8 Si 5.8 Al 0.4 Cr 0.12 Fe 0.9 K 0.25 Ke 0.8 C 13 S 0.16 S0, 0.02 C1 1.3 F 0.5 P0, 40 N03 6.6 C13 5.04 HC03 0.33 The cations greater than 12 wre drivermined by A^2 and these lass han 12 by existing on a write and leach of the sludge. Eighteen percent of the sludge was soluble in water, and 362 soluble in NET of the sludge was soluble in water, and 362 soluble in A2-7 B6-21 | Rocky Flats Plant | | | |
| C.T. Hewait 374 Account No. Tile Disc $7-24-81$ Lab. No. MB1-(109 Reported by Approved $(167.75.1000)(160.1$ | Griden, Colerente 80401 | | | |
| C.T. Hewite 3744 File Reported by Approved <u>(1): Miller</u> A.H. Hiller A.H. Hil | / | Account No. | | Date Lab. No. |
| Reported by Approved $(A, B, B,$ | G.T. Hewitt 374 | | 371 | 7-14-81 M81-1109 |
| Approved <u>Approved</u> <u>A.H. Hiller</u> Aver Hiller Aver Hi | File | | | Reported by |
| Approved Approved Approved A.M. Hiller A.M. Hiller A.M. Hiller A.M. Hiller A.M. Hiller A.M. Hiller Approved Approved A.M. Hiller Approved A.M. Hiller Approved Approved A.M. Hiller Approved Approved Approved A.M. Hiller Approved Approved Approved A.M. Hiller Approved Approved Approved Approved A.M. Hiller Approved | | | | |
| note Description To Waste Sludge - Dried studge avsis flexitis characterization of the 374 waste sludge was requested. The nalysis of a composited sample is given. All results are in 2. Ca 11 Mg 3.4 Si 5.8 A1 0.4 Cr 0.12 Fe 0.9 K 0.25 Na 0.8 C 13 S 0.36 S0 0.02 C1 1.3 F 0.5 P04 40 NO3 6.6 CO3 0.04 NO3 6.6 CO3 0.04 NO3 0.33 The cations greater than 12 were determined by A^2 and these less than 15 by greater to sharpy. The amons, except for HCO3. CO3. and NO3 were determined on a mitrin and leach of the sludge. Zigfteen percent of the sludge was soluble in water, and 362 soluble in nitrin anid. A2-7 B6-21 | | | | Approved |
| note Description 74 Waste Sludge - Dried sludge alvess Results characterization of the 374 waste sludge was requested. The nalysis of a composited sample is given. All results are in 2. Ca 11 Mg 3.4 Si 5.8 Al 0.4 Cr 0.12 Fe 0.9 K 0.25 Ha 0.8 C 13 S 0.36 S0 0.02 C1 1.3 F 0.5 F0 40 NO3 6.6 C3 0.04 MCO3 0.33 The cations greater than i2 were determined by A^2 and these less than 12 by emission spectroscopy. The among, except for HCO3, CO3, and NO3 were determined on a mittin arid leach of the sludge. Zigfteen percent of the sludge was soluble in water, and 362 soluble in nitric arid. A2-7 B6-21 | | | | christin Tillly |
| The vasie Sludge - Dried studge alvest Results characterization of the 374 waste sludge was requested. The nalysis of a composited sample is given. All results are in 2. Ca 11 Mg 3.8 Si 5.8 Al 0.4 Cr 0.12 Fe 0.9 K 0.25 Na 0.8 C 13 S 0.36 S0 0.02 Cl 1.3 F 0.5 P0 40 MC3 6.6 CG 3 5.04 HCC3 0.13 The cations greater than i2 were determined by A^2 and these less than 12 by emission spectroscopy. The anions, except for HCO3, CO3, and NO3 were determined on a nitit arid leach of the sludge 3 Eighteen percent of the sludge was voluble in water, and 362 voluble in nitric arid. A2-7 B6-21 | | | ana an | A.M. Niller |
| The vasue Sludge - Dried shudge invest Results characterization of the 374 waste sludge was requested. The malysis of a composited sample is given. All results are in 2. Ca 11 Mg 3.a St 5.8 Al 0.4 Cr 0.12 Fe 0.9 K 0.255 Na 0.8 C 13 S 0.4 C1 1.3 F 0.5 P0.4 40 NO3 5.6 C3 0.04 NC3 5.6 C3 0.3 The cations greater than it were determined by A^2 and those less than 12 by emission spectroscopy. The anions, except for HC03. C03. Ind NO3 were determined on a mitter and leach of the sludge. Eighteen percent of the sludge was soluble in water, and 362 soluble in Shirt arid. A2-7 B6-21 | npie Description | | | |
| average of the state of the state studge was requested. The naises of a composited sample is given. All results are in Z. $ \begin{array}{c} Ca & 11 \\ Mg & 3.8 \\ Si & 5.8 \\ Al & 0.4 \\ Cr & 0.12 \\ Fe & 0.9 \\ K & 0.25 \\ Ma & 0.8 \\ C & 13 \\ S & 0.36 \\ SO_{0} & 0.02 \\ Cl & 1.3 \\ F & 0.5 \\ PO_{4} & 40 \\ NO_{3} & 6.6 \\ CO_{3} & c.04 \\ HCO_{3} & 0.33 \\ \end{array} $ The cations greater than if were determined by A^{2} and these less than 12 by emission spectroscopy. The anions, except for HCO_{3}, CO_{3} and NC_{6} were determined on a nitive arid leath of the studge. Eighteen percent of the studge was sniuble in water, and 362 soluble in nitrie arid. $ \begin{array}{c} A2-7 \\ B6-21 \\ \end{array} $ | 74 Waste Sludge | - Dried | t she | د |
| where determined on a minim and leach of the sludge was soluble in water, and $3b_2$ soluble in 2.6 $M_{1} = 10$ $M_{2} = 11$ $M_{2} = 3.8$ $M_{1} = 0.4$ $C_{1} = 0.12$ $F_{1} = 0.23$ $M_{2} = 0.23$ $M_{3} = 0.8$ $C_{1} = 1.3$ $F_{1} = 0.5$ $PO_{4} = 40$ $NO_{3} = 6.6$ $CO_{3} = 0.04$ $HCO_{3} = 0.31$ The cations greater than i2 were determined by A^{2} and these less than 12 by emission spectroscopy. The anions, except for HCO_{3} , CO_{3} , and NO_{3} were determined on a minim and leach of the sludge. Eighteen percent of the sludge was soluble in water, and $3b_{2}$ soluble in minimaria. A2-7 B6-21 | • | | J | |
| Harvest Hesuits a characterisation of the 374 waste sludge was requested. The inalysis of a composited sample is given. All results are in Z. Ca II Hg 3.6 Si 5.8 Al 0.4 Cr 0.12 Fe 0.9 K 0.23 Na 0.8 C 13 S 0.16 S0 0.02 Cl 1.3 F 0.5 P0 40 NO3 5.6 CO3 0.04 NCO3 0.03 The cations greater than iZ were determined by A^2 and these less than 1Z by emission spectroscopy. The anions, except for HCO3, CO3, and NO3 were determined on a minist and leach of the sludge, SigPreen percent of the sludge was soluble in water, and 302 soluble in nitric acid. A2-7 B6-21 | ومعتقوبية عبية المتناولية ومعاودتها فستنهيه | و د | يوهير ديار ماريك | |
| characterization of the 374 waste sludge was requested. The imalysis of a composited sample is given. All results are in Z. | ISIAZIZ MGZUILZ | • | | |
| the cations greater than if were determined by A^2 and those less than R_3 by emission of a composited sample is given. All results are in f_1 . Ca 11 Mg 3.4 Si 5.8 Al 0.4 Cr 0.12 Fe 0.9 K 0.25 Na 0.8 C 13 S 0.36 S0.4 0.02 Cl 1.3 F 0.5 P0.4 40 NO3 6.6 CO3 0.06 HCO3 0.33 The cations greater than if were determined by A^2 and those less than R_3 were determined on a minimarial leach of the sludge. Eighteen percent of the sludge was soluble in water, and 362 soluble in mitrir arid. A2-7 B6-21 | | | | |
| malysis of a composited sample is given. All results are in 7. Ca 11 Ng 3.8 Si 5.8 Al 0.4 Cr 0.12 Fe 0.9 K 0.25 Na 0.8 C 13 S 0.36 S04 0.02 Cl 1.3 F 0.5 P04 40 N03 6.6 C03 0.04 N03 0.35 The cations greater than i2 were determined by A^2 and these less than 12 by emission spectroscopy. The amions, except for HCO_3 . CO ₃ , and NO ₃ were determined on a mitrix acid leach of the sludge. EigPreen percent of the sludge was soluble in water, and 362 soluble in nitrir acid. A2-7 B6-21 | characterization | of the 3 | 74 waste | sludge was requested. The |
| Ca 11 Hg 3.8 Si 5.8 A1 0.4 Cr 0.12 Fe 0.9 K 0.25 N= 0.8 C 13 S 0.36 S0.4 0.02 C1 1.3 F 0.5 P0.4 40 N0.5 6.6 C3 0.04 nc0.3 0.04 nc0.3 0.04 nc0.3 0.03 The calions greater than iZ were determined by Λ^2 and those less than 1% by emission spectroscopy. The anions, except for HC0.4 CO.4 and NO.5 were determined on a mitrix acid leach of the sludge. Sighteen percent of the sludge was soluble in water, and 36% soluble in mitrix acid. A2-7 B6-21 | nalysis of a compo | sited same | mple is p | given. All results are in %. |
| The cations greater than iZ were determined by A^2 and these less than IL by emission spectroscopy. The anions, except for HCO ₃ , CO ₃ , and NO ₃ were determined on a minim and leach of the sludge. Eighteen percent of the sludge was soluble in water, and 362 soluble in mitrir anid. A2-7 B6-21 | | - | | |
| $ \begin{array}{c} \mathbf{x} \mathbf{y} \mathbf{y} \mathbf{z} \mathbf{z} \mathbf{z} \mathbf{z} \mathbf{z} \mathbf{z} \mathbf{z} z$ | | | | |
| At 0.4 Cr 0.12 Fe 0.9 K 0.25 Na 0.8 C 13 S 0.36 SO ₄ 0.02 Ct 1.3 F 0.5 PO ₄ 40 NCO ₃ 6.6 CO ₃ 0.04 NCO ₃ 0.33 The cations greater than iZ were determined by Λ^2 and these less than 1Z by emission spectroscopy. The anions, except for HCO ₃ , CO ₃ , and NO ₃ were determined on a mitrix acid leach of the sludge. Eighteen percent of the sludge was soluble in water, and 3bZ soluble in nitric acid. A2-7 B6-21 | • | 5 i | 3.8 5.8 | •• |
| $\frac{Cr}{Fe} = 0.12$ Fe = 0.9 K = 0.25 Na = 0.8 C = 13 S = 0.36 SO ₄ = 0.02 Cl = 1.3 F = 0.5 PO ₄ = 40 NO ₃ = 6.6 CO ₃ = 0.04 NCO ₃ = 0.33 The cations greater than iZ were determined by A ² and those less than 1% by emission spectroscopy. The anions, except for HCO ₃ , CO ₃ , and NO ₃ were determined on a mitrix acid leach of the studge. Eighteen percent of the studge was soluble in water, and 362 soluble in mitrix acid. A2-7 B6-21 | | A1 | 0.4 | |
| Fe 0.9 K 0.25 Na 0.8 C 13 S 0.36 S04 0.02 C1 1.3 F 0.5 P04 40 N03 6.6 C03 0.03 The cations greater than i2 were determined by Λ^2 and those less than 1% by emission spectroscopy. The anions, except for HC03, C03, and N03 were determined on a mitrix acid leach of the sludge. Eighteen percent of the sludge was soluble in water, and 362 soluble in nitric acid. A2-7 B6-21 | | Cr | 0.12 | |
| K 0.25 Na 0.8 C 13 S 0.36 S0 0.02 Cl 1.3 F 0.5 P04 40 N03 6.6 C03 0.04 HCO3 0.33 The cations greater than iZ were determined by Λ^2 and these less than 1Z by emission spectroscopy. The anions, except for HCO3, CO3, and NG3 were determined on a mitrix acid leach of the sludge. ³ Eighteen percent of the sludge was soluble in water, and 362 soluble in mitrit acid. A2-7 B6-21 | - | Fe | 0.9 | |
| Na C.8 C 13 S 0.36 SO ₄ 0.02 Cl 1.3 F 0.5 PO ₄ 40 NO ₃ 6.6 CO ₃ 0.04 NCO ₃ 0.33 The cations greater than 12 were determined by A^2 and those less than 12 by emission spectroscopy. The anions, except for HCO ₃ , CO ₃ , and NO ₃ were determined on a nitrit acid leach of the sludge. Eighteen percent of the sludge was soluble in water, and 362 soluble in nitrit acid. A2-7 B6-21 | | K | U.25 | |
| $C = 13$ $S = 0.36$ $SO_4 = 0.02$ $C1 = 1.3$ $F = 0.5$ $PO_4 = 40$ $NO_3 = 6.6$ $CO_3 = 0.04$ $HCO_3 = 0.33$ The cations greater than iZ were determined by A ² and these less than 1% by emission spectroscopy. The anions, except for HCO_3, CO_3, and NO_3 were determined on a mitrix arid leach of the sludge. Eighteen percent of the sludge was soluble in water, and 362 soluble in mitrix arid. $A2-7$ $B6-21$ | | NA | 8.0 | |
| S 0.36 SO ₄ 0.02 Cl 1.3 F 0.5 PO ₄ 40 NO ₃ 6.6 CO ₃ 0.04 HCO ₃ 0.33 The cations greater than iZ were determined by Λ^2 and these less than 1% by emission spectroscopy. The anions, except for HCO ₃ , CO ₃ , and NO ₂ were determined on a mirrir acid leach of the sludge. Eighteen percent of the sludge was soluble in water, and 36% soluble in mitric acid. A2-7 B6-21 | | C | 13 | |
| $SO_4 = 0.02$ C1 = 1.3 F = 0.5 PO_4 = 40 NO_3 = 6.6 CO_3 = 0.04 HCO_3 = 0.33 The cations greater than iZ were determined by A ² and these less than 1% by emission spectroscopy. The anions, except for HCO_3, CO_3, and NO_3 were determined on a minim and leach of the sludge. Eighteen percent of the sludge was sninble in water, and 36% soluble in nitric acid. A2-7 B6-21 | | S | 0.36 | |
| F 0.5 P04 40 N03 6.6 C03 0.04 HC03 0.33 The cations greater than 12 were determined by Λ^2 and these less than 12 by emission spectroscopy. The amions, except for HC03, C03, and N03 were determined on a mitric acid leach of the sludge. Eighteen percent of the sludge was soluble in water, and 362 soluble in mitric acid. A2-7 B6-21 | | so ₄ | 0.02 | |
| Po ₄ 40 NO ₃ 6.6 CO ₃ 0.04 HCO ₃ 0.33 The cations greater than 12 were determined by Λ^2 and these less than 12 by emission spectroscopy. The anions, except for HCO ₃ , CO ₃ , and NO ₃ were determined on a minic acid leach of the sludge. ³ Eighteen percent of the sludge was soluble in water, and 362 soluble in mitric acid. A2-7 B6-21 | | L L F | 1.3 | |
| NO ₃ ⁴ 6.6 CO ₃ 0.04 HCO ₃ 0.33 The cations greater than iZ were determined by Λ^2 and those less than 1% by emission spectroscopy. The anions, except for HCO ₃ , CO ₃ , and NO ₃ were determined on a mitrix acid leach of the sludge. Eighteen percent of the sludge was soluble in water, and 362 soluble in mitric acid. A2-7 B6-21 | | PO. | 40 | |
| $\frac{co_3^3}{3} = 0.04$ $\frac{nco_3}{3} = 0.33$ The cations greater than 12 were determined by A ² and these less than 12 by emission spectroscopy. The anions, except for HCO ₃ , CO ₃ , and NO ₃ were determined on a mitrix acid leach of the studge. Eighteen percent of the studge was soluble in water, and 362 soluble in a nitrix acid. $A2-7$ B6-21 | | NO. | 6.6 | |
| $\frac{1000}{1000}$ 0.33 The cations greater than iZ were determined by Λ^2 and these less than 1% by emission spectroscopy. The anions, except for HCO ₃ . CO ₃ , and NO ₃ were determined on a mitric acid leach of the sludge. Eighteen percent of the sludge was soluble in water, and 302 soluble in nitric acid. A2-7B6-21 | | ເດິ | r 0/ | |
| The cations greater than iZ were determined by Λ^2 and these less than 1% by emission spectroscopy. The anions, except for HCO ₃ , CO ₃ , and NO ₃ were determined on a mitric acid leach of the sludge. Eighteen percent of the sludge was soluble in water, and 36% soluble in nitric acid. A2-7 B6-21 | | כ | 0.33 | |
| The cations greater than if were determined by A^2 and these less than 1% by emission spectroscopy. The anions, except for HGO_3 , GO_3 , and NG_3 were determined on a nitric acid leach of the sludge. Eighteen percent of the sludge was soluble in water, and 36% soluble in nitric acid. A2-7 B6-21 | | | | |
| than 1% by emission spectroscopy. The anions, except for HCO ₃ , CO ₃ , and NC ₃ were determined on a mitrix acid leach of the sludge. Eighteen percent of the sludge was soluble in water, and 36% soluble in nitric acid. A2-7 B6-21 | | 1003 | | 1 |
| and NG, were determined on a unifficial leach of the sludge. Eighteen percent of the sludge was soluble in water, and 36% soluble in mitric acid. A2 - 7 B6-21 | The cations greate | r than if | Z wore de | ecormined by Λ^2 and these less |
| A2 - 7 B6-21 | The cations greate than 1% by emission | r than if n spectre | Z unte da oscopy. | ecommined by Λ^2 and these less The autons, except for HCO ₃ , CO ₃ , |
| A2 - 7 B6-21 | The calions greate than 1% by emissio and NG3 were deter | r than is n spectre mined on | Z were da oscopy. - a mittio | evermined by Λ^2 and these less The anions, except for HCO_3 , CO_3 , a avid leach of the sludge. Eighteen is water and 367 soluble in |
| A2 - 7 B6-21 | The calions greate than 1% by emissio and NG ₃ were deter percent of the shu nitric acid. | r than 1 n spectre mined on dge was | Z шого da обсорут – а віттіс боїобіс | ecormined by A ² and these less The anions, except for HGO ₃ , GO ₃ , a acid leach of the sludge. Eighteen in water, and 36% soluble in |
| A2 - 7 B6-21 | The cations greate than 1% by emission and NG ₃ were deter percent of the shu nitric acid. | noo3 r than if w spectro mined on dge was | Z were da oscopy. a nitrit soluble | ecormined by A ² and these less The anions, except for HCO ₃ , CO ₃ , a acid leach of the sludge. Eighteen in water, and 362 soluble in |
| A2 - 7 B6-21 | The cations greate than 1% by emission and NO ₃ were deter percent of the slu nitric acid. | noog r than i u spectro mined on idge was | Z were da oscopy. a mitrie sniabie | ecormined by A ² and those less. The autons, except for HCO ₃ , GO ₃ , a acid leach of the sludge. Eighteen in water, and 36% soluble in |
| A2 - 7 B6-21 | The cations greate than 1% by emission and NG3 were deter percent of the shu nitric acid. | noo3 r than if w spectro mined on dge was | Z wore da oscopy. a nitrio soluble | ecormined by A ² and those less The autons, except for HCO ₃ , CO ₃ , actid leach of the sludge. Eighteen in water, and 36% soluble in |
| A2 - 7 B6-21 | The cations greate than 1% by emission and NG ₂ were deter percent of the slu nitric arid. | noo3 r than i u spectro mined on idge was | Z were da oscopy. a mitrig soluble | ecormined by Λ^2 and those less The anions, except for HCO ₃ , GO ₃ , a acid leach of the sludge. Eighteen in water, and 362 soluble in |
| A2 - 7 B6-21 | The cations greate than 1% by emission and NG ₃ were deter percent of the shu nitric acid. | noog r than if or spectro mined on idge was | Z where do oscopy. a withit shiubit | ecormined by A ² and those less The autons, except for HCO ₃ , CO ₃ , actid leach of the sludge. Eighteen in water, and 36% soluble in |
| B6-21 | The cations greate than 1% by emission and NG ₃ were deter percent of the shu nitric acid. | noo3 r than if u spectro mined on dge was | Zurre da oscopy. a niti in soinbie | ecormined by A ² and those less The autons, except for HCO ₃ , GO ₃ , a acid leach of the sludge. Eighteen in water, and 36% soluble in |
| B6-21 | The cations greate than 1% by emission and NG ₃ were deter percent of the slu nitric acid. | noo3 or than i ou spectro mined on odge was | Zurredi oscony. a nitrio sniubie | A2-7 |
| | The cations greate than 1% by emission and NG3 were deter percent of the sim nitric acid. | noo3 er than if en spectro mined on idge was | Zurre da oscopy. a nitrio sniubie | A2-7 |

i

8801 , 4 [inqA 100021996 (200) Golden, Colorado 60402-0464 ISROUSFIEIRI 151 108 Oc nonavogred insentantal liewacon **Hewsoor** enoliamyQ esens avshemA Anoit PACEY FIRM PLACE ATTACHMENT 3 W 1 P P - 0 1 04 - 0 C 0 R - 0 1 9 2

Contractor to U.S. Department of Energy

88-RF-1089

PAGE ·

BAGE

-<u>10</u>

DOE, RFAD тэрьпьМ ветА nsmstidW .3 trsdfA

ENGINEERING PARAMETERS FOR ROCKY FLATS WASTE FORMS

This information is for the attention of W. C. Rask.

TRUPACT-II container. thirteen Rocky Flats waste forms, which will be transported in the distribution, dated March 1, 1988. Information is included for all that were requested in the letter from J. S. Tollison to Attached are the engineering parameters for Rocky Flats waste forms

With your approval please forward to DOE/AL. Waste Transportation. Ji⊞ Alexander at (303) 966-7585 or Jeff Faynter at (303) 966-5252. If you have questions regarding the enciosed information, contact

Rerespace Operations Rocky Flats Plant Vaste Operations TSEAR ANDAISN .A .3

.on3

Orig. and 3 cc - A. E. Whiteman

| · | · - |
|-------------|--|
| | |
| | 100 - 17CL - 23C |
| <u> </u> | |
| - ZX | |
| | 3140 |
| 107 | 2011 h |
| 121 121C | VINT CLASSING |
| | SECHET SOMPTOLIAL |
| ম্ | CINSSING |
| -1-1- | |
| # | |
| | |
| | |
| সমি | HICKI- FT |
| · X | VI JYHNGX |
| | CONNES' CONTROL |
| | |
| | ······································ |
| ! ==== | WI ZBOOSVIJA |
| · • | |
| ; === | 716 4844391 |
| | HEARDA' H'T |
| | TO ADDA HT ADDA HT TO ADDA EV TO ADDA EV TO ADDA EV TO ADDA EV MINE C OT |
| | HEMBA BT INVINOUT EY TOTOENDING OF SHITE OT HOLEY OT |
| | Та 2000 година и 1000 година и 10000 година и 1000 година и 1000 година и 1000 година и 1000 година |
| | HEMBA BT INVINOUT EY TOTOENBRUC DE MARGE DTV HOENTER HOENTER HOENTER HOENTER HOENTER HOENTER |
| | HENRY BY HENRY BY TOTOENDING OF HENRY TY HOLE OT HOLE TY HOLE OT HOLE OT HOLE TT HOLE OT HOLE OT HOL |
| | HEMBA BT INVINOU EY INVINOU EY FINE BA FINE BA HOLLITE HOLLITE HOLLITE HOLLITE HOLLITE HOLLITE HOLLITE HOLLITE HOLLITE HOLLITE |
| | |
| | |
| | |
| | INTERPORT OF CONTROL O |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |

CH 2,10NIDDICD TOBINOD SEECO

us i line terrer de

I - 5∀

ENGINEERING FARAMETERS FOR TRUPACT-II

Waste Stream - - TRU SOLIDIFIED ORGANIC WASTE (WF-112)

For data in Section 1, Secondary Container, and Section 2, Arrangement of Secondary Containers, see the General Engineering Farameters for TRUFACT II.

J WASTE MATERIAL INFORMATION:

- 7.1 <u>Structural:</u>
- 5.1.1 Maximum and Minimum Weight -

Drums: 750 lb max. / 530 lb avg. / 200 lb. min. (including the weight of the drum)

- 3.1.2 <u>Acceptable Projectile Envelope</u> - NA, solid monolith cast in the liner inside the drum.
- 3.2 <u>Thermal:</u>

1

3.2.1 <u>Quantity of Radionuclides</u> - - Isotopic Composition (Mix Group 9, TRUPACT-II Spec.):

| <u>Isotope</u> | Fraction |
|----------------|----------|
| Pu-238 | TRACE |
| Pu-239 | 0.930 |
| Pu-240 | 0.058 |
| Pu-241 | 0.004 |
| Pu-242 | TRACE |
| Am-241 | TRACE |
| OTHER | 0.007 |
| | |

Max. radionuclides (Weapons Grade Pu): 200 grams/drum Maximum decay heat (Pu): 0.4 watts/drum (Am): 0.3 watts/drum Total: 0.7 watts/drum

| 5.2.2 | <u>Chemical Form</u> | | <u>min.</u> | | <u>Max.</u> | | <u>ave.</u> | |
|--|---|-----|-------------|----|-------------|-----|-------------|--|
| | oils | 10 | % | 30 | % | | | |
| | trichloroethane and trichlorotrifluoroethane | 5 | 7. | 10 | 7. | | | |
| | carbon tetrachloride emulsifier (a polvethyl | 2 | % | 5 | 7 | | | |
| | glycol ester) | 5 | 7. | 10 | 7. | | | |
| | water | 5 | 7. | 15 | 7. | | | |
| gypsum cement total liquid (32 gallon | gypsum cement | 40 | %. | 50 | 7 | 200 | 16 | |
| | A3 - | . 2 | | | 250 | 16 | | |





memorandum

Carlsbad Area Office Carlsbad, New Mexico 88221

DATE:

τ

April 4. 1996

ATTN OF: CAO:NTP:DW:96-1126

SUBJECT:

Estimate of Cement Content in TRU Solidified Waste Forms Scheduled for Disposal in WIPP

TO:

Les Shephard, Director, SNL

Attached is a summary of the best estimate of portland cement in stored and projected volumes of solidified waste streams listed in Revision 2 of the Transuranic (TRU) Waste Baseline Inventory Report (TWBIR). This information was requested from the TWBIR team in support of the Performance Assessment team.

These values have been scaled (similar to the methodology used for waste material parameters in the TWBIR) to the full volume of the Waste Isolation Pilot Plant (WIPP) repository. The total estimated weight of portland cement in these scaled solidified waste forms is 8.54E+06 kg. Dividing this value by 6.2E+06 ft³ (-175,600 m³), the maximum capacity of WIPP, yields a portland cement density in the overall combined contact-handled (CH) and remote-handled (RH) transuranic (TRU) waste of 48.6 kg/m³. The portland cement reported is both reacted and unreacted cement in the waste. There are no data available to estimate the percentage of reacted versus unreacted cement.

The basic methodology was to perform a sort of the Revision 2 database that supports the TWBIR for all Solidified Inorganic and Solidified Organic waste streams. This sort resulted in 221 waste streams. Some waste streams were eliminated from further consideration for the following reasons:

- Data about most Rocky Flats waste streams (both residue and nonresidue waste streams) are for waste in current form only and not in final form. The item description code (IDC) for many particulate waste streams will change to final form because the waste is in a cemented final form. A total of 91 current-form RF TRU waste streams were eliminated because of this constraint. (the final form of these waste streams, however, is included in the portland cement estimate.)
- The Solidified Inorganic waste streams listed from Savannah River Site are all vitrified and therefore do not contain any portland cement. A total of 20 waste streams were eliminated because of this constraint.


Les Shephard

If you have any questions concerning the attached information, please contact Mr. Russ Bisping of my staff at (505) 234-7446.

Don Watkins Manager National TRU Program

Attachment

cc w/attachment: M. McFadden, CAO K. Hunter, CAO R. Bisping, CAO P. Drez, CTAC J. Harvill, CTAC L. Sanchez, SNL M. Chu, SNL M. Marietta, SNL

Calculation Summary

At the bottom of Table 1 the total kilograms of portland cement is summarized for CH-TRU and RH-TRU waste for both stored plus projected waste (in "Total kg" column) and projected only waste (in "Projected kg" column). The TOTAL SCALED portland cement is calculated as follows:

CH-TRU "Total kg" + 2.05 * CH-TRU "Projected kg" + RH-TRU "Total Kg" = TOTAL SCALED kg of portland cement, or

5.28E+06 + 2.05(1.34E+06) + 5.05E+05 = 8.54E+06 kg portland cement

The total density of portland cement is calculated as follows:

 $8.54E + 06 \text{ kg}/175,600 \text{ m}^3 = 48.6 \text{ kg}/\text{ m}^3 \text{ portland cement}$



Table 1. Estimate of Portland Cement in TRU Waste for Disposal in the WIPP

| A STATE OF THE PROVEMENT OF STATES | BE FETCER LET WE PURE DET DE | | Sell diffication | _Commt (hg/m) | Stared (m) | Projected (m) | Total kg | Projected by | |
|------------------------------------|--|---------------|---------------------------------------|-----------------|--|--|--|--------------------|---|
| WOIZ 10 MIRY | | [POISON CE | All(11 | ! /? | | | | | |
| | RII Solidified | norgenici | vaporitic sal/dudze(3) | | <u></u> |].8 | 261.16 | 131:30 | |
| W012.12 NIRU | RII | aarganice | per ge/particuletet(3) | | 0 | Q.2 | | 61.79 | |
| (316-127 MIRH | Rij Solidiged | 001800 St | · · · · · · · · · · · · · · · · · · · | | | 81 | | X | |
| /228,856 MIKU | nii | IDOT SANICE | · · · · · · · · · · · · · · · · · · · | | 19 <u>}</u> 3} | | | · 8 | |
| VOI6 MIRU | Ris | Laarganice | | 397 7 | 18]]] | 206.5 | | E1197 2 | |
| /317, 1029 MIRU | Ril | Premica | entra(3) | | | | | | |
| YOJA MERU | Ci Soldile | antenica | Cid int DACKIUS | | ······································ | ¥ | | | |
| LOOS NITRI | Cil Solidified | Inorganice 1 | ticic acid liquide(1) | | | 0.4 | | | |
| 146 699 MIRU | CII Solidilles | 1091 Stoles | asymped IN-W216.98 | 271. | 2.3 | 9 | <u>\$72.74</u> | ¥ | |
| 159 1072 MIRY | | 2 tointgion | Autilic Hallandta | | 0.7 | | 31.12 | ō | |
| | Chi Solidifio | | | | | 0 | | 8 | |
| 166 928 MITRI | CII Solidified | norgenics | 18 | 12.6 | <u>56.8</u> | | | V | |
| 174.1002 MIRN 1 | CII Solidified | Inorgenics | | 0(9) - n/al | [| | | | |
| 174 154 NITRI | | Ingrennist | | | | | | | |
| 177 156 NIRU | C11 Solidified | Inorgenist iE | Qrco(4) | | | <u> </u> | | ñ | |
| 19. 1011 NITRI) | Cil Solidified | norgenice | J <u>R(?)</u> [· | | | <u>X</u> I · | !******** | ŏ | |
| 179 ISU MITRIA | CH | 19918ANIST | K 22 | 308.4 | | | 2964 4 | Q | |
| | Solidified | Inorganica A | saumed [N-W222.116 | 102 | | | | <u>8</u> | |
| MIRU C | Solidicied | norgenice | 8 | | | | | | |
| 111 160 MIRU | CH Solidilled | laerenice | S | | | ă l | 436684 32 | 9 | |
| 16 875 MIBH | n in in it is in it is i | Ingreenice B | | | | Q . | 1}22222-2291- | Į | |
| JI 909 NIRI | Soliditied | notgenics A | sevmed RE-M10007 | /]] [] | <u>]</u> []; <u>-</u>] | | ¹ ₹} <u>8</u> 1 · <u>1</u> ₹ · | ŏ | |
| 219 914 MIRI | Splidified | Ingranica | India 21448c(1) | | | č i | | Ó | |
| | | laoseanice B | | 301,1 | | § - | | § | |
| Silii Miru | Solidilled | Botganics | 8(6) | | | | | Š | |
| 221 927 MIRI | | narsanice | B(<u>•</u>) | <u></u> <u></u> | | ð _ | | ĝ | |
| | ii Solidilled | laurganica Di | R | | | 0 - | | 9 | |
| Si ioi MIRU C | Solidified | norganica D | B | | | X - | | Xi · | |
| | Solidified | norganice DI | R astidification used | | | ·¥[- | | ¥ | |
| 247 523 MIRU | | nortanice iDi | atomaceous earth(1) | D/9 | | ······································ | | | [|
| 263 520 MIRU C | Solidilied | notsenics ing | solidilication used | | | a - | | | |
| TIS 601 NITRU | 1 | narsanise | 2 301 Assumed INW 219. 172 | <u></u> | | ¥ - | | / | í |
| 111 661 IBV C | | norganica (Pi | 8951181 | | | | | | |
| -114 IGi 1611 E | Solidified | not genics | reniculite(2) | | | | [- | [- | |
| 'HI II. HBU | 11 | aareanke | miculke(?) | | | | - 181-131- | | |
| 344 1912 . IBN | II | nargnaks | avid(3) | <u></u> | 8.51- | | | Į. | |
| 111 JIF HRU | Solidified | 801888 CE | | | 9]_ | Q | | Ŷ]- | |
| 375 1096 JRU | Solidilied | norganica Si | dge-Assumed IN-W216.875 | <u>JU</u> , | | | | - 16 Sibili | |
| | 11 | norganice | <u>{</u> | | 1075 46151- | | 1038363.6681- | 181132.681 | |
| VOOG NERI C | Soldified | | | | <u> </u> | | 10016.2361 | | |
| 006 TRU C | il Solidified | nerganke | b | | 4,228 | !]: <u>{</u>] | <u>410%, 94</u> | 4122 <u>3,6</u> _ | |
| nai ibil | II Solidified | anreanice (10 | 2) | (901 | | | | | |

}

and the second second

.... . .

Table 1. Estimate of Portland Cement in TRU Waste for Disposal in the WIPP

)

)

| 12 .8 111 | HA HAR Y COLUMN FILM A | | 181 191 2 11 19 | NET SERVICE | Solidification | Comment (ka/m²) | Stored (m) | Profested (m ²) | Taske | Professed La | - <u> </u> |
|---------------|------------------------|--------------|-----------------|----------------------|--|-------------------|--|-----------------------------|--------------|----------------|------------|
| (00) | TRU | | Solidille | norgenist | Sludge - Assumed IN.W 179. 1014 | 391. | 13 | 0 | 1655 6 | | õ |
| /021 | TRU | cii | Solidilied | norganics | Mainty debris | | | | [| | |
| 110001 | MIBU | CII | Solidilled | luarganics | | 117 | 11 | Q | 691 12 | · | ō |
| 10007 | | | | noi 800 kB | | | I ?]] | | | | 9 |
| 10001 | MIRU | ĊH | Solidified | Inorganice | | | <u> </u> 2} | | | 47 }; | á l |
| 10107 | MIBU | | Solidified | nargenice | | · 11.1 | 73.3 | 41.6 | 1745.5 | 2212 7 | 2 |
| 10806 | ATRU . | | | Ingistanica | | | | | <u></u> | | §} |
| 19823 | TRU | či | Solidified | Ingranice | | | 7 | | 528.5 | | |
| [0806 | TRU | <u>CII</u> | Solidified | Inorganics | ·· B B | | | 202.2 | 14789.92 | 14780.1 | 2 |
| 10802 | | či – | Solidified | Ingreenice | BIR(6) | 68.32 | ¥ | | | 1778 3 | <u> </u> |
| 2#1 | NITRU | ĉij | Solidified | Inerganice | Vermiculite(2) | | | | | | |
| | IRU | CH | Solidified | Inorganice | Vermiculite(2) | D(\$ | | | | | |
| 039 | MIRU | či | Solidified | Organica | resins(3) | 73.1 | 0.2 | Ō | 14.62 | | j |
| 157 144 | MIRI | <u>cii</u> | Solidified | Oxganica | PR(6) | | | ĝ | 2313,7 2 | | |
| 151 906 | MIRU | <u>CH</u> | Solidified | Organice | BIR(6) | 217. <u>10</u> 8 | 103.7 | | _24830'1510 | | · |
| 164 153 | MIRU | čil | Soliditied | Digenice | Environtonel) | | | | | | |
| 197 149 | MIBH | <u>cii</u> | Solidified | Prenist | | 0/9 | | | · | | |
| 109 510 | ATTRU | SI - | Solidified | Dreenice | calc-silicate(12) | U!!! p/a | / / / / / / / / / / / / / / / / / | | | | ····· |
| 111 151 | MIRU | çii | Soliditied | Digenica | [[seins(]] | | | Q | 2434 211 | | [] |
| _117 758 | MIRU | CH | Solidified | Prganice | [[[]]]]]]]]]]]]]]]]]]]]]]]]]]]]]]]]]]] | | | | | · X | I I |
| | MIRU | | Solidified | Dreenics | resins(3) | | | Š | 5 61 | Ž | |
| άų. | TRU | | Solidilied | Zigen ci | (10) | 691 | | | 21205.8 | 20166.3 | |
| ,19 1000.1 | MTRU | | Solidified | Trganice | Envirostonel | <u>0/8</u> | | | | | |
| 10801 | MIRU | | Solidified (| zrgenice | Envicostone(11) | n/e | | | | | |
| t0113 | MIRU (| | Solidified (| Arganics | | | | | | 1172.1 | |
| 200 | ATŘÍ C | · · · · · · | Solidified (| Zrganica Dreanica | pin (14) | | ¥[| | 1149:3 | 1 # 49:1 | |
| 202 | MTRU | | Solidified (| tranisa | Mainly debris | 0/4 | | | | | |
| 285 | MIRU | | Solidified (| rienics | Conweb peda(\$) | D/Q - | | | | | |
| j]č | TATRU R | | Salidified C | rganice | Conveb peris(15) | D/9 | | | | | |
| 172 | MIRI | | Solidined | 1840 SR | PCB westel (6) | | ······································ | | | | |
| 338 | MIRE E | | Solidified C | nglaist | Conweb pads(15) | D/9] D/9] | | | | | |
| - 344 | MIBU | | Solidined C | rganics | Diatome coops certh(2) | | | | | | |
| 115 | MIRI | ∰ | Solidified 6 | ranica | Conweb pada(15) | <u>0/a</u>] | | | - | | |
| 361 | TRU | | Solidified C | nganica | Vermiculite(9) | D(?)D(?) | · · · · · · · · · · · · · · · · · · · | · | - - | | |
| 380 | iku | | Solidified C | rganica | Distomaceous earth(7) | n/4 | | | | | |
| | [·· [· | | | | | <u> </u> <u>B</u> | II TRI Tole | I | 505118.2 | | ks |
| · | | | | | | [C | 11- IKU 101M | | 5262063.767 | 1344061.99 | Kg |
| · · · | | |] | | | <i>.</i> | | | | | |
| | l | | L | | | | | | | | |
| Cement equ | als Stored + Project | ed plus 2.05 | times Projec | ted for CH TR | l) | <u> </u> [| OTAL SCALED | 2 | | 8542531.067 | <u> </u> |
| us Stored + | Projected for RII-TI | RU | 1 | | | | | | | | |

}

Table 1. Estimate of Portland Cement in TRU Waste for Disposal in the WIPP

| The PRINT NAME AND A RECTAL OF A REAL AND A DECIDE OF MILE STOLEN AND A DECIDE AND A DECIDE AND A DECIDE | Solidification | | _Stared (m). | _Projected (m) | Total kg | Projected kg | · |
|---|-----------------------------------|----------|--------------|----------------|----------|--------------|-----|
| | | | | | [| I | |
| It waste, does not contain any portland cement | | | | | | | · · |
| scapsulated metal waste, does not contain any portland cement | | | | | | ······ | |
| ssume RF-MT0806 for final form cement density | | | · | | | | |
| orco (clay) is used as sorbent not portland cement | | | | | | | |
| te portland cement for this waste stream in the BIR occurs in the "Other | Inorganic Material* | | | | | | |
| ily 61 % of the solidification agent reported as cement in the TWBIR is | portland cement | | · | ···· | | | |
| stomscous earth is used as the sorbent in this waste stream | | | | | | | |
| aster of Paris used as solidification agent | | | | | · | | |
| ermiculite used as sorbent in this waste stream | L | | | | | | |
| lasis for portland cement are values reported in TWBIR supplemented v | vith information provided by LANL | | | | | | |
| for previous WIPP memo on nitrate, sulfate, and phosphate | l | | | | | | |
| olidification agent is Envirostone (a gypsum based process) that does n | ot contain portland cement | | | | | | |
| olidification agent is a calcium-silicate process that does not use portlar | id cement | | | | | | · |
| Dil Dri is used as sorbent | | | | | | | |
| olidified organics is paint, contains no portland cement | | | | | | | |
| olidification agent/sorbent is conwed pads (plastic fiber absorbent) +/- | vermiculite | | | | | | |
| CB containing waste, excluded from current WIPP inventory | <u> </u> | I | I | | I | | |

)

Attachment 4 of 4

.

1

APPENDIX C



APPENDIX C

SITE-SPECIFIC STORED RADIONUCLIDE INVENTORIES

| Nuclide | ARCO | ARMY | ETEC | HANF | INEL | LBL |
|---------|----------|----------|----------|----------|----------|----------|
| Ac225 | | | 2.23E-15 | 1.31E-01 | 1.52E+00 | 5.45E-06 |
| Ac227 | | 1.98E-15 | 4.08E-14 | 1.02E-04 | 3.86E-02 | 1.35E-19 |
| Ac228 | | | 2.87E-18 | 5.60E-02 | 3.08E-01 | 1.69E-19 |
| Ag109m | | | | | | |
| Ag110 | | | | 5.08E-10 | 3.55E-09 | |
| Ag110m | | | | 3.81E-08 | 2.67E-07 | |
| Am241 | | | 5.19E-01 | 4.73E+03 | 9.01E+04 | 9.17E-02 |
| Am243 | | | | 9.02E-02 | 3.80E-01 | 3.85E-02 |
| Am245 | | | | | 1.12E-09 | 3.60E-14 |
| At217 | | | 2.23E-15 | 1.31E-01 | 1.52E+00 | 5.45E-06 |
| Ba137m | | | 1.99E-01 | 6.46E+02 | 5.71E+01 | |
| Bi210 | 5.22E-15 | | 2.05E-15 | 5.30E-06 | 2.70E-02 | 8.96E-03 |
| Bi211 | | 1.98E-15 | 4.08E-14 | 1.02E-04 | 3.87E-02 | 1.35E-19 |
| Bi212 | | | 1.10E-18 | 5.19E-02 | 2.62E+01 | 8.59E-20 |
| Bi213 | | | 2.23E-15 | 1.31E-01 | 1.52E+00 | 5.45E-06 |
| Bi214 | 6.86E-13 | | 4.56E-14 | 3.15E-05 | 4.80E-02 | 3.37E-02 |
| Bk249 | | | | | 7.70E-05 | 2.48E-09 |
| Bk250 | | | | | | 8.68E-08 |
| C14 | | | | 1.60E+00 | 1.66E-01 | |
| Cd109 | | | | | | |
| Cd113m | | | | 1.25E-09 | 3.20E-08 | |
| Ce144 | T | | | 4.41E-03 | 3.15E-02 | |

CH Curies on a Site-by-Site¹ Basis (Decayed to the End of 1995)

¹Argonne National Laboratory-East, Argonne National Laboratory-West, and Teledyne Brown Engineering are not included because no data were received. Data from Sandia National Laboratory-Albuquerque are reported under RH-TRU waste because although the final waste form is expected to be CH-TRU waste, the stored waste is remotely handled at the site.

C - 1

| Nuclide | ARCO | ARMY | ETEC | HANF | INEL | LBL |
|---------|------|----------|----------|----------|----------|----------|
| Cf249 | | | | | 1.02E-02 | 3.10E-03 |
| Cf250 | | | | | | 1.97E-04 |
| Cf251 | | | | | | |
| Cf252 | | | | 3.55E-05 | 2.19E-03 | |
| Cm242 | | | | | 2.73E-08 | |
| Cm243 | | | | 1.52E-02 | | |
| Cm244 | | | | 3.70E+03 | 4.91E+02 | 8.70E-02 |
| Cm245 | | | | 1.71E-03 | 9.09E-06 | 2.27E-06 |
| Cm246 | | | | | 1.53E-03 | 4.83E-07 |
| Cm247 | | | | | | |
| Cm248 | - | | | 8.13E-09 | 4.73E-07 | |
| Co58 | | | | | 1.22E-14 | |
| Co60 | | | | | 6.23E+01 | |
| Cs134 | | | | 2.45E-04 | 1.20E-03 | |
| Cs135 | | | | 1.91E-07 | 8.08E-06 | |
| Cs137 | | | 2.11E-01 | 6.83E+02 | 6.04E+01 | |
| Es254 | | | | | | 8.67E-08 |
| Eu150 | | | | | 3.50E-05 | |
| Eu152 | | | | 7.34E-07 | 1.62E-01 | |
| Eu154 | | | | 6.22E-05 | 6.42E-01 | |
| Eu155 | | | | 1.06E-03 | 3.82E-01 | |
| Fe55 | | | | | 1.91E-05 | |
| Fe59 | | | | | 3.38E-21 | |
| Fr221 | | | 2.23E-15 | 1.31E-01 | 1.52E+00 | 5.45E-06 |
| Fr223 | | 2.73E-17 | 5.63E-16 | 1.41E-06 | 5.33E-04 | 1.86E-21 |
| НЗ | | | | | 8.02E-01 | |
| 1129 | | | | | | |

C - 2

| Nuclide | ARCO | ARMY | ETEC | HANF | INEL | LBL |
|---------|----------|----------|----------|----------|----------|----------|
| Kr85 | | | | | | |
| Mn54 | | | | | 8.49E-04 | |
| Nb95 | | | | 1.80E-11 | 2.38E-09 | |
| Nb95m | | | | 6.00E-14 | 7.95E-12 | |
| Ni59 | | | | | | |
| Ni63 | | | | | 9.06E-05 | |
| Np237 | | | 9.49E-07 | 2.72E-01 | 8.53E-01 | 6.32E-06 |
| Np239 | | | | 9.02E-02 | 3.80E-01 | 3.85E-02 |
| Np240m | | | | 5.84E-10 | 3.50E-14 | |
| Pa231 | | 1.88E-13 | 6.72E-13 | 4.84E-04 | 1.33E-05 | 1.99E-18 |
| Pa233 | | | 9.49E-07 | 2.72E-01 | 8.53E-01 | 6.32E-06 |
| Pa234 | | | 6.06E-17 | 7.62E-03 | 1.50E-04 | 2.40E-14 |
| Pa234m | | | 4.66E-14 | 5.86E+00 | 1.16E-01 | 1.84E-11 |
| РЬ209 | | | 2.23E-15 | 1.31E-01 | 1.52E+00 | 5.45E-06 |
| РЬ210 | 5.22E-15 | | 2.05E-15 | 5.30E-06 | 2.70E-02 | 8.96E-03 |
| Pb211 | | 1.98E-15 | 4.08E-14 | 1.02E-04 | 3.87E-02 | 1.35E-19 |
| Pb212 | | | 1.10E-18 | 5.19E-02 | 2.62E+01 | 8.59E-20 |
| Pb214 | 6.86E-13 | | 4.56E-14 | 3.15E-05 | 4.80E-02 | 3.37E-02 |
| Pd107 | | | | 2.82E-08 | 1.19E-06 | |
| Pm147 | | | | 4.78E-02 | 2.63E+00 | |
| Po210 | 1.42E-15 | | 2.05E-15 | 5.30E-06 | 2.70E-02 | 8.96E-03 |
| Po211 | | 5.53E-18 | 1.14E-16 | 2.87E-07 | 1.08E-04 | 3.78E-22 |
| Po212 | | | 7.04E-19 | 3.32E-02 | 1.68E+01 | |
| Po213 | | | 2.19E-15 | 1.28E-01 | 1.49E+00 | 5.33E-06 |
| Po214 | 6.86E-13 | | 4.56E-14 | 3.15E-05 | 4.80E-02 | 3.37E-02 |
| Po215 | | 1.98E-15 | 4.08E-14 | 1.02E-04 | 3.87E-02 | 1.35E-19 |
| Po216 | | | 1.10E-18 | 5.19E-02 | 2.62E+01 | 8.59E-20 |

| Nuclide | ARCO | ARMY | ETEC | HANF | INEL | LBL |
|---------|----------|----------|----------|----------|----------|----------|
| Po218 | 6.86E-13 | 1.40E-11 | 4.56E-14 | 3.15E-05 | 4.80E-02 | 3.37E-02 |
| Pr144 | | | | 4.36E-03 | 3.12E-02 | |
| Pu236 | | | | | 1.04E-02 | |
| Pu238 | 3.70E+02 | | 1.11E-01 | 8.05E+04 | 5.98E+04 | 2.32E-04 |
| Pu239 | | 1.80E+01 | 1.79E+00 | 2.63E+04 | 4.01E+04 | 8.45E-06 |
| Pu240 | | | 6.12E-01 | 6.15E+03 | 9.82E+03 | 5.14E-03 |
| Pu241 | | | 6.22E+00 | 3.78E+04 | 1.50E+05 | 4.48E-07 |
| Pu242 | | | 5.00E-05 | 3.80E-01 | 9.45E-01 | 1.01E-02 |
| Pu243 | | | | | | |
| Pu244 | | | | 5.85E-10 | 3.50E-14 | |
| Ra223 | | 1.98E-15 | 4.08E-14 | 1.02E-04 | 3.87E-02 | 1.35E-19 |
| Ra224 | | | 1.10E-18 | 5.19E-02 | 2.62E+01 | 8.59E-20 |
| Ra225 | | | 2.23E-15 | 1.31E-01 | 1.52E+00 | 5.45E-06 |
| Ra226 | 6.86E-13 | | 4.56E-14 | 3.15E-05 | 4.80E-02 | 3.37E-02 |
| Ra228 | | | 2.87E-18 | 5.60E-02 | 3.08E-01 | 1.69E-19 |
| Rh106 | | | | 2.17E-03 | 1.12E-02 | |
| Rn219 | | 1.98E-15 | 4.08E-14 | 1.02E-04 | 3.87E-02 | 1.35E-19 |
| Rn220 | | | 1.10E-18 | 5.19E-02 | 2.62E+01 | 8.59E-20 |
| Rn222 | 6.86E-13 | | 4.56E-14 | 3.15E-05 | 4.80E-02 | 3.37E-02 |
| Ru106 | | | | 2.17E-03 | 1.12E-02 | |
| Sb125 | | | | 5.91E-04 | 3.53E-03 | |
| Sb126 | | | | 5.13E-08 | 2.17E-06 | |
| Sb126m | | ` | | 3.67E-07 | 1.55E-05 | |
| Se79 | | | | 1.66E-07 | 7.00E-06 | |
| Sm151 | | | | 6.14E-04 | 2.39E-02 | |
| Sn119m | | | | 2.95E-07 | 2.10E-06 | |
| Sn121m | | | | 1.20E-05 | 4.38E-04 | |

| Nuclide | ARCO | ARMY | ETEC | HANF | INEL | LBL |
|---------|----------|----------|----------|------------------|----------|----------|
| Sn126 | | | | 3.67E-07 | 1.55E-05 | |
| Sr90 | | | 2.00E-01 | 6.92E+02 | 1.96E+00 | |
| Tc99 | | | | 9.51E-06 | 2.16E-03 | |
| Te125m | | | | 1.44E-04 | 8.62E-04 | |
| Te127 | | | | 3.95 E-09 | 1.02E-07 | |
| Te127m | | | | 4.03E-09 | 1.04E-07 | |
| Th227 | | 1.95E-15 | 4.02E-14 | 1.01E-04 | 3.82E-02 | 1.33E-19 |
| Th228 | | | 1.10E-18 | 5.19E-02 | 2.62E+01 | 8.59E-20 |
| Th229 | | | 2.23E-15 | 1.31E-01 | 1.52E+00 | 5.45E-06 |
| Th230 | 4.75E-09 | | 5.25E-11 | 8.11E-03 | 2.08E-02 | 1.50E-13 |
| Th231 | _ | 1.77E-08 | 1.06E-08 | 1.71E+00 | 6.17E-02 | 3.32E-14 |
| Th232 | | | 1.61E-17 | 6.71E-02 | 3.30E-01 | 5.33E-19 |
| Th234 | | | 4.66E-14 | 5.86E+00 | 1.16E-01 | 1.84E-11 |
| T1207 | | 1.97E-15 | 4.07E-14 | 1.02E-04 | 3.86E-02 | 1.34E-19 |
| TI208 | | | 3.95E-19 | 1.86E-02 | 9.42E+00 | 3.09E-20 |
| TI209 | | | 4.83E-17 | 2.82E-03 | 3.28E-02 | 1.18E-07 |
| U232 | | | | | 2.53E+01 | |
| U233 | | | 1.20E-11 | 8.00E+01 | 8.99E+02 | 4.81E-03 |
| U234 | 1.05E-03 | | 1.93E-06 | 5.37E+01 | 6.17E+00 | 4.73E-09 |
| U235 | | 1.77E-08 | 1.06E-08 | 1.71E+00 | 6.17E-02 | 3.32E-14 |
| U236 | | _ | 1.09E-07 | 2.49E-03 | 5.27E-03 | 1.81E-09 |
| U237 | | | 1.53E-04 | 9.26E-01 | 3.67E+00 | 1.10E-11 |
| U238 | | | 4.66E-14 | 5.86E+00 | 1.16E-01 | 1.84E-11 |
| U240 | | | | 5.84E-10 | 3.50E-14 | |
| Y90 | | | 2.00E-01 | 6.92E+02 | 1.96E+00 | |
| Zr93 | | | | 2.14E-06 | 9.06E-05 | |
| Zr95 | | | | 8.09E-12 | 1.07E-09 | |

¹Argonne National Laboratory-East, Argonne National Laboratory-West, and Teledyne Brown Engineering are not included because no data were received. Data from Sandia National Laboratory-Albuquerque are reported under RH-TRU waste because although the final waste form is expected to be CH-TRU waste, the stored waste is remotely handled at the site.

C - 5

| Nuclide | ARCO | ARMY | ETEC | HANF | INEL | LBL |
|------------------|----------|----------|----------|----------|----------|----------|
| Total by Site | 3.70E+02 | 1.80E+01 | 1.01E+01 | 1.62E+05 | 3.51E+05 | 5.08E-01 |
| | | | | | | |
| Nuclide | LANL | LLNL | MOUND | MURR | NTS | ORNL |
| Ac225 | 8.06E-02 | 9.81E-13 | | 1.59E-13 | 2.41E-03 | 2.07E-01 |
| Ac227 | 2.32E-01 | 3.32E-10 | 4.13E-12 | 1.83E-17 | 2.09E-04 | 9.85E-03 |
| Ac228 | 1.59E-03 | 1.60E-16 | | | 1.90E-16 | 7.12E-04 |
| Ag109m | 6.56E+00 | | | | | |
| Ag110 | 2.87E-11 | | | | 5.55E-11 | |
| Ag110m | 2.16E-09 | | | | 4.18E-09 | |
| Am241 | 9.11E+03 | 1.44E+02 | | 3.24E-01 | 2.84E+02 | 1.61E+03 |
| Am243 | 3.83E+00 | 2.45E-02 | | | 1.22E+00 | 1.16E+01 |
| Am245 | 1.95E-15 | | | | 5.29E-14 | 1.49E-10 |
| At217 | 8.06E-02 | 9.81E-13 | | 1.59E-13 | 2.41E-03 | 2.07E-01 |
| Ba137m | 4.55E+01 | 1.57E-06 | | | 3.41E-01 | 2.20E+03 |
| Bi210 | 2.80E-01 | 2.38E-13 | 7.23E-10 | | 6.69E-02 | 1.26E+00 |
| Bi211 | 2.32E-01 | 3.32E-10 | 4.13E-12 | 1.83E-17 | 2.09E-04 | 9.85E-03 |
| Bi212 | 1.32E-03 | 6.13E-17 | | | 1.64E-02 | 2.84E-01 |
| Bi213 | 8.06E-02 | 9.81E-13 | | 1.59E-13 | 2.41E-03 | 2.07E-01 |
| Bi214 | 9.04E-01 | 9.47E-12 | 6.88E-09 | 1.94E-22 | 2.50E-01 | 6.49E+00 |
| Bk249 | 1.35E-10 | | | | 3.65E-09 | 1.03E-05 |
| Bk250 | | | | | 4.11E-11 | 9.51E-13 |
| C14 | 2.00E-07 | | | | 2.50E-04 | |
| Cd109 | 6.55E+00 | | | | | |
| Cd113m | 7.42E-07 | | | | 6.50E-09 | |
| Ce144 | 3.04E-04 | | | | 7.88E-04 | |
| Cf249 | 9.64E-04 | | | | 1.14E-02 | 2.82E-02 |

| Nuclide | LANL | LLNL | MOUND | MURR | NTS | ORNL |
|---------|----------|----------|----------|----------|----------|----------|
| Cf250 | | | | | 3.18E-01 | 1.49E-03 |
| Cf251 | 1.58E-03 | | | | | |
| Cf252 | | | | | 1.70E-02 | 1.60E-01 |
| Cm242 | 3.42E-17 | 1.70E-04 | | | | 1.39E-03 |
| Cm243 | 1.09E+00 | | | | | |
| Cm244 | 1.57E+02 | 6.54E+01 | | | 2.28E+02 | 1.06E+03 |
| Cm245 | 1.60E-06 | | | | 9.44E-06 | 3.35E-05 |
| Cm246 | 4.01E-02 | 5.22E-04 | | | 6.14E-04 | 1.60E-05 |
| Cm247 | 1.34E-09 | | | | | |
| Cm248 | | | | | 3.57E-06 | 2.55E-02 |
| Co58 | 1.22E-13 | | | | | |
| Co60 | 2.14E-04 | | | | _ | 1.84E-06 |
| Cs134 | 4.24E-03 | | | | 4.03E-04 | |
| Cs135 | 2.05E-04 | | | | 1.20E-06 | |
| Cs137 | 4.81E+01 | 1.66E-06 | | | 3.60E-01 | 2.33E+03 |
| E\$254 | | | | | 4.11E-11 | |
| Eu150 | | | | | | |
| Eu152 | 4.18E-04 | 1.33E-06 | | | 1.06E+00 | 6.18E-04 |
| Eu154 | 2.45E-02 | 5.25E-07 | | | 4.28E-01 | |
| Eu155 | 2.31E-01 | | | | 3.80E-03 | |
| Fe55 | | | | | | |
| Fe59 | 1.35E-16 | | | | | 1.87E-07 |
| Fr221 | 8.06E-02 | 9.81E-13 | | 1.59E-13 | 2.41E-03 | 2.07E-01 |
| Fr223 | 3.20E-03 | 4.58E-12 | 5.70E-14 | 2.53E-19 | 2.89E-06 | 1.36E-04 |
| Н3 | | | | | 6.46E-02 | |
| 1129 | | | | | | |
| Kr85 | | | | | 1.96E-01 | |

C - 7

| Nuclide | LANL | LLNL | MOUND | MURR | NTS | ORNL |
|---------|----------|----------|----------|----------|----------|----------|
| Mn54 | 5.48E-08 | | | | | |
| Nb95 | 1.76E-11 | | | | 1.51E-17 | |
| Nb95m | 5.89E-14 | | | | 5.05E-20 | |
| Ni59 | | | | | | |
| Ni63 | | | | | | 1.09E-04 |
| Np237 | 3.24E-02 | 4.71E-04 | | 2.28E-04 | 5.78E-03 | 7.27E-01 |
| Np239 | 3.83E+00 | 2.45E-02 | | | 1.22E+00 | 1.16E+01 |
| Np240m | 1.94E-07 | | | | 9.99E-07 | 1.10E-09 |
| Pa231 | 1.24E-03 | 1.54E-08 | 3.24E-11 | 8.98E-16 | 5.00E-04 | 3.14E-01 |
| Pa233 | 3.24E-02 | 4.71E-04 | | 2.28E-04 | 5.78E-03 | 7.27E-01 |
| Pa234 | 3.07E-05 | 3.94E-05 | | 1.51E-10 | 2.13E-07 | 5.54E-05 |
| Pa234m | 2.36E-02 | 3.03E-02 | | 1.16E-07 | 1.64E-04 | 4.26E-02 |
| Pb209 | 8.06E-02 | 9.81E-13 | | 1.59E-13 | 2.41E-03 | 2.07E-01 |
| Ръ210 | 2.80E-01 | 2.38E-13 | 7.23E-10 | | 6.69E-02 | 1.26E+00 |
| Pb211 | 2.32E-01 | 3.32E-10 | 4.13E-12 | 1.83E-17 | 2.09E-04 | 9.85E-03 |
| РЬ212 | 1.32E-03 | 6.13E-17 | | | 1.64E-02 | 2.84E-01 |
| Pb214 | 9.04E-01 | 9.47E-12 | 6.88E-09 | 1.94E-22 | 2.50E-01 | 6.49E+00 |
| Pd107 | 3.03E-05 | | | | 1.78E-07 | |
| Pm147 | 2.00E+00 | | | | 1.05E-01 | 1.94E-02 |
| Po210 | 2.80E-01 | 1.97E-13 | 7.23E-10 | | 6.69E-02 | 1.26E+00 |
| Po211 | 6.50E-04 | 9.28E-13 | 1.16E-14 | 5.13E-20 | 5.86E-07 | 2.76E-05 |
| Po212 | 8.48E-04 | 3.93E-17 | | | 1.05E-02 | 1.82E-01 |
| Po213 | 7.89E-02 | 9.60E-13 | | 1.55E-13 | 2.36E-03 | 2.02E-01 |
| Po214 | 9.04E-01 | 9.47E-12 | 6.87E-09 | | 2.50E-01 | 6.49E+00 |
| Po215 | 2.32E-01 | 3.32E-10 | 4.13E-12 | 1.83E-17 | 2.09E-04 | 9.85E-03 |
| Po216 | 1.32E-03 | 6.13E-17 | | | 1.64E-02 | 2.84E-01 |
| Po218 | 9.05E-01 | 9.47E-12 | 6.88E-09 | 1.94E-22 | 2.50E-01 | 6.49E+00 |

C - 8

| Nuclide | LANL | LLNL | MOUND | MURR | NTS | ORNL |
|---------|----------|----------|----------|----------|----------|----------|
| Pr144 | 3.00E-04 | | | | 7.79E-04 | |
| Pu236 | 5.37E-17 | | | | | |
| Pu238 | 1.15E+05 | 7.65E+01 | 1.53E+03 | | 1.95E+02 | 3.50E+03 |
| Pu239 | 7.69E+04 | 1.58E+02 | 2.98E+01 | 2.46E-02 | 2.76E+03 | 1.01E+03 |
| Pu240 | 1.00E+02 | 6.44E+01 | | | 1.88E+01 | 9.48E+02 |
| Pu241 | 1.62E+03 | 1.63E+03 | | 6.32E-03 | 2.40E+02 | 4.79E+04 |
| Pu242 | 4.85E+02 | 2.02E-02 | | | 8.70E-02 | 2.37E-01 |
| Pu243 | 1.34E-09 | | | | | |
| Pu244 | 1.94E-07 | | | | 1.00E-06 | 1.10E-09 |
| Ra223 | 2.32E-01 | 3.32E-10 | 4.13E-12 | 1.83E-17 | 2.09E-04 | 9.85E-03 |
| Ra224 | 1.32E-03 | 6.13E-17 | | | 1.64E-02 | 2.84E-01 |
| Ra225 | 8.06E-02 | 9.81E-13 | | 1.59E-13 | 2.41E-03 | 2.07E-01 |
| Ra226 | 9.05E-01 | 9.47E-12 | 6.88E-09 | 1.94E-22 | 2.50E-01 | 6.49E+00 |
| Ra228 | 1.59E-03 | 1.60E-16 | | | 1.90E-16 | 7.12E-04 |
| Rh106 | 9.97E-04 | | | | 8.76E-04 | |
| Rn219 | 2.32E-01 | 3.32E-10 | 4.13E-12 | 1.83E-17 | 2.09E-04 | 9.85E-03 |
| Rn220 | 1.32E-03 | 6.13E-17 | | | 1.64E-02 | 2.84E-01 |
| Rn222 | 9.05E-01 | 9.47E-12 | 6.88E-09 | 1.94E-22 | 2.50E-01 | 6.49E+00 |
| Ru106 | 9.97E-04 | | | | 8.76E-04 | |
| Sb125 | 4.67E-02 | | | | 1.37E-03 | |
| Sb126 | 5.52E-05 | | | | 3.23E-07 | |
| Sb126m | 3.94E-04 | | | | 2.31E-06 | |
| Se79 | 1.78E-04 | | | | 1.04E-06 | |
| Sm151 | 6.00E-01 | | | | 3.75E-03 | |
| Sn119m | 1.66E-08 | | | | 2.97E-08 | |
| Sn121m | 1.09E-02 | | | | 7.17E-05 | |
| Sn126 | 3.94E-04 | | | | 2.31E-06 | |

| Nuclide | LANL | LLNL | MOUND | MURR | NTS | ORNL |
|---------|----------|----------|----------|----------|----------|----------|
| Sr90 | 4.44E+01 | | | | 3.10E-01 | 1.48E+03 |
| Тс99 | 1.02E-02 | | | | 5.99E-05 | 1.78E+01 |
| Te125m | 1.14E-02 | | | | 3.33E-04 | |
| Te127 | 7.45E-10 | | | | 2.29E-12 | |
| Te127m | 7.60E-10 | | | | 2.34E-12 | |
| Th227 | 2.29E-01 | 3.27E-10 | 4.07E-12 | 1.81E-17 | 2.06E-04 | 9.72E-03 |
| Th228 | 1.32E-03 | 6.13E-17 | | | 1.64E-02 | 2.84E-01 |
| Th229 | 8.06E-02 | 9.81E-13 | | 1.59E-13 | 2.41E-03 | 2.07E-01 |
| Th230 | 4.90E-04 | 3.06E-08 | 3.35E-06 | 1.35E-18 | 9.98E-07 | 2.45E-04 |
| Th231 | 5.27E-01 | 5.93E-04 | 2.68E-07 | 4.45E-11 | 6.15E-05 | 8.15E-03 |
| Th232 | 2.29E-03 | 9.37E-16 | | | 8.19E-16 | 8.57E-04 |
| Th234 | 2.36E-02 | 3.03E-02 | | 1.16E-07 | 1.64E-04 | 4.26E-02 |
| TI207 | 2.31E-01 | 3.31E-10 | 4.12E-12 | 1.83E-17 | 2.09E-04 | 9.82E-03 |
| TI208 | 4.76E-04 | 2.20E-17 | | | 5.89E-03 | 1.02E-01 |
| TI209 | 1.74E-03 | 2.12E-14 | | 3.43E-15 | 5.20E-05 | 4.47E-03 |
| U232 | 5.50E-18 | | | | 1.65E-02 | 2.90E-01 |
| U233 | 4.46E+01 | 5.95E-09 | | 1.78E-09 | 1.81E+00 | 1.77E+02 |
| U234 | 5.84E+00 | 3.17E-03 | 5.52E-02 | 2.98E-13 | 1.25E-02 | 1.57E+01 |
| U235 | 5.27E-01 | 5.93E-04 | 2.68E-07 | 4.45E-11 | 6.15E-05 | 8.15E-03 |
| U236 | 2.99E-06 | 7.63E-06 | | | 4.20E-06 | 3.40E-04 |
| U237 | 3.98E-02 | 4.00E-02 | | 1.55E-07 | 5.88E-03 | 1.18E+00 |
| U238 | 2.36E-02 | 3.03E-02 | | 1.16E-07 | 1.64E-04 | 4.26E-02 |
| U240 | 1.94E-07 | | | | 9.99E-07 | 1.10E-09 |
| Y90 | 4.45E+01 | | | | 3.10E-01 | 1.48E+03 |
| Zr93 | 2.30E-03 | | | | 1.35E-05 | |
| Zr95 | 7.93E-12 | | | | 6.81E-18 | |

1. XX

C - 10

| Nuclide | LANL | LLNL | MOUND | MURR | NTS | ORNL |
|------------------|----------|----------|----------|----------|----------|----------|
| Total by Site | 2.03E+05 | 2.14E+03 | 1.56E+03 | 3.55E-01 | 3.74E+03 | 6.38E+04 |

³Argonne National Laboratory-East, Argonne National Laboratory-West, and Teledyne Brown Engineering are not included because no data were received. Data from Sandia National Laboratory-Albuquerque are reported under RH-TRU waste because although the final waste form is expected to be CH-TRU waste, the stored waste is remotely handled at the site.

C - 11

| Nuclide | PAD | PANT | RFETS | RF-Res | SR-On | SR-Off | TOTAL |
|---------|----------|----------|----------|----------|----------|----------|----------|
| Ac225 | 4.01E-07 | | 3.55E-11 | 2.14E-09 | 1.31E-05 | 1.02E-10 | 1.94E+00 |
| Ac227 | 1.27E-12 | 4.83E-17 | 1.58E-10 | 1.62E-08 | 3.70E-07 | 2.89E-10 | 2.80E-01 |
| Ac228 | | | 1.49E-14 | 7.07E-13 | 1.01E-02 | 2.13E-14 | 3.76E-01 |
| Ag109m | | | | | | | 6.56E+00 |
| Ag110 | | | | | | | 4.14E-09 |
| Ag110m | | | | | | | 3.11E-07 |
| Am241 | | | 1.10E+04 | 1.19E+05 | 3.58E+03 | 1.20E+02 | 2.40E+05 |
| Am243 | | | | | 7.55E-01 | | 1.80E+01 |
| Am245 | | | | | | | 1.27E-09 |
| At217 | 4.01E-07 | | 3.55E-11 | 2.14E-09 | 1.31E-05 | 1.02E-10 | 1.94E+00 |
| Ba137m | | | | | 7.11E+00 | | 2.96E+03 |
| Bi210 | | | 4.54E-12 | 1.08E-09 | 1.69E-07 | 9.40E-07 | 1.65E+00 |
| Bi211 | 1.27E-12 | 4.83E-17 | 1.58E-10 | 1.62E-08 | 3.70E-07 | 2.89E-10 | 2.81E-01 |
| Bi212 | | | 4.98E-15 | 3.82E-13 | 9.18E-03 | 1.93E-14 | 2.66E+01 |
| Bi213 | 4.01E-07 | | 3.55E-11 | 2.14E-09 | 1.31E-05 | 1.02E-10 | 1.94E+00 |
| Bi214 | | | 9.77E-11 | 1.35E-08 | 1.51E-06 | 5.79E-06 | 7.72E+00 |
| Bk249 | | | | | | | 8.73E-05 |
| Bk250 | | | | | | | 8.68E-08 |
| C14 | | | | | | | 1.77E+00 |
| Cd109 | | | | | | | 6.55E+00 |
| Cd113m | | | | | | | 7.81E-07 |
| Ce144 | | | | | 8.72E-13 | | 3.70E-02 |
| Cf249 | | | | | | | 5.39E-02 |
| Cf250 | | | | | | | 3.20E-01 |
| Cf251 | | | | | | | 1.58E-03 |
| Cf252 | | | | | 3.62E-01 | | 3.61E+01 |
| Cm242 | T | | | | | | 1.56E-03 |

| Nuclide | PAD | PANT | RFETS | RF-Res | SR-On | SR-Off | TOTAL |
|---------|----------|----------|----------|----------|----------|----------|----------|
| Cm243 | | | | | | | 1.11E+00 |
| Cm244 | | | | | 6.15E+02 | | 6.32E+03 |
| Cm245 | | | | | | | 1.77E-03 |
| Cm246 | | | | | | | 4.28E-02 |
| Cm247 | | | | | | | 1.34E-09 |
| Cm248 | | | | | 1.61E-04 | | 3.35E-02 |
| Co58 | | | | | | | 1.34E-13 |
| Co60 | | | | | 3.56E-01 | | 6.27E+01 |
| Cs134 | | | | | 3.18E-06 | | 6.09E-03 |
| Cs135 | | | | | | | 2.15E-04 |
| Cs137 | | | | | 7.52E+00 | | 3.12E+03 |
| Es254 | | | | | | | 8.68E-08 |
| Eu150 | | | | | | | 3.50E-05 |
| Eu152 | | | | | | | 1.22E+00 |
| Eu154 | | | | | 2.83E-04 | | 1.10E+00 |
| Eu155 | | | | | 3.13E-06 | | 6.18E-01 |
| Fe55 | | | | | | | 1.91E-05 |
| Fe59 | | | | | | | 1.87E-07 |
| Fr221 | 4.01E-07 | | 3.55E-11 | 2.14E-09 | 1,31E-05 | 1.02E-10 | 1.94E+00 |
| Fr223 | 1.75E-14 | 6.67E-19 | 2.19E-12 | 2.23E-10 | 5.10E-09 | 3.99E-12 | 3.87E-03 |
| НЗ | | | | | | | 8.66E-01 |
| 1129 | | | | | 1.17E-07 | | 1.17E-07 |
| Kr85 | | | | | | | 1.96E-01 |
| Mn54 | | | | | 1.00E-10 | | 8.49E-04 |
| Nb95 | | | | | | | 2.41E-09 |
| Nb95m | | | | | | | 8.06E-12 |
| Ni59 | | | | | 1.25E-03 | | 1.25E-03 |

¹Argonne National Laboratory-East, Argonne National Laboratory-West, and Teledyne Brown Engineering are not included because no data were received. Data from Sandia National Laboratory-Albuquerque are reported under RH-TRU waste because although the final waste form is expected to be CH-TRU waste, the stored waste is remotely handled at the site.

C - 13

| Nuclide | PAD | PANT | RFETS | RF-Res | SR-On | SR-Off | TOTAL |
|---------|----------|----------|----------|----------|----------|----------|----------|
| Ni63 | | | | | 1.53E-01 | | 1.53E-01 |
| Np237 | 5.49E+01 | | 1.70E-02 | 3.19E-01 | 8.59E+00 | 3.58E-03 | 6.58E+01 |
| Np239 | | | | | 7.55E-01 | | 1.80E+01 |
| Np240m | | | | | 1.59E-11 | | 1.20E-06 |
| Pa231 | 2.09E-11 | 2.31E-15 | 2.70E-09 | 1.59E-07 | 1.68E-06 | 1.65E-09 | 3.16E-01 |
| Pa233 | 5.49E+01 | | 1.70E-02 | 3.19E-01 | 8.59E+00 | 3.58E-03 | 6.58E+01 |
| Pa234 | | | 1.94E-17 | 9.23E-12 | 7.37E-06 | 5.26E-08 | 7.90E-03 |
| Pa234m | | | 1.49E-14 | 7.10E-09 | 5.67E-03 | 4.04E-05 | 6.08E+00 |
| РЬ209 | 4.01E-07 | | 3.55E-11 | 2.14E-09 | 1.31E-05 | 1.02E-10 | 1.94E+00 |
| РЬ210 | | | 4.54E-12 | 1.08E-09 | 1.69E-07 | 9.40E-07 | 1.65E+00 |
| Pb211 | 1.27E-12 | 4.83E-17 | 1.58E-10 | 1.62E-08 | 3.70E-07 | 2.89E-10 | 2.81E-01 |
| Pb212 | | | 4.98E-15 | 3.82E-13 | 9.18E-03 | 1.93E-14 | 2.66E+01 |
| Pb214 | | | 9.77E-11 | 1.35E-08 | 1.51E-06 | 5.79E-06 | 7.72E+00 |
| Pd107 | | | | | | | 3.17E-05 |
| Pm147 | | | | | 1.24E-05 | | 4.79E+00 |
| Po210 | | | 4.50E-12 | 1.08E-09 | 1.69E-07 | 9.40E-07 | 1.65E+00 |
| Po211 | 3.56E-15 | 1.35E-19 | 4.43E-13 | 4.53E-11 | 1.04E-09 | 8.10E-13 | 7.86E-04 |
| Po212 | | | 3.19E-15 | 2.45E-13 | 5.88E-03 | 1.24E-14 | 1.70E+01 |
| Po213 | 3.93E-07 | | 3.47E-11 | 2.10E-09 | 1,28E-05 | 9.94E-11 | 1.90E+00 |
| Po214 | | | 9.77E-11 | 1.35E-08 | 1.51E-06 | 5.79E-06 | 7.72E+00 |
| Po215 | 1.27E-12 | 4.83E-17 | 1.58E-10 | 1.62E-08 | 3.70E-07 | 2.89E-10 | 2.81E-01 |
| Po216 | | | 4.98E-15 | 3.82E-13 | 9.18E-03 | 1.93E-14 | 2.66E+01 |
| Po218 | | | 9.77E-11 | 1.35E-08 | 1.51E-06 | 5.79E-06 | 7.73E+00 |
| Pr144 | | | | | 8.62E-13 | | 3.66E-02 |
| Pu236 | | | | | | | 1.04E-02 |
| Pu238 | | | 3.43E+02 | 8.09E+03 | 2.86E+05 | 2.01E+05 | 7.56E+05 |
| Pu239 | 5.57E+01 | 5.55E-02 | 9.98E+03 | 1.84E+05 | 9.13E+03 | 1.58E+02 | 3.51E+05 |

| Nuclide | PAD | PANT | RFETS | RF-Res | SR-On | SR-Off | TOTAL |
|---------|----------|----------|----------|----------|----------|----------|----------|
| Pu240 | | | 7.22E+03 | 4.22E+04 | 2.21E+03 | 7.97E+01 | 6.88E+04 |
| Pu241 | | | 5.23E+04 | 7.22E+05 | 6.02E+04 | 1.73E+03 | 1.07E+06 |
| Pu242 | | | 9.63E-05 | 5.33E+00 | 3.75E-01 | | 4.93E+02 |
| Pu243 | | | | | | | 1.34E-09 |
| Pu244 | | | | | 1.59E-11 | | 1.20E-06 |
| Ra223 | 1.27E-12 | 4.83E-17 | 1.58E-10 | 1.62E-08 | 3.70E-07 | 2.89E-10 | 2.81E-01 |
| Ra224 | | | 4.98E-15 | 3.82E-13 | 9.18E-03 | 1.93E-14 | 2.66E+01 |
| Ra225 | 4.01E-07 | | 3.55E-11 | 2.14E-09 | 1.31E-05 | 1.02E-10 | 1.94E+00 |
| Ra226 | | | 9.77E-11 | 1.35E-08 | 1.51E-06 | 5.79E-06 | 7.73E+00 |
| Ra228 | | | 1.49E-14 | 7.07E-13 | 1.01E-02 | 2.13E-14 | 3.76E-01 |
| Rh106 | | | | | 1.84E-10 | | 1.52E-02 |
| Rn219 | 1.27E-12 | 4.83E-17 | 1.58E-10 | 1.62E-08 | 3.70E-07 | 2.89E-10 | 2.81E-01 |
| Rn220 | | | 4.98E-15 | 3.82E-13 | 9.18E-03 | 1.93E-14 | 2.66E+01 |
| Rn222 | | | 9.77E-11 | 1.35E-08 | 1.51E-06 | 5.79E-06 | 7.73E+00 |
| Ru106 | | | | | 1.84E-10 | | 1.52E-02 |
| Sb125 | | | | | 2.60E-05 | | 5.22E-02 |
| Sb126 | | | | | 2.41E-08 | | 5.78E-05 |
| Sb126m | | | | | 1.72E-07 | | 4.12E-04 |
| Se79 | | | | | | | 1.86E-04 |
| Sm151 | | | | | 3.13E-04 | | 6.28E-01 |
| Sn119m | | | | | | | 2.44E-06 |
| Sn121m | | | | | | | 1.14E-02 |
| Sn126 | | | | | 1.72E-07 | | 4.12E-04 |
| Sr90 | | | | | 6.98E+00 | | 2.22E+03 |
| Tc99 | | | | | 4.50E-06 | | 1.78E+01 |
| Te125m | | | | | 6.34E-06 | | 1.27E-02 |
| Te127 | | | | | | | 1.07E-07 |

C - 15

| Nuclide | PAD | PANT | RFETS | RF-Res | SR-On | SR-Off | TOTAL |
|---------|----------|----------|----------|----------|----------|----------|----------|
| Te127m | | | | | | | 1.09E-07 |
| Th227 | 1.25E-12 | 4.77E-17 | 1.56E-10 | 1.60E-08 | 3.64E-07 | 2.85E-10 | 2.77E-01 |
| Th228 | | | 4.98E-15 | 3.82E-13 | 9.18E-03 | 1.93E-14 | 2.66E+01 |
| Th229 | 4.01E-07 | | 3.55E-11 | 2.14E-09 | 1.31E-05 | 1.02E-10 | 1.94E+00 |
| Th230 | | | 1.16E-07 | 8.88E-06 | 6.87E-04 | 1.66E-03 | 3.20E-02 |
| Th231 | 3.29E-07 | 1.09E-10 | 4.78E-05 | 1.56E-03 | 5.83E-03 | 1.04E-05 | 2.31E+00 |
| Th232 | | | 1.02E-13 | 2.55E-12 | 2.13E-02 | 4.79E-14 | 4.22E-01 |
| Th234 | | | 1.49E-14 | 7.10E-09 | 5.67E-03 | 4.04E-05 | 6.08E+00 |
| T1207 | 1.27E-12 | 4.82E-17 | 1.58E-10 | 1.61E-08 | 3.69E-07 | 2.88E-10 | 2.80E-01 |
| T1208 | | | 1.79E-15 | 1.37E-13 | 3.30E-03 | 6.94E-15 | 9.55E+00 |
| T1209 | 8.67E-09 | | 7.66E-13 | 4.63E-11 | 2.83E-07 | 2.19E-12 | 4.19E-02 |
| U232 | | | | | | | 2.56E+01 |
| U233 | 1.42E-03 | | 1.95E-07 | 6.56E-06 | 8.93E-03 | 1.78E-07 | 1.20E+03 |
| U234 | | | 4.81E-03 | 2.03E-01 | 1.06E+01 | 1.50E+01 | 1.07E+02 |
| U235 | 3.29E-07 | 1.09E-10 | 4.78E-05 | 1.56E-03 | 5.83E-03 | 1.04E-05 | 2.31E+00 |
| U236 | | | 9.17E-04 | 1.07E-02 | 4.77E-02 | 1.12E-04 | 6.75E-02 |
| U237 | | | 1.28E+00 | 1.77E+01 | 1.48E+00 | 4.23E-02 | 2.64E+01 |
| U238 | | | 1.49E-14 | 7.10E-09 | 5.67E-03 | 4.04E-05 | 6.08E+00 |
| U240 | | | | | 1.59E-11 | | 1.20E-06 |
| Y90 | | | | | 6.98E+00 | | 2.22E+03 |
| Zr93 | | | | | | | 2.41E-03 |
| Zr95 | | | | | | | 1.09E-09 |
| TOTAL | 1.66E+02 | 5.55E-02 | 8.08E+04 | 1.08E+06 | 3.62E+05 | 2.03E+05 | 2.51E+06 |

ABBREVIATIONS

| ARCO | ARCO Medical Center, Pennsylvania |
|---------------|---|
| ARMY | US Army Materiel Command |
| ETEC | Energy Technology Engineering Center |
| HANE | Hanford |
| INEL | Idaho National Engineering Laboratory |
| KAPL | Knolls Atomic Power Laboratory |
| LANL | Los Alamos National Laboratory |
| LBL | Lawrence Berkeley Laboratory |
| LLNL | Lawrence Livermore National Laboratory |
| Mound | Mound Facility |
| MURR | University of Missouri |
| NTS | Nevada Test Site |
| ORNL | Oak Ridge National Laboratory |
| PAD | Paducah |
| PANT | Pantex |
| RFETS | Rocky Flats Environmental Technology Site (All waste except residues) |
| RF-Res | Rocky Flats Environmental Technology Site - Residues Only |
| SR-On | Savannah River Site, waste generated on-site |
| SR-Off | Savannah River Site, waste that was generated off-site but currently stored at Savannah River |
| | |

| RH Curies on a Site-by-Site ¹ B | lasis |
|--|-------|
| (Decayed to the End of 199 | (5) |

| Nuclide | ETEC | HANF | INEL | KAPL | LANL |
|---------|----------|----------|----------|----------|----------|
| Ac225 | 3.07E-18 | 5.45E-04 | 1.76E-04 | 4.11E-18 | |
| Ac227 | 1.05E-16 | 1.70E-05 | 2.61E-07 | 1.35E-18 | 4.60E-07 |
| Ac228 | | 1.60E-03 | 3.87E-05 | | |
| Ag110 | | | 4.13E-09 | | 9.88E-10 |
| Ag110m | | | 3.11E-07 | | 7.43E-08 |
| Am241 | 5.85E-02 | 1.93E+02 | 4.68E+01 | 5.07E-02 | |
| Am243 | | | 6.91E-04 | | |
| Am245 | | | | | |
| At217 | 3.07E-18 | 5.45E-04 | 1.76E-04 | 4.11E-18 | |
| Ba137m | 2.48E+00 | 6.61E+03 | 1.80E+03 | 5.40E+01 | 1.28E+02 |
| Bi210 | | 2.33E-07 | 6.06E-12 | 1.87E-16 | 5.61E-17 |
| Bi211 | 1.05E-16 | 1.70E-05 | 2.61E-07 | 1.35E-18 | 4.60E-07 |
| Bi212 | | 1.49E-03 | 2.65E-05 | | |
| Bi213 | 3.07E-18 | 5.45E-04 | 1.76E-04 | 4.11E-18 | |
| Bi214 | | 1.16E-06 | 3.26E-10 | 1.24E-14 | 7.25E-15 |
| Bk249 | | | | | |
| C14 | | | 4.00E-02 | | |
| Cd113m | | | 1.15E-07 | | 8.88E-07 |
| Ce144 | | | 3.98E+00 | 1.56E+00 | 1.60E-02 |
| Cf249 | | | | | |
| Cf250 | | | | | |
| Cf252 | | | | | |
| Cm243 | | | 1.45E-02 | | |
| Cm244 | | | 9.63E-02 | | |
| Cm245 | | | | | |
| Cm246 | | | | | |
| Cm248 | | | | | |

¹Argonne National Laboratory-West is not included in this table because no radionuclide data were received fr. the site.

| Nuclide | ETEC | HANF | INEL | KAPL | LANL |
|---------|----------|----------|----------|----------|----------|
| Co58 | | | 4.37E-11 | | |
| Co60 | 2.30E+00 | 3.36E+02 | 1.31E+01 | 2.75E-01 | 4.17E+00 |
| Cr51 | | | 1.08E-05 | | |
| Cs134 | | | 5.38E+01 | 4.73E+00 | 2.42E-02 |
| Cs135 | | | 2.36E-05 | | 1.91E-04 |
| Cs137 | 2.62E+00 | 6.98E+03 | 1.90E+03 | 5.71E+01 | 1.35E+02 |
| Eu152 | | | 1.14E-01 | | 5.09E-04 |
| Eu154 | | | 7.92E-01 | 1.40E+00 | 3.50E-02 |
| Eu155 | | | 3.35E-01 | 1.81E-01 | 1.77E+00 |
| Fe55 | | | 5.97E-01 | | |
| Fr221 | 3.07E-18 | 5.45E-04 | 1.76E-04 | 4.11E-18 | |
| Fr223 | 1.45E-18 | 2.35E-07 | 3.60E-09 | 1.87E-20 | 6.34E-09 |
| Н3 | | | 1.43E-01 | | |
| Kr85 | | | 5.95E+00 | | |
| Mn54 | | | 8.31E-02 | | |
| Nb95 | | | 5.28E-12 | | 2.14E-14 |
| Nb95m | | | 1.76E-14 | | 7.15E-17 |
| Ni63 | | | 3.50E+00 | | |
| Np237 | 2.26E-08 | 1.58E-03 | 8.10E-04 | 2.25E-08 | |
| Np239 | | | 6.91E-04 | | |
| Np240m | | | | | |
| Pa231 | 6.68E-15 | 6.21E-05 | 1.42E-06 | 7.51E-17 | 2.39E-06 |
| Pa233 | 2.26E-08 | 1.58E-03 | 8.10E-04 | 2.25E-08 | |
| Pa234 | | 1.33E-05 | 1.80E-06 | 4.48E-18 | 2.60E-08 |
| Pa234m | | 1.03E-02 | 1.38E-03 | 3.45E-15 | 2.00E-05 |
| РЬ209 | 3.07E-18 | 5.45E-04 | 1.76E-04 | 4.11E-18 | |
| Pb210 | | 2.33E-07 | 6.06E-12 | 1.87E-16 | 5.61E-17 |

¹Argonne National Laboratory-West is not included in this table because no radionuclide data were received from the site.

| Nuclide | ETEC | HANF | INEL | KAPL | LANL |
|---------|----------|----------|----------|----------|----------|
| Pb211 | 1.05E-16 | 1.70E-05 | 2.61E-07 | 1.35E-18 | 4.60E-07 |
| Pb212 | | 1.49E-03 | 2.65E-05 | | |
| Pb214 | | 1.16E-06 | 3.26E-10 | 1.24E-14 | 7.25E-15 |
| Pd107 | | | 3.49E-06 | | 2.83E-05 |
| Pm147 | | | 1.49E+01 | 4.34E+00 | 1.13E+01 |
| Po210 | | 2.33E-07 | 4.06E-12 | 8.21E-17 | 1.60E-17 |
| Po211 | 2.94E-19 | 4.77E-08 | 7.30E-10 | 3.78E-21 | 1.29E-09 |
| Po212 | | 9.54E-04 | 1.70E-05 | | |
| Po213 | 3.00E-18 | 5.33E-04 | 1.72E-04 | 4.02E-18 | |
| Po214 | | 1.16E-06 | 3.26E-10 | 1.24E-14 | 7.25E-15 |
| Po215 | 1.05E-16 | 1.70E-05 | 2.61E-07 | 1.35E-18 | 4.60E-07 |
| Po216 | | 1.49E-03 | 2.65E-05 | | |
| Po218 | | 1.16E-06 | 3.26E-10 | 1.24E-14 | 7.25E-15 |
| Pr144 | | | 3.93E+00 | 1.54E+00 | 1.59E-02 |
| Pu238 | | 4.67E+01 | 6.09E+01 | 9.27E-01 | 3.90E+00 |
| Pu239 | 4.00E-01 | 3.35E+02 | 2.98E+01 | 3.30E-03 | 9.28E+01 |
| Pu240 | | 1.67E+02 | 1.13E+01 | 3.10E-03 | |
| Pu241 | | 4.67E+03 | 4.82E+01 | 7.77E-01 | |
| Pu242 | | 4.92E-03 | 1.01E-03 | 1.56E-05 | |
| Pu244 | | | | | |
| Ra223 | 1.05E-16 | 1.70E-05 | 2.61E-07 | 1.35E-18 | 4.60E-07 |
| Ra224 | | 1.49E-03 | 2.65E-05 | | |
| Ra225 | 3.07E-18 | 5.45E-04 | 1.76E-04 | 4.11E-18 | |
| Ra226 | | 1.16E-06 | 3.26E-10 | 1.24E-14 | 7.25E-15 |
| Ra228 | | 1.60E-03 | 3.87E-05 | | |
| Rh106 | | | 6.64E-02 | 4.98E-01 | 3.38E-01 |
| Rn219 | 1.05E-16 | 1.70E-05 | 2.61E-07 | 1.35E-18 | 4.60E-07 |

¹Argonne National Laboratory-West is not included in this table because no radionuclide data were received fruithe site.



| Nuclide | ETEC | HANF | INEL | KAPL | LANL |
|---------|----------|----------|----------|----------|----------|
| Rn220 | | 1.49E-03 | 2.65E-05 | | |
| Rn222 | | 1.16E-06 | 3.26E-10 | 1.24E-14 | 7.25E-15 |
| Ru106 | | | 6.64E-02 | 4.98E-01 | 3.38E-01 |
| Sb125 | | | 9.81E-01 | 5.33E-01 | 2.79E+00 |
| Sb126 | | | 6.35E-06 | | 5.15E-05 |
| Sb126m | | | 4.53E-05 | | 3.68E-04 |
| Se79 | | | 2.05E-05 | | 1.66E-04 |
| Sm151 | | | 7.23E-02 | | 5.82E-01 |
| Sn119m | | | 2.33E-06 | | 5.20E-07 |
| Sn121m | | | 1.36E-03 | | 1.09E-02 |
| Sn126 | | | 4.53E-05 | | 3.68E-04 |
| Sr90 | 2.62E+00 | 6.46E+03 | 1.70E+03 | 5.70E+01 | 1.24E+02 |
| Ta182 | | | 1.49E-07 | | |
| Tc99 | | | 1.18E-03 | | 9.54E-03 |
| Te125m | | | 2.39E-01 | 1.30E-01 | 6.88E-01 |
| Te127 | | | 5.78E-09 | | 1.31E-10 |
| Te127m | | | 5.91E-09 | | 1.34E-10 |
| Th227 | 1.03E-16 | 1.68E-05 | 2.57E-07 | 1.33E-18 | 4.53E-07 |
| Th228 | | 1.49E-03 | 2.65E-05 | | |
| Th229 | 3.07E-18 | 5.45E-04 | 1.76E-04 | 4.11E-18 | |
| Th230 | | 2.42E-04 | 1.37E-06 | 4.36E-11 | 5.01E-11 |
| Th231 | 4.73E-10 | 1.46E-01 | 5.41E-03 | 4.53E-12 | 8.78E-03 |
| Th232 | | 1.96E-03 | 7.51E-05 | 4.68E-21 | |
| Th234 | | 1.03E-02 | 1.38E-03 | 3.45E-15 | 2.00E-05 |
| TI207 | 1.05E-16 | 1.70E-05 | 2.60E-07 | 1.35E-18 | 4.58E-07 |
| TI208 | | 5.35E-04 | 9.52E-06 | | |
| TI209 | 6.63E-20 | 1.18E-05 | 3.79E-06 | 8.88E-20 | |

¹Argonne National Laboratory-West is not included in this table because no radionuclide data were received from the site.

| Nuclide | ETEC | HANF | INEL | KAPL | LANL |
|---------|----------|----------|------------------|----------|----------|
| U233 | 6.55E-14 | 4.15E-01 | 3.91E-01 | 7.62E-14 | |
| U234 | | 1.29E+00 | 1.51E-01 | 4.98E-06 | 1.11E-05 |
| U235 | 4.73E-10 | 1.46E-01 | 5.41E-03 | 4.53E-12 | 8.78E-03 |
| U236 | | 8.63E-05 | 3.52E-06 | 1.24E-10 | |
| U237 | | 1.15E-01 | 1. 18E-03 | 1.91E-05 | |
| U238 | | 1.03E-02 | 1.38E-03 | 3.45E-15 | 2.00E-05 |
| U240 | | | | | |
| Y90 | 2.62E+00 | 6.46E+03 | 1.70E+03 | 5.70E+01 | 1.24E+02 |
| Zr93 | | | 2.65E-04 | | 2.15E-03 |
| Zr95 | | | 2.38E-12 | | 9.64E-15 |
| TOTAL | 1.31E+01 | 3.23E+04 | 7.39E+03 | 2.43E+02 | 6.30E+02 |

| Nuclide | NTS | ORNL | SRS | SNL/NM | WVDP | TOTAL |
|---------|----------|----------|----------|----------|----------|----------|
| Ac225 | 8.80E-14 | 3.02E-01 | 2.96E-15 | 6.40E-18 | 7.44E-15 | 3.03E-01 |
| Ac227 | 9.88E-13 | 7.17E-04 | 4.20E-13 | 2.77E-20 | | 7.35E-04 |
| Ac228 | 3.63E-18 | 8.73E-02 | | | | 8.89E-02 |
| Ag110 | | | | | | 5.12E-09 |
| Ag110m | | | | | х | 3.85E-07 |
| Am241 | 4.85E-01 | 2.41E+02 | 6.79E-02 | 1.02E-02 | 5.39E-01 | 4.82E+02 |
| Am243 | | 9.98E-05 | 1.60E-05 | | | 8.07E-04 |
| Am245 | | 8.61E-16 | | | | 8.61E-16 |
| At217 | 8.80E-14 | 3.02E-01 | 2.96E-15 | 6.40E-18 | 7.44E-15 | 3.03E-01 |
| Ba137m | | 9.25E+03 | 6.49E+00 | | 5.06E+01 | 1.79E+04 |
| Bi210 | | 2.39E-07 | 1.24E-16 | | 1.51E-12 | 4.72E-07 |
| Bi211 | 9.88E-13 | 7.19E-04 | 4.20E-13 | 2.77E-20 | | 7.37E-04 |
| Bi212 | 2.08E-18 | 8.51E-02 | | | | 8.66E-02 |

¹Argonne National Laboratory-West is not included in this table because no radionuclide data were received frc the site.



| Nuclide | NTS | ORNL | SRS | SNL/NM | WVDP | TOTAL |
|---------|----------|----------|----------|----------|----------|----------|
| Bi213 | 8.80E-14 | 3.02E-01 | 2.96E-15 | 6.40E-18 | 7.44E-15 | 3.03E-01 |
| Bi214 | | 1.66E-06 | 1.64E-14 | 7.34E-20 | 2.38E-11 | 2.82E-06 |
| Bk249 | | 5.94E-11 | | | | 5.94E-11 |
| C14 | | 6.12E+00 | | | | 6.15E+00 |
| Cd113m | | | | | | 1.00E-06 |
| Ce144 | | 1.20E+01 | | | | 1.75E+01 |
| Cf249 | | 1.34E-02 | | | | 1.34E-02 |
| Cf250 | 1.81E-01 | | | | | 1.81E-01 |
| Cf252 | | 3.86E+00 | | | | 3.86E+00 |
| Cm243 | | 1.48E+02 | | | | 1.48E+02 |
| Cm244 | 1.55E+02 | 9.44E+02 | 4.68E+00 | | | 1.10E+03 |
| Cm245 | | 4.39E-06 | | | | 4.39E-06 |
| Cm246 | 3.95E-04 | | | | | 3.95E-04 |
| Cm248 | | 6.14E-04 | | | | 6.14E-04 |
| Co58 | | | | | | 4.37E-11 |
| Co60 | | 6.17E+02 | | | | 9.73E+02 |
| Cr51 | | | | | | 1.08E-05 |
| Cs134 | | 9.56E+00 | | | | 6.81E+01 |
| Cs135 | | | | | | 2.15E-04 |
| Cs137 | | 9.78E+03 | 6.86E+00 | | 5.35E+01 | 1.89E+04 |
| Eu152 | | 3.66E+03 | | | | 3.66E+03 |
| Eu154 | | 1.77E+03 | | | | 1.77E+03 |
| Eu155 | | 3.51E+02 | | | | 3.53E+02 |
| Fe55 | | | | | | 5.97E-01 |
| Fr221 | 8.80E-14 | 3.02E-01 | 2.96E-15 | 6.40E-18 | 7.44E-15 | 3.03E-01 |
| Fr223 | 1.36E-14 | 9.90E-06 | 5.80E-15 | 3.82E-22 | | 1.01E-05 |
| НЗ | | 7.71E-02 | | | | 2.20E-01 |

^{*} ¹Argonne National Laboratory-West is not included in this table because no radionuclide data were received from the site.

| Nuclide | NTS | ORNL | SRS | SNL/NM | WVDP | TOTAL |
|---------|----------|----------|-------------------|----------|----------|----------|
| Kr85 | | | | | | 5.95E+00 |
| Mn54 | | | | | | 8.31E-02 |
| Nb95 | | 2.01E+00 | | | | 2.01E+00 |
| Nb95m | | 6.72E-03 | | | | 6.72E-03 |
| Ni63 | | | | | | 3.50E+00 |
| Np237 | 3.19E-06 | 8.39E+00 | 1.43E-05 | 1.01E-08 | 1.49E-06 | 8.39E+00 |
| Np239 | | 9.98E-05 | 1.60E-05 | | | 8.07E-04 |
| Np240m | | 6.62E-11 | | | | 6.62E-11 |
| Pa231 | 6.39E-12 | 8.11E-05 | 2.67E-11 | 5.21E-19 | | 1.47E-04 |
| Pa233 | 3.19E-06 | 8.39E+00 | 1.43E-05 | 1.01E-08 | 1.49E-06 | 8.39E+00 |
| Pa234 | 3.31E-21 | 3.96E-02 | | | | 3.96E-02 |
| Pa234m | 2.54E-18 | 3.05E+01 | | | | 3.05E+01 |
| Ръ209 | 8.80E-14 | 3.02E-01 | 2 .96E -15 | 6.39E-18 | 7.44E-15 | 3.03E-01 |
| РЬ210 | | 2.39E-07 | 1.24E-16 | | 1.51E-12 | 4.72E-07 |
| Pb211 | 9.88E-13 | 7.19E-04 | 4.20E-13 | 2.77E-20 | | 7.37E-04 |
| Pb212 | 2.08E-18 | 8.51E-02 | | | | 8.66E-02 |
| Pb214 | | 1.66E-06 | 1.64E-14 | 7.34E-20 | 2.38E-11 | 2.82E-06 |
| Pd107 | | | | | | 3.18E-05 |
| Pm147 | | | 1.34E+00 | | | 3.19E+01 |
| Po210 | | 2.39E-07 | 3.40E-17 | | 1.51E-12 | 4.72E-07 |
| Po211 | 2.77E-15 | 2.01E-06 | 1.18E-15 | 7.74E-23 | | 2.06E-06 |
| Po212 | 1.34E-18 | 5.45E-02 | | | | 5.55E-02 |
| Po213 | 8.61E-14 | 2.95E-01 | 2.89E-15 | 6.26E-18 | 7.28E-15 | 2.96E-01 |
| Po214 | | 1.66E-06 | 1.64E-14 | 7.34E-20 | 2.38E-11 | 2.82E-06 |
| Po215 | 9.88E-13 | 7.19E-04 | 4.20E-13 | 2.77E-20 | | 7.37E-04 |
| Po216 | 2.08E-18 | 8.51E-02 | | | | 8.66E-02 |
| Po218 | | 1.66E-06 | 1.64E-14 | 7.34E-20 | 2.38E-11 | 2.82E-06 |

¹Argonne National Laboratory-West is not included in this table because no radionuclide data were received fruithe site.

C - 24

| Nuclide | NTS | ORNL | SRS | SNL/NM | WVDP | TOTAL |
|---------|----------|----------|----------|----------|----------|----------|
| Pr144 | | 1.18E+01 | | | | 1.73E+01 |
| Pu238 | | 2.82E+01 | 8.83E+00 | 4.92E-06 | 1.98E+01 | 1.69E+02 |
| Pu239 | 2.36E+00 | 9.86E+01 | 1.06E-02 | 2.00E-06 | | 5.59E+02 |
| Pu240 | 2.54E-01 | 1.07E+00 | 5.06E-04 | | | 1.79E+02 |
| Pu241 | 6.60E-05 | 3.98E-07 | | | | 4.71E+03 |
| Pu242 | 4.27E-09 | | | | | 5.94E-03 |
| Pu244 | | 6.63E-11 | | | | 6.63E-11 |
| Ra223 | 9.88E-13 | 7.19E-04 | 4.20E-13 | 2.77E-20 | | 7.37E-04 |
| Ra224 | 2.08E-18 | 8.51E-02 | | | | 8.66E-02 |
| Ra225 | 8.80E-14 | 3.02E-01 | 2.96E-15 | 6.40E-18 | 7.44E-15 | 3.03E-01 |
| Ra226 | | 1.66E-06 | 1.64E-14 | 7.34E-20 | 2.38E-11 | 2.82E-06 |
| Ra228 | 3.63E-18 | 8.73E-02 | | | | 8.89E-02 |
| Rh106 | | 3.21E+01 | | | | 3.30E+01 |
| Rn219 | 9.88E-13 | 7.19E-04 | 4.20E-13 | 2.77E-20 | | 7.37E-04 |
| Rn220 | 2.08E-18 | 8.51E-02 | | | | 8.66E-02 |
| Rn222 | | 1.66E-06 | 1.64E-14 | 7.34E-20 | 2.38E-11 | 2.82E-06 |
| Ru106 | | 3.21E+01 | | | | 3.30E+01 |
| Sb125 | | | | | | 4.30E+00 |
| Sb126 | | | | | | 5.78E-05 |
| Sb126m | | | | | | 4.13E-04 |
| Se79 | | | | | | 1.86E-04 |
| Sm151 | | | | | | 6.55E-01 |
| Sn119m | | | | | | 2.85E-06 |
| Sn121m | | | | | | 1.23E-02 |
| Sn126 | | | | | | 4.13E-04 |
| Sr90 | | 3.53E+04 | 6.85E+00 | | 1.96E+01 | 4.36E+04 |
| Ta182 | | | | | | 1.49E-07 |

^{*} ¹Argonne National Laboratory-West is not included in this table because no radionuclide data were received from the site.

C - 25

| Nuclide | NTS | ORNL | SRS | SNL/NM | WVDP | TOTAL |
|------------------|----------|----------|----------|----------|----------|----------|
| Тс99 | | | | | | 1.07E-02 |
| Te125m | | | | | | 1.06E+00 |
| Te127 | | | | | | 5.91E-09 |
| Te127m | | | | | | 6.04E-09 |
| Th227 | 9.74E-13 | 7.09E-04 | 4.14E-13 | 2.73E-20 | | 7.27E-04 |
| Th228 | 2.08E-18 | 8.51E-02 | | | | 8.66E-02 |
| Th229 | 8.80E-14 | 3.02E-01 | 2.96E-15 | 6.40E-18 | 7.44E-15 | 3.03E-01 |
| Th230 | | 6.64E-04 | 1.13E-10 | 2.54E-16 | 1.92E-08 | 9.07E-04 |
| Th231 | 3.71E-08 | 5.53E-01 | 1.26E-06 | 9.85E-15 | | 7.13E-01 |
| Th232 | 1.24E-17 | 9.92E-02 | 1.24E-22 | | | 1.01E-01 |
| Th234 | 2.54E-18 | 3.05E+01 | | | | 3.05E+01 |
| TI207 | 9.85E-13 | 7.17E-04 | 4.19E-13 | 2.76E-20 | | 7.35E-04 |
| TI208 | 7.49E-19 | 3.06E-02 | | | | 3.11E-02 |
| TI209 | 1.90E-15 | 6.52E-03 | 6.39E-17 | 1.38E-19 | 1.61E-16 | 6.54E-03 |
| U233 | 1.40E-10 | 4.36E+02 | 6.26E-11 | 6.67E-14 | 2.76E-11 | 4.36E+02 |
| U234 | 2.02E-23 | 1.02E+01 | 2.51E-05 | 2.81E-11 | 4.94E-04 | 1.17E+01 |
| U235 | 3.71E-08 | 5.53E-01 | 1.26E-06 | 9.85E-15 | | 7.13E-01 |
| U236 | 5.24E-08 | 2.82E-01 | 7.54E-12 | | | 2.82E-01 |
| U237 | 1.62E-09 | 9.74E-12 | | | | 1.16E-01 |
| U238 | 2.54E-18 | 3.05E+01 | | | | 3.05E+01 |
| Ų240 | | 6.62E-11 | | | | 6.62E-11 |
| Y90 | | 3.53E+04 | 6.85E+00 | | 1.96E+01 | 4.36E+04 |
| Zr93 | | | | | | 2.41E-03 |
| Zr95 | | 9.06E-01 | | | | 9.06E-01 |
| Total by Site | 1.58E+02 | 9.81E+04 | 4.20E+01 | 1.02E-02 | 1.64E+02 | 1.39E+05 |

¹Argonne National Laboratory-West is not included in this table because no radionuclide data were received from the site.



ABBREVIATIONS

ETEC Energy Technology Engineering Center

- HANF Hanford
- INEL Idaho National Engineering Laboratory
- KAPL Knolls Atomic Power Laboratory
- LANL Los Alamos National Laboratory
- NTS Nevada Test Site
- ORNL Oak Ridge National Laboratory
- SRS Savannah River Site
- SNL/NM Sandia National Laboratory-Albuquerque
- WVDP West Valley Demonstration Plant

¹Argonne National Laboratory-West is not included in this table because no radionuclide data were received from the site.

APPENDIX D



Sandia National Laboratories

Managed and Operand by Sandia Corporation a Lockheed Martin Corporation Albuquerque, New Mexico 87185-1328

date : June 20, 1996 to : Russ Bisping (DOE/NTP/CAO) from : L. C. Sanchez, Org 6848, MS-1328, PH-(505)848-0685, Fax-848-0705

subject : Correction for Cf252 Decayed Inventory

Per a request from the TWBIR Team [CH-1], a detailed check was made on the data that was used to perform decay calculations for the stored Cf252 inventory from the Hanford site [SNL-1]. The result of the data check was that the undecayed Cf252 stored CH-TRU inventory for the year 1982 should be 1.08E-03 Ci. The value that was erroneously used for the decay calculations was 1.08E+03 Ci. This means that the Cf252 and it principal decay daughters (Cm248 and Pu240) are overestimated (see Table 1). The WIPP disposal radionuclide inventory in the electronic database should be adjusted to correct these errors. Since Cf252 has a halflife less than 20 yr and the buildup (ingrowth) activities of Cm248 and Pu240 are very small, they have a negligible effect of the EPA Unit calculations (i.e., activity loading) - they represent a change in the calculated EPA Unit of less than one part in a million (see Table 4 of Ref. SNL-3). Thus, it not necessary for SNL WIPP PA CCA calculations to re-adjust the activity loading values presented in Refs. CCA-2 and CCA-3.

| Table 1. Activity Calculations Performed With Analytical Solution to BATEMAN Equation (a) | | | | | | | |
|---|--------------------------|--|--|--|--|--|--|
| Nuclide | | | Solution Using Analytical Solution to Bateman Equation, Decayed to the Year 1995 | | | | |
| ID (b) | Decay Mode (c) | Half- Life [sec] (d) | Existing Inventory [Curies] (e) | Correct Inventory [Curies] (f) | | | |
| Cf252 Cm248 Pu240 | α.γ.SF α.SF α.γ.SF | 8.3250E+07 1.0700E+13 2.0630E+11 | 3.5482E+01 (g) 8.1266E-03 (h) 8.2980E-06 (i) | 3.5482E-05 8.1266E-09 8.2980E-12 | | | |

(a) Calculations correspond only to the 1982 inventory of stored Cf252 at Hanford for CH-TRU. These values indicated that the decay calculations of Ref. SNL-1 overestimated the inventory (on a WIPP-Scale basis) of Cf252 (and to an lesser extent for the first two daughters of Cf252, namely - Cm248 and Pu240). The calculations presented here correspond to 1.08E+03 Ci for the "Existing Inventory" and 1.08E-03 Ci for the "Correct Inventory" at year = 1982 for the undecayed stored Hanford CH-TRU. The Existing Inventory value was that value used in Ref. SNL-1. Activity values presented here for Cm248 and Pu240 correspond only to ingrowth activities from Cf252 only.

Exceptional Service in the National Interest

- (b) Radionuclides are Cf252 and its first two daughters (these are incorporated into the WIPP PA database to yield Ref. CCA-2).
- (c) Decay mode information taken from Ref. GE-1.
- (d) Halflife values are those incorporated in ORIGEN2, see Ref. SNL-2.
- (e) "Existing Inventory" values correspond to 1982 Hanford CH-TRU inventory (originating from Cf252) decayed to the base year of 1995. The undecayed Cf252 inventory was 1.08E+03. Decay calculations were performed using Ref. KA-1b.
- (f) "Correct Inventory" values correspond to 1982 Hanford CH-TRU inventory (originating from Cf252) decayed to the base year of 1995. The undecayed Cf252 inventory was 1.08E-03. Decay calculations were performed using Ref. KA-1b.
- (g) Using this value in the TWBID resulted in a total decayed WIPP-Scale stored CI252 inventory of 36.1 Ci [Ref. CH-2] (98.3 % of this value was from the incorrect value from the 1982 Hanford inventory). Thus, correcting the undecayed 1982 Hanford value for CI252 will result in a substantial lowering of the stored and projected inventory of CI252. Since CI252 has a halfife less than 20 yr, it does not contribute to the EPA Unit value and does not effect WIPP PA CCA calculations.
- (b) Using this ingrowth value in the TWBID resulted in a total decayed WIPP-Scale stored Cm248 inventory of 3.35E-02 Ci [Ref. CH-2] (24.3 % of this value was from the incorrect value from the 1982 Hanford inventory). Thus, correcting the undecayed 1982 Hanford value for Cf252 will result in a substantial lowering of the stored and projected inventory of Cm248. Since the total activity change due to the ingrowth of Cm248 from Cf252 is very small it has a negligible contribution to the EPA Unit (see Table 4 of Ref. SNL-3) and does not affect PA calculations.
- (1) Using this ingrowth value in the TWBID resulted in a total decayed WIPP-Scale stored Cm248 inventory of 6.87E+04 Ci [Ref. CH-2] (less than 2.0E-08 % of this value was from the incorrect value from the 1982 Hanford inventory). Thus, correcting the undecayed 1982 Hanford value for Cf252 will result in a negligible lowering of the stored and projected inventory of Pu240 (or any further decay daughters from Cf252). Since the total activity change due to the ingrowth of Pu240 from Cf252 is very small it has a negligible contribution to the EPA Unit (see Table 4 of Ref. SNL-3) and does not affect PA calculations.

REFERENCES

[CCA-2]

DOE (U.S. Department of Energy), CAO Memorandum from: Don Watkins (Manager, National TRU Program); to: Les E. Shephard (SNL/NM Director, Nuclear Waste Management Programs Center); Subject: "Revised Radionuclide Data in Support of the Compliance Certification Application"; Dated: June 4, 1996; CAO:NTP:RBL:96-1174.

[CCA-3]

DOE (U.S. Department of Energy), CAO Memorandum from: Don Watkins (Manager, National TRU Program); to: Dr. Les E. Shephard (Nuclear Waste Management Programs Center Director, SNL/NM); Subject: "Preliminary Activities for Selected Radionuclides for CH-TRU Waste Streams"; dated: June 12, 1996. Data originally presented in: Personal communications S. Chakraborti (SAIC/CTAC); May 1, 1996.

- [CH-1]Personal communications (phone) with Sayan Chakraborti (SAIC/CTAC); Subject: "Error in Cf252 Inventory"; date: June 19, 1996.
- [CH-2]Personal communications (fax) from Sayan Chakraborti (SAIC/CTAC); Subject: "Stored and Projected WIPP Total Curies"; date: April 19, 1996.
- [GE-1]General Electric Company (Nuclear Energy Operations), Nuclides and Isotopes, Fourteenth Edition, 1989.
- [KA-1]I. Kaplan; Nuclear Physics (Second Edition); Addison-Wesley Publishing Company; Reading, Massachusetts, 1964.

[KA-1b]

Ibid., Equations 10-26 and 10-27, pg. 243.

[SNL-1]

Sandia National Laboratories; "WIPP PA Analysis Report for ORIGEN2 Suite", Document Version 1.00, WBS # 1.1.6.1, April 22, 1996; SWCF-A:1.4.01.6:SFT:QA:ORIGEN2 Suite.
[SNL-2]

Sandia National Laboratories; Memo from: L.C. Sanchez (Org 6741), to: M. Martell (Org 6749); Subject: "Radionuclide Half-lives and Specific Activities Obtained From ORIGEN2 Data"; dated: March 28, 1996.

[SNL-3]

Sandia National Laboratories; Memo from: L.C. Sanchez (Org 6741), to: Distribution; Subject: "Identification of Important Radionuclides Used in 1996 CCA WIPP Performance Assessment"; dated: April 25, 1996.

LCS:6848:lcs/(96-2113)

Copy to: Sayan Chakraborti [SAIC/CTAC] Paul Drez [DEA/CTAC] MS-1328, H. Jow [Dept. 6848] MS-1328, J. Garner [Dept. 6849] MS-1328, Day File [Dept. 6848] MS-1328, L.C. Sanchez [Dept. 6848] File - SWCF-A WBS 1.1.6.2; PA; PBWAC - WIPP ACTIVITY